REPORT NO. UCB/EERC-80/01 JANUARY 1980

EARTHQUAKE ENGINEERING RESEARCH CENTER

EARTHQUAKE RESPONSE OF CONCRETE GRAVITY DAMS INCLUDING HYDRODYNAMIC AND FOUNDATION INTERACTION EFFECTS

by

ANIL K. CHOPRA P. CHAKRABARTI SUNIL GUPTA

A report on research conducted under Contract DACW 73-71-C-0051 with the Office of the Chief of Engineers, Washington, D.C. and Grants ATA74-20554 and ENV76-80073 from the National Science Foundation.



COLLEGE OF ENGINEERING

UNIVERSITY OF CALIFORNIA · Berkeley, California

Any opinions, findings, and conclusions or recommendations expressed in this publication are those of the authors and do not necessarily reflect the views of the National Science Foundation.

See back of report for up to date listing of EERC reports.

REPORT DOCUMENTATION	1. REPORT NO.	. 2.	3. Recipient's Accession No.	
Title and Subtitle	NSF/ RA-80019	<u> </u>	A Decort Date	
Earthquak	ce Response of Concrete	Gravity Dams Ind	January 1980	
Hydrodyna	amic and Foundation inte	raction Effects	6.	·····
Author(s) A.K. Chopra, I	P. Chakrabarti and S. Gu	pta	8. Performing Organization Rept UCB/EERC-80/01	No.
Performing Organization Name a	ind Address		10. Project/Task/Work Unit No.	
Iniversity of Califo	ornia Richmond Field St	ation		
47th and Hoffman Bly	vd.		II. Contract(C) or Grant(G) No.	
Richmond, California	a 94804			
			(G) $ATA/4 = 20554$ ENV76 = 80073	
2. Sponsoring Organization Name a	and Address	·····	13. Type of Report & Period Cov	red
National Science Fou	undation			
1800 G Street, N.W.				
Washington, D. C. 2	20550		14.	
E. Supplementany Notes				
. Supplementary notes				
represented as a fini in the upstream direct structural displacement chosen as normal mode	ite element system, the ction, and the foundatio ents of the dam are expr es of an associated unda	fluid domain, as on rock region as ressed as a lines mped dam-foundat	s a continuum of infinite l s a viscoelastic halfplane. ar combination of Ritz vect tion system. The modal dis	engt Th ors, plac
ments due to earthqua using Fast Fourier Tr displacements.	ake motion are computed are notion procedures. Th	by synthesizing e stress respons	their complex frequency re ses are calculated from the	spon. mod
The response of a motion is presented a of the dam, foundation effects of structure- effects of the two so conclusions presented	idealized dam cross-sect for a range of important on rock and impounded wa -water interaction and s ources of interaction, o d.	ions to harmonia system parameta ter. Based on tructure-foundat on dynamic respon	c horizontal or vertical gr ers characterizing the prop these results, the separate tion interaction, and the c nse of dams are investigate	cound perti combi ed an
	•			
				, ··

Release Unlimited

20. Security Class (This Page)

22. Price

.

EARTHQUAKE RESPONSE OF CONCRETE GRAVITY DAMS INCLUDING HYDRODYNAMIC AND FOUNDATION INTERACTION EFFECTS

by

Anil K. Chopra

P. Chakrabarti

Sunil Gupta

A Report on Research Conducted Under Contract DACW 73-71-C-0051 with the Office of the Chief of Engineers, Washington, D.C. and Grants ATA74-20554 and ENV76-80073 from the National Science Foundation

> Report No. UCB/EERC-80/01 Earthquake Engineering Research Center University of California Berkeley, California

> > January 1980

. -•

1 1 1

ABSTRACT

A general procedure for analysis of the response of concrete gravity dams, including the dynamic effects of impounded water and flexible foundation rock, to the transverse (horizontal) and vertical components of earthquake ground motion is presented. The problem is reduced to one in two dimensions, considering the transverse vibration of a monolith of the dam. The system is analyzed under the assumption of linear behavior for the concrete, foundation rock and water.

The complete system is considered as composed of three substructures -the dam, represented as a finite element system, the fluid domain, as a continuum of infinite length in the upstream direction, and the foundation rock region as a viscoelastic halfplane. The structural displacements of the dam are expressed as a linear combination of Ritz vectors, chosen as normal modes of an associated undamped dam-foundation system. The effectiveness of this analytical formulation lies in its being able to produce excellent results by considering only a few Ritz vectors. The modal displacements due to earthquake motion are computed by synthesizing their complex frequency responses using Fast Fourier Transform procedures. The stress responses are calculated from the modal displacements.

An example analysis is presented to illustrate results obtained from this analytical procedure. Computation times for several analyses are presented to illustrate effectiveness of the procedure.

The response of idealized dam cross-sections to harmonic horizontal or vertical ground motion is presented for a range of important system parameters characterizing the properties of the dam, foundation rock and impounded water. Based on these results, the separate effects of structure-water interaction and structure-foundation interaction, and the combined effects of the two sources of interaction, on dynamic response of dams are investigated, leading to the following conclusions.

Each source of interaction generally has significant effect on the complex frequency response functions for the dam. The fundamental resonant frequency of the dam decreases and its apparent damping increases because of structure-foundation interaction. The higher resonant frequencies and associated damping are affected similarly but to a lesser degree. These effects

- i -

are qualitatively similar whether the reservoir is empty or full, except at the resonant frequencies of the fluid domain. Because of hydrodynamic effects, the response curves are complicated in the neighborhood of the natural frequencies of water in the reservoir; the resonant frequencies of the dam are reduced -- the fundamental frequency by a significant amount but the higher resonant frequencies by relatively little; and the fundamental mode exhibits highly resonant behavior. The hydrodynamic effects in the dam response are qualitatively similar whether the foundation rock is rigid or flexible. The fundamental resonant frequency of the dam is reduced by roughly the same degree, independent of the foundation material properties. However, the apparent damping at the fundamental resonant frequency is dominated by effects of structure-foundation interaction and varies little with the depth of water. The fundamental resonant frequency of the dam is reduced by each of the two sources of interaction, with the influence of water usually being larger. However, there are no general trends regarding the comparative effects of water and foundation on the higher resonant frequencies or on the resonant responses of the dam. The response of the dam, without water, to vertical ground motion is small relative to that due to horizontal ground motion, but it becomes relatively significant when the hydrodynamic effects are included.

The displacement and stress responses of Pine Flat Dam to the S69E component of the Taft ground motion only, and to the S69E and vertical components acting simultaneously, are presented. For each of these excitations, the response of the dam is analyzed four times corresponding to the following four sets of assumptions: (1) rigid foundation, hydrodynamic effects excluded; (2) rigid foundation, hydrodynamic effects included; (3) flexible foundation, hydrodynamic effects excluded; and (4) flexible foundation, hydrodynamic effects included. These results lead to the following conclusions.

The displacements and stresses of Pine Flat Dam due to the Taft ground motion are increased significantly because of hydrodynamic effects. Compared to the response of the dam including only hydrodynamic effects, the stresses in the upper parts of the dam are significantly increased due to structure-foundation interaction. Stresses at the heel of the dam are increased to a lesser extent because of the stress relaxation due to foundation flexibility. The influence of structure-foundation interaction and structure-water interaction on the response of a dam depends in part on

- ii -

the change in the earthquake response spectrum ordinate associated with changes in frequency and apparent damping due to these interaction effects. As such, these effects would depend on the resonant frequencies of the dam and the shape of the earthquake response spectrum in a neighborhood of these resonant frequencies. Although considerable stresses in Pine Flat Dam are caused by the vertical component of Taft ground motion, they partially cancel the stresses due to the horizontal component, resulting in reduced response when both ground motion components are considered simultaneously. The contribution of the vertical component of ground motion to the total response of a dam, including hydrodynamic effects, depends on the relative phasing of the responses to horizontal and vertical ground motion, which in turn depends on the phasing of the ground motion components and the vibration properties of the dam.

TABLE OF CONTENTS

			Page
ABS	FRACT		i
TABI	LE OF	CONTENTS	v
1.	INTRO	DDUCTION	1
	1.1	Objectives	1
	1.2	Review of Past Work	1
	1.3	Scope of this Report	4
	1.4	Acknowledgments	5
2.	SYSTI	EM AND GROUND MOTION	7
	2.1	Dam-Water-Foundation System	7
	2.2	Ground Motion	8
3.	ANALY	YSIS OF GRAVITY DAMS	11
	3.1	Introductory Note	11
	3.2	Governing Equations	11
	3.3	Earthquake Response Analysis	13
		3.3.1 Time Domain Analysis	13
		3.3.2 Frequency Domain Analysis	15
4.	ANALY	YSIS OF GRAVITY DAMS INCLUDING DAM-WATER INTERACTION	17
	4.1	Introductory Note	17
	4.2	Governing Equations	17
		4.2.1 Substructure 1: Dam	17
		4.2.2 Substructure 2: Fluid Domain	19
	4.3	Response Analysis for Harmonic Horizontal Ground Motion	20
	4.4	Response Analysis for Harmonic Vertical Ground Motion .	24
	4.5	Response to Arbitrary Ground Motion	27

Page

5.	ANAL	YSIS OF DAMS INCLUDING DAM-FOUNDATION INTERACTION	29
	5.1	Introductory Note	29
	5.2	Frequency Domain Equations in Nodal Point Coordinates	29
		5.2.1 Substructure 1: Dam	29
		5.2.2 Substructure 2: Foundation Rock Region	30
		5.2.3 Dam-Foundation System	32
		5.2.4 Computational Requirements	34
	5.3	Reduction of Degrees of Freedom	34
	5.4	Response to Arbitrary Ground Motion	36
6.	ANAL ACTI	YSIS OF DAMS INCLUDING HYDRODYNAMIC AND FOUNDATION INTER- ON EFFECTS	39
	6.1	Introductory Note	39
	6.2	Frequency Domain Equations	39
		6.2.1 Substructure 1: Dam	39
		6.2.2 Substructure 2: Foundation Rock Region	41
		6.2.3 Dam-Foundation System	44
		6.2.4 Reduction of Degrees of Freedom	44
		6.2.5 Substructure 3: Fluid Domain	47
		6.2.6 Dam-Water-Foundation System	53
		6.2.7 Singularities of Response	56
	6.3	Response to Arbitrary Ground Motion	56
	6.4	Static Stress Analysis	57
	6.5	Special Cases	62
		6.5.1 Dam-Foundation System, No Water	62
		6.5.2 Dam on Rigid Foundation, with Water	62
		6.5.3 Dam on Rigid Foundation, No Water	63
	6.6	Computer Program	64

.

7.	PREL	IMINARY RESULTS FOR PARAMETER SELECTION
	7.1	Scope of the Chapter
	7.2	Systems, Ground Motions, and Response Quantities 65
		7.2.1 Systems
		7.2.2 Ground Motions
		7.2.3 Response Quantities
	7.3	Comparison of Plane Strain and Plane Stress Assumption 68
	7.4	Elastic Moduli Parameters
	7 5	Damping Models and Parameters 70
	1.5	7.5.1 Dam 70
		7.5.2 Daurdation
	7 6	Number of Convoluted Coordinates 72
0	7.6	Number of Generalized Coordinates
8.	COMP	LEX FREQUENCY RESPONSES
	8.1	Scope of Chapter
	8.2	Systems, Ground Motions, Cases Analyzed and Response Quantitites
		8.2.1 Systems
		8.2.2 Ground Motions
		8.2.3 Cases Analyzed
		8.2.4 Response Quantities
	8.3	Complex Frequency Responses
		8.3.1 Dam-Water Interaction Effects 83
		8.3.2 Dam-Foundation Interaction Effects 90
		8.3.3 Dam-Water and Dam-Foundation Interaction Effects . 90
9.	EART	HQUAKE RESPONSE OF PINE FLAT DAM
	9.1	Scope of the Chapter
	9.2	Pine Flat Dam, Ground Motion, Cases Analyzed and Response Results

Page

9.2.1 Pine Flat Dam	7
9.2.2 Ground Motion	.1
9.2.3 Cases Analyzed	.1
9.2.4 Response Results	.4
9.3 Dam-Water Interaction Effects	.5.
9.4 Dam-Foundation Interaction Effects	0
9.5 Dam-Water and Dam-Foundation Interaction Effects 12	4
10. CONCLUSIONS	3
REFERENCES	7
APPENDIX A: USERS GUIDE TO COMPUTER PROGRAM	1
APPENDIX B: LISTING OF EAGD COMPUTER PROGRAM 15	9
EERC REPORT LISTING	9

Page

1. INTRODUCTION

1.1 Objectives

It is extremely important to design dams which store large quantities of water to safely withstand earthquakes, particularly in view of the catastrophic consequences of dam failure. The damage to Koyna Dam, in India, which was designed according to standard, widely accepted procedures, shows that concrete gravity dams are not as immune to earthquake damage as has commonly been believed [1,2].

Reliable analytical procedures are necessary to design earthquake resistant dams, and to evaluate the safety of existing dams during future earthquakes. These procedures should provide the capability of evaluating the dynamic deformations and stresses in a dam subjected to a given ground motion. Special attention should be given to the interaction of the dam with the impounded water and with the foundation rock or soil. These factors complicate the otherwise routine finite element analysis of concrete gravity dams.

The objectives of the present study are: (a) to develop reliable and effective techniques for analyzing the response of concrete gravity dams to earthquake ground motion, including effects of dam-water interaction and of dam-foundation interaction, and (b) to examine the significance of these interaction effects in earthquake response of dams.

1.2 Review of Past Work

During the past 25 years, the finite element method has become the standard procedure for analysis of all types of civil engineering structures. Early in its development, it became apparent that this method had unique potentialities in the evaluation of stress in dams, and many of its earliest civil engineering applications concerned special problems associated with such structures [3,4]. The earliest dynamic finite element analyses of civil engineering structures involved the earthquake response analysis of earth dams [5]. Using the finite element method and excluding hydrodynamic effects, a number of questions were studied concerning the response of concrete gravity Cams to earthquakes. These questions were prompted by the structural damage, caused by the December 1967 earthquake, to Koyna Dam.

- 1 -

The analysis of hydrodynamic pressures, due to horizontal ground motion, on <u>rigid</u> dams started with Westergaard's pioneering work of 1933 [7]. Continuing with the assumption of a rigid dam, more comprehensive analyses of hydrodynamic pressures on the dam face, due to both horizontal and vertical components of ground motion, have been developed [8,12].

In the finite element analyses mentioned, hydrodynamic effects were not considered; whereas in the studies of hydrodynamic pressures, the dam was assumed to be rigid. Additional hydrodynamic pressures will result from deformations of the upstream face of the dam, and the structural deformations in turn will be affected by the hydrodynamic pressures on the upstream face. To break this closed cycle of cause and effect, the problem formulation must recognize the dynamic interaction between the dam and water.

Finite element analysis of the complete dam-water system is one possible approach to including these interaction effects. Application of standard analysis procedures with nodal point displacements as the degrees-of-freedom was only partly successful [13]. A different finite element formulation of the complete system, in which displacements are considered as the unknowns at the nodal points for the dam, and pressures as the unknowns at the nodal points for the water, has been applied to small problems [14]; but this approach appears to require prohibitive computational effort for practical problems.

The more effective approach is to treat the dam-water system as composed of two substructures -- dam and fluid domain -- coupled through the interaction forces and appropriate continuity conditions at the face of the dam. A series of studies [15-18] led to a general analysis procedure [19] and computer program [20] for dynamic analyses of dams, including dam-water interaction. This approach conveniently permits different models to be used for the dam and water. The dam may be idealized by the finite element method, which has the ability to handle systems of arbitrary geometry. At the same time, the fluid domain may be treated as a continuum, an approach which is ideally suited to the simple geometry but great upstream extent of the impounded water [19]. The fluid domain may also be idealized by the finite element method in conjunction with infinite elements [21]. When the substructure method is employed, along with transformation of the governing equations to generalized coordinates

- 2 --

associated with vibration modes of the dam alone, the analysis procedure is very efficient, and little additional computational effort is required to include the hydrodynamic effects [19].

Utilizing such an analysis procedure, it was shown that the dam-water interaction and water compressibility have a significant influence on the dynamic behavior of concrete gravity dams and their responses to earthquake ground motion [22]. Because of hydrodynamic effects, the vertical component of ground motion is more important in the response of gravity dams than in other classes of structures [18,22]. A simplified analysis procedure has been developed which includes the dam-water interaction effects in the computation of lateral earthquake forces for dam design [23].

The effects of dam-foundation interaction can most simply be included in dynamic analysis of dams by including, in the finite element idealization, foundation rock or soil above a rigid horizontal boundary. The response of such a finite element system to excitation specified at the bottom, rigid boundary is then analyzed by standard procedures. Such an approach leads to enormous computational requirements and the reliability of results in some cases is questionable. For sites where similar materials extend to large depths and there is no obvious "rigid" boundary such as a soil-rock interface, the location of the rigid boundary introduced in the analysis is often quite arbitrary, and it may significantly distort the response.

These difficulties can be overcome by using the substructure method [24, 25], wherein the dam and the foundation rock or soil region are considered as two substructures of the complete system. The dam may be idealized as a finite element system which has the ability to represent arbitrary geometry and material properties. The foundation rock or soil region may be idealized as either a continuum (a viscoelastic halfspace for example), or as a finite element system, whichever is appropriate for the site. The halfspace idealization permits accurate modelling of sites where similar materials extend to large depths. For sites where layers of soil or soft rock overlie harder rock at shallow depths, finite element idealization of the foundation region would be appropriate. The governing equations for the two substructures are combined by imposing equilibrium and compatibility requirements at the base of the dam. These equations make direct use of free field ground motion specified at the dam-foundation interface. The resulting equations are transformed to generalized Ritz coordinates; the displacements are expressed as a

- 3 -

linear combination of the first few vibration modes of an associated damfoundation system, thus leading to a very efficient solution.

The preceding review discussed the dam-water system and dam-foundation system separately. In each of these two cases, the impounded water and the foundation, respectively, modify the vibration properties of the dam and may significantly affect its response. However, the two problems actually are coupled and the results obtained by separate analyses will, in general, be invalid. There is need, therefore, for developing techniques for analysis of complete dam-water-foundation systems, and assessing simultaneously, the interaction due to both water and foundation. Some work on this problem has been reported in recent years [26,27]. The substructure methods developed for separately considering hydrodynamic and foundation interaction effects in the response of dams appear to be ideally suited for and extendable to complete analyses which simultaneously include both types of interaction.

1.3 Scope of this Report

Chapter 2 describes how the dam-water-foundation system and earthquake ground motion are idealized in this study. Also included in this chapter are the assumptions underlying the procedure, developed in later chapters, for analysis of earthquake response of concrete gravity dams.

For the reader's convenience, a summary is given of the procedures available for analyzing earthquake response of concrete gravity dams under restrictive conditions. Chapter 3 summarizes the standard finite element method for analysis of dams on rigid foundations with no water stored in their reservoirs. Chapter 4 summarizes the substructure method for including the effects of dam-water interaction in the analysis procedure. (This method treats the impounded water and dam on rigid foundation as two substructures of the total system.) Also based on the substructure concept is the procedure for including effects of interaction between the dam and its flexible foundation on the earthquake response of dams without water, summarized in Chapter 5.

Because the substructure concept has proven to be effective in separately including the effects of dam-water interaction and of dam-foundation interaction in the analysis, it is extended in Chapter 6 to develop an analysis procedure simultaneously including various effects of the water and the foundation. These include effects arising from interaction between the dam

- 4 -

and foundation, dam and water, water and foundation, and interaction among all three substructures -- dam, water, and foundation. The computer program developed to implement the procedure for analysis of dam-water-foundation systems is briefly described in this chapter; the user's guide and program listing is included in Appendices A and B.

Using this computer program, results of several preliminary analyses are presented in Chapter 7 with the aim of defining the important system and analysis parameters for the subsequent study of dynamic response behavior of dams. The behavioral study is in two parts, separated into Chapters 8 and 9.

The response of idealized dam cross-sections to harmonic horizontal or vertical ground motion is presented in Chapter 8 for a range of the important systems parameters characterizing the properties of the dam, foundation rock, and impounded water. With the aid of these results, the separate effects of dam-water interaction and dam-foundation interaction, and the combined effects of the two sources of interaction on dynamic response of dams are investigated.

Chapter 9 presents the responses of Pine Flat Dam to the S69E component of the Taft ground motion only; and to the S69E and vertical components acting simultaneously. For each of these excitations, the response of the dam is analyzed four times corresponding to the following four sets of assumptions: (1) rigid foundation, hydrodynamic effects excluded; (2) rigid foundation, hydrodynamic effects included; (3) flexible foundation, hydrodynamic effect excluded; and (4) flexible foundation, hydrodynamic effects included. These results provide insight into the effects of dam-water and dam-foundation interaction, considered separately or together, in the earthquake response of dams.

Chapter 10 briefly summarizes the significant conclusions which may be drawn from this investigation of dam-water and dam-foundation interaction effects in earthquake response of concrete gravity dams.

1.4 Acknowledgements

This study was initially supported by the Office of the Chief of Engineers, Department of the Army, Washington, D.C., under Contract No. DACW73-71-C-0051. Completion of this study was made possible by Grants ATA74-20554 and ENV76-80073 from the National Science Foundation. The authors are grateful for the support from both agencies.

- 5 -

Perhaps a few words regarding the history of this study would be of interest to the sponsors and the reader. The study was initiated in 1974. Dr. P. Charkabarti participated in formulating the essential aspects of the analysis procedure and developed the computer programs during 1974 and 1975, at which time he left Berkeley to accept a position in India. Thereafter, Dr. G. Dasgupta was involved in the study for a short time, incorporating some improvements in the analysis procedure and computer program. His efforts were then diverted to developing dynamic stiffness matrices for the foundation required in obtaining the numerical results for this study [32]. At this juncture, in 1976, Sunil Gupta, a graduate research assistant, became associated with the project. He produced the large volume of computer results presented in this report, and assisted in interpretating these results and preparing the report. Because there was some discontinuity between the times Chakrabarti, Dasqupta, and Gupta worked on this project; because considerable time was required to develop the results on dynamic stiffness matrices for the foundation; and for other reasons beyond our control, our efforts on this study have been sporadic. This resulted in considerable, unanticipated delay in completion of the study.

2. SYSTEM AND GROUND MOTION

2.1 Dam-Water-Foundation System

Vibration tests on Pine Flat Dam [28] indicate that at small vibration amplitudes a concrete gravity dam will behave like a solid even though there is some slippage between monoliths. Thus, at the beginning of an earthquake, the behavior of a dam can be best described by a three-dimensional model. However, at large amplitudes of motion, the inertia forces are much larger than the shear forces that can be transmitted across joints between monoliths. The monoliths slip and tend to vibrate independently, as evidenced by the spalled concrete and increased water leakage at the joints of Koyna Dam during the Koyna earthquake of December 11, 1967. Consequently, two-dimensional models of individual monoliths appear to be more appropriate than threedimensional models for predicting the response of concrete gravity dams to the strong phase of intense ground motion. However, a model more complicated than either of these two models will be necessary to describe the behavior of a concrete gravity dam through the complete amplitude range.

Because the dimensions and dynamic properties of the various monoliths differ, the effects of the dam on deformations and stresses in the foundation will vary along the length of the dam. Thus even with two-dimensional models for the dam, a three-dimensional model would seem necessary for the foundation. If the dam were to behave as a solid without slippage between monoliths, and all of its properties and the ground motion did not vary along the length, it would be appropriate to assume the dam as well as the foundation to be in plane strain. A plane stress model which applies to a thin sheet-like body seems obviously inappropriate for a continuum foundation.

Although the basic concepts underlying the analysis procedure presented in this work are applicable under more general conditions, the procedure is specifically developed for two-dimensional systems; thus its application is restricted to systems in generalized plane stress or plane strain. Although neither of the two models are strictly applicable, the former is better for the dam and the latter for the foundation. However, in order to define the dam-foundation system on a consistent basis, the same model should be employed for both substructures. Results from the two models are compared in Sec. 7.3 to provide a basis for choosing one for parameter studies.

- 7 -

A cross-section of the system considered is shown in Fig. 2.1. The system consists of a concrete gravity dam supported on the horizontal surface of a viscoelastic half plane and impounding a reservoir of water. The system is analyzed under the assumption of linear behavior for the concrete, foundation soil or rock, and water. The dam is idealized as a two-dimensional finite element system, thus making it possible to consider arbitrary geometry and variation of material properties. However, certain restrictions on the geometry are imposed to permit solutions for the foundation and fluid domains treated as continua. For the purposes of determining hydrodynamic effects, and only for this purpose, the upstream face of the dam is assumed to be vertical. This is reasonable for an actual concrete gravity dam, because the upstream face is vertical or almost vertical for most of its height, and the hydrodynamic pressures on the dam face are insensitive to small departures of the face slope from the vertical. For the purpose of including structure-foundation interaction effects, the foundation surface is assumed to be horizontal; thus the base of the dam as well as the reservoir bottom is assumed to be horizontal. An actual system can usually be idealized to conform to this assumption.

2.2 Ground Motion

The excitation for the two-dimensional dam-foundation-water system is defined by the two components of free-field ground acceleration in the plane of a cross-section of the dam: the horizontal component transverse to the dam axis, $a_{\alpha}^{X}(t)$, and the vertical component, $a_{\alpha}^{Y}(t)$.



ł

FIG. 2.1 DAM-WATER-FOUNDATION SYSTEM

- 9 -

ι.

3. ANALYSIS OF GRAVITY DAMS

3.1 Introductory Note

The standard finite element method for analysis of gravity dams on rigid foundations with no water stored in the reservoir is summarized in this chapter. In later chapters, the analysis will be extended to include effects of impounded water and foundation flexibility.

3.2 Governing Equations

Consider a monolith of a concrete gravity dam on a rigid foundation, with no water in the reservoir, subjected to earthquake ground motion which does not vary across the base of the monolith. The equations of motion for such a dam monolith idealized as a planar, two-dimensional finite element system are:

$$\underline{m\ddot{r}} + \underline{c}\dot{r} + \underline{k}r = -\underline{m}\underline{l}^{X} a_{g}^{X}(t) - \underline{m}\underline{l}^{Y} a_{g}^{Y}(t)$$
(3.1)

In Eq. 3.1, \underline{m} , \underline{k} and \underline{c} are the mass, stiffness, and viscous damping matrices for the finite element system; \underline{r} is the vector of nodal point displacements, relative to the free-field displacement:

$$\underline{\mathbf{r}}^{\mathrm{T}} = \langle \mathbf{r}_{1}^{\mathrm{x}} \mathbf{r}_{1}^{\mathrm{y}} \mathbf{r}_{2}^{\mathrm{x}} \mathbf{r}_{2}^{\mathrm{y}} \dots \mathbf{r}_{n}^{\mathrm{x}} \mathbf{r}_{n}^{\mathrm{y}} \dots \mathbf{r}_{N}^{\mathrm{x}} \mathbf{r}_{N}^{\mathrm{y}} \rangle$$

where r_n^x and r_n^y are the x- and y- components of displacement of nodal point n and the number of nodal points above the base is N; \dot{r} and \ddot{r} are, respectively, the nodal point velocity and acceleration vectors.

> $\{\mathbf{1}^{\mathbf{X}}\}^{\mathrm{T}} = \langle 1 \ 0 \ 1 \ 0 \ \dots \ 1 \ 0 \ \dots \ 1 \ 0 \ \dots \ 1 \ 0 \rangle$ $\{\mathbf{1}^{\mathbf{Y}}\}^{\mathrm{T}} = \langle 0 \ 1 \ 0 \ 1 \ \dots \ \dots \ 0 \ 1 \ \dots \ 0 \ 1 \rangle$

 $a_g^{x}(t)$ and $a_g^{y}(t)$ are the x (horizontal) and y (vertical) components of the free-field ground acceleration.

The stiffness and mass matrices of the structure are obtained from the corresponding matrices for the individual finite elements by direct assembly

procedures. The element stiffness matrices are derived using quadratic interpolation function for displacements, whereas the element mass matrices are based on a lumped mass approximation.

Energy dissipation in structures, even in the linear range of vibration, is due to various complicated phenomena. Because it is not possible to mathematically describe each of these sources of energy dissipation, it is customary to define damping in terms of damping ratios in the natural modes of vibration. Damping ratios for similar structures obtained from both the analysis of harmonic vibration tests, and responses recorded during earthquakes, are used as a basis for assigning the modal damping ratios. The modal damping ratios provide a complete description of damping properties for purposes of linear analysis and the damping matrix need not be defined explicitly.

Traditionally, analysis of dynamic repsonse of structures has been carried out directly in the time domain. For such analyses, viscous damping is the most convenient representation of energy dissipation in the structure. When effects of dam-water interaction and dam-foundation interaction are included, the analysis is most effectively carried out by a substructure method [3,4]. As will be seen in Chapters 4,5 and 6, the substructure method is best formulated in the frequency domain. In such a formulation, constant hysteretic damping is a preferable representation -- for conceptual as well as computational reasons -- especially when structure-soil interaction effects are included [5].

The Fourier transform of Eq. 2.1 provides the governing equation in the frequency domain

$$\left[-\omega^{2}\underline{m} + i\omega\underline{c} + \underline{k}\right]\hat{\underline{r}}(\omega) = -\underline{m}\underline{l}^{x}\hat{a}_{g}^{x}(\omega) - \underline{m}\underline{l}^{y}\hat{a}_{g}^{y}(\omega)$$
(3.2)

in which the Fourier transform of f(t) is denoted by $f(\omega)$. For constant hysteretic damping, Eq. 2.2 becomes

$$\left[-\omega^{2}\underline{\mathbf{m}} + (\mathbf{1} + i\eta)\underline{\mathbf{k}}\right]\underline{\hat{\mathbf{r}}}(\omega) = -\underline{\mathbf{ml}}^{\mathbf{X}} \hat{\mathbf{a}}_{g}^{\mathbf{X}}(\omega) - \underline{\mathbf{ml}}^{\mathbf{Y}} \hat{\mathbf{a}}_{g}^{\mathbf{Y}}(\omega)$$
(3.3)

where η is the constant hysteretic damping factor.

Damping ratios as obtained from forced vibration tests on dams are

essentially independent of the mode number [28]. If ξ is the damping ratio appropriate for all the natural modes of vibration of the dam, $\eta = 2\xi$ would be an appropriate value for the constant hysteretic damping factor [29]. With the damping coefficient so related, essentially the same response will be obtained for a lightly damped system with either damping mechanism.

3.3 Earthquake Response Analysis

3.3.1 Time Domain Analysis

The earthquake response of a dam is obtained by solving the equations of motion (Eq. 3.1). These equations of motion in nodal point coordinates may be solved either directly or after transformation to modal coordinates [30]. The latter approach, commonly known as the mode superpositon method, is applicable if the response is within the linear range. This method is advantageous for calculating the earthquake response of many types of structures, because the response is essentially due to the first few modes of vibration.

The first step in the mode superposition analysis procedure is to obtain the lower few natural frequencies and mode shapes of vibration of the dam by solving the eigenvalue problem:

$$\underline{k} \ \underline{\phi} = \omega^2 \ \underline{m} \ \underline{\phi} \tag{3.4}$$

The equations of motion (Eq. 3.1) are uncoupled by the transformation

$$\underline{r}(t) = \sum_{n=1}^{2N} \underline{\phi}_n Y_n(t)$$
(3.5)

provided the damping matrix <u>c</u> satisfies certain restrictions [6]. In Eq. 3.5, $Y_n(t)$ is the generalized coordinate and ϕ_n the mode shape for the nth natural mode of vibration. The uncoupled equation for the nth mode of vibration is

$$M_{n}\ddot{Y}_{n} + C_{n}\dot{Y}_{n} + K_{n}Y_{n} = -L_{n}^{X}a_{g}^{X}(t) - L_{n}^{Y}a_{g}^{Y}(t)$$
(3.6)

where

$$M_n = \Phi_n^T \underline{m} \Phi_n$$
, $C_n = \xi_n (2M_n \omega_n)$, $K_n = \omega_n^2 M_n$, ω_n and ξ_n are the natural

circular frequency and damping ratio for the nth mode, $L_n^x = \phi_n^T \underline{\mathbf{ml}}^x$, and $L_n^y = \phi_n^T \underline{\mathbf{ml}}^y$.

Equation 3.6 may be solved for $Y_n(t)$ by a step-by-step integration method [30]. After computation has been repeated for all modes, the nodal displacement vector $\underline{r}(t)$ can be obtained from Eq. 3.5. In practice, it is generally sufficient to solve the equations of motion only for the lower few modes, because the contributions of the higher modes to the total response are small.

The stresses $\frac{\sigma}{p}(t)$ in finite element p at any instant of time are related to the nodal displacement $\underline{r}_{p}(t)$ for that element by

$$\sigma_{p}(t) = T_{p} r_{p}(t)$$
 (3.7)

where the stress transformation matrix $\frac{T}{p}$ is based on the interpolation functions for the element as well as its elastic properties. The stresses throughout the dam at any instant of time are determined from the nodal point displacements by application of the above transformation to each finite element.

The initial stresses, before the earthquake, should be added to the stresses due to earthquake excitation, determined by the procedures presented in the preceding sections, to obtain the total stresses in the dam. Excluding the temperature and creep stresses in concrete, only the gravity loads need to be considered. The equations of static equilibrium are formulated:

$$K \mathbf{r} = \mathbf{R} \tag{3.8}$$

where <u>R</u> is the vector of static loads due to the weight of the dam and hydrostatic pressures. Solution of these algebraic equations results in the displacement vector <u>r</u>. Static stresses are then determined from the displacements by applying Eq. 3.7 to each finite element.

Analytical predictions, based on the procedure summarized above, for the perfomance of Koyna Dam correlated well with the damage experienced by the dam during the Koyna earthquake of December 11, 1976 [2].

3.3.2 Frequency Domain Analysis

An alternative approach to solving the modal equations of motion (Eq. 3.6) exists in which, instead of step-by-step integration in the time domain, the complex frequency responses are superposed in the frequency domain utilizing the Fourier integral [30]. The complex frequency response function $\bar{z}(\omega)$ for a response quantity z(t) has the property that, when the excitation is the real part of $e^{i\omega t}$, then the response is the real part of $\bar{z}(\omega)e^{i\omega t}$.

The response in the nth mode of vibration due to the horizontal ground motion will be denoted by Y_n^x , and that due to the vertical ground motion by Y_n^y . The response to excitation $a_g^\ell(t) = e^{i\omega t}$ can then be expressed as

$$Y_n^{\ell} = \overline{Y}_n^{\ell} e^{i\omega t}$$
 where $\ell = x$ or y.

Substitution in Eq. 3.6 leads to

$$(-\omega^{2}M_{n} + i\omega C_{n} + K_{n}) \overline{Y}_{n}^{\ell}(\omega) = -L_{n}^{\ell}; \ \ell = x, y \qquad (3.9)$$

from which

$$\tilde{Y}_{n}^{\ell}(\omega) = \frac{-L_{n}^{\ell}}{-\omega^{2}M_{n} + i\omega C_{n} + K_{n}}, \ \ell = x, y \qquad (3.10)$$

If the system has constant hysteretic damping instead of viscous damping, starting from Eq. 3.3, it can be shown that Eqs. 3.9 and 3.10 become

$$[-\omega^{2}M_{n} + (1 + i\eta)K_{n}] \overline{y}_{n}^{\ell}(\omega) = -L_{n}^{\ell}; \ \ell = x, y \qquad (3.11)$$

and

$$\bar{\bar{x}}_{n}^{\ell}(\omega) = \frac{-L_{n}^{\ell}}{-\omega^{2}M_{n} + (1 + i\eta)K_{n}}; \ \ell = x, y \qquad (3.12)$$

The response to arbitrary ground motion is the superpostion of responses to individual harmonic components of the excitation through the Fourier integral:

$$\bar{\mathbf{Y}}_{n}^{\ell}(t) = \frac{1}{2\pi} \int_{-\infty}^{\infty} \bar{\mathbf{Y}}_{n}^{\ell}(\omega) A_{g}^{\ell}(\omega) e^{i\omega t} d\omega; \ \ell = x, y \qquad (3.13)$$

- 15 -

in which $A_q^{\ell}(\omega)$ is the Fourier transform of $a_q^{\ell}(t)$:

$$A_{g}^{\ell}(\omega) = \int_{0}^{d} a_{g}^{\ell}(t) e^{-i\omega t} dt; \quad \ell = x, y \qquad (3.14)$$

where d is the duration of the ground motion.

The response in the nth vibration mode due to simultaneous action of the horizontal and vertical components of ground motion is

$$Y_{n}(t) = Y_{n}^{x}(t) + Y_{n}^{y}(t)$$
 (3.15)

Repeating this procedure for all the necessary values of n, the displacement responses may be obtained by superpostion of the modal responses (Eq. 3.5) and the stress responses by calculating the stresses associated with these displacements (Eq. 3.7). The initial stresses are added to determine the total stresses.

Until the development of the Fast Fourier Transform (FFT) algorithm [31], numerical evaluation of integrals such as those in Eqs. 3.13 and 3.14 required prohibitive amounts of computer time and the errors in the results could not be predicted accurately. As a result, step-by-step integration in the time domain has conventionally been used for response analysis. With the FFT algorithm, integrals of Eqs. 3.13 and 3.14 can be evaluated accurately and efficiently. As a result, the frequency domain approach can now be employed advantageously for analysis of dynamic response of structures. It provides an alternative approach for analysis of dams without water. As will be seen in Chapters 4,5, and 6, the analysis is best formulated by the frequency domain approach when dam-water and dam-foundation interaction effects are included. 4. ANALYSIS OF GRAVITY DAMS INCLUDING DAM-WATER INTERACTION

4.1 Introductory Note

Summarized in this chapter is the procedure presented earlier [19], for including the effects of dam-water interaction in earthquake response analyses of concrete gravity dams. The concept underlying this procedure is to treat the impounded water and the dam as two substructures of the total system. With this approach, the dam may be idealized by the finite element method, which has the ability to represent systems of arbitrary geometry and material properties (see Chapter 3). At the same time, the impounded water may be treated as a continuum, an approach which is ideally suited to the simple idealized geometry but great upstream extent of the impounded water. The equations of motion for the two substructures are coupled by including the forces of interaction between the water and the face of the dam.

4.2 Governing Equations

4.2.1 Substructure 1: Dam

A gravity dam supported on a rigid foundation and storing water to a given depth is shown in Fig. 4.1. Including hydrodynamic effects, the equations of motion are, for a dam monolith idealized as a planar two-dimensional finite element system (Chapter 3), subjected to horizontal and vertical components of ground motion:

$$\underline{\mathbf{m}} \ \underline{\ddot{\mathbf{r}}} + \underline{\mathbf{c}} \ \underline{\ddot{\mathbf{r}}} + \underline{\mathbf{k}} \ \underline{\mathbf{r}} = -\underline{\mathbf{ml}}^{\mathbf{X}} \ \mathbf{a}_{g}^{\mathbf{X}}(t) - \underline{\mathbf{ml}}^{\mathbf{Y}} \ \mathbf{a}_{g}^{\mathbf{Y}}(t) + \underline{\mathbf{R}}_{h}(t)$$
(4.1)

This is identical to Eq. 3.1 except for the inclusion of the hydrodynamic loads $\underline{R}_{h}(t)$ on the upstream face of the dam.

Whereas Eq. 4.1 is valid for dams of arbitrary geometry, it is assumed that for the purpose of defining hydrodynamic effects, and only for this purpose, the upstream face of the dam is vertical. Considering that the upstream face of concrete gravity dams is vertical or almost vertical for most of the height, this is a reasonable assumption. With this assumption and because the hydrodynamic loads act only on the upstream face, only those loads in $\frac{R}{h}(t)$ that correspond to the x-degrees of freedom of the nodal points on the upstream face are non-zero. The sub-vector of the non-zero loads is denoted by $\frac{R_{h}^{xf}}{h}(t)$. The superscript "xf" indicates that only the

- 17 -



FIG. 4.1 CONCRETE GRAVITY DAM WITH IMPOUNDED WATER. THE HYDRODYNAMIC PRESSURES ON THE UPSTREAM FACE OF THE DAM ARE $p_c(y,t)$

x-component of the loads acting on the face of the dam are included in the vector.

The procedure for analyzing the response of dams without including the effects of water, presented in Chapter 3, utilized the mode-superpostion concept. Although classical natural modes of vibration do not exist for the dam when hydrodynamic effects are included, expansion of displacements, including hydrodynamic effects, in terms of natural modes of vibration with-out water can still be used advantageously. Thus,

$$r(t) \simeq \sum_{n=1}^{J} \phi_n Y_n(t)$$
 (4.2)

The vectors ϕ_n , n = 1,2...2N are linearly independent and span the vector space of dimension 2N. Thus, the expansion of Eq. 4.2 is exact if the contributions of all the 2N natural modes of vibration are included. However, even when hydrodynamic effects are included, the contributions of the lower modes of vibration are expected to be more significant and relatively few terms, e.g. J << 2N, would suffice in Eq. 4.2.

Substituting Eq. 4.2 into Eq. 4.1 and utilizing the orthogonality property of mode shapes, the equation governing the generalized displacement Y_n (t) in the nth mode is

$$M_{n}\ddot{Y}_{n} + C_{n}\dot{Y}_{n} + K_{n}Y_{n} = -L_{n}^{X}a_{g}^{X}(t) - L_{n}^{Y}a_{g}^{Y}(t) + R_{hn}(t)$$
(4.3)

This is the same as Eq. 3.6, except for the additional term $R_{hn}(t) = \frac{\Phi_{n-h}^{TR}(t)}{\Phi_{n-h}^{TR}(t)} = \frac{\Phi_{n-h}^{xf}}{\Phi_{n-h}^{xf}(t)} \frac{R_{h}^{xf}(t)}{\Phi_{n-h}^{xf}(t)}$ in which the vector Φ_{n-h}^{xf} consists of the x-components of the displacements of the nodal points on the upstream face of the dam.

4.2.2 Substructure 2: Fluid Domain

The motion of the dam-water system is considered to be two-dimensional, i.e., it is the same for any vertical plane perpendicular to the axis of the dam. Assuming water to be linearly compressible and neglecting its internal viscosity, the small amplitude, irrotational motion of the fluid is governed by the two-dimensional wave equation

$$\frac{\partial^2 p}{\partial x^2} + \frac{\partial^2 p}{\partial y^2} = \frac{1}{C^2} \frac{\partial^2 p}{\partial t^2}$$
(4.4)

- 19 -

in which p(x,y,t) is the hydrodynamic pressure (in excess of the hydrostatic pressure) and C is the velocity of sound in water. The hydrodynamic pressure acting on the upstream face of the dam is $p_{c}(y,t) \equiv p(o,y,t)$.

4.3 Response Analysis for Harmonic Horizontal Ground Motion

Specializing the equations of motion of the dam in terms of generalized modal displacements (Eq. 4.3) for the case when the ground motion acts only in the horizontal (x) direction leads to

$$M_{n}\ddot{Y}_{n}^{X} + C_{n}\dot{Y}_{n}^{X} + K_{n}Y_{n}^{X} = -L_{n}^{X}a_{g}^{X}(t) + R_{hn}^{X}(t)$$
(4.5)

where $L_n^{\mathbf{X}} = \Phi_n^{\mathbf{T}} \underline{\mathbf{ml}}^{\mathbf{X}}$ and $\mathbf{R}_{hn}^{\mathbf{X}}(t) = \{\Phi_n^{\mathbf{X}\mathbf{f}}\}^{\mathbf{T}} \underline{\mathbf{R}}_h^{\mathbf{X}\mathbf{f}}(t)$. The vector $\underline{\mathbf{R}}_h^{\mathbf{X}\mathbf{f}}(t)$ consists of forces, associated with hydrodynamic pressures acting in the horizontal direction, at the nodal points on the vertical upstream face of the dam. These hydrodynamic pressures are governed by the two-dimensional wave equation (Eq. 4.4) and the following boundary conditions:

•Pressure at the free surface is zero, implying that the effects of waves at the free surface are ignored;

•Vertical motion at the base of the reservoir is zero; and

•Horizontal component of motion of the fluid boundary x = 0 is the same as the horizontal motion of the upstream face of the dam.

The responses to harmonic ground acceleration $a_g^{x}(t) = e^{i\omega t}$ can be expressed as follows:

Generalized Displacements, $Y_n^x(t) = \overline{Y}_n^x(\omega) e^{i\omega t}$ (4.6a)

Generalized Accelerations, $\ddot{Y}_{n}^{x}(t) = \ddot{Y}_{n}^{x}(\omega)e^{i\omega t}$ (4.6b)

Horizontal accelerations at the upstream face,

$$\ddot{\mathbf{u}}(t) = \{\hat{\mathbf{1}} + \sum_{j=1}^{J} \phi_{j}^{\mathbf{x}f} \ \ddot{\vec{\mathbf{y}}}_{j}^{\mathbf{x}}(\omega)\} e^{i\omega t}$$
(4.6c)

in which 1 is a vector with all elements equal to unity.

Hydrodynamic pressures,
$$p_c^{x}(y,t) = \overline{p}_c^{x}(y,\omega)e^{i\omega t}$$
 (4.6d)

Strictly for purposes of notational convenience, the vector $\underline{\phi}_{j}^{\text{xf}}$ is replaced by its continuous analog function $\underline{\phi}_{j}(y)$. The total horizontal acceleration

of the fluid boundary at x = 0, the upstream face of the dam, may now be represented by the function

$$\ddot{u}(0,y,t) = \{1 + \sum_{j=1}^{J} \phi_{j}(y) \quad \ddot{\vec{y}}_{j}^{x}(\omega) \} e^{i\omega t}$$

$$(4.7)$$

With reference to Fig. 4.1, the boundary conditons for the wave equation are

$$p(\mathbf{x},\mathbf{H},\mathbf{t}) = 0$$

$$\frac{\partial p}{\partial \mathbf{y}}(\mathbf{x},0,\mathbf{t}) = 0$$

$$\frac{\partial p}{\partial \mathbf{x}}(0,\mathbf{y},\mathbf{t}) = -\frac{\mathbf{w}}{g}\left\{1 + \sum_{j=1}^{J} \phi_{j}(\mathbf{y}) \ \tilde{\mathbf{y}}_{j}^{\mathbf{x}}(\omega)\right\} e^{i\omega t}$$

$$(4.8)$$

Because the governing equation as well as the boundary conditions are linear, the principal of superpostion applies. The complex frequency response function for $p_c^{\mathbf{x}}(y,t)$ can therefore be expressed as

$$\vec{p}_{c}^{x}(y,\omega) = \vec{p}_{0}^{x}(y,\omega) + \sum_{j=1}^{J} \vec{y}_{j}^{x}(\omega) \vec{p}_{j}(y,\omega)$$

$$(4.9)$$

where $p_0^x(y,t) = \bar{p}_0^x(y,\omega)e^{i\omega t}$ is the solution of the wave equation at x = 0 (Eq. 4.4) for the following boundary conditions:

$$p(\mathbf{x},\mathbf{H},\mathbf{t}) = 0$$

$$\frac{\partial p}{\partial y}(\mathbf{x},0,\mathbf{t}) = 0$$

$$\frac{\partial p}{\partial x}(0,y,\mathbf{t}) = -\frac{w}{g}e^{i\omega t}$$
(4.10)

and $p_j(y,t) = \bar{p}_j(y,\omega)e^{i\omega t}$ is the solution of the wave equation at x = 0 for the following boundary conditions:

$$p(x,H,t) = 0$$

$$\frac{\partial p}{\partial y}(x,0,t) = 0$$

$$\frac{\partial p}{\partial x}(0,y,t) = -\frac{w}{g}\phi_{j}(y)e^{i\omega t}$$
(4.11)

Solutions of Eq. 4.4 for the above two sets of boundary conditons are

presented in Eqs. 4.12 and 4.13

$$\bar{\mathbf{p}}_{0}^{\mathbf{x}}(\mathbf{y},\omega) = \frac{4w}{\pi g} \sum_{\ell=1}^{\infty} \frac{(-1)^{\ell}}{(2\ell-1)\sqrt{\lambda_{\ell}^{2} - \omega^{2}/C^{2}}} \cos \lambda_{\ell} \mathbf{y}$$
(4.12)

$$\bar{p}_{j}(y,\omega) = -\frac{2w}{gH} \sum_{\ell=1}^{\infty} \frac{I_{j\ell}}{\sqrt{\lambda_{\ell}^{2} - \omega^{2}/C^{2}}} \cos \lambda_{\ell} y \qquad (4.13)$$

in which $\lambda_{\ell} = (2\ell-1) \pi/2H$ and $I_{j\ell} = \int_{0}^{H} \phi_{j}(y) \cos \lambda_{\ell} y \, dy$. The complex frequency response function $\bar{p}_{0}^{x}(y,\omega)$ is for the hydrodynamic pressures on the dam when the excitation to the fluid domain is the horizontal acceleration of the ground and the dam is rigid. The corresponding function is $\bar{p}_{j}^{x}(y,\omega)$ when the excitation is the acceleration of the dam in the jth mode of vibration and there is no motion of the base of the reservoir.

The integrals $I_{j\ell}$ are computed as a summation using the elements of the vector $\underline{\phi}_{j}^{xf}$ instead of the continuous function $\phi_{j}(y)$ which was introduced only for notational convenience. The nodal force vectors are $\underline{R}_{0}^{x}(t) = \overline{R}_{0}^{x}(\omega)e^{i\omega t}$ and $\underline{R}_{j}(t) = \overline{R}_{j}(\omega)e^{i\omega t}$, where $\overline{R}_{0}^{x}(\omega)$ and $\overline{R}_{j}^{x}(\omega)$ are the static equivalents of the corresponding pressure functions $\overline{p}_{0}^{x}(y,\omega)$ and $\overline{p}_{j}(y,\omega)$, respectively. They may be computed directly by using the principle of virtual work with the displacements between nodal points defined by the finite element interpolation functions. The complex frequency response function for the total force vector $\underline{R}_{h}^{xf}(t)$ is, from Eq. 4.9,

$$\bar{\bar{R}}_{h}^{xf}(\omega) = \bar{\bar{R}}_{0}^{x}(\omega) + \sum_{j=1}^{J} \bar{\bar{Y}}_{j}^{x}(\omega) \bar{\bar{R}}_{j}(\omega)$$
(4.14)

The hydrodynamic forces on the dam have thus been expressed in terms of the unknown generalized coordinate responses $\frac{\ddot{z}_{i}^{x}}{\dot{z}_{i}}(\omega)$.

With the aid of Eq. 4.14 and substituting $\tilde{\tilde{Y}}_{j}^{x}(\omega) = -\omega^{2} \tilde{\tilde{Y}}_{j}^{x}(\omega)$, Eq. 4.5 can be written as follows for the excitation $a_{q}^{x}(t) = e^{i\omega t}$:

$$\left[-\omega^{2}M_{n} + i\omega C_{n} + K_{n}\right] \bar{Y}_{n}^{x}(\omega) = -L_{n}^{x} + \left\{\phi_{n}^{xf}\right\}^{T} \left\{\bar{\bar{R}}_{0}^{x}(\omega) - \omega^{2}\sum_{j=1}^{J} \bar{\bar{Y}}_{j}^{x}(\omega)\bar{\bar{R}}_{j}(\omega)\right\}$$
(4.15)

- 22 -
The set of equations 4.15 for $n = 1, 2 \dots J$ may be rearranged and expressed in matrix form as

$$\begin{bmatrix} \mathbf{S}_{11}(\omega) & \mathbf{S}_{12}(\omega) & \cdots & \mathbf{S}_{1J}(\omega) \\ \mathbf{S}_{21}(\omega) & \mathbf{S}_{22}(\omega) & \cdots & \mathbf{S}_{2J}(\omega) \\ \vdots & & & \\ \mathbf{S}_{J1}(\omega) & \mathbf{S}_{J2}(\omega) & \cdots & \mathbf{S}_{JJ}(\omega) \end{bmatrix} \begin{bmatrix} \overline{\mathbf{Y}}_{1}^{\mathbf{X}}(\omega) \\ \overline{\mathbf{Y}}_{2}^{\mathbf{X}}(\omega) \\ \vdots \\ \overline{\mathbf{Y}}_{J}^{\mathbf{X}}(\omega) \end{bmatrix} = \begin{cases} \mathbf{L}_{1}^{\mathbf{X}}(\omega) \\ \mathbf{L}_{2}^{\mathbf{X}}(\omega) \\ \vdots \\ \vdots \\ \mathbf{L}_{J}^{\mathbf{X}}(\omega) \end{cases}$$
(4.16)

in which

$$S_{nj}(\omega) = \omega^{2} \{ \underline{\phi}_{n}^{xf} \}^{T} \overline{\underline{R}}_{j}(\omega) \text{ for } n \neq j$$

$$S_{nn}(\omega) = [-\omega^{2}M_{n} + i\omega C_{n} + K_{n}] + \omega^{2} \{ \underline{\phi}_{n}^{xf} \}^{T} \overline{\underline{R}}_{n}(\omega) \} \begin{cases} n = 1, 2 \dots J \\ j = 1, 2 \dots J \end{cases}$$

$$L_{n}^{x}(\omega) = -L_{n}^{x} + \{ \underline{\phi}_{n}^{xf} \}^{T} \overline{\underline{R}}_{0}^{x}(\omega) \qquad (4.17)$$

The coefficient matrix $\underline{S}(\omega)$ in Eq. 4.17 is a frequency dependent matrix which relates the generalized displacement vector $\underline{\underline{Y}}^{\mathbf{X}}(\omega)$ with the corresponding generalized load vector $\underline{\underline{L}}^{\mathbf{X}}(\omega)$. Unlike classical modal analysis, the matrix $\underline{\underline{S}}(\omega)$ is not diagonal because the vectors $\underline{\Phi}_{\mathbf{n}}$ are not the natural vibration modes of the dam-fluid system; they simply are the natural vibration modes of the dam without water. It can be shown that $\underline{S}(\omega)$ is a symmetric matrix.

With one change, Eqs. 4.16 and 4.17 are valid even if damping for the dam is idealized as constant hysteretic damping instead of viscous damping. The expression for the diagonal elements of $S(\omega)$ changes to

$$\mathbf{S}_{nn}(\omega) = \left[-\omega^2 \mathbf{M}_n + (\mathbf{1} + \mathbf{i}\eta) \mathbf{K}_n\right] + \omega^2 \left\{\underline{\phi}_n^{\mathbf{x}\mathbf{f}}\right\}^{\mathrm{T}} \overline{\underline{\mathbf{R}}}_n(\omega)$$
(4.18)

Solution of Eq. 4.16 for all values of excitation frequency would give the complete frequency response functions for all the generalized coordinates $\bar{Y}_{n}^{x}(\omega)$, n = 1, 2 ... J. The frequency responses for generalized accelerations may be obtained from the generalized displacements as

$$\bar{\tilde{Y}}_{n}^{\mathbf{x}}(\omega) = -\omega^{2} \bar{\tilde{Y}}_{n}^{\mathbf{x}}(\omega)$$
(4.19)

- 23 -

The complex frequency response functions for the vectors of nodal displacements and accelerations relative to the ground motion are obtained by superposition of the contributions of the various modes of vibration (Eq. 4.2):

$$\vec{\underline{r}}^{\mathbf{X}}(\omega) \simeq \sum_{n=1}^{J} \phi_{n} \vec{\underline{r}}_{n}^{\mathbf{X}}(\omega)$$
(4.20)

$$\overline{\underline{\vec{x}}}_{\underline{r}}^{\mathbf{x}}(\omega) \simeq -\omega^{2} \sum_{n=1}^{J} \underline{\phi}_{n} \, \overline{\vec{\mathbf{y}}}_{n}^{\mathbf{x}}(\omega)$$
(4.21)

4.4 Response Analysis for Harmonic Vertical Ground Motion

Specializing the equations of motion of the dam in terms of generalized modal displacements (Eq. 4.3) for the case when the ground motion acts only in the vertical (y) direction leads to:

$$M_{n}\ddot{Y}_{n}^{Y} + C_{n}\dot{Y}_{n}^{Y} + K_{n}Y_{n}^{Y} = -L_{n}^{Y}a_{g}^{Y}(t) + R_{hn}^{Y}(t)$$
(4.22)

where $L_n^{y} = \phi_n^{T} \underline{m} \underline{l}^{y}$ and $R_{hn}^{y}(t) = \{\phi_n^{xf}\}^{T} \underline{R}_{h}^{xf}(t)$. The vector $\underline{R}_{h}^{xf}(t)$ consists of forces, associated with hydrodynamic pressures acting in the horizontal direction, at the nodal points on the vertical upstream face of the dam. These hydrodynamic pressures are governed by the two-dimensional wave equation (Eq. 4.4) together with the following boundary conditions (see Fig. 4.1):

•Pressure at the free surface is zero, implying that the effects of waves at the free surface are ignored;

•Vertical component of motion of the fluid boundary y = 0, the base of the reservoir, is prescribed by the vertical component of ground acceleration; and

•Horizontal component of motion of the fluid boundary x = 0 is the same as the horizontal motion of the upstream face of the dam.

The responses to harmonic ground acceleration $a_g^Y(t) = e^{i\omega t}$ can be expressed as follows:

Generalized displacements,
$$Y_n^{Y}(t) = \overline{Y}_n^{Y}(\omega)e^{i\omega t}$$
 (4.23a)

Generalized accelerations, $\ddot{Y}_{n}^{Y}(t) = \bar{\ddot{Y}}_{n}^{Y}(\omega) e^{i\omega t}$ (4.23b)

Horizontal accelerations at the upstream face,

$$\underbrace{\ddot{u}}(t) = \begin{cases} J \\ \sum \phi_{j=1}^{xf} \quad \overline{\ddot{y}}_{j}^{Y}(\omega) \end{cases} \dot{e}^{i\omega t}$$
(4.23c)

Hydrodynamic pressures, $p_{c}^{Y}(y,t) = \bar{p}_{c}^{Y}(y,\omega)e^{i\omega t}$ (4.23d)

Replacing the vector ϕ_j^{xf} by its continuous analog function, $\phi_j(y)$, as before, the horizontal acceleration of the fluid boundary at x = 0, the upstream face of the dam, is represented by the function:

$$\ddot{u}(0,y,t) = \begin{cases} J \\ \Sigma \phi_{j}(y) \ddot{Y}_{j}^{y}(\omega) \end{cases} e^{i\omega t}$$
(4.24)

Then the boundary conditions for the wave equation are

$$P(\mathbf{x}, \mathbf{H}, \mathbf{t}) = 0$$

$$\frac{\partial p}{\partial \mathbf{y}}(\mathbf{x}, 0, \mathbf{t}) = -\frac{\mathbf{w}}{\mathbf{g}} e^{\mathbf{i}\omega\mathbf{t}}$$

$$\frac{\partial p}{\partial \mathbf{x}}(0, \mathbf{y}, \mathbf{t}) = -\frac{\mathbf{w}}{\mathbf{g}} \left\{ \sum_{j=1}^{J} \phi_{j}(\mathbf{y}) \ \overline{\mathbf{y}}_{j}^{\mathbf{y}}(\omega) \right\} e^{\mathbf{i}\omega\mathbf{t}}$$

$$(4.25)$$

Because the governing equations as well as boundary conditions are linear, the principle of superpositions applies. The complex frequency response function for $p_C^Y(y,t)$ can therefore be expressed as

$$\overline{p}_{c}^{Y}(y,\omega) = \overline{p}_{0}^{Y}(y,\omega) + \sum_{j=1}^{J} \overline{\tilde{y}}_{j}^{Y}(\omega) \overline{p}_{j}(y,\omega)$$

$$(4.26)$$

where $p_0^{Y}(y,t) = \bar{p}_0^{Y}(y,\omega)e^{i\omega t}$ is the solution of the wave equation (Eq. 4.4) for the following boundary conditions

$$p(\mathbf{x}, \mathbf{H}, \mathbf{t}) = 0$$

$$\frac{\partial p}{\partial \mathbf{y}}(\mathbf{x}, 0, \mathbf{t}) = -\frac{\mathbf{w}}{\mathbf{g}} e^{\mathbf{i}\omega\mathbf{t}}$$

$$\frac{\partial p}{\partial \mathbf{x}}(0, \mathbf{y}, \mathbf{t}) = 0$$

$$(4.27)$$

- 25 -

and $p_j(y,t) = \bar{p}_j(y,\omega)e^{i\omega t}$ is the solution of the wave equation (Eq. 4.4) at x = 0 for the following boundary conditions:

$$p(x,H,t) = 0$$

$$\frac{\partial p}{\partial y}(x,0,t) = 0$$

$$\frac{\partial p}{\partial x}(0,y,t) = -\frac{w}{g}\phi_{j}(y)e^{i\omega t}$$

$$(4.28)$$

Solution of Eq. 4.4 for the boundary conditions of Eq. 4.27 is

$$\bar{p}_{0}^{Y}(y,\omega) = \frac{w}{g} \frac{C}{\omega} - \frac{(1+\alpha) \sin \frac{\omega(H-y)}{C}}{(1+\alpha) \cos \frac{\omega H}{C} + i(1-\alpha) \sin \frac{\omega H}{C}}$$
(4.29)

In this equation, the reflection coefficient

$$\alpha = \frac{\binom{C_{r}w_{r}}{C_{w}}}{\binom{C_{r}w_{r}}{C_{w}}} + 1}$$
(4.30)

where w_r and C_r are the unit weight and P-wave velocity for the rock material at the bottom of the reservoir. For rigid rock, $\alpha = 1$ indicating that hydrodynamic pressure waves are completely reflected at the bottom of the reservoir. For deformable rock, $\alpha < 1$ implying that hydrodynamic pressure waves impinging at the reservoir bottom are partially reflected back in the fluid domain and partially refracted into rock. The complex frequency response function $\bar{p}_0^{Y}(y,\omega)$ is for the hydrodynamic pressures on the dam when the excitation to the fluid domain is the vertical acceleration of the ground and the dam is rigid.

The solution of Eq. 4.4 for the boundary conditions of Eq. 4.28 has been presented earlier in Eq. 4.13. As mentioned before, $\bar{p}_j(y,\omega)$ is the complex frequency response function for the hydrodynamic pressure when the excitation is the acceleration of the dam in the jth mode of vibration and there is no motion of the base.

The nodal force vectors are $\underline{R}_{0}^{Y}(t) = \overline{\underline{R}}_{0}^{Y}(\omega) e^{i\omega t}$ and $\underline{R}_{j}(t) = \overline{\underline{R}}_{j}(\omega) e^{i\omega t}$, where $\overline{\underline{R}}_{0}^{Y}(\omega)$ and $\overline{\underline{R}}_{j}(\omega)$ are the static equivalents of the corresponding pressure functions $\overline{p}_{0}^{Y}(y,\omega)$ and $\overline{p}_{j}(y,\omega)$ respectively. The complex frequency response function for the total force vector $\underline{R}_{h}^{xf}(t)$ is, from Eq. 4.25,

$$\bar{\mathbf{R}}_{\mathbf{h}}^{\mathbf{xf}}(\omega) = \bar{\mathbf{R}}_{\mathbf{0}}^{\mathbf{y}}(\omega) + \sum_{\mathbf{j=1}}^{\mathbf{J}} \bar{\mathbf{y}}_{\mathbf{j}}^{\mathbf{y}}(\omega) \bar{\mathbf{R}}_{\mathbf{j}}(\omega)$$
(4.31)

With the aid of Eq. 4.31 and substituting $\overline{Y}_{j}^{Y}(\omega) = -\omega^{2} \overline{Y}_{-j}^{Y}(\omega)$, Eq. 4.22 can be written as follows for the excitation $a_{q}^{Y}(t) = e^{i\omega t}$:

$$\left[-\omega^{2}M_{n} + i\omega C_{n} + \kappa_{n}\right] \overline{\tilde{Y}}_{n}^{Y}(\omega) = -L_{n}^{Y} + \left\{\phi_{n}^{xf}\right\}^{T} \left\{\overline{\tilde{R}}_{0}^{Y}(\omega) - \omega_{j=1}^{2} \overline{\tilde{Y}}_{j}^{Y}(\omega) \overline{\tilde{R}}_{j}(\omega)\right\} \quad (4.32)$$

The set of Eqs. 4.32 for n = 1, 2... J after rearranging may be expressed in matrix form as

$$\underline{s}(\omega) \, \underline{\tilde{Y}}^{Y}(\omega) = \underline{L}^{Y}(\omega) \tag{4.33}$$

in which the coefficient matrix $\underline{S}(\omega)$ is identical to the case when the excitation was the horizontal component of ground motion (Eqs. 4.16-4.18), and

$$L_{n}^{Y}(\omega) = -L_{n}^{Y}(\omega) + \{\phi_{n}^{xf}\}^{T} \bar{R}_{0}^{Y}(\omega), n = 1, 2, ... J$$
(4.34)

Solution of Eq. 4.33 for all values of excitation frequency would give the complex frequency response functions for generalized displacements $\bar{Y}_{n}^{Y}(\omega)$. The frequency responses for generalized accelerations

$$\vec{\tilde{Y}}_{n}^{Y}(\omega) = -\omega^{2}\vec{\tilde{Y}}_{n}^{Y}(\omega)$$
(4.35)

The complex frequency response functions for the vectors of nodal displacements and accelerations relative to the ground motion are given by expressions similar to Eqs. 4.20 and 4.21:

$$\underline{\bar{r}}^{Y}(\omega) \simeq \sum_{n=1}^{J} \Phi_{n} \overline{\bar{Y}}_{n}^{Y}(\omega)$$
(4.36)

$$\overline{\vec{r}}^{Y}(\omega) \simeq -\omega^{2} \sum_{n=1}^{J} \phi_{n} \ \overline{\vec{Y}}^{Y}_{n}(\omega)$$
(4.37)

4.5 Response to Arbitrary Ground Motion

Once the complex frequency response functions $\bar{Y}_{n}^{x}(\omega)$ and $\bar{Y}_{n}^{y}(\omega)$, n = 1, 2,...J, have been determined by solving Eqs. 4.16 and 4.33 for an appropriate range of excitation frequency ω , the responses to arbitrary ground motion can be obtained as the superposition of responses to individual harmonic components of the excitation through the Fourier integral:

$$\overline{Y}_{n}^{\ell}(t) = \frac{1}{2\pi} \int_{-\infty}^{\infty} \int_{n}^{\infty} \overline{Y}_{n}^{\ell}(\omega) A_{g}^{\ell}(\omega) e^{i\omega t} d\omega; \ \ell = x, y$$
(4.38)

in which $A_g^{\ell}(\omega)$ is the Fourier Transform of $a_g^{\ell}(t)$:

$$A_{g}^{\ell}(\omega) = \int_{0}^{d} a_{g}^{\ell}(t) e^{-i\omega t} dt; \ \ell = x, y \qquad (4.39)$$

where d is the duration of ground motion.

The combined response $Y_n(t)$ to horizontal and vertical components of ground motion acting simultaneously is

$$Y_{n}(t) = Y_{n}^{x}(t) + Y_{n}^{y}(t)$$
 (4.40)

Repeating this procedure for all the necessary values of n, the displacement response is obtained by transforming back from generalized to nodal point coordinates

$$\underline{r}(t) \simeq \sum_{n=1}^{J} \underline{\phi}_n Y_n(t) \qquad (4.41)$$

The stresses $\underline{\sigma}_p$ (t) in finite element p at any instant of time are replated to the nodal displacements \underline{r}_p (t) for that element by

$$\underline{\sigma}_{p}(t) = \underline{T}_{p}\underline{r}_{p}(t) \qquad (4.42)$$

where \underline{T}_{-p} is the stress transformation matrix for the finite element p. At any instant of time, the stresses throughout the dam are determined from the nodal point displacements by applying the transformation of Eq. 4.42 to each finite element.

The total stresses are the sum of the dynamic stresses determined above and the intial static stresses due to the weight of the dam and hydrostatic pressures (Sec. 3.3).

A digital computer program was developed [20] to implement the analysis summarized in this chapter. Studies using this computer program have led to a better understanding of hydrodynamic effects in earthquake response of concrete gravity dams [18]. 5. ANALYSIS OF DAMS INCLUDING DAM-FOUNDATION INTERACTION

5.1 Introductory Note

This chapter summarizes the procedure presented earlier [24,25] for including the effects of structure-foundation interaction, i.e., interaction between the structure and underlying soil or rock, in earthquake response analyses of concrete gravity dams without water. This procedure to include the dam-foundation interaction effects is based on the same basic substructure concept utilized to include the dam-water interaction. The soil or rock region and the dam are considered as two substructures of the combined system. The dam may be idealized by the finite element method, which has the ability to represent systems of arbitrary geometry and material properties (see Chapters 3 and 4). The soil or rock region may be idealized as either a finite element system or as a continuum, for example, as a viscoelastic halfspace. The halfspace idealization permits accurate modelling of sites where essentially similar rocks extend to large depths. For sites where layers of soil or soft rock are underlain by harder rock at shallow depth, finite element idealization of the soil region would be appropriate. The governing equations for the two substructures are combined by imposing equilibrium and compatibility requirements at the base of the dam, which is the structure-foundation interface.

5.2 Frequency Domain Equations in Nodal Point Coordinates

5.2.1 Substructure 1: Dam

The equations of motion for the dam idealized as a planar, two-dimensional finite element system (Chapter 3) are:

$$\underline{m}_{c-c} \ddot{r} + \underline{c}_{c-c} \dot{r} + \underline{k}_{c-c} r = -\underline{m}_{c-c} \underline{l}_{c-c}^{X} a_{g}^{X}(t) - \underline{m}_{c-c} \underline{l}_{c-c}^{Y} a_{g}^{Y}(t) + \underline{R}_{c}(t)$$
(5.1)

in which $\underline{\mathbf{m}}_{c}$, $\underline{\mathbf{c}}_{c}$, and $\underline{\mathbf{k}}_{c}$ are the mass, damping, and stiffness matrices for the finite element system; $\underline{\mathbf{r}}_{c}$ is the vector of nodal point displacements relative to the free-field earthquake displacement at the base (Fig. 5.1):

$$\underline{r}_{c}^{T} = \langle r_{1}^{X} r_{1}^{Y} r_{2}^{X} r_{2}^{Y} \dots r_{n}^{X} r_{n}^{Y} \dots r_{N+N_{b}}^{X} r_{N+N_{b}}^{Y} \rangle$$

in which $r_n^{\boldsymbol{x}}$ and $r_n^{\boldsymbol{y}}$ are the \boldsymbol{x} and \boldsymbol{y} components of the displacement of nodal

- 29 -

point n; N is the number of nodal points above the base, N_{b} is the number on the base, and the total number is N+N_b;

$$\{\underline{1}_{c}^{x}\}^{T} = \langle 1 \ 0 \ 1 \ 0 \ . \ . \ . \ 1 \ 0 \ . \ . \ . \ 1 \ 0 \rangle$$
$$\{\underline{1}_{c}^{y}\}^{T} = \langle 0 \ 1 \ 0 \ 1 \ . \ . \ . \ 0 \ 1 \ . \ . \ . \ 0 \ 1 \rangle$$

 $\frac{R}{C}(t)$, the vector of forces due to structure-foundation interaction, will have non zero elements only at the base of the dam; $a_g^X(t)$ and $a_g^Y(t)$ are the x and y components of the specified free-field ground acceleration, assumed to be identical at all nodal points on the dam base.

For harmonic ground acceleration in the x (horizontal) or y (vertical) directions $a_g^{\ell}(t) = e^{i\omega t}$, $\ell = x$ or y, the displacements and interaction forces can be expressed in terms of their complex frequency responses: $\underline{r}_c(t) = \frac{\overline{r}_c^{\ell}(\omega)}{\underline{r}_c}(\omega) e^{i\omega t}$ and $\underline{R}_c(t) = \overline{\underline{R}}_c^{\ell}(\omega) e^{i\omega t}$ and the governing equation as

$$\left[-\omega^{2}\underline{m}_{c} + i\omega\underline{c}_{c} + \underline{k}_{c}\right]\underline{\bar{r}}_{c}^{\ell}(\omega) = -\underline{m}_{c}\underline{l}_{c}^{\ell} + \underline{\bar{R}}_{c}^{\ell}(\omega); \ \ell = x \text{ or } y$$
(5.2)

If the structure has constant hysteretic damping, Eq. 5.2 becomes

$$\left[-\omega^{2}\underline{m}_{c} + (1 + i\eta)\underline{k}_{c}\right]\underline{\vec{r}}_{c}^{\ell}(\omega) = -\underline{m}_{c}\underline{1}_{c}^{\ell} + \underline{\vec{R}}_{c}^{\ell}(\omega); \ \ell = x \text{ or } y$$
(5.3)

Partitioning \underline{r}_c into \underline{r} , the vector of displacements of nodal points above the base, and \underline{r}_b the vector of interaction displacements at the base (Fig. 5.1), Eq. 5.3 can be expressed as

$$\begin{pmatrix} -\omega^{2} \begin{bmatrix} \underline{m} & 0 \\ 0 & \underline{m}_{b} \end{bmatrix} + (1+i\eta) \begin{bmatrix} \underline{k} & \underline{k}_{b} \\ (\underline{k}_{b})^{T} & \underline{k}_{bb} \end{bmatrix} \begin{pmatrix} \underline{\bar{r}}^{\ell}(\omega) \\ \underline{\bar{r}}^{\ell}_{b}(\omega) \end{pmatrix} = - \begin{pmatrix} \underline{m} & \underline{l} \\ \underline{m} & \underline{l} \\ \underline{m}_{b} & \underline{l}_{b} \end{pmatrix} + \begin{pmatrix} \underline{0} \\ \underline{R}^{\ell}_{b}(\omega) \end{pmatrix}$$
(5.4)

5.2.2 Substructure 2: Foundation Rock Region

The dynamics of the foundation soil or rock region are considered separately. The interaction forces, $\frac{R}{-f}(t)$, at the foundation surface can be expressed in terms of the corresponding displacements, $\frac{r}{-f}(t)$, relative to the free-field earthquake ground displacement [24]:



SUBSTRUCTURE 2: FOUNDATION ROCK REGION

FIG. 5.1 SUBSTRUCTURE REPRESENTATION OF DAM-FOUNDATION SYSTEM

$$\underline{\mathcal{S}}_{f}(\omega) \quad \underline{\bar{r}}_{f}(\omega) = \overline{\mathbb{R}}_{f}(\omega) \tag{5.5}$$

where $\underline{S}_{f}(\omega)$ is the complex valued, dynamic stiffness matrix for the foundation soil region. The ijth element of this matrix, $\underline{S}_{f}(\omega)_{ij}$ is defined in Fig. 5.2 where displacements as shown have been imposed at nodal points within the base of the dam and tractions outside the base are zero.

The foundation stiffness matrix $\underline{\mathscr{G}}_{\mathbf{f}}(\omega)$ is to be determined by a separate analysis of the substructure representing the foundation soil or rock region. For sites where formations of soft rock or soil overlie hard rock, it would be reasonable to hypothesize that the boundary between the two types of rock is rigid. Then the foundation region may be idealized as a finite element system, in which material non-homogeneity and irregular geometry can be conveniently handled. Techniques for determining $\underline{\mathscr{S}}_{\mathbf{f}}(\omega)$ for a finite element idealization of the foundation soil region are discussed elsewhere [29]. For sites where similar rock extends to large depths, it may be more appropriate to idealize the foundation as a viscoelastic half-plane. Assuming that the dam is supported at the surface of a viscoelastic half-plane with homogeneous material properties, $\underline{\mathscr{S}}_{\mathbf{f}}(\omega)$ can be determined from the available data and procedures [32]. These results, in contrast to those from many earlier studies, are not restricted to a rigid plate at the base of the structure.

5.2.3 Dam-Foundation System

Equilibrium of interaction forces between the two substructures -- dam and foundation -- at the base of the dam requires that

$$\overline{\underline{R}}_{\mathbf{b}}^{\ell}(\omega) = -\overline{\underline{R}}_{\mathbf{f}}(\omega)$$
 (5.6)

Similarly, compatibility of interaction displacements in the two substructures at the dam base requires that

$$\frac{\bar{r}_{b}^{\ell}(\omega)}{\bar{r}_{f}} = \bar{r}_{f}(\omega)$$
(5.7)

Consequently, Eq. 5.5 becomes

$$\overline{\underline{R}}_{b}^{\ell}(\omega) = -\underline{s}_{f}(\omega) \ \underline{\underline{r}}_{b}^{\ell}(\omega)$$
(5.8)



FIG. 5.2 DEFINITION OF $\mathscr{S}_{f}(\omega)_{ij}$, THE ij-ELEMENT OF THE DYNAMIC STIFFNESS MATRIX $\underline{\mathscr{S}}_{f}(\omega)$ FOR THE FOUNDATION REGION which is substituted in Eq. 5.4 to obtain

$$\left[-\omega_{\underline{m}_{C}}^{2} + (1 + i\eta)k_{\underline{c}} + \tilde{\underline{\mathcal{B}}}_{\underline{f}}(\omega)\right]_{\underline{r}_{C}}^{-\ell}(\omega) = -\underline{m}_{\underline{c}}\underline{1}_{\underline{c}}^{\ell}$$
(5.9)

where

$$\widetilde{\underline{\mathcal{B}}}_{\underline{f}}(\omega) = \begin{bmatrix} \underline{0} & \underline{0} \\ \\ \\ \underline{0} & \underline{\mathcal{B}}_{\underline{f}}(\omega) \end{bmatrix}$$
(5.10)

For a particular excitation frequency ω , Eq. 5.9 represents $2(N+N_b)$ algebraic equations with complex-valued coefficients, in the unknown nodal point displacements $\frac{r\ell}{r_c}$.

5.2.4 Computational Requirements

Analysis of structural response to earthquake motion via the frequency domain requires that the complex frequency response functions be determined for a range of frequencies over which the ground motion and structural response have significant components. For each excitation frequency, this entails solution of Eq. 5.9, a set of as many algebraic equations as the number of degrees of freedom for the dam. Finite element systems with a few hundred degrees of freedom are typically employed to represent concrete gravity dams. Enormous computational effort would be required for repeated solutions of these equations for many values of the excitation frequecy. Furthermore, the entire displacement vector, \underline{r}_{c}^{ℓ} , at each ω has to be stored for future Fourier synthesis, requiring very large amounts of computer storage. Fourier synthesis calculations would have to be performed for each of the few hundred displacements, requiring large computational effort even when the very efficient FFT procedure is employed. The need for judiciously reducing the number of degrees of freedom in the analysis is therefore apparent.

5.3 Reduction of Degrees of Freedom

At any instant of time, the structural displacements in the DOF above the base may be separated into static displacements due to the interaction displacements at the base and the remaining dynamic displacements. The

- 34 -

latter can be expressed as a linear combination of the first few natural modes of vibration of the structure on a fixed base. This has been demonstrated to be effective in reducing the number of DOF for one-dimensional structural systems on rigid footing supported on the surface of a halfspace and also for complex structures on a deformable base. If J vibration modes are adequate to represent the structural response, the number of unknowns in Eq. 5.9 would be reduced to $J+2N_{\rm b}$, the latter being the number of DOF at the base. While this approach is conceptually appealing, the number of unknowns can not be reduced below $2N_{\rm b}$, the number of DOF at the base, which in finite element models of dams might be of the order of 20.

For such systems the number of unknowns in Eq. 5.9 can be most effectively reduced by use of the Ritz concept. The displacements \underline{r}_c are expressed as linear combinations of the Ritz vectors, chosen as the normal modes of an associated undamped structure-foundation system. The associated system considered is one in which $\underline{\mathcal{S}}_f(\omega)$ is replaced by a frequency-independent value, say the static value $\underline{\mathcal{S}}_f(0)$. The frequencies λ_n and mode shapes $\underline{\psi}_n$ of the associated system are solutions of the eigenvalue problem

$$\begin{bmatrix} \underline{k}_{c} + \tilde{\underline{\beta}}_{f}(0) \end{bmatrix} \underline{\psi}_{n} = \lambda_{n}^{2} \underline{m}_{c} \underline{\psi}_{n}$$
(5.11)

where $\tilde{\underline{\beta}}_{f}(0)$ was defined in Eq. 5.10.

If the first J modes $\underline{\psi}_1$, $\underline{\psi}_2$, ... $\underline{\psi}_J$ are considered as the Ritz vectors, the structural displacements are expressed as

$$\underline{\underline{r}}_{c}^{\ell}(t) \simeq \sum_{j=1}^{J} z_{j}^{\ell}(t) \underline{\Psi}_{j} \quad \ell = \mathbf{x}, \mathbf{y}$$
(5.12)

where Z 's are the generalized coordinates. In terms of the complex frequency response functions, Eq. 5.12 is

$$\underline{\underline{r}}_{c}^{\ell}(\omega) = \sum_{j=1}^{J} \underline{\underline{z}}_{j}^{\ell}(\omega) \underline{\psi}_{j}$$
(5.13)

Introducing the transformations of Eq. 5.13 into Eq. 5.9, premultiplying by ψ_n^T , and utilizing the orthogonality property of modes of the associated structure-foundation system with respect to the "stiffness" and mass matrices of Eq. 5.11, results in

$$\underline{S}(\omega) \quad \underline{\overline{Z}}^{\ell}(\omega) = \underline{L}^{\ell}(\omega) \tag{5.14}$$

- 35 -

In Eq. 5.14, the diagonal elements of $S(\omega)$ are

$$\mathbf{s}_{nn}(\omega) = \left[-\omega^{2} + (1+i\eta)\lambda_{n}^{2}\right] \underline{\psi}_{n}^{\mathrm{T}} \underline{\mathbf{m}}_{c} \underline{\psi}_{n} + \psi_{n}^{\mathrm{T}} \left[\widetilde{\underline{\boldsymbol{z}}}_{f}(\omega) - (1+i\eta) \widetilde{\underline{\boldsymbol{z}}}_{f}(0) \right] \underline{\psi}_{n} \qquad (5.15a)$$

whereas the off-diagonal elements are

$$\mathbf{s}_{nj}(\omega) = \underline{\psi}_{n}^{\mathrm{T}} \left[\tilde{\underline{\mathcal{S}}}_{f}(\omega) - (1 + i\eta) \; \tilde{\underline{\mathcal{S}}}_{f}(0) \right] \underline{\psi}_{j}$$
(5.15b)

where n and j = 1,2,3,...,J; $\overline{Z}(\omega)$ is a column vector of the complex frequency response functions for the generalized coordinates Z_n ; $\frac{l}{n}$ is a column vector with the nth element

$$L_{n}^{\ell} = -\underline{\psi}_{n}^{\mathrm{T}} \underline{\mathbf{m}}_{c} \underline{\mathbf{1}}_{c}^{\ell}; \quad \ell = \mathbf{x}, \mathbf{y}$$
 (5.15c)

For a particular excitation frequency ω , Eq. 5.14 represents J simultaneous algebraic equations in the generalized coordinates $\overline{Z}_{n}(\omega)$, n = 1, 2, ..., J.

If all the $2(N+N_{\rm b})$ vibration modes of the associated structurefoundation system were included in the Ritz method, the results obtained by solving the simultaneous equations 5.14 would be identical to the "exact" results from solution of Eq. 5.9. There would, however, be no advantage in choosing to solve Eq. 5.14 instead of Eq. 5.9 because the number of equations is not reduced nor are they uncoupled. On the other hand, if sufficiently accurate results can be obtained with $J<<2(N+N_{\rm b})$, there would be profound computational advantages in working with Eq. 5.14.

That this is indeed the case has been demonstrated for one-dimensional structural systems on rigid footings supported on the surface of a halfspace [33] and also for concrete gravity dams [24]. The number of generalized coordinates that should be included in analysis depends on the properties of the structure-foundation system, the response quantities, and the frequency range of interest. The number necessary in analysis of concrete gravity dams studied in this report is determined in Section 7.6.

5.4 Response to Arbitrary Ground Motion

The complex frequency response functions $\overline{Z}_{n}^{\mathbf{X}}(\omega)$ and $\overline{Z}_{n}^{\mathbf{Y}}(\omega)$, n = 1, 2,..., J, are determined by solving Eq. 5.14 for an appropriate range of excitation frequency ω . The response to arbitrary ground motion can be computed using the equations of Sec. 4.5 with the following changes in notation: replace Y_n^{ℓ} by Z_n^{ℓ} , \underline{r} by \underline{r}_c , and $\underline{\phi}_n$ by $\underline{\psi}_n$.

ł ł { ł 1 1 ł ł ł ł ł 1 ł ł Ł ł 1

6. ANALYSIS OF DAMS INCLUDING HYDRODYNAMIC AND FOUNDATION INTERACTION EFFECTS

6.1 Introductory Note

The substructure concept was employed to separately include the effects of dam-water interaction (Chapter 4) and dam-foundation interaction (Chapter 5) in the analysis. Using the same concept the analysis procedure is extended to analyze earthquake response of dams, simultaneously including various effects of the water and the foundation. These include effects arising from interaction between the dam and foundation, dam and water, water and foundation, and from interaction among all three substructures -dam, water and foundation.

6.2 Frequency Domain Equations

6.2.1 Substructure 1: Dam

A cross-section of the dam-water-foundation system is shown in Fig. 2.1. The equations of motion for the dam idealized as a planar, two-dimensional finite element system are:

$$\underline{m}_{c-c} \dot{r}_{c-c} + \underline{c}_{c-c} \dot{r}_{c-c} + \underline{k}_{c-c} = -\underline{m}_{c-c-c} \underline{l}_{g}^{X}(t) - \underline{m}_{c-c-c-q} \underline{l}_{g}^{Y}(t) + \underline{R}_{c}(t)$$
(6.1)

in which \underline{m}_{c} , \underline{c}_{c} , and \underline{k}_{c} are the mass, damping and stiffness matrices for the finite element system; \underline{r}_{c} is the vector of nodal point displacements relative to the free-field earthquake displacement at the base (Fig. 6.1):

$$\underline{r}_{c}^{T} = \langle r_{1}^{X} r_{1}^{Y} r_{2}^{X} r_{2}^{Y} \dots r_{n}^{X} r_{n}^{Y} \dots r_{N+N_{b}}^{X} r_{N+N_{b}}^{Y} \rangle$$

in which r_n^x and r_n^y are the x and y components of the displacement of nodal point n; N is the number of nodal points above the base, N_b is the number of nodal points on the base, and the total number is N+N_b;

$$\{\underline{1}_{c}^{\mathbf{X}}\}^{\mathrm{T}} = \langle 1 \ 0 \ 1 \ 0 \ . \ . \ . \ 1 \ 0 \ . \ . \ . \ . \ 1 \ 0 \rangle$$
$$\{\underline{1}_{c}^{\mathbf{Y}}\}^{\mathrm{T}} = \langle 0 \ 1 \ 0 \ 1 \ . \ . \ . \ 0 \ 1 \ . \ . \ . \ 0 \ 1 \rangle$$

The force vector $\frac{R}{-c}(t)$ includes hydrodynamic forces $\frac{R}{-h}(t)$ at the upstream face of the dam and forces $\frac{R}{-b}(t)$ at the base of the dam due to structure-

- 39 -



FIG. 6.1 SUBSTRUCTURE REPRESENTATION OF DAM-WATER-FOUNDATION SYSTEM

foundation interaction; $a_g^X(t)$ and $a_g^Y(t)$ are the x and y components of the specified free-field ground acceleration, assumed to be identical at all nodal points on the dam base.

For harmonic ground acceleration in the x (horizontal) or y (vertical) directions $a_g^{\ell}(t) = e^{i\omega t}$, $\ell = x$ or y, the displacements and forces can be expressed in terms of their complex frequency responses: $\underline{r}_{c}(t) = \frac{r}{r_{c}}^{\ell}(\omega)e^{i\omega t}$, $\underline{R}_{c}(t) = \frac{\bar{R}_{c}^{\ell}(\omega)e^{i\omega t}$, $\underline{R}_{h}^{\ell}(t) = \frac{\bar{R}_{h}^{\ell}(\omega)e^{i\omega t}$ and $\underline{R}_{b}^{\ell}(\omega) = \frac{\bar{R}_{b}^{\ell}(\omega)e^{i\omega t}$. The governing equation, Eq. 6.1, then becomes

$$\left[-\omega_{\underline{m}_{c}}^{2} + i\omega_{\underline{c}_{c}}^{2} + \underline{k}_{c}\right] \underline{\bar{r}}_{c}^{\ell}(\omega) = -\underline{m}_{\underline{c}} \underline{1}_{c}^{\ell} + \underline{\bar{R}}_{c}^{\ell}(\omega) ; \ell = x \text{ or } y \qquad (6.2)$$

If the structure has constant hysteretic damping, Eq. 6.2 becomes

$$\left[-\omega^{2}\underline{m}_{c} + (1+i\eta)\underline{k}_{c}\right]\underline{r}_{c}^{\ell}(\omega) = -\underline{m}_{c}\underline{l}_{c}^{\ell} + \underline{R}_{c}^{\ell}(\omega) ; \ell = x \text{ or } y \qquad (6.3)$$

Partitioning \underline{r}_{c} into \underline{r} , the vector of displacements of nodal points above the base, and \underline{r}_{b} , the vector of interaction displacements at the base (Fig. 6.1), Eq. 6.3 can be expressed as

$$\begin{pmatrix} -\omega^{2} \begin{bmatrix} \underline{m} & \underline{0} \\ \\ \underline{0} & \underline{m}_{b} \end{bmatrix} + (1 + i\eta) \begin{bmatrix} \underline{k} & \underline{k}_{b} \\ \\ (\underline{k}_{b})^{T} & \underline{k}_{bb} \end{bmatrix} \end{pmatrix} \begin{pmatrix} \underline{\tilde{r}}^{\ell}(\omega) \\ \\ \underline{\tilde{r}}^{\ell}(\omega) \\ \\ \underline{\tilde{r}}^{\ell}(\omega) \end{pmatrix} = - \begin{pmatrix} \underline{m} \underline{1}^{\ell} \\ \\ \underline{m} \underline{1}^{\ell} \\ \\$$

The forces at the base of the dam, $\overline{R}_{b}^{\ell}(\omega)$, are next expressed in terms of interaction displacements by appropriate analyses of substructure 2, the foundation (Sec. 6.2.2). Later (Sec. 6.2.5) the hydrodynamic forces $\overline{R}_{h}(\omega)$ are expressed in terms of displacements or accelerations at the upstream face of the dam by appropriate analyses of substructure 3, the fluid region.

6.2.2 Substructure 2: Foundation Rock Region

The forces acting at the surface of the foundation include the forces R_f at the base of the dam due to structure-foundation interaction and the hydrodynamic forces Q_h at the bottom of the reservoir. Beyond a certain distance upstream of the dam, the hydrodynamic pressures will become small enough to be negligible. Only the significant forces are included in Q_h . For unit harmonic ground acceleration these forces can be expressed in

terms of their complex frequency response functions $\underline{R}_{f}(t) = \overline{\underline{R}}_{f}(\omega)e^{i\omega t}$ and $\underline{Q}_{h}(t) = \overline{\underline{Q}}_{h}(\omega)e^{i\omega t}$. The corresponding displacements relative to the freefield ground displacement are $\underline{r}_{f}(t) = \overline{\underline{r}}_{f}(\omega)e^{i\omega t}$ and $\underline{q}(t) = \overline{\underline{q}}(\omega)e^{i\omega t}$. The forces and displacements can be related through the complex valued, dynamic foundation stiffness matrix $\beta(\omega)$:

$$\begin{bmatrix} \underline{\mathscr{Z}}_{rr}(\omega) & \underline{\mathscr{Z}}_{rq}(\omega) \\ \underline{\mathscr{Z}}_{rq}^{T}(\omega) & \underline{\mathscr{Z}}_{qq}(\omega) \end{bmatrix} \quad \begin{cases} \bar{\mathbf{r}}_{f} \\ \bar{\mathbf{q}} \\ \bar{\mathbf{q}} \end{bmatrix} = \begin{cases} \bar{\mathbf{R}}_{-f} \\ \bar{\mathbf{Q}}_{h} \end{cases}$$
(6.5)

The ijth element of this matrix, $\underline{S}_{ij}(\omega)$, is defined in Fig. 6.2 where displacements as shown have been imposed at nodal points contained in \underline{r}_{f} and \underline{q} and tractions outside these nodal points are zero.

The second matrix equation from Eq. 6.5 can be expressed as

$$\bar{\underline{q}} = \underline{\underline{s}}_{qq}^{-1}(\omega) \left[\bar{\underline{Q}}_{h} - \underline{\underline{s}}_{rq}^{T}(\omega) \ \bar{\underline{r}}_{f} \right]$$
(6.6)

Substituting this expression into the first matrix equation from Eq. 6.5 leads to

$$\begin{bmatrix}\underline{\mathscr{S}}_{rr}(\omega) & -\underline{\mathscr{S}}_{rq}(\omega) & \underline{\mathscr{S}}_{qq}^{-1}(\omega) & \underline{\mathscr{S}}_{rq}^{T}(\omega)\end{bmatrix} = \underline{\bar{r}}_{f} = \underline{\bar{R}}_{f} - \underline{\mathscr{S}}_{rq}(\omega) & \underline{\mathscr{S}}_{qq}^{-1}(\omega) & \underline{\mathscr{Q}}_{h}$$

or

$$\underline{\mathcal{S}}_{f}(\omega) \, \underline{\tilde{r}}_{f} = \underline{\tilde{R}}_{f} - \underline{\mathcal{S}}_{rq}(\omega) \, \underline{\mathcal{S}}_{qq}^{-1}(\omega) \, \underline{\mathcal{Q}}_{h}$$
(6.7a)

where

$$\underline{\mathcal{S}}_{f}(\omega) = \underline{\mathcal{S}}_{rr}(\omega) - \underline{\mathcal{S}}_{rq}(\omega) \ \underline{\mathcal{S}}_{qq}^{-1}(\omega) \ \underline{\mathcal{S}}_{rq}^{T}(\omega)$$
(6.7b)

The matrix $\underline{\mathcal{S}}_{f}(\omega)$ of Eq. 6.7b is the same as the matrix $\underline{\mathcal{S}}_{f}(\omega)$ in Eq. 5.5. In particular, without hydrodynamic forces, $\underline{Q}_{h} = \underline{0}$ and Eq. 6.6 reduces to Eq.5.5.

The dynamic stiffness matrices for the foundation appearing in Eqs. 6.5, 6.7a, and 6.7b are to be determined by a separate analysis of the substructure representing the foundation soil or rock region. At many sites of concrete gravity dams, similar rocks extend to large depths, and it is therefore appropriate to idealize the foundation as a viscoelastic half plane. The surface of the half plane is assumed to be horizontal and its material properties to be homogeneous. Available data and methods [32] are utilized in this investigation to determine dynamic stiffness for such idealized foundations.



FIG. 6.2 DEFINITION OF $\mathscr{F}_{f}(\omega)_{ij}$, THE ij-ELEMENT OF THE DYNAMIC STIFFNESS MATRIX $\underline{\mathscr{P}}_{f}(\omega)$ FOR THE FOUNDATION REGION

6.2.3 Dam-Foundation System

Equilibrium of interaction forces between the two substructures -- dam and foundation -- at the base of the dam requires that

$$\bar{\bar{R}}_{\rm b}^{\ell}(\omega) = -\bar{\bar{R}}_{\rm f}(\omega) \tag{6.8}$$

Similarly, compatibility of interaction displacements at the dam base in the two substructures requries that

$$\bar{\mathbf{f}}_{\mathbf{b}}^{\ell}(\omega) = \bar{\mathbf{f}}_{\mathbf{f}}(\omega) \tag{6.9}$$

Consequently, Eq. 6.6 because

$$\overline{\underline{R}}_{\mathbf{b}}^{\ell}(\omega) = -\underline{\mathscr{S}}_{\mathbf{f}}(\omega) \quad \overline{\underline{r}}_{\mathbf{b}}^{\ell}(\omega) - \underline{\mathscr{S}}_{\mathbf{rq}} \quad \underline{\mathscr{S}}_{\mathbf{qq}}^{-1} \quad \underline{\mathscr{Q}}_{\mathbf{h}}$$
(6.10)

Thus, the forces $\underline{\bar{R}}_{b}^{\ell}(\omega)$ at the base of the dam due to structure-foundation interaction have been expressed in terms of the interaction displacements $\underline{\bar{r}}_{b}^{\ell}(\omega)$ and hydrodynamic forces \underline{Q}_{h} through dynamic stiffness matrices for the foundation region. Substituting Eq. 6.10 into Eq. 6.4 leads to

$$\begin{pmatrix} -\omega^{2} \begin{bmatrix} \bar{m} & 0 \\ 0 & \bar{m}_{b} \end{bmatrix} + (1 + i\eta) \begin{bmatrix} \bar{k} & \bar{k}_{b} \\ \bar{k}_{b}^{T} & \bar{k}_{bb} \end{bmatrix} + \begin{bmatrix} 0 & 0 \\ 0 & \bar{s}_{f}(\omega) \end{bmatrix} \end{pmatrix} \begin{cases} \bar{r}^{\ell}(\omega) \\ \bar{r}^{\ell}_{b}(\omega) \end{cases}$$

$$= - \begin{cases} \bar{m} \underline{1}^{\ell} \\ \bar{m}_{b} \underline{1}^{\ell}_{b} \end{cases} + \begin{cases} \bar{R}^{\ell}_{h}(\omega) \\ -\bar{s}_{rq} - \bar{s}_{qq} - \bar{q}_{q} - \bar{q}_{h} \end{cases}$$

$$(6.11)$$

or

$$\left[-\omega^{2} \underline{m}_{c} + (1+in_{c})\underline{k}_{c} + \tilde{\underline{\mathcal{S}}}_{f}(\omega)\right]\underline{\overline{r}}_{c}^{\ell}(\omega) = -\underline{m}_{c}\underline{1}_{c}^{\ell} + \tilde{\underline{R}}_{c}(\omega)$$
(6.12)

where

$$\tilde{\underline{\mathbf{g}}}_{f}(\omega) = \begin{bmatrix} \underline{\mathbf{0}} & \underline{\mathbf{0}} \\ \\ \underline{\mathbf{0}} & \underline{\mathbf{s}}_{f}(\omega) \end{bmatrix}$$
(6.13)

The hydrodynamic forces $\overline{\underline{R}}_h$ and $\overline{\underline{Q}}_h$ will be expressed in terms of accelerations of the upstream face of the dam and the bottom of the reservoir by analyses of the fluid region, to be described later.

6.2.4 Reduction of Degrees of Freedom

For each excitation frequency Eq. 6.11 is a set of $2(N+N_b)$ algebraic equations, as many as the number of DOF of the dam, which typically would

- 44 -

be a few hundred. Enormous computational effort would be required for repeated solutions of these equations for many values of the excitation frequency and subsequent Fourier synthesis of the harmonic responses in each DOF. Thus, it is important to judiciously reduce the number of unknowns in the analysis. As mentioned in Chapter 5, two approaches have proven to be effective in reducing the number of DOF.

In the first approach [25, 33], the structural displacements $\underline{r}^{\ell}(t)$ at some time t in the DOF above the base are separated into two parts: static displacements due to base displacements $\underline{r}^{\ell}_{b}(t)$ at the same time plus the remaining "dynamic" displacements; and the latter are expressed as a linear combination of the first few natural modes of vibration of the structure on a fixed base. If J vibration modes are adequate to represent the structural response, the number of unknowns in Eq. 6.11 would reduce to $J+2N_{b}$, where $2N_{b}$ = the number of DOF at the base. While this approach is conceptually appealing, the number of uknowns cannot be reduced below $2N_{b}$, which in finite element models of dams may be, say, 20.

The second approach [24,33], based on the Ritz concept, is more effective for such systems. The displacements \underline{r}_c are expressed as linear combinations of the Ritz vectors, chosen as the normal modes of an associated undamped structure-foundation system. The associated system chosen here and in earlier studies is one in which $\underline{\mathcal{S}}_f(\omega)$ is replaced by a frequency independent value, say the static value $\underline{\mathcal{S}}_f(0)$. The vibration frequencies λ_n and mode shapes $\underline{\psi}_n$ of the associated system are solutions of the eigenvalue problem

$$\left[\underline{\mathbf{k}}_{\mathbf{c}} + \widetilde{\underline{\mathbf{\mathcal{S}}}}_{\mathbf{f}}(\mathbf{0})\right] \underline{\boldsymbol{\psi}}_{\mathbf{n}} = \lambda_{\mathbf{n}-\mathbf{c}-\mathbf{n}}^{2} \boldsymbol{\psi}$$
(6.14)

where $\tilde{\underline{\mathcal{S}}}_{f}(0)$ was defined in Eq. 6.13.

If the first J modes $\underline{\psi}_1, \underline{\psi}_2, \ldots, \underline{\psi}_j$ are considered as the Ritz vectors, the structural displacements are expressed as

$$\underline{\underline{r}}_{c}^{\ell}(t) = \sum_{j=1}^{J} z_{j}^{\ell}(t) \underline{\psi}_{j}$$
(6.15)

where Z 's are the generalized coordinates. In terms of the complex frequency response functions, Eq. 6.15 is

$$\bar{\underline{r}}_{c}^{\ell}(\omega) = \sum_{j=1}^{J} \bar{z}_{j}^{\ell}(\omega) \underline{\psi}_{j}$$
(6.16)

- 45 -

Equation 6.16 contains the following equation for the displacements at the base of the structure:

$$\vec{\underline{r}}_{b}^{\ell}(\omega) = \sum_{j=1}^{J} \vec{z}_{j}^{\ell}(\omega) \underline{\psi}_{bj}$$
(6.17)

where $\underline{\Psi}_{bj}$ is the subvector of $\underline{\Psi}_{j}$ corresponding to the base DOF. Because $\underline{r}_{f} = \underline{r}_{b}$, and $\underline{\Psi}_{j}$ were determined from Eq. 6.14 involving $\underline{\mathcal{S}}_{f}(0)$, the static foundation stiffness matrix, \underline{q} may be expressed in terms of generalized coordinates by setting $\omega = 0$ in Eq. 6.6 and ignoring the hydrodynamic loads Q; thus

$$\underline{q} = -\underline{s}_{qq}^{-1}(0) \ \underline{s}_{rq}^{T}(0) \underline{r}_{b}$$
 (6.18)

Combining Eqs. 6.17 and 6.18 leads to

$$\mathbf{q} = \sum_{j=1}^{J} \bar{z}_{j}^{\ell}(\omega) \underline{\chi}_{j}$$
(6.19)

where

$$\chi_{j} = -g_{qq}^{-1}(0) g_{rq}^{T}(0) \psi_{bj}$$
(6.20)

Introducing the transformation of Eq. 6.16 into Eq. 6.12, premultiplying by $\underline{\psi}_n^{\mathrm{T}}$ and utilizing the orthogonality property of eigenvectors of the associated structure-foundation system with respect to the stiffness and mass matrices of Eq. 6.14, results in

$$\underline{S}(\omega) \quad \underline{Z}^{\ell}(\omega) = \underline{L}^{\ell}(\omega) \tag{6.21}$$

In Eq. 6.21, the diagonal elements of $\underline{S}(\omega)$ are

$$\mathbf{S}_{nn}(\omega) = \left[-\omega^{2} + (1+i\eta)\lambda_{n}^{2}\right] \underline{\psi}_{n-c}^{\mathrm{T}} \underline{\psi}_{n} + \underline{\psi}_{n}^{\mathrm{T}} \left[\widetilde{\underline{\mathscr{S}}}_{f}(\omega) - (1+i\eta) \widetilde{\underline{\mathscr{S}}}_{f}(0) \right] \underline{\psi}_{n}$$
(6.22a)

whereas the off-diagonal elements are

$$\mathbf{s}_{nj}(\omega) = \underline{\psi}_{n}^{\mathrm{T}} \left[\underline{\tilde{\mathbf{Z}}}_{f}(\omega) - (1+i\eta) \underline{\tilde{\mathbf{Z}}}_{f}(0) \right] \underline{\psi}_{j}$$
(6.22b)

where n and $j = 1, 2, 3 - - J; \overline{Z}^{\ell}(\omega)$ is a column vector of the complex frequency response functions for the generalized coordinates $Z_n^{\ell}; \underline{L}^{\ell}$ is a column vector with the nth element ℓ ℓ \mathcal{L}^{ℓ} \mathcal{L}^{τ} is a column vector (1.11)

$$L_{n}^{\ell} = -L_{n}^{\ell} + \underline{\Psi}_{n-c}^{\mathrm{T}}(\omega)$$
(6.23)

where

$$L_{n}^{\ell} = \Psi_{n-c-c}^{T} \frac{1}{c}$$
(6.24)

For a particular excitation frequency ω , Eq. 6.21 represents J simultaneous algebraic equations in the generalized coordinates $Z_n^{\ell}(\omega)$, n = 1, 2 - - J. These equations need to be solved for several hundred values of the excitation frequency to determine the complex frequency responses.

Equation 6.21 would contain $2(N+N_b)$ equations if all the vibration modes of the associated structure-foundation system where included as Ritz vectors. Solution of these equations would then be identical to the solution of Eq. 6.11 in nodal point coordinates. There would, however, be no advantage in choosing to solve Eq. 6.21 instead of Eq. 6.11 because the number of equations is not reduced nor are the equations uncoupled by the transformation. However, there would be profound computational advantages in working with Eq. 6.21 if sufficiently accurate results can be obtained by including only a few Ritz vectors. The number necessary depends on the properties of the structure-foundation system, the response quantities, and the frequency range of interest; for concrete dams this number is determined in Section 7.6.

6.2.5 Substructure 3: Fluid Domain

The motion of the dam-water system is assumed to be two-dimensional, and the same for any vertical plane perpendicular to the axis of the dam (Chapter 2). Assuming water to be linearly compressible and neglecting its internal viscosity, the small amplitude irrotational motion of water is governed by the two-dimensional wave equation

$$\frac{\partial^2 \mathbf{p}}{\partial \mathbf{x}^2} + \frac{\partial^2 \mathbf{p}}{\partial \mathbf{y}^2} = \frac{1}{c^2} \frac{\partial^2 \mathbf{p}}{\partial t^2}$$
(6.25)

in which p(x,y,t) is the hydrodynamic pressure (in excess of the hydrostatic pressure) and the velocity of sound in water $C = \sqrt{gK/w}$ where K and w are bulk modulus and unit weight of water, respectively, and g = acceleration of gravity. For harmonic ground motion $a_{\sigma}^{\ell}(t) = e^{i\omega t}$, $\ell = x$ or y,

$$p(x,y,t) = \bar{p}(x,y,\omega)e^{i\omega t}$$
(6.26)

where \bar{p} is the complex frequency response function for the pressure and Eq. 6.25 becomes the Helmholtz equation:

$$\frac{\partial^2 \bar{p}}{\partial x^2} + \frac{\partial^2 \bar{p}}{\partial y^2} + \frac{\omega^2}{c^2} \bar{p} = 0$$
 (6.27)

- 47 -

Whereas Eqs. 6.11 and 6.17 are valid for dams of arbitrary geometry, it is assumed that for purposes of defining hydrodynamic loads, and only for this purpose, the upstream face of the dam is vertical. This is a reasonable assumption, given that the upstream face of concrete gravity dams is vertical or almost vertical for most of the height.

Dynamic water pressures are generated by horizontal motions of the vertical upstream face of the dam and by vertical motions of the horizontal bottom of the reservoir. The vector of horizontal accelerations at the nodal points on the upstream face of the dam is

$$\underbrace{\ddot{u}}(t) = \left\{ \underbrace{\hat{l}}_{j=1} + \sum_{j=1}^{J} \underbrace{\Psi_{j}^{xf}}_{j} \overline{\ddot{z}}_{j}^{x}(\omega) \right\} e^{i\omega t}$$
(6.28a)

if the excitation is horizontal ground motion $a_{\alpha}^{x}(t) = e^{i\omega t}$; and

$$\underbrace{\ddot{\mathbf{u}}}_{\mathbf{i}}(t) = \left\{ \sum_{j=1}^{J} \underbrace{\psi}_{j}^{\mathbf{x}f} \, \ddot{\mathbf{z}}_{j}^{\mathbf{y}}(\omega) \right\} e^{\mathbf{i}\omega t}$$
(6.28b)

if the excitation is vertical ground motion $a_g^{Y}(t) = e^{i\omega t}$. In Eq. 6.28 the vector ψ_j^{xf} is a subvector of ψ_j containing only the elements corresponding to the x-DOF of the nodal points on the upstream face of the dam. From Eq. 6.19, the vector of vertical accelerations at the nodal points at the bottom of the reservoir is

$$\vec{\underline{v}}(t) = \left\{ \sum_{j=1}^{J} \chi_{j}^{Y} \vec{\overline{z}}_{j}^{X}(\omega) \right\} e^{i\omega t}$$
(6.29a)

if the excitation is horizontal ground motion; and

$$\mathbf{\vec{v}}(t) = \left\{ \mathbf{1} + \sum_{j=1}^{J} \chi_{j}^{Y} \, \mathbf{\vec{z}}_{j}^{Y}(\omega) \right\} e^{\mathbf{i}\omega t}$$
(6.29b)

if the excitation is vertical ground motion, where χ_j^y is a subvector of χ_j containing only the elements corresponding to the y-DOF of the nodal points at the bottom of the reservoir.

Strictly for purposes of notational convenience, the vectors $\underline{\psi}_{j}^{\text{xf}}$ and χ_{j}^{Y} are replaced by their continuous analogue functions $\psi_{j}(y)$ and $\chi_{j}(x)$, respectively in Eq. 6.28 to obtain

$$\ddot{u}(0,y,t) = \left[1 + \sum_{j=1}^{J} \psi_{j}(y) \ \ddot{z}_{j}^{x}(\omega)\right] e^{i\omega t}$$
(6.30a)

$$\ddot{u}(0,y,t) = \begin{bmatrix} J & \psi_{j}(y) & \overline{\ddot{z}}^{Y}_{j}(\omega) \end{bmatrix} e^{i\omega t}$$
(6.30b)

and in Eq. 6.29 to obtain

$$\ddot{\mathbf{v}}(\mathbf{x},0,t) = \left[\sum_{j=1}^{J} \chi_{j}(\mathbf{x}) \ \ddot{\ddot{\mathbf{z}}}_{j}^{\mathbf{x}}(\omega)\right] e^{i\omega t}$$
(6.31a)

$$\ddot{\mathbf{v}}(\mathbf{x},0,t) = \left[1 + \sum_{j=1}^{J} \chi_{j}(\mathbf{x}) \ \ddot{\mathbf{z}}_{j}^{Y}(\omega)\right] e^{i\omega t}$$
(6.31b)

When the excitation is horizontal ground motion $a_g^{x}(t) = e^{i\omega t}$, neglecting effects of the waves at the free surface of water [11], the boundary conditions for Eq. 6.27 governing \bar{p} are (with reference to Fig. 6.1)

$$\overline{p}(\mathbf{x}, \mathbf{H}, \omega) = 0$$

$$\frac{\partial \overline{p}}{\partial y}(\mathbf{x}, 0, \omega) = -\frac{w}{g} \int_{j=1}^{J} \chi_{j}(\mathbf{x}) \tilde{z}_{j}^{\mathbf{x}}(\omega)$$

$$\frac{\partial \overline{p}}{\partial x}(0, y, \omega) = -\frac{w}{g} \left[1 + \sum_{j=1}^{J} \psi_{j}(y) \tilde{z}_{j}^{\mathbf{x}}(\omega) \right]$$

$$(6.32)$$

For the case of vertical ground motion $a_g^{Y}(t) = e^{i\omega t}$ the boundary conditions are

$$\overline{p}(\mathbf{x}, \mathbf{H}, \omega) = 0$$

$$\frac{\partial \overline{p}}{\partial y}(\mathbf{x}, 0, \omega) = -\frac{w}{g} \left[\mathbf{1} + \sum_{j=1}^{J} \chi_{j}(\mathbf{x}) \ \overline{z}_{j}^{Y}(\omega) \right]$$

$$\frac{\partial \overline{p}}{\partial \mathbf{x}}(0, \mathbf{y}, \omega) = -\frac{w}{g} \left[\sum_{j=1}^{J} \psi_{j}(\mathbf{y}) \ \overline{z}_{j}^{Y}(\omega) \right]$$

$$(6.33)$$

Because the governing equation as well as the boundary conditions are linear, the principle of superposition applies. The complex frequency response functions $\overline{p}^{X}(x,y,\omega)$ and $\overline{p}^{Y}(x,y,\omega)$, associated respectively with x and y components of ground motion, can therefore be expressed as

$$\bar{p}^{\ell}(\mathbf{x},\mathbf{y},\omega) = \frac{-\ell}{P_0}(\mathbf{x},\mathbf{y},\omega) + \sum_{j=1}^{J} \bar{\tilde{z}}_{j}^{\ell}(\omega) \left[\bar{p}_{j}^{f}(\mathbf{x},\mathbf{y},\omega) + \bar{p}_{j}^{b}(\mathbf{x},\mathbf{y},\omega) \right] \quad (6.34)$$

In Eq. 6.34, $\overline{p}_0^{\mathbf{x}}(\mathbf{x},\mathbf{y},\omega)$ is the solution of Eq. 6.27 for the following boundary conditions:

$$\bar{p}(\mathbf{x}, \mathbf{H}, \omega) = 0$$

$$\frac{\partial \bar{p}}{\partial y}(\mathbf{x}, 0, \omega) = 0$$

$$\frac{\partial \bar{p}}{\partial x}(0, y, \omega) = -\frac{w}{g};$$
(6.35)

 $\bar{p}_0^Y(x,y,\omega)$ is the solution of Eq. 6.27 for the following boundary conditions:

$$\overline{p}(\mathbf{x}, \mathbf{H}, \omega) = 0$$

$$\frac{\partial \overline{p}}{\partial y}(\mathbf{x}, 0, \omega) = -\frac{w}{g}$$

$$\frac{\partial \overline{p}}{\partial x}(0, y, \omega) = 0 ;$$

$$(6.36)$$

 $\bar{p}_{j}^{f}(x,y,\omega)$ is the solution of Eq. 6.27 for the following boundary conditions:

$$\left. \begin{array}{l} \overline{p}(\mathbf{x},\mathbf{H},\boldsymbol{\omega}) = 0 \\ \\ \frac{\partial \overline{p}}{\partial \mathbf{y}}(\mathbf{x},0,\boldsymbol{\omega}) = 0 \\ \\ \frac{\partial \overline{p}}{\partial \mathbf{x}}(0,\mathbf{y},\boldsymbol{\omega}) = -\frac{w}{g}\psi_{j}(\mathbf{y}); \end{array} \right\}$$

$$(6.37)$$

and $\vec{p}_{j}^{b}(x,y,\omega)$ is the solution of Eq. 6.27 for the following boundary conditions:

$$\left. \begin{array}{l} \overline{p}(\mathbf{x},\mathbf{H},\omega) = 0 \\ \frac{\partial \overline{p}}{\partial \mathbf{y}}(\mathbf{x},0,\omega) = -\frac{w}{g}\chi_{j}(\mathbf{x}) \\ \frac{\partial \overline{p}}{\partial \mathbf{x}}(0,\mathbf{y},\omega) = 0 \end{array} \right\}$$
(6.38)

The complex frequency response functions \bar{p}_0^x and \bar{p}^y are for the hydrodynamic pressures due to accelerations of the ground in the horizontal and vertical directions, respectively, and the dam is rigid; $\bar{p}_j^f(x,y,\omega)$ is the corresponding function when the excitation is the horizontal acceleration $\psi_j(y)$ of the dam in the jth vibration mode of the associated dam-foundation system, with no motion of the reservoir bottom; $\bar{p}_j^b(x,y,\omega)$ is due to vertical acceleration $\chi_j(x)$ of the reservoir bottom, associated with the jth vibration mode of the associated dam-foundation mode of the associated dam.

Solution of Eq. 6.27 in fluid domains of infinite extent in the upstream direction (Fig. 6.1) and the boundary conditions of Eqs. 6.35, 6.37 and 6.36 have been reported earlier and are presented in Eqs. 6.39, 6.40 and 6.41, respectively:

$$\bar{p}_{0}^{x}(x,y,\omega) = \frac{4\omega}{\pi g} \sum_{\ell=1}^{\infty} \frac{(-1)^{\ell}}{(2\ell-1)\sqrt{\lambda_{\ell}^{2} - \frac{\omega^{2}}{c^{2}}}} \exp\left(x/\lambda_{\ell}^{2} - \frac{\omega^{2}}{c^{2}}\right) \cos \lambda_{\ell} y \quad (6.39)$$

$$\bar{p}_{j}^{f}(x,y,\omega) = -\frac{2\omega}{gH} \sum_{\ell=1}^{\infty} \frac{I_{j\ell}}{\sqrt{\lambda_{\ell}^{2} - \frac{\omega^{2}}{c^{2}}}} \exp\left(x/\lambda_{\ell}^{2} - \frac{\omega^{2}}{c^{2}}\right) \cos \lambda_{\ell} y \quad (6.40)$$

in which $\lambda_{\ell} = (2\ell-1) \pi/2H$ and $I_{j\ell} = \int_{0}^{H} \frac{\psi_{j}(y) \cos \lambda_{\ell} y}{y} dy$.

$$\overline{p}_{0}^{Y}(x,y,\omega) = \frac{wC}{g\omega} \frac{(1+\alpha) \sin \frac{\omega(H-y)}{C}}{(1+\alpha) \cos \frac{\omega H}{C}}$$
(6.41)

where the reflection coefficient

$$\alpha = \frac{\binom{C_{r} w_{r} / Cw}{-1}}{\binom{C_{r} w_{r} / Cw}{+1}}$$
(6.42)

where w_r and C_r are the unit weight and P-wave velocity for the foundation rock at the bottom of the reservoir; $\alpha = 1$ for rigid rock indicating that hydrodynamic pressure waves are completely reflected at the reservoir bottom; and $\alpha < 1$ for deformable rock implying that hydrodynamic pressure waves impinging at the reservoir bottom are partially reflected back in the fluid domain and partially refracted into rock.

Solution of Eq. 6.27 in fluid domains of infinite extent in the upstream direction (Fig. 6.1) and the boundary conditions of Eq. 6.38 may be obtained by procedures similar to those used in obtaining the above solutions of Eqs. 6.39 and 6.41 [34] and by employing the Fourier transform with respect to the spatial coordinate X. Such a general solution is not necessary, however, because the resulting pressures have little influence on the response of the dam (see discussion preceeding Eqs. 6.52).

The complex frequency response function for $R_{-h}^{f}(t)$, the vector of total hydrodynamic forces at the upstream face of the dam, is, from Eq. 6.34,

$$\bar{\underline{R}}_{h}^{f}(\omega) = \bar{\underline{R}}_{0}^{\ell}(\omega) + \sum_{j=1}^{J} \bar{\bar{z}}_{j}^{\ell}(\omega) \left[\bar{\underline{R}}_{j}^{f}(\omega) + \bar{\underline{R}}_{j}^{b}(\omega) \right]$$
(6.43)

in which $\bar{\mathbb{R}}_{0}^{\ell}(\omega)$, $\bar{\mathbb{R}}_{j}^{f}(\omega)$, and $\bar{\mathbb{R}}_{j}^{b}(\omega)$ are the nodal forces statically equivalent to the corresponding pressure functions at the dam face: $\bar{\mathbb{p}}_{0}^{\ell}(0,y,\omega)$, $\bar{p}_{j}^{f}(0,y,\omega)$, and $\bar{p}_{j}^{b}(0, y, \omega)$, respectively. They are computed by using the principle of virtual work with the displacements between nodal points defined by the interpolation functions used in the finite element idealization of the dam. Because the pressures act in the horizontal direction on a vertical upstream face, elements of \mathbb{R}_{h}^{f} that correspond to the y-DOF will be zero. Similarly, from Eq. 6.34, the complex frequency response function for $\underline{Q}_{h}(t)$, the vector of hydrodynamic forces at the reservoir bottom, is

$$\bar{\underline{Q}}_{h}(\omega) = \bar{\underline{Q}}_{0}^{\ell}(\omega) + \sum_{j=1}^{J} \bar{\bar{z}}_{j}^{\ell}(\omega) \left[\bar{\underline{Q}}_{j}^{f}(\omega) + \bar{\underline{Q}}_{j}^{b}(\omega) \right]$$
(6.44)

in which $\bar{Q}_{0}^{\ell}(\omega)$, $\bar{Q}_{j}^{f}(\omega)$, and $\bar{Q}_{j}^{b}(\omega)$ are the nodal forces statically equivalent to the corresponding pressure functions at the reservoir bottom: $\bar{p}_{0}^{\ell}(x,0,\omega)$, $\bar{p}_{j}^{f}(x,0,\omega)$ and $\bar{p}_{j}^{b}(x,0,\omega)$, respectively. Because the pressures act in the vertical direction on a horizontal bottom of the reservoir, elements of \bar{Q}_{h} that correspond to the x-DOF will be zero.

Thus, by analyses of the fluid domain for appropriate conditions, the hydrodynamic forces in Eq. 6.21 have been expressed in terms of the unknown generalized accelerations $\overline{Z}_{i}^{\ell}(\omega)$ and by substituting

. - 52 -

$$\overline{\overline{z}}_{j}^{\ell}(\omega) = -\omega^{2} \overline{z}_{j}^{\ell}(\omega)$$
(6.45)

in terms of the unknown generalized displacements.

6.2.6 Dam-Water-Foundation System

The nth equation contained in Eq. 6.21 is

$$s_{nn}(\omega)\bar{z}_{n}^{\ell}(\omega) + \sum_{\substack{j=1\\ j\neq n}}^{J} s_{nj}(\omega) \bar{z}_{j}^{\ell}(\omega)$$

$$= -L_{n}^{\ell} + \{\underline{\psi}_{n}^{f}\}^{T} \underline{\bar{R}}_{h}^{f}(\omega) - \{\underline{\psi}_{bn}\}^{T} \underline{\mathscr{S}}_{rq-qq-h}^{J-1}(\omega)$$

$$(6.46)$$

in which $S_{nj}(\omega)$ were defined in Eq. 6.22, L_n^{ℓ} in Eq. 6.24, and $\underline{\psi}_n^{\mathbf{f}}$ is the subvector of $\underline{\psi}_n$ containing only the elements corresponding to the face (f) DOF. Expressing hydrodynamic forces $\underline{\tilde{R}}_h^{\mathbf{f}}(\omega)$ and $\underline{\tilde{Q}}_h(\omega)$ in terms of generalized displacements from Eqs. 6.43 - 6.45 leads to

$$S_{nn}(\omega)\overline{z}_{n}^{\ell}(\omega) + \sum_{\substack{j=1\\j\neq n}}^{J} S_{nj}(\omega) z_{j}^{\ell}(\omega)$$

$$= -L_{n}^{\ell} + \{\underline{\psi}_{n}^{f}\}^{T} \left[\overline{\underline{R}}_{0}^{\ell}(\omega) - \omega^{2} \sum_{j=1}^{J} \overline{z}_{j}^{\ell}(\omega) \{ \overline{\underline{R}}_{j}^{f}(\omega) + \overline{\underline{R}}_{j}^{b}(\omega) \} \right]$$

$$- \{\underline{\psi}_{bn}\}^{T} \underline{\mathscr{B}}_{rq}(\omega) \underline{\mathscr{B}}_{qq}^{-1}(\omega) \left[\overline{\underline{\varrho}}_{0}^{\ell}(\omega) - \omega^{2} \sum_{j=1}^{J} \overline{z}_{j}^{\ell}(\omega) \{ \overline{\underline{\varrho}}_{j}^{f}(\omega) + \overline{\underline{\varrho}}_{j}^{b}(\omega) \} \right]$$

$$(6.47)$$

The set of equations 6.47 for $n = 1, 2 \dots J$ may be rearranged and expressed in matrix form as

$$\begin{bmatrix} \tilde{s}_{11}(\omega) & \tilde{s}_{12}(\omega) & \dots & \tilde{s}_{1J}(\omega) \\ \tilde{s}_{21}(\omega) & \tilde{s}_{22}(\omega) & \dots & \tilde{s}_{2J}(\omega) \\ \vdots & & & \\ \tilde{s}_{J1}(\omega) & \tilde{s}_{J2}(\omega) & \dots & \tilde{s}_{JJ}(\omega) \end{bmatrix} \begin{bmatrix} \bar{z}_{1}^{\ell}(\omega) \\ \bar{z}_{2}^{\ell}(\omega) \\ \vdots \\ \bar{z}_{2}^{\ell}(\omega) \\ \vdots \\ \bar{z}_{1}^{\ell}(\omega) \end{bmatrix} = \begin{bmatrix} \tilde{L}_{1}^{\ell}(\omega) \\ \tilde{L}_{2}^{\ell}(\omega) \\ \vdots \\ \vdots \\ \bar{z}_{1}^{\ell}(\omega) \end{bmatrix}$$
(6.48)

- 53 -

in which

$$\widetilde{S}_{nj}(\omega) = S_{nj}(\omega) + \omega^{2} \left\{ \underline{\psi}_{n}^{f} \right\}^{T} \left\{ \overline{R}_{j}^{f}(\omega) + \overline{R}_{j}^{b}(\omega) \right\} - \omega^{2} \left\{ \underline{\psi}_{bn} \right\}^{T} \underbrace{\mathcal{B}}_{rq}(\omega) \underbrace{\mathcal{B}}_{qq}^{-1}(\omega) \left\{ \overline{\underline{Q}}_{j}^{f}(\omega) + \overline{\underline{Q}}_{j}^{b}(\omega) \right\}$$
(6.49)

valid for diagonal as well as off-diagonal elements of $S\left(\omega\right)$ and

$$\widetilde{\mathbf{L}}_{n}^{\ell}(\omega) = -L_{n}^{\ell} + \left\{ \underline{\psi}_{n}^{f} \right\}^{T} \widetilde{\underline{R}}_{0}^{\ell}(\omega) - \left\{ \underline{\psi}_{bn} \right\}^{T} \underline{\mathscr{I}}_{rq}(\omega) \underline{\mathscr{I}}_{qq}^{-1}(\omega) \underline{\widetilde{\mathcal{I}}}_{0}^{\ell}(\omega)$$
(6.50)

Equation 6.49, after substitution of Eq. 6.22, becomes

$$\widetilde{S}_{nn}(\omega) = \left[-\omega^{2} + (1+i\eta)\lambda_{n}^{2}\right] \underline{\psi}_{n=c}^{T}\underline{\psi}_{n} + \underline{\psi}_{n}^{T} \left[\underline{\widetilde{\mathcal{S}}}_{f}(\omega) - (1+\eta)\underline{\widetilde{\mathcal{S}}}_{f}(0)\right]\underline{\psi}_{n} \\ + \omega^{2}\left\{\underline{\psi}_{n}^{f}\right\}^{T}\left\{\overline{R}_{-n}^{f}(\omega) + \overline{R}_{-n}^{b}(\omega)\right\} \\ - \omega^{2}\left\{\underline{\psi}_{bn}\right\}^{T} \underbrace{\mathcal{S}}_{rq}(\omega) \quad \underline{\mathcal{S}}_{-qq}^{-1}(\omega)\left\{\underline{\widetilde{Q}}_{n}^{f}(\omega) + \underline{\widetilde{Q}}_{n}^{b}(\omega)\right\}$$
(6.51a)

and

$$\widetilde{\mathbf{S}}_{nj}(\omega) = \underline{\Psi}_{n}^{\mathrm{T}} \left[\widetilde{\underline{\mathscr{B}}}_{\mathbf{f}}(\omega) - (1+i\eta) \widetilde{\underline{\mathscr{B}}}_{\mathbf{f}}(0) \right] \underline{\Psi}_{j} + \omega^{2} \left\{ \underline{\Psi}_{n}^{\mathrm{f}} \right\}^{\mathrm{T}} \left\{ \overline{\underline{\mathsf{R}}}_{j}^{\mathrm{f}}(\omega) + \overline{\underline{\mathsf{R}}}_{j}^{\mathrm{b}}(\omega) \right\} - \omega^{2} \left\{ \underline{\Psi}_{bn}^{\mathrm{T}} \right\}^{2} \underline{\mathscr{B}}_{rq}(\omega) - \underbrace{\mathscr{B}}_{qq}^{-1}(\omega) \left\{ \overline{\underline{\mathsf{Q}}}_{j}^{\mathrm{f}}(\omega) + \overline{\underline{\mathsf{Q}}}_{j}^{\mathrm{b}}(\omega) \right\}$$
(6.51b)

Equations 6.48 and 6.49 contain effects of the foundation and water in various forms:

- Dam-foundation interaction effects appear in eigenvalues λ_n and eigenvectors of the associated dam-foundation system and through the foundation stiffness matrix $\underline{\mathcal{S}}_{\underline{f}}(\omega)$ (Eqs. 6.7b and 6.13).
- Additional hydrodynamic loads $\frac{R_0^{\ell}}{0}$ on the dam due to the free-field ground motions computed on the assumption of a rigid dam.
- Dam-water interaction effects appear through the hydrodynamic forces $\begin{array}{c} R_{j}^{f} \end{array}$ on the face of the dam and $\begin{array}{c} Q_{j}^{f} \end{array}$ on the reservoir bottom due to deformational motions of the dam.
- · Water-foundation interaction effects appear through the hydrodynamic

forces \mathbb{R}_{j}^{b} on the face of the dam and \mathbb{Q}_{j}^{b} on the reservoir bottom due to deformational motions of the reservoir bottom.

• Dam-water-foundation interaction effects appear through the hydrodynamic forces \underline{Q}_{0}^{ℓ} , \underline{Q}_{j}^{f} , and \underline{Q}_{j}^{b} at the bottom of the reservoir which influences the forces at the dam-foundation interface.

It can be argued and shown through numerical examples that several terms in Eqs. 6.49 - 6.51 are relatively small and can be dropped without introducing any significant errors. One group of such terms arises from the hydrodynamic forces Q_0^{ℓ} , Q_j^{f} and Q_j^{b} at the reservoir bottom which are due to the various excitations mentioned earlier. The other such term involves R_{j}^{b} , the hydrodynamic forces at the upstream face of the dam due to deformational motions of the reservoir bottom.

Dropping these terms from Eqs. 6.49 - 6.51 leads to their simplified version:

$$\widetilde{\underline{S}}(\omega)\overline{\underline{Z}}^{\ell}(\omega) = \widetilde{\underline{L}}^{\ell}(\omega), \ \ell=x, \ y$$
(6.52)

in which

$$\widetilde{\mathbf{S}}_{nn}(\omega) = \left[-\omega^{2} + (1+i\eta)\lambda_{n}^{2}\right] \underline{\boldsymbol{\psi}}_{n}^{\mathrm{T}} \underline{\boldsymbol{\psi}}_{n}^{\mathrm{T}} + \underline{\boldsymbol{\psi}}_{n}^{\mathrm{T}} \left[\underline{\boldsymbol{\tilde{\mathcal{S}}}}_{\mathbf{f}}(\omega) - (1+i\eta) \underline{\boldsymbol{\tilde{\mathcal{S}}}}_{\mathbf{f}}(0) \right] \underline{\boldsymbol{\psi}}_{n} + \omega^{2} \left\{ \underline{\boldsymbol{\psi}}_{n}^{\mathrm{f}} \right\}^{\mathrm{T}} \underline{\tilde{\mathbf{R}}}_{n}^{\mathrm{f}}(\omega)$$

$$(6.53a)$$

$$\widetilde{S}_{nj}(\omega) = \underline{\Psi}_{n}^{T} \left[\underbrace{\widetilde{\mathscr{B}}}_{f}(\omega) - (1+i\eta) \underbrace{\widetilde{\mathscr{B}}}_{f}(0) \right] \underline{\Psi}_{j} + \omega^{2} \left\{ \underline{\Psi}_{n}^{f} \right\}^{T} \underbrace{\mathbb{R}}_{j}^{f}(\omega)$$
(6.53b)

$$\widetilde{\mathbf{L}}_{\mathbf{n}}^{\ell}(\omega) = -\underline{\Psi}_{\mathbf{n}-\mathbf{c}-\mathbf{c}}^{\mathbf{T}} + \left\{ \underline{\Psi}_{\mathbf{n}}^{\mathbf{f}} \right\}^{\mathbf{T}} \overline{\mathbf{R}}_{\mathbf{0}}^{\ell}(\omega)$$
(6.53c)

These are the final equations governing the response of the dam to horizontal or vertical ground motions, applied separately. Included in these equations are the hydrodynamic effects and dam-foundation interaction effects that are significant in the response of the dam. Equation 6.52 represents J algebraic equations in the unkown complex frequency response functions in the generalized coordinates $\overline{Z}_{j}^{\ell}(\omega)$, j = 1, 2, --- J, corresponding to the J Ritz vectors included in the analysis. The coefficient matrix $\underline{S}(\omega)$ is to be determined for each excitation frequency and simultaneous solution of the equations leads to the values of \overline{Z}_{j}^{ℓ} at that excitation frequency. Repeated solution for several hundred excitation frequencies, covering the frequency range over which the ground motion and structural response have significant components, leads to the complete functions $\overline{Z}_{j}^{\ell}(\omega)$.

- 55 -

The number of Ritz vectors that need be included in the analysis depends on the system properties, the response quantities, and frequency range of interest. The number necessary in analysis of concrete gravity dams is determined in Sec. 7.6. As will be seen later, a few generalized coordinates (ten to twenty) are sufficient in the analysis of dams idealized as finite element systems with say 200 DOF, thus leading to very efficient procedures.

6.2.7 Singularities of Response

Elements of matrix $\tilde{\underline{S}}(\omega)$ and load vector $\underline{\tilde{L}}(\omega)$ are to be computed for a frequency ω by using Eq. 6.53 together with the pressure functions \overline{p}_0^{ℓ} and \overline{p}_j of Eqs. 6.39 and 6.40. The pressure functions, and hence the corresponding nodal forces $\underline{\tilde{R}}_0^{\ell}$ and $\underline{\tilde{R}}_j$, are unbounded at $\omega = \omega_m^r$ where $\omega_m^r = (2m-1)\pi C/2H$, is the mth resonant frequency of the fluid domain. Consequently, the elements of $\underline{\tilde{S}}(\omega)$ and $\underline{\tilde{L}}_{\star}^{\star}(\omega)$ are unbounded at these frequencies.

When J = 1, i.e. when only one vibration mode of the dam is considered in the analysis, the J equations of Eq. 6.53 reduce to one equation and the response at $\omega = \omega_m^r$ can be obtained through a limiting process. However, when J > 1, i.e. when more than one mode is considereed in the analysis, the limiting process yields a system of equations such that $\underline{\tilde{S}}(\omega)$ is singular at $\omega = \omega_m^r$. In particular, all the M equations become identical to each other and no solution can be obtained.

This degeneracy of the equations may be considered as a limitation of the substructure method of analysis. However, this limitation is of no practical consequence in obtaining earthquake responses. Numerical values for the frequency responses may be obtained for values of ω arbitrarily close to ω_m^r . The singularities at ω_m^r constitute a discrete set and will therefore not affect the values of the Fourier integrals which will lead to earthquake responses (Sec. 6.3).

6.3 Response to Arbitrary Ground Motion

Once the complex frequency response functions $\overline{z}_n^{\mathbf{x}}(\omega)$ and $\overline{z}_n^{\mathbf{y}}(\omega)$, n = 1,2, ..., J, have been determined by solving Eq. 6.52 for an appropriate range of excitation frequency ω , the responses to arbitrary ground motion can be obtained as the superposition of responses to individual harmonic components of the excitation through the Fourier integral:

$$\overline{z}_{n}^{\ell}(t) = \frac{1}{2\pi} \int_{-\infty}^{\infty} \overline{z}_{n}^{\ell}(\omega) A_{g}^{\ell}(\omega) e^{i\omega t} d\omega; \quad \ell = x, y \quad (6.54)$$

in which $A_g^{\ell}(\omega)$ is the Fourier transform of $a_g^{\ell}(t)$:

$$A_{g}^{\ell}(\omega) = \int_{0}^{d} a_{g}(t) e^{-i\omega t} dt; \quad \ell = x, y \qquad (6.55)$$

where d is the duration of ground motion.

The combined response $Z_n(t)$ to horizontal and vertical components of ground motions acting simultaneously is

$$Z_{n}(t) = Z_{n}^{X}(t) + Z_{n}^{Y}(t)$$
 (6.56)

Repeating this procedure for all the necessary values of n, the displacement response is obtained by transforming back from generalized to nodal point coordinates

$$\underline{\mathbf{r}}_{c}(t) = \sum_{n=1}^{J} \mathbf{Z}_{n}(t) \underline{\boldsymbol{\psi}}_{n}$$
(6.57)

The stresses $\sigma_p(t)$ in finite element p at any instant of time are related to the nodal point displacements $r_p(t)$ for that element by

$$\sigma_{\rm p}(t) = T_{\rm p} r_{\rm p}(t)$$
 (6.58)

where \underline{T}_{p} is the stress transformation matrix for the finite element p. At any instant of time, the stresses throughout the dam are determined from the nodal point displacements by applying the transformations of Eq. 6.58 to each finite element.

6.4 Static Stress Analysis

The standard procedure for analysis of stresses due to static loads (Chapter 3) applies to structures on rigid base. Including foundation flexibility, the equilibrium equations for a dam subjected only to hydrostatic pressures and forces associated with its own weight can be obtained as a

- 57 -

special case of the equations of dynamics (Eq. 6.11)

$$\begin{bmatrix} \underline{k} & \underline{k}_{b} \\ (\underline{k}_{b})^{\mathrm{T}} & \underline{k}_{bb} + \underline{g}_{f}(0) \end{bmatrix} \begin{cases} \underline{r} \\ \underline{r}_{b} \end{cases} = \begin{cases} \underline{R} \\ \underline{0} \\ \underline{0} \end{cases}$$
(6.59)

The vector of static loads is denoted by R and the effects of foundation deformations are represented by the foundation stiffness matrix at zero frequency.

In principle, Eq. 6.59 may be directly solved for the desired nodal point displacements resulting from the gravity loads. In practice, this direct approach may be unsatisfactory because the displacements \underline{r} and $\underline{r}_{\underline{b}}$ include large rigid body components which have no effect on the stresses in the dam. To avoid this difficulty, the total displacements are expressed as the sum of rigid body displacements plus an increment due to deformations, i.e.,

$$\begin{cases} \underline{r} \\ \underline{r}_{b} \end{cases} = \begin{cases} \underline{r}_{0} \\ \underline{r}_{0b} \end{cases} + \begin{cases} \underline{\tilde{r}} \\ \underline{\tilde{r}}_{b} \end{cases}$$
(6.60)

in which the subscript "⁰" refers to rigid body displacements, and the "tilde" identifies the relative displacement components. The rigid body and relative displacements at the base of the dam are shown in Fig. 6.3.

Introducing Eq. 6.60 into Eq. 6.59 yields:

$$\begin{bmatrix} \underline{k} & \underline{k}_{b} \\ (\underline{k}_{b})^{\mathrm{T}} & \underline{k}_{bb} + \underline{\beta}_{f}(0) \end{bmatrix} \begin{cases} \underline{r}_{0} \\ \underline{r}_{0b} \end{cases} + \begin{bmatrix} \underline{k} & \underline{k}_{b} \\ (\underline{k}_{b})^{\mathrm{T}} & \underline{k}_{bb} + \underline{\beta}_{f}(0) \end{bmatrix} \begin{cases} \underline{\tilde{r}} \\ \underline{\tilde{r}}_{b} \end{cases} = \begin{cases} \underline{R} \\ \underline{0} \end{cases}.$$
(6.61)

But since the rigid body displacements produce no forces in the structure, this may be written as

$$\begin{bmatrix} \underline{0} & \underline{0} \\ \underline{0} & \underline{\beta}_{f}(0) \end{bmatrix} \begin{pmatrix} \underline{r} \\ \underline{r}_{0b} \end{pmatrix} + \begin{bmatrix} \underline{k} & \underline{k}_{b} \\ (\underline{k}_{b})^{T} & \underline{k}_{bb} + \underline{\beta}_{f}(0) \end{bmatrix} \begin{pmatrix} \underline{\tilde{r}} \\ \underline{\tilde{r}}_{b} \end{pmatrix} = \begin{pmatrix} \underline{R} \\ \underline{0} \end{pmatrix}$$
(6.62)
Designating the foundation forces associated with rigid body displacements by $\underline{R}_{\rm Ob},$ where

$$\underline{\mathbf{R}}_{\mathrm{Ob}} = \underline{\beta}_{\mathbf{f}}(\mathbf{0}) \underline{\mathbf{r}}_{\mathrm{Ob}}$$
(6.63)

Eq. 6.62 may be written as follows:

$$\begin{bmatrix} \underline{k} & \underline{k}_{b} \\ \\ \underline{k}_{b}^{T} & \underline{k}_{bb} + \underline{s}_{f}(0) \end{bmatrix} \begin{cases} \widetilde{\underline{r}} \\ \\ \widetilde{\underline{r}}_{b} \end{cases} = \begin{cases} \underline{R} \\ \\ -\underline{R}_{Ob} \end{cases}$$
(6.64)

Equation 6.64 thus provides a means to solve directly for the relative displacements; the rigid body displacements enter only in the evaluation of the rigid body foundation forces $\frac{R}{-0b}$. To evaluate these forces, it is convenient to express the rigid body displacements r_{-0b} in terms of

$$\mathbf{E}^{cg} = \begin{cases} \mathbf{u}^{cg} \\ \mathbf{v}^{cg} \\ \mathbf{\theta}^{cg} \end{cases}$$

the three components of displacement of the center of gravity (c.g) of the base (Fig. 6.4):

$$\underline{\mathbf{r}}_{0b} = \underline{\mathbf{D}} \, \underline{\mathbf{r}}_{-cg} \tag{6.65}$$

where the transformation matrix \underline{D} involves only the geometry of the location of the base nodal points and the c.g of the base.

Similarly, the resultant base forces $\frac{R}{-cg}$ corresponding with the displacements $\frac{r}{-cg}$ can be expressed in terms of the nodal forces $\frac{R}{-0b}$ as follows:

$$\underline{\mathbf{R}}_{\mathbf{cg}} = \underline{\mathbf{p}}^{\mathrm{T}} \underline{\mathbf{R}}_{\mathrm{Ob}}$$
(6.66)

where

$$\mathbf{R}_{cg} = \left\{ \begin{array}{c} \mathbf{U} \\ \mathbf{V} \\ \mathbf{M} \end{array} \right\}$$

- 59 -







FIG. 6.4 RIGID BODY DISPLACEMENTS AT THE CENTER OF GRAVITY OF THE DAM BASE

in which

- U = resultant horizontal force
- V = resultant vertical force
- M = resultant moment about the center of gravity

Introducing Eqs. 6.63 and 6.65 into Eq. 6.66 yields

$$\underline{\mathbf{R}}_{cg} = \underline{\mathbf{D}}^{\mathrm{T}} \underline{\mathbf{R}}_{0b} = \underline{\mathbf{D}}^{\mathrm{T}} \underline{\mathbf{\mathcal{S}}}_{f}(0) \underline{\mathbf{r}}_{0b} = \underline{\mathbf{D}}^{\mathrm{T}} \underline{\mathbf{\mathcal{S}}}_{f}(0) \underline{\mathbf{p}} \underline{\mathbf{r}}_{cg}$$

from which

$$\mathbf{R}_{-cg} = \mathbf{\mathcal{I}}_{f0} \mathbf{\underline{r}}_{cg} \tag{6.67}$$

if the symbol

$$\underline{\mathbf{\mathcal{S}}}_{\mathbf{f}\mathbf{O}} = \underline{\mathbf{p}}^{\mathrm{T}} \, \underline{\mathbf{\mathcal{S}}}_{\mathbf{f}}(\mathbf{O}) \, \underline{\mathbf{p}} \tag{6.68}$$

is used to represent the rigid body foundation stiffness matrix.

Solving Eq. 6.67 for the rigid body displacements:

$$\underline{\mathbf{r}}_{cg} = \underbrace{\mathbf{g}}_{f0}^{-1} \underbrace{\mathbf{R}}_{cg}$$

and making use of Eqs. 6.65 and 6.63, the nodal forces $\frac{R_{ob}}{r_{ob}}$ associated with rigid body displacements are expressed in terms of $\frac{R_{ob}}{r_{cg}}$, the resultant forces at the base due to all applied loads R:

$$\frac{R_{0b}}{R_{0b}} = \frac{g}{f}(0) \frac{D}{D} \frac{g^{-1}}{f_{0}} \frac{R_{cg}}{R_{cg}}$$
(6.69)

With this result the right hand side of Eq. 6.64 is known. This equation can be solved for the relative nodal displacements, from which all desired stresses in the dam can be found, without any specific consideration of the rigid body displacements.

6.5 Special Cases

6.5.1 Dam-Foundation System, No Water

The frequency-domain equations for the dam, without any impounded water and including dam-foundation interaction effects, can be obtained from the general Eqs. 6.52 and 6.33 by simply dropping the hydrodynamic terms $\underline{R}_{0}^{\ell}(\omega)$ and $\underline{R}_{1}^{f}(\omega)$. The resulting equations are

$$\underline{s}(\omega) \underline{z}^{\ell}(\omega) = \underline{L}^{\ell}(\omega); \quad \ell = \mathbf{x}, \mathbf{y}$$
(6.70)

in which

$$\mathbf{S}_{nn}(\omega) = \left[-\omega^{2} + (1+i\eta)\lambda_{n}^{2}\right] \underline{\boldsymbol{\psi}}_{n-c}^{\mathrm{T}} \underline{\boldsymbol{\psi}}_{n} + \underline{\boldsymbol{\psi}}_{n}^{\mathrm{T}} \left[\widetilde{\boldsymbol{\mathcal{Z}}}_{f}(\omega) - (1+i\eta)\widetilde{\boldsymbol{\mathcal{Z}}}_{f}(0) \right] \underline{\boldsymbol{\psi}}_{n}$$
(6.71a)

$$s_{nj}(\omega) = \underline{\psi}_{n}^{T} \left[\widetilde{\underline{\mathscr{S}}}_{f}(\omega) - (1+i\eta) \widetilde{\underline{\mathscr{S}}}_{f}(0) \right] \underline{\psi}_{j}$$
(6.71b)

$$\mathbf{L}_{n}^{\ell} = -\underline{\Psi}_{n-c-\ell}^{\mathrm{T}} \underline{1}_{c}^{\ell}$$
(6.71c)

Equations 6.70 and 6.71 are the same as those presented in Chapter 5 for dynamic response of dams including structure-foundation interaction.

6.5.2 Dam on Rigid Foundation, with Water

In this case, the base DOF can be eliminated from the analysis; the associated structure-foundation system defined in Sec. 6.2.4 is simply the dam on rigid foundation. Denoting the mass and stiffness matrices of the dam on rigid base by \underline{m} and \underline{k} , the natural vibration frequencies ω_n and corresponding mode shapes $\underline{\phi}_n$ of this system are solutions of the eigenvalue problem

$$\underline{k}\underline{\phi}_{n} = \omega_{n}^{2}\underline{m}\underline{\phi}_{n} \tag{6.72}$$

The structural displacements are expressed as a linear combination of the first J mode shapes:

$$\underline{\underline{r}}^{\ell}(t) = \sum_{j=1}^{J} \underline{Y}_{j}^{\ell}(t) \underline{\phi}_{j} \text{ and } \underline{\underline{r}}^{\ell}(\omega) = \sum_{j=1}^{J} \underline{\overline{Y}}_{j}^{\ell}(\omega) \underline{\phi}_{j}$$
(6.73)

Thus, the frequency domain equations for the dam on a rigid foundation,

including hydrodynamic effects, can be obtained by appropriately specializing Eqs. 6.52 and 6.53. The resulting equations are

$$\underline{S}(\omega) \quad \underline{\underline{Y}}^{\ell}(\omega) = \underline{L}^{\ell}(\omega) \quad ; \quad \ell = x, y \quad (6.74)$$

in which

$$S_{nn}(\omega) = \left[-\omega^{2} + (1+i\eta)\omega_{n}^{2}\right] \Phi_{n-\Phi}^{T} + \omega^{2} \left\{\Phi_{n}^{f}\right\}^{T} \overline{E}_{n}^{f}(\omega)$$
(6.75a)

$$\mathbf{S}_{nj}(\omega) = \omega^{2} \{ \underline{\phi}_{n}^{\mathbf{f}} \}^{\mathrm{T}} \underline{\mathbf{R}}_{j}^{\mathbf{f}}(\omega)$$
(6.75b)

$$\mathbf{L}_{n}^{\ell}(\omega) = - \underline{\phi}_{n-1}^{\mathrm{T}} \underline{1}^{\ell} + \{\underline{\phi}_{n}^{\mathrm{f}}\}^{\mathrm{T}} \underline{\overline{\mathbf{R}}}_{0}^{\ell}(\omega)$$
(6.75c)

These equations are equivalent to those in Chapter 4 (Eqs. 4.16-4.18 and 4.34-4.35) developed directly for dams with water on rigid foundation.

6.5.3 Dam on Rigid Foundation, No Water

Equations governing complex frequency responses of a dam without water are obtained by further specializing Eqs. 6.74 and 6.75, simply by setting to zero the hydrodynamic forces $\overline{\mathbb{R}}_{0}^{\ell}(\omega)$ and $\overline{\mathbb{R}}_{j}(\omega)$. The coefficients of Eq. 6.74 are then given by:

$$S_{nn}(\omega) = \left[-\omega^2 + (1+i\eta)\omega_n^2\right] \Phi_n^{T} \Phi_n \qquad (6.76a)$$

$$S_{nj}(\omega) = 0 \tag{6.76b}$$

$$L_{n}^{\ell}(\omega) = -\phi_{n-1}^{T} \underline{l}^{\ell}$$
(6.76c)

The matrix $\underline{S}(\omega)$ is now diagonal since the chosen Ritz vectors are the natural modes of vibration of the system considered.

Comparing Eqs. 6.76 and 6.75, it is apparent that $\underline{S}(\omega)$ and $\underline{L}(\omega)$ are modified due to the presence of water. The diagonal terms of $\underline{S}(\omega)$ are modified by an additional frequency-dependent mass. Off-diagonal mass terms appear because the Ritz vectors, the natural modes of vibration of the dam (without water), are not the vibration modes of the combined dam-water system, and they become coupled due to hydrodynamic interaction effects. An additional term appears in $\underline{L}^{\ell}(\omega)$ due to hydrodynamic loads. These equations are equivalent to those in Chapter 3 (Eq. 3.11) derived directly for the dam without water on rigid foundation.

6.6 Computer Program

Based on the analytical procedures developed in this chapter, a computer program has been written in FORTRAN IV to numerically evaluate the responses of concrete gravity dams, including various effects of both the water and the foundation. These include effects arising from interaction between the dam and foundation, dam and water, water and foundation, and from interaction among all three substructures--dam, water, and foundation. The dynamic stiffness matrix for the foundation is not computed within the computer program. Thus it should be computed externally and supplied to the computer program. Available data and methods [32] may be utilized to determine the dynamic stiffness matrix for a foundation region idealized as a viscoelastic half plane.

The computer program is capable of analyzing the response of a dam for four conditions: Dam on rigid foundation excluding hydrodynamic effects, dam on rigid foundation including hydrodynamic effects, dam on flexible foundation excluding hydrodynamic effects, and dam on flexible foundation including hydrodynamic effects.

The response of dam-water-foundation systems, idealized as described in Chapter 2, to horizontal and vertical components of ground motion can be analyzed. The dam is treated as an assemblage of two-dimensional finite elements; the impounded water and foundation are treated as continua.

The output from the computer program includes the complex frequency response functions describing the response to harmonic ground motions and the complete time-history of displacements and stresses throughout the dam. Because the program is capable of including any number of Ritz shape vectors, results can be obtained to any desired degree of accuracy.

The users guide and listing of the program are included in Appendices A and B.

7. PRELIMINARY RESULTS FOR PARAMETER SELECTION

7.1 Scope of the Chapter

Several preliminary analyses were carried out to assist in the selection of parameters defining the system and also those necessary to carry out the analyses for the dynamic response study presented in Chapter 8. Results from two analyses of a dam foundation system, assuming the system to be in plane stress in one and in plane strain in the other, are presented to provide a basis for selecting one of the two assumptions. Similarly, response of the dam is presented and compared for two damping models, viscous damping and constant hysteretic damping. With the aid of response results, the minimum number of parameters -- those that have the most important influence on the dam response -- defining the elastic modulii of the dam and foundation are identified. The number of generalized coordinates necessary to obtain accurate responses over the desired range of excitation frequency is determined with the aid of response results obtained with varying numbers of these coordinates.

7.2 Systems, Ground Motions, and Response Quantities

7.2.1 Systems

The dam is idealized as having a triangular cross section with a vertical upstream face and a downstream face with a slope of 0.8:1. The dam is assumed to be homogeneous with linearly elastic and isotropic properties for mass concrete: Young's modulus $E_s = 2 \text{ or } 4 \text{ million psi}$; Poisson's ratio = 0.2; and unit weight = 155 lbs/ft³. Representative values for the latter two parameters have been chosen. The possible small variation around the chosen values will have little influence on the dam response. Two damping assumptions are considered: Viscous damping with damping ratio $\xi_s = 0.05$ in each natural mode of vibration of the dam alone (without water) on a fixed base, and constant hysteretic damping with the energy loss coefficient $\eta_c = 0.1$.

The finite element idealization for a monolith of the dam (Fig. 7.1) consists of 20 quadrilateral elements and 26 nodal points which provide 42 degrees of freedom on a fixed base and 52 on a flexible base.

The dam monolith is supported on the surface of a homogeneous, isotro-



FIG. 7.1 FINITE ELEMENT IDEALIZATION OF A DAM MONOLITH WITH TRIANGULAR CROSS-SECTION

pic, linearly viscoelastic half space. The following properties are assumed for the foundation materials: $E_f = 2E_s$; Poisson's ratio = 1/3; and unit weight, $w_f = 165 \text{ lbs/ ft}^3$. The latter two parameter values are not varied because the response is insensitive to the small variations in those parameter values that can occur in practical problems. The foundation soil or rock is idealized as a constant hysteretic solid with several values of energy loss coefficient $\eta_f = 0.1$, 0.25, and 0.5, or as a Voigt solid with energy loss coefficient $\xi_f = 0.1$. Impounded water in the reservoir has a constant depth H and is idealized as extending to infinity in the upstream direction. The unit weight of water w = 62.5 lbs/ft³ and the velocity of sound in water = 4720 ft/sec. Two values of the depth of water H, relative to the height of the dam H_s, are considered: $H/H_s = 0$ (no water) and 1 (full water).

The dam and foundation both are assumed to be in a state of generalized plane stress or plane strain.

7.2.2 Ground Motions

The excitation for the dam-water-foundation system is defined by the two components of free-field ground motion in the plane of a monolith of the dam: the horizontal component transverse to the dam axis; and the vertical component. Each component of ground acceleration is assumed to be harmonic, $a_g^X(t) = a_g^Y(t) = e^{i\omega t}$, with the excitation frequency to be varied over a relevant range.

7.2.3 Response Quantities

The response of the dam to harmonic free field ground motion was determined by the analytical procedures presented in Chapter 6. Five mode shapes on fixed base or generalized coordinates, as appropriate, were included in the analysis. Although the analysis led to response at all nodal points, only the results for a few selected degrees-of-freedom are presented. For dams on rigid foundations, the complex frequency response functions for horizontal acceleration at the crest of the dam are presented. In addition, the complex frequency response functions for the horizontal acceleration at the center of the base of the dam and for vertical accelerations at the upstream and downstream edges of the base are presented when the interaction between the structure and flexible foundation material is included. The absolute value (or modulus) of the complex-valued acceleration response is plotted against normalized excitation frequency parameter ω/ω_1 , where ω_1 = fundamental fixed base natural frequency of the dam alone.

7.3 Comparison of Plane Strain and Plane Stress Assumption

Because, at the present time, the analysis procedure presented in Chapter 6 has been implemented only for two-dimensional systems, its application is restricted to systems in plane stress or plane strain. Although, as mentioned in Chapter 2, neither of the two models are strictly valid, the former is better for the dam and the latter for the foundation. However, in order to define the dam-foundation system on a consistent basis, the same model should be employed for both substructures.

The results of two analyses of the dam-foundation system described in Section 7.2.1, assuming the system to be in plane stress in one and in plane strain in the other, are presented in Fig. 7.2. Constant hysteretic damping with energy loss coefficient $\eta = 0.1$ was assumed for the dam and foundation materials. The results presented are for systems with $E_f/E_s = 2$, independent of the E_f and E_s value separately. It is obvious from Fig. 7.2 that the response of the dam, including effects of dam-foundation interaction, is essentially the same under two assumptions -- plane stress or plane strain.

The dam-foundation system is assumed to be in plane stress for all the results in this and later chapters. As discussed in Chapter 2, based on physical behavior, this assumption is appropriate for describing the behavior of a concrete gravity dam vibrating at large amplitudes of motion, but not for the foundation. However, based on the above results, the same assumption may be made for the foundation behavior without significant loss in accuracy.

7.4 Elastic Moduli Parameters

The complex frequency acceleration response functions describe the steady-state acceleration response of dams to harmonic excitations. These functions for a dam on a rigid foundation and no water, plotted against normalized excitation frequency parameter ω/ω_1 , apply to dams of all heights provided they all have the same cross-sectional shape. Furthermore, they are independent of the unit weight w_e and Young's modulus E_e for the dam



2 RESPONSE OF A DAM WITHOUT WATER, INCLUDING DAM-FOUNDATI INTERACTION, TO HARMONIC HORIZONTAL GROUND ACCELERATION OBTAINED UNDER TWO ASSUMPTIONS FOR THE SAME SYSTEM: (1) PLANE STRESS; AND (2) PLANE STRAIN concrete. It can be concluded from the equations of motion for the dam, including the dam-foundation interaction but excluding hydrodynamic effects, that the complex frequency response functions plotted against normalized excitation frequency do not depend separately on E_s and E_f but only the ratio E_f/E_s . These functions for dams on rigid foundations vary significantly with E_s when hydrodynamic effects are considered [22]. They vary even for dams on flexible foundations: for a fixed value of E_f/E_s , the value of E_s has influence on the response of the dam including dam-water interaction effects. Figure 7.3 demonstrates that this influence is minor in the response to horizontal ground motion but significant in the response to vertical ground motion.

7.5 Damping Models and Parameters

7.5.1 Dam

The analytical procedures presented in Chapters 3-6 included two models -- viscous damping and constant hysteretic damping -- for energy dissipation in the dam. Viscous damping had been selected earlier [22], in studying hydrodynamic effects in earthquake response of dams. Constant hysteretic damping is preferable for conceptual as well as computational reasons, especially when structure-soil interaction is included in the analysis [29]. Therefore, all numerical results presented in Chapters 8 and 9 are for dams with constant hysteretic damping. In order to provide a basis for comparing these results with those presented earlier [22], response results using both damping models are presented in this section and compared.

Using the analysis procedure in Chapter 6, the steady-state response of the idealized dam monolith (Fig. 7.1) to harmonic ground acceleration in the horizontal direction is determined for several cases. Assuming the concrete properties listed in Section 7.2.1, with Young's modulus = 4×10^6 psi, and considering the foundation to be rigid, the complex frequency response functions for the dam were determined for the following four cases. (The first five vibration modes were included in the analysis.)

- 1. Viscous damping, ξ_s = 0.05 for all vibration modes; no water
- 2. Constant hysteretic damping, $\eta_s = 0.10$; no water
- 3. Viscous damping, $\xi_s = 0.05$ for all vibration modes; full water
- 4. Constant hysteretic damping, $\eta_s = 0.10$; full water



FIG. 7.3 INFLUENCE OF YOUNG'S MODULUS ${\rm E_S}$ OF DAM CONCRETE ON RESPONSE OF A DAM, INCLUDING EFFECTS OF WATER AND FOUNDATION, TO HARMONIC GROUND MOTION. ${\rm E_f/E_S}$ IS THE SAME IN BOTH CASES.

The selected values of damping factor n in constant hysteretic damping and viscous damping ratio ξ_s for all modes of vibration are related by $n_s = 2\xi_s$. As a result, the energy dissipated per vibration cycle in each natural mode of vibration at the resonant excitation frequency would be the same for the two damping models. The energy dissipated per cycle in constant hysteretic damping is independent of excitation frequency, but varies linearly with frequency for viscous damping.

Because the energy dissipated at the fundamental resonant frequency of the dam alone is the same for the two damping models and the selected parameters, the dam response is independent of the type of damping (Fig. 7.4). At the fundamental resonant frequency of the dam, including hydrodynamic effects, the energy dissipated in constant hysteretic damping will be the same but reduced in viscous damping, compared to the energy dissipated at the natural frequency of the dam alone. As a result, the resonant response is significantly smaller for the dam with constant hysteretic damping, compared to the results for viscous damping (Fig. 7.4). The differences in resonant responses for the two damping models are directly related to the reduction in resonant frequencies due to dam-water interaction effects. Because such reduction is relatively small for higher resonant frequencies, the corresponding resonant responses are affected less by the damping model (Fig. 7.4). At excitation frequencies not close to resonant frequencies, damping has little effect on the response and it is essentially the same with the two damping models.

Constant hysteretic damping with $n_s = 0.1$ is assumed for the dam in obtaining all subsequent results in Chapters 7, 8, and 9.

7.5.2 Foundation

Numerical results and analytical procedures have been presented to determine dynamic stiffness matrices for a viscoelastic half-plane of Voigt or constant hysteretic materials [32]. In this section, computed responses of the dam including structure-soil interaction effects are presented to aid in selecting the damping model and parameters for the foundation material.

The response of the idealized dam monolith (Fig. 7.1) without water but including dam-foundation interaction effects is computed by the procedures of Chapter 6. The following properties are selected for the dam in addition to those listed in Section 7.2.1: Young's modulus $E_s = 4 \times 10^6$ psi, constant hysteretic damping factor $\eta_s = 0.1$. The foundation modulus E_f is twice that of the dam and, in addition to the properties of the foundation material listed

- 72 -





FIG. 7.4

in Section 7.2.1, the following are selected: $E_f/E_s = 2$, energy loss coefficients $\xi_f = 0.1$ for the Voigt solid; and $n_f = 0.1$ for the constant hysteretic solid. The two dam-foundation systems analyzed have identical properties with one exception: the foundation is of Voigt material in one case, and constant hysteretic material in the other case. Including five generalized coordinates in the analysis, the complex frequency response functions for the dam were obtained for the two systems. The results presented in Fig. 7.5a demonstrate that the damping model for the foundation has little effect on the resonant frequencies of the dam or on the response except at the fundamental resonant frequency. The fundamental resonant response of the dam is somewhat larger with Voigt foundation material.

For a Voigt solid, the energy loss per cycle of harmonic vibration is proportional to the excitation frequency. However, over a considerable range of frequencies, rocks and soils exhibit energy loss essentially independent of the frequency of vibration. Such materials are better idealized as constant hysteretic solids. This is the model adopted for the foundation material for the dam responses presented in Chapters 8 and 9. Based on the above discussion, however, those results are also indicative of the response of dams on Voigt foundation materials.

Using the properties mentioned above for the dam and foundation, the analysis was repeated for two other values of damping coefficient for a constant hysteretic foundation: $n_f = 0.25$ and 0.50. The complex frequency response functions for the dam with constant hysteretic damping, $n_g = 0.1$, on a constant hysteretic foundation are presented in Fig. 7.5b for three values of the damping factor $n_f = 0.1$, 0.25, and 0.50. The responses are insensitive to the foundation damping except in a neighborhood of the fundamental resonant frequency. Even this resonant response is not influenced greatly by damping; increasing $n_f = 0.1$ to 0.25 reduces the resonant response by only 13%. Damping in the foundation is modeled as a constant hysteretic solid with $n_f = 0.1$ in obtaining all subsequent results in Chapters 7, 8, and 9. The assumed value of damping factor is appropriate for sound rocks at sites of major concrete dams. Because the dam response is not very sensitive to foundation damping, however, those results are approximately indicative of the dam response with other levels of foundation damping.



FIG. 7.5 RESPONSE OF A DAM WITHOUT WATER TO HARMONIC, HORIZONTAL GROUND ACCELERATION: (A) INFLUENCE OF DAMPING MODEL, VOIGT OR CONSTANT HYSTERETIC, FOR THE FOUNDATION. (B) INFLUENCE OF DAMPING COEFFICIENT $\eta_{\rm f}$ OF A CONSTANT HYSTERETIC FOUNDATION.

7.6 Number of Generalized Coordinates

The governing equations in the frequency domain were transformed from the physical nodal point displacement coordinates to generalized coordinates (Chapters 3-6). As many of these generalized coordinates should be included in the analysis as necessary to obtain accurate responses over a desired range of excitation frequencies. The required number of generalized coordinates will generally be a small fraction of the number of degrees-offreedom in the finite element idealization of the dam.

The natural modes of vibration of the dam alone were selected as the generalized coordinates for analysis of the dam on rigid foundation, with or without water (Chapters 3 and 4). In order to accurately obtain the response of the dam without water on a rigid base, all the modes of vibration should be included which have natural frequencies within and close to the range of excitation frequenices of interest, and contribute significantly to the response in this frequency range. Accelerograph records accurately reproduce ground motion components with frequencies up to approximately 25 cps. The first few, say five, natural vibration modes, would generally be sufficient to obtain accurate responses of concrete gravity dams -- they have relatively high natural frequencies -- in this frequency range. Dam-water interaction significantly reduces the fundamental resonant frequency but has little influence on the higher resonant frequencies. Thus, usually, the same number of natural vibration modes would be sufficient for analysis of the response of the dam, with or without water.

The eigenvectors (vibration mode shapes) of an associated dam-foundation system, the dam with the foundation characterized by the stiffness matrix $\underline{\mathscr{G}}_{\mathbf{f}}(0)$ equal to the dynamic stiffness matrix $\underline{\mathscr{G}}_{\mathbf{f}}(\omega)$ at $\omega = 0$, were selected as the generalized coordinates for response analysis of the dam including dam-foundation interaction effects (Chapter 5). Unlike the natural modes of vibration of the dam on a rigid base, these generalized coordinates are not physically meaningful. Thus the number of these generalized coordinates that need to be included in the analysis is not as obvious. The complex frequency response functions for the dam with horizontal ground acceleration as the excitation was obtained from four analyses using the procedure of Chapters 5 and 6, including 5, 7, 19, and 15 generalized coordinates, respectively. The results presented in Fig. 7.6 are for the idealized dam





monolith, without water, but including dam-foundation interaction effects. The first 8 eigenvalues Ω_{η} (vibration frequenices) of the associated damfoundation system are identified along the frequency axis. It is apparent that, in order to obtain satisfactory results for responses over some frequency range, all the generalized coordinates associated with eigenvalues within the frequency range plus a few, say three, more should be included in the analysis. Based on this criterion, the frequency-range represented in earthquake ground motion, and the fact that complex frequency response functions decrease at larger excitation frequencies, ten generalized coordinates were considered sufficient for the results presented in Chapters 8 and 9.

8.1 Scope of Chapter

The analysis procedure developed in Chapter 6 can be used to evaluate the response of concrete gravity dams subjected to earthquake ground motion. In this procedure, the complex frequency response functions describing the response to harmonic ground motion are determined, followed by the Fourier transform procedures to compute responses to arbitrary ground motion. In this chapter, the complex frequency response functions for an idealized cross section representative of concrete gravity dams are determined for a range of the important system parameters characterizing the properties of the dam, foundation soil or rock, and impounded water. The effects of damwater interaction and of dam-foundation interaction on the dynamic response of the dam are studied.

8.2 Systems, Ground Motions, Cases Analyzed and Response Quantities

8.2.1 Systems

The idealized cross section considered as representative of concrete gravity dams is a triangle with a vertical upstream face and a downstream face with a slope of 0.8:1. The dam is assumed to be homogeneous with linearly elastic and isotropic properties for mass concrete: Young's modulus $E_s = 3$, 4 or 5 million psi; Poisson's ratio = 0.2; and unit weight = 155 lbs/cu.ft. Energy dissipation in the structure is represented by constant hysteretic damping, with the energy loss coefficient η_s selected as 0.1. This is equivalent to a damping ratio of 0.05 in all the natural modes of vibration of the dam alone (without water) on a fixed base.

The finite element idealization for a monolith of the dam is shown in Fig. 8.1, consisting of 20 quadrilateral elements and 26 nodal points, which provide 42 degrees of freedom on a fixed base and 52 on a flexible base.

The dam is supported on the surface of a homogeneous, isotropic, linearly viscoelastic half space. For the foundation material, several values for the Young's modulus E_f are considered, Poisson's ratio = 1/3, and unit weight = w_f = 165 lbs/cu.ft. The foundation soil or rock is idealized as a constant hysteretic solid with energy loss coefficient η_f = 0.1.

- 79 -



FIG. 8.1 FINITE ELEMENT IDEALIZATION OF A DAM MONOLITH WITH TRIANGULAR CROSS-SECTION

Impounded water in the reservoir has constant depth H and is idealized as extending to infinity in the upstream direction. Several different values for H, the depth of water, relative to H_s, the height of dam, are considered: $H/H_s = 0$ (empty reservoir), 0.8 (partially filled reservoir), and 1 (full reservoir). The unit weight of water w = 62.5 lbs/cu.ft. and the velocity of sound in water = 4720 ft/sec.

The dam and foundation are assumed to be in a state of generalized plane stress. This assumption, although not most appropriate for the foundation material, is dictated by the expected behavior of the dam (see Chapter 2).

8.2.2 Ground Motions

The excitation for the dam-water-foundation system is defined by the two components of free-field ground motion in the plane of a monolith of the dam: the horizontal component transverse to the dam axis, and the vertical component. Each component of ground acceleration is assumed to be harmonic, $a_g^X(t) = a_g^Y(t) = e^{i\omega t}$, with the excitation frequency to be varied over a relevant range.

8.2.3 Cases Analyzed

As concluded in Chapter 7, the most important parameters influencing the complex frequency response functions for dams on rigid foundations are E, the Young's modulus of elasticity for mass concrete in the structure, and H/H, the ratio of water depth to dam height. When dam-foundation interaction is considered, an additional parameter is important: E_{f}/E_{c} , the ratio of Young's moduli for the foundation and dam materials. For a fixed value of E_f/E_c , the value of E_c influences the response of the structure to a minor extent for horizontal ground motion and significantly for vertical ground motion. However, to keep the number of cases analyzed to a minimum, the E_ value is not varied. Several dam-water-foundation systems were defined by values for parameters shown in Table 8.1. The complex frequency response functions for each case were determined. The following figures are presented in a manner to facilitate study of: the effects of structure-water interaction and structure-foundation interaction, considered separately; and the combined effects of these two sources of interaction, on the response of the dam.

CASES ANALYZED RESPONSE TO HORIZONTAL AND VERTICAL GROUND MOTION SEPARATELY: TABLE 8.1

		FOUNDATION PR	DPERTIES	WATER IN RESE	RVOIR
CASE	E _S - psi	Rigid or Flexible	$\mathrm{E_{f}/^{E_{s}}}$	Condition	H/H _S
	¥ny*	Rigid	8	Empty	0
7	4x10 ⁶	Rigid	8	Partially Filled	0.8
m	4x10 ⁶	Rigid	8	Full	r-1
4	3x10 ⁶	Rigid	8	Full	ы
2	5x10 ⁶	Rigid	8	Full	
9	Any*	Flexible	4	Empty	0
7	4x10 ⁶	Flexible	4	Full	
æ	Any*	Flexible	7	Empty	0
Q	4 x 106	. Flexible	7	Partially Filled	0.8
10	4x10 ⁶	Flexible	7	Full	
	Any*	Flexible		Empty	0
12	4 x 106	Flexible	1	Full	1

Results for all cases are presented in a form applicable to dams of all heights with the idealized triangular cross-section and chosen ${\rm E}_{\rm S}$ value. Results for these cases, when presented in an appropriately normalized form, are valid for all E_s. *

H = depth of water in reservoir H_{S} = height of the dam E_{S} = Young's modulus of elasticity of concrete in the dam E_{f} = Young's modulus of elasticity of foundation material

The hydrodynamic pressures due to vertical ground motion depend on, among other parameters, the reflection coefficient at the reservoir bottom $\alpha = [(C_{rw_r}/Cw) - 1]/[(C_{rw_r}/Cw) + 1]$ where w_r and C_r are the unit weight and P-wave velocity for the rock, w and C are the unit weight and sound velocity in water. It can be easily seen that $\alpha \leq 1$ with the maximum value corresponding to rigid rock. For $E_s = 4 \times 10^6$ psi and $E_f/E_s = \infty$, 4, 2, 1, and the values of w and w_r mentioned earlier, the values of α are 1, 0.85, 0.80 and 0.72, respectively.

8.2.4 Response Quantities

The response of the dam to harmonic free-field ground motion was determined by the analytical procedures presented in Chapter 6. Ten generalized coordinates were included in the analysis. Based on Sec. 7.6, the resulting complex frequency response functions should be accurate for excitation frequencies up to approximately 25 cps. Although the analysis led to response at all nodal points, only the results for a few selected degrees-of-freedom are presented. For dams on rigid foundations, the complex frequency response functions for horizontal and vertical accelerations at the crest of the dam are presented. In addition, the complex frequency response functions for horizontal acceleration at the center of the base of the dam and for the vertical accelerations at the upstream edge of the base are presented when interaction between the dam and flexible foundation material is included.

All the complex frequency response functions presented are for accelerations relative to the prescribed free-field ground acceleration. They are not direct measures of deformations within the dam. The absolute value (or modulus) of the complex valued acceleration response is plotted against the normalized excitation frequency parameter ω/ω_1 , where $\omega_1 =$ fundamental fixed-base natural frequency of the dam alone. When presented in this form, the results apply to dams of all heights with the idealized triangular cross section and chosen E_s value. Furthermore, if the reservoir is empty, the plotted results are also independent of the E_s value.

8.3 Complex Frequency Responses

8.3.1 Dam-Water Interaction Effects

In order to identify the effects of water on the dynamic response of the dam, the results of analysis cases 1, 2, and 3 (Table 8.1) to horizontal and vertical ground motion, separately, are presented in Figs. 8.2 and 8.3,

- 83 -



FIG. 8.2 HYDRODYNAMIC EFFECTS IN THE RESPONSE OF DAMS TO HARMONIC, HORIZONTAL GROUND MOTION. RESULTS ARE PRESENTED FOR VARYING WATER DEPTH: CASES 1, 2, AND 3 OF TABLE 8.1



FIG. 8.3 HYDRODYNAMIC EFFECTS IN THE RESPONSE OF DAMS TO HARMONIC, VERTICAL GROUND MOTION. RESULTS ARE PRESENTED FOR VARYING WATER DEPTH: CASES 1, 2, AND 3 OF TABLE 8.1

respectively. The response curve for the dam without water $(H/H_S = 0)$ is representative of a multi-degree-of-freedom system with constant mass, stiffness, and damping parameters. However, dam-water interaction introduces frequency dependent terms in the equations of motion of the dam, resulting in complicated shapes of the response curves. The response behavior is especially complicated at excitation frequencies in the neighborhood of the natural frequencies of water in the reservoir. In particular, the response curve has a double resonant peak near ω_1 and ω_1^r , which is especially pronounced for the case of vertical ground motion. At these frequencies the hydrodynamic terms in the equations of motion become unbounded but the response has finite limits (see Section 6.2.7). Based on the results of Figs. 8.2 and 8.3, the following observations can be made.

Presence of water results in a significant decrease in the fundamental resonant frequency of the dam but relatively little decrease in the higher resonant frequencies. At an excitation frequency ω smaller than ω_1^r , the fundamental resonant frequency of water in the reservoir, the effect of damwater interaction is equivalent to an added mass and real-valued load; their magnitude depends on the excitation frequency. This added mass, in addition to reducing the fundamental resonant frequency of the dam, has the indirect effect of reducing the apparent damping ratio for the fundamental mode, resulting in narrower band width at resonance and larger resonant response. Some of the reduction in the resonant band width is, however, a consequence of the double resonant peak. At $\omega > \omega_1^r$, the effect of dam-water interaction is equivalent to frequency dependent additional mass, damping and load; the load is complex valued for horizontal and vertical ground motion except that it is real valued for vertical ground motion with α = 1. The added mass is relatively small, which has relatively little effect on the higher resonant frequencies, but the added damping is significant, resulting in decrease in the response at the higher resonant frequencies. The fundamental resonant frequency of the dam, including hydrodynamic effects, is less than both the fundamental natural frequency of the dam with no water ω_1 , and of water in the reservoir, ω_1^r .

Comparing the results for cases 1 ($H/H_S = 0$), 2 ($H/H_S = 0.8$), and 3 ($H/H_S = 1$), it is seen that the fundamental resonant frequency is affected most by water in the upper parts of the dam height. At higher excitation frequencies, there is little difference in the responses for $H/H_S = 0.8$ and 1, except that the response to vertical ground motion in the neighborhood of the second resonant frequency is strongly affected by increasing H/H_S from 0.8 to 1.

The vertical and horizontal motions at the crest of the dam are obviously related by the horizontal and vertical components of the vibration mode shapes. Vertical motion at the crest of the dam, while smaller than the horizontal motion, as expected, is not negligible for either horizontal or vertical ground motions.

Comparing the responses of the dam to horizontal and vertical ground motions (Figs. 8.2 and 8.3), it is apparent, consistent with common view, that without water the response to vertical ground motion is relatively small. However, with a full reservoir the response to vertical ground motion is almost as large as that due to horizontal ground motion when $\alpha < 1$ and it is even larger when $\alpha = 1$. The response to vertical ground motion is large because the hydrodynamic forces act in the horizontal direction on the vertical upstream face of the dam, causing significant lateral response of the dam. Hydrodynamic effects are seen to have especially large influence on the response to vertical ground motion.

The response of the dam alone without water $(H/H_S = 0)$, when presented in the form of Figs. 8.2 and 8.3, is independent of the modulus of elasticity Es of the concrete. Similarly, the response curves including hydrodynamic effects, do not vary with E_s if water compressibility is neglected [22]. However, results from analysis of cases 3, 4, and 5 (Table 8.1) presented in Figs. 8.4 and 8.5 demonstrate that the E value affects the response functions when compressibility of water is included. This effect is primarily on the fundamental resonant frequency and on the response in the neighborhood of this frequency. The fundamental resonant frequency of the dam decreases due to hydrodynamic effects to a greater degree for the larger values of E_S. Increase in E_S causes larger amplification of response but over a narrower frequency band. The resonant response to horizontal ground motion is affected little by variations in Es but is influenced substantially for vertical ground motion. At higher excitation frequencies, the response functions for both components of ground motion are insensitive to Es. With decreasing Es the effects of water compressibility on response become smaller and the response approaches the incompressible case.

- 87 -



FIG. 8.4 INFLUENCE OF YOUNG'S MODULUS E_S OF DAM CONCRETE ON RESPONSE OF DAMS TO HARMONIC, HORIZONTAL GROUND MOTION. E_S IS VARIED AS SHOWN: CASES 3, 4, AND 5 OF TABLE 8.1



FIG. 8.5 INFLUENCE OF YOUNG'S MODULUS E_S OF DAM CONCRETE ON RESPONSE OF DAMS TO HARMONIC, VERTICAL GROUND MOTION. E_S IS VARIED AS SHOWN: CASES 3,4 AND 5 OF TABLE 8.1

8.3.2 Dam-Foundation Interaction Effects

Complex frequency response functions for the dam, including effects of dam-foundation interaction, are presented in Figs. 8.6 and 8.7 for varying foundation stiffness (cases 1, 6, 8 and 11 in Table 8.1). When presented in this form, these functions do not depend separately on E_s and E_f but only on the ratio E_f/E_s . Results are presented for four values; $E_f/E_s = \infty$, 4, 2 and 1. The first represents rigid foundation material, thus reducing the system to dam on fixed base, and the smallest value represents the same elastic moduli for the foundation material and dam concrete.

Unlike water in the reservoir, the half plane does not have any resonant frequencies; the foundation impedances are smooth, slowly varying functions of the frequency [32]. As a result, structure-foundation interaction affects the response of the dam in a simpler manner than does structure-water interaction. As the E_f/E_S ratio decreases, which for a fixed E_S implies decrease of foundation modulus, the fundamental resonant frequency of the dam decreases; the response at the crest of the dam at this frequency decreases and the frequency band width at resonance increases, implying an increase in the apparent damping of the structure (Figs. 8.6 and 8.7). Accompanying this decrease in the resonant response at the crest of the dam is an increasing response at the base of the dam--which is, however a small fraction of the response at the crest -- with increasingly flexible soils (Figs. 8.6 and 8.7). Similar influences of decreasing the E_f/E_s ratio is observed at higher resonant frequencies, resulting in reduced response at the crest of the dam and increased response at the base over a wide range of excitation frequencies; however, the higher resonant frequencies are decreased to a lesser degree by structure-foundation interaction. The above mentioned effects of structure-foundation interaction are similar in response to horizontal and to vertical ground motions.

8.3.3 Dam-Water and Dam-Foundation Interaction Effects

In order to understand the effects of structure-water interaction and of structure-foundation interaction on the complex frequency response functions for the dam, the results of several cases in Table 8.1 are presented in different ways.

The results of analysis cases 8, 9, and 10 for horizontal and vertical









ground motion are presented in Figs. 8.8 and 8.9. These results, for a fixed $E_f/E_s = 2$, indicate the influence of varying depths of water in the reservoir. Whereas the response function for the dam without water $(H/H_{e} = 0)$ is influenced by structure-foundation interaction effects, for the other two cases ($H/H_5 = 0.8$, 1) dam-water interaction is also involved. Qualitatively, the effects of water on the response at the crest of the dam are generally similar whether the foundation material is rigid or flexible. However, some of the effects are relatively small. Whereas the fundamental resonant frequency is decreased by roughly the same degree as before, the apparent damping for the fundamental vibration mode is now dominated by effects of structure-foundation interaction and varies little with the depth of water. The effects of dam-water interaction on the response of a dam on flexible foundation are significant primarily at excitation frequencies in a neighborhood of the fundamental resonant frequency. At higher frequencies, dam-water interaction has little influence on the response of the dam on flexible foundation medium, except locally near the resonant frequencies of the reservoir. The principal effect of dam-water interaction on the response at the base of the dam permitted by flexible foundation medium is to decrease the fundamental resonant frequency and to increase the amplitude of response in the neighborhood of this resonant frequency. At higher frequencies the response at the base of the dam is not affected strongly by the water, except locally at the resonant frequencies of the water contained in the reservoir.

The conclusion presented in Sec. 8.3.1 is also confirmed by the results of Figs. 8.8 and 8.9. Without water the response of the dam to vertical ground motion is a small fraction of the response to horizontal ground motion, but with water the two responses at the fundamental resonant frequency are similar; at some higher frequencies the response to vertical ground motion exceeds that due to horizontal ground motion. Because of the hydrodynamic effects, the vertical ground motion is especially significant in the response of concrete gravity dams to earthquakes.

Complex frequency response functions for the dam with the reservoir full of water and varying modulus for foundation rock (cases 3, 7, 10, and 12 in Table 8.1) are presented in Figs. 8.10 and 8.11 for horizontal and vertical ground motion, respectively. Results are presented for a fixed value of $E_s = 4 \times 10^6$ psi and four values of $E_f/E_s = \infty$, 4, 2, and 1. By

- 93 -
















comparing the results with those presented in Figs. 8.6 and 8.7, it is apparent that the effects of dam-foundation interaction are generally similar, independent of the effects of dam-water interaction.

As the E_{f}/E_{s} ratio decreases, which for a fixed E_{s} implies an increasingly flexible foundation, the fundamental resonant frequency decreases. The response at the crest of the dam at this frequency decreases and the frequency bandwidth at resonance increases, implying an increase in the apparent damping of the structure. The response behavior near the fundamental resonant frequency is complicated, especially for vertical ground motion, because of the double resonant peak created by hydrodynamic effects. The response at the base of the dam, although only a small fraction of the response at the crest of the dam, is also affected by $E_{\rm f}/E_{\rm s}$. In the case of horizontal ground motion, the base accelerations at and near the fundamental resonant frequency increase with decreasing E_{f}/E_{s} , but there is no clear trend in the case of vertical ground motion. Similar influence of increasingly flexible soils is observed at higher resonant frequencies, resulting in reduced response at the crest of the dam and increased response at the base over a wide range of excitation frequencies. Structure-foundation interaction has little influence on the higher resonant frequencies and tends to suppress the response at the dam crest at these frequencies.

At excitation frequencies $\omega = \omega_n^r$, the resonant frequencies of the water in the reservoir, the response at the crest of the dam is controlled by damwater interaction and is essentially independent of the properties of the foundation rock (Figs. 8.10 and 8.11).

In order to compare the effects of structure-water interaction and structure-foundation interaction on the complex frequency response functions for the dam, results from analyses of four systems are presented together: dam on rigid foundation with no water, dam on rigid foundation with full reservoir, dam on flexible foundation with no water, dam on flexible foundation with full reservoir. The results for the first two systems (cases 1 and 3 in Table 8.1) along with those for the latter two systems for $E_{f/E_{S}} = 4$ (cases 6 and 7), $E_{f/E_{S}} = 2$ (cases 8 and 10), and $E_{f/E_{S}} = 1$ (cases 11 and 12) are presented in Figs. 8.12 to 8.14 for horizontal ground motion and Figs. 8.15 to 8.17 for vertical ground motion.









- 100 -

















The fundamental resonant frequency of the dam decreases because of damwater interaction and dam-foundation interaction, with the influence of the water being larger. For the most flexible foundations considered ($E_f = E_s$), the decrease in the fundamental frequency due to dam-foundation interaction is about the same as that due to dam-water interaction. This trend may or may not exist for higher resonant frequencies.

As mentioned earlier, the response at the crest of the dam is increased at the fundamental resonant frequency, but, in many cases, decreased at higher resonant frequencies because of dam-water interaction. These effects of dam-water interaction are similar whether the foundation is rigid or flexible for all values of E_f/E_c considered.

As mentioned earlier, the response at the crest of the dam is reduced and motions at the base of the dam, relative to the free-field motion, are possible due to flexibility of the soil. These effects of dam-foundation interaction are similar whether the reservoir is empty or full. The above mentioned effects of dam-foundation interaction tend to increase as E_f/E_s decreases, i.e., for a fixed E_c , as the foundation rock becomes increasingly flexible.

Comparing the effects of dam-water interaction and dam-foundation interaction on the resonant responses at the crest of the dam to horizontal ground motion (Figs. 8.12 to 8.14), it is apparent that the effects of the stiff rock ($E_f/E_s = 4$) are similar to those of the water, but for softer rocks ($E_f/E_s = 2$ and 1) the effects of dam-foundation interaction are relatively more significant. Whereas the amplitude of the response at the crest of the dam at the fundamental resonant frequency is increased by dam-water interaction but reduced due to dam-foundation interaction, both sources of interaction contribute towards a decrease in the response at higher resonant frequencies.

Because, as mentioned earlier, dam-water interaction influences the dam response to vertical ground motion to a much greater degree than it influences the response to horizontal ground motion, the relative effect of dam-water interaction and dam-foundation interaction on dam response to vertical ground motion (Figs. 8.15 to 8.17) is different than the above observations from responses to horizontal ground motion (Figs. 8.12 to 8.14). At the fundamental resonant frequency, the response of the dam is affected by dam-water interaction to a greater degree than it is affected by dam-foundation interaction. However, at higher resonant frequencies, the influence of dam-foundation interaction is greater. Whereas the amplitude of the response at the crest of the dam at the fundamental resonant frequency is increased by dam-water interaction but reduced due to dam-foundation interaction, both sources of interaction contribute towards a decrease in the response at higher resonant frequencies, but with one exception: Vertical acceleration response at the crest of the dam is increased by dam-water interaction even at the second resonant frequency.

The effects of water on the response at the base of the dam, permitted by a flexible foundation, are closely tied to the corresponding effects on the responses at the crest of the dam. Dam-water interaction has the effect of decreasing the resonant frequencies for the base response and modifying the response amplitudes. These effects are relatively large in the responses to vertical ground motion.

9. EARTHQUAKE RESPONSE OF PINE FLAT DAM

9.1 Scope of the Chapter

Chapter 8 presented the effects of dam water interaction and of damfoundation interaction, separately or together, on the complex frequency response functions of idealized gravity dams with triangular cross-section. This chapter presents the responses of Pine Flat Dam to a selected earthquake ground motion, analyzed using the procedures of Chapter 6 under various assumptions. The objective of this chapter is to identify the effects of dam-water interaction and dam-foundation interaction on the response of the dam.

9.2 Pine Flat Dam, Ground Motion, Cases Analyzed and Response Results

9.2.1 Pine Flat Dam

Located on the King's River near Fresno, California, Pine Flat Dam consists of thirty-six 50 ft and one 40 ft wide monoliths. The crest length is 1840 ft and the height of the tallest monolith is 400 ft. The dam is shown in Fig. 9.1 and its downstream elevation is shown in Fig. 9.2.

The tallest, non-overflow monolith of the dam with water at El. 951.00 is selected for purposes of this study. The two-dimensional finite element idealization for this monolith in plane stress is shown in Fig. 9.3, consisting of 136 quadrilateral elements with 162 nodal points. The nine nodal points at the base of the dam are equally spaced, as required for developing the dynamic stiffness matrix for the foundation [32]. This finite element system has 306 degrees of freedom analysis of the dam on rigid base. In analyses considering the foundation flexibility and the resulting motion at the dam-foundation interface, the system has 324 degrees of freedom. mass concrete in the dam is assumed to be a homogeneous, isotropic, elastic solid with the following properties: Young's modulus of elasticity = 3.25 million psi, unit weight = 155 pcf, and Poisson's ratio = 0.2. The selected elastic modulus, determined by forced vibration tests on the dam [28], is different from that used in earlier analyses [22]. The damping factors associated with the modes of vibration of the dam, determined from the above mentioned forced vibration tests, were in the range of 2 to 3.5 percent of critical damping. A constant hysteretic damping coefficient of 0.1,

- 107 -



FIG. 9.1 PINE FLAT DAM





- 109 -



FIG. 9.3 FINITE ELEMENT IDEALIZATION OF TALLEST, NON-OVERFLOW MONOLITH OF PINE FLAT DAM

which corresponds to a 5 percent damping factor in all modes of vibration of the dam, has been selected. This is considered appropriate for the much larger motions and stress levels expected during strong earthquake ground shaking.

The foundation rock is idealized as a homogenous, isotropic, viscoelastic half plane (in plane stress) of constant hysteretic solid with the following properties: Young's modulus of elasticity = 10 million psi, a value which is appropriate for the granites and basalts at the site; Poisson's ratio = 1/3, unit weight = 165 pcf, and damping coefficient for constant hysteretic solid = 0.05. The P-wave velocity for this rock = 10,266 ft/sec.

The following properties are assumed for water: unit weight = 62.5 pcf, wave velocity = 4720 ft/sec. For these properties of water and rock, the wave reflection coefficient at the bottom of the reservoir is α = 0.817.

9.2.2 Ground Motion

The ground motion recorded at the Taft Lincoln School Tunnel during the Kern County, California, earthquake of July 21, 1952, is selected as the excitation for analyses of Pine Flat Dam. The ground motion acting transverse to the axis of the dam and in the vertical direction is defined by the S69E and vertical components of the recorded motion, respectively. These two components of the recorded ground motion and the maximum values of acceleration are shown in Fig. 9.4.

9.2.3 Cases Analyzed

Using the computer program described in Appendix A based on the analysis procedure presented in Chapter 6, responses of the tallest monolith of Pine Flat Dam to the selected ground motion are analyzed. The dam monolith and the foundation are assumed to be in plane stress.

Responses of the dam to the S69E component of the Taft ground motion only and to the S69E and vertical components acting simultaneously are analyzed. For each of these excitations the response of the dam is separately analyzed four times, corresponding to the following four sets of assumptions for the foundation and hydrodynamic effects.



GROUND MOTION AT TAFT LINCOLN SCHOOL TUNNEL; KERN COUNTY, CALIFORNIA, EARTHQUAKE, JULY 21, 1952 FIG. 9.4



FLEXIBLE FOUNDATION, AND DAM WITH FULL RESERVOIR ON FLEXIBLE FOUNDATION COMPUTED FOR FOUR CONDITIONS: DAM WITH NO WATER ON RIGID FOUNDATION, DAM WITH FULL RESERVOIR ON RIGID FOUNDATION, DAM WITH NO WATER ON RESPONSES OF PINE FLAT DAM TO HARMONIC, HORIZONTAL GROUND MOTION 9.5 FIG.

ABSOLUTE VALUE OF HORIZONTAL ACCELERATION

- 113 -

Case	Foundation	Hydrodynamic Effects	
1	Rigid	Excluded	
2	Rigiā	Included	
3	Flexible	Excluded	
4	Flexible	Included	

In order to consider hydrodynamic effects realistically, compressibility of water is included. The displacements and stresses due to the weight of the dam and to hydrostatic pressures are included in all analyses. As mentioned in Chapter 6, the rigid body displacements of the dam due to a deformable foundation have been excluded.

All the vibration modes or generalized coordinates, as appropriate, necessary to obtain accurate results for complex frequency response up to excitation frequencies of approximately 20 cps were included in the analysis. The first five natural modes of vibration of the dam were included in the analyses assuming a rigid base. The first ten generalized coordinates, were included in analyses considering dam-foundation interaction effects (see Sec. 6.2.6).

9.2.4 Response Results

The complex frequency response functions for the horizontal acceleration at the crest of Pine Flat Dam, with horizontal ground motion as the excitation, for the four sets of assumptions for foundation and hydrodynamic effects (Sec. 9.2.3) are presented in Fig. 9.5. From these results, the fundamental resonant period of vibration and the effective damping, determined by the half-power band width method, are as follows:

Case	Foundation	Hydrodynamic Effects	Fundamental Resonance	
			Vibration Period sec.	Damping Factor
1	Rigid	Excluded	0.317	5.0
2	Rigid	Included	0.397	3.2
3	Flexible	Excluded	0.341	6.7
4	Flexible	Included	0.429	5.2
1	1		1	ł

- 114 -

These vibration periods are identified on the response spectrum of the S69E component of the Taft ground motion (Fig. 9.6). The spectrum ordinates at these vibration periods corresponding to damping ratio of 5 percent can be observed in the same figure.

The response of Pine Flat Dam was determined for each of the two excitations listed in Sec. 9.2.2 and the four sets of assumptions in Sec. 9.2.3. In each case, the computer results of dynamic analysis represent the total response, including the effects of the weight of the dam and hydrostatic pressures. These results consisted of the complete time-history of the horizontal and vertical components of displacement of all the nodal points and of three components of stress in all the finite elements. For each of the cases analyzed only a small part of the total result is presented. The maximum crest displacement and maximum tensile stresses in three critical parts of the monolith are summarized in Table 9.1. Presented in Figs. 9.7 to 9.17 are the time-history of displacements at nodal points 1, 73, and 118, located at different levels on the upstream face, and at nodal points 154, 158, and 162 at the base when foundation flexibility is considered; and the distribution of envelope values of maximum principal stress (maximum tensile stress or minimum compressive stress) during the earthquake.

9.3 Dam-Water Interaction Effects

Without hydrodynamic effects, the response of the dam is typical of a multi-degree-of-freedom system with mass, stiffness, and damping properties independent of excitation frequency. Dam-water interaction introduces frequency dependent terms in the equations of motion, resulting in complicated response curves, especially in the neighborhood of the resonant frequencies for the impounded water (Fig. 9.5). A decrease in the fundamental resonant frequency of the dam is apparent from a comparison of the responses with and without hydrodynamic effects.

When the excitation is the S69E component of Taft ground motion, the inclusion of hydrodynamic effects increases the maximum displacement at the crest of the dam from 1.38" to 1.83" (Fig. 9.7). The maximum tensile stresses in the dam are increased at the upstream face from 153 to 223 psi, from 208 to 254 psi at the downstream face, and from 257 to 366 psi at the heel (Fig. 9.8). The area enclosed by a particular stress contour increases due to hydrodynamic effects, indicating that stresses exceed the value

- 115 -

TABLE 9.1 SUMMARY OF RESPONSES OF PINE FLAT DAM TO TAFT GROUND MOTION

.

4

- 116 -



FIG. 9.6 RESPONSE SPECTRUM FOR THE S69E COMPONENT OF TAFT GROUND MOTION. THE FUNDAMENTAL VIBRATION PERIOD OF PINE FLAT DAM COMPUTED FOR THE FOUR SETS OF ASSUMPTIONS (SEC. 9.2.4) FOR FOUNDATION AND HYDRODYNAMIC EFFECTS ARE AS NOTED



DISPLACEMENT RESPONSE OF PINE FLAT DAM ON RIGID FOUNDATION TO S69E COM-PONENT OF TAFT GROUND MOTION: (A) HYDRODYNAMIC EFFECTS EXCLUDED, AND (B) HYDRODYNAMIC EFFECTS INCLUDED FIG. 9.7

- 118 -





corresponding to that stress contour over a larger portion of the monolith.

Consistent with the observations in Sec. 8.3.1, when hydrodynamic effects are excluded, the response of the dam is only slightly increased by the contribution of the vertical component of ground motion (compare Figs. 9.7-9.8 with 9.9-9.10). But, contrary to the observations in Sec. 8.3.1 and earlier results for Pine Flat Dam (18), the displacements and stresses in the dam with hydrodynamic effects included are slightly decreased by the contributions of the vertical component of ground motion (compare Figs. 9.7-9.8 with 9.9-9.10).

Earlier results (18) for the response of Pine Flat Dam subjected to the same Taft ground motion but assuming a different value of Young's modulus, demonstrated that the vertical component of ground motion causes considerable increase in the response of the dam. The vertical ground motion causes lateral hydrodynamic forces resulting in significant lateral displacements and associated stresses. Because the principal change in the system properties and ground motion from the earlier results to those presented here is in the value of assumed Young's modulus for the dam (the other change is the damping model discussed in Sec. 7.5.1); it is surprising that the results presented here are not consistent with the earlier conclusions. In order to resolve what appears to be an anomaly, the earthguake stresses (excluding initial static stresses) due to horizontal and vertical ground motions, separately, and including hydrodynamic effects, for selected finite elements (Nos. 40, 41, 129 in Fig. 9.3), are presented in Fig. 9.11. Although considerable stresses are caused by vertical ground motion, they partially cancel the stresses due to horizontal ground motion, resulting in reduced response when both ground motion components are considered simultaneously. The contribution of the vertical component of ground motion to the total response of a dam, including hydrodynamic effects, therefore depends on the relative phasing of the responses to horizontal and vertical ground motion, which in turn, depends on the phasing of the ground motion components and the vibration properties of the dam.

9.4 Dam-Foundation Interaction Effects

The foundation impedances for a half plane are smooth, slowly varying functions of the excitation frequency. However, the hydrodynamic terms are



DISPLACEMENT - INCHES

(A) HYDRO-DISPLACEMENT RESPONSE OF PINE FLAT DAM ON RIGID FOUNDATION TO S69E AND DYNAMIC EFFECTS EXCLUDED, AND (B) HYDRODYNAMIC EFFECTS INCLUDED VERTICAL COMPONENTS, SIMULTANEOUSLY, OF TAFT GROUND MOTION: FIG. 9.9



ENVELOPE VALUES OF MAXIMUM PRINCIPAL (MAXIMUM TENSILE OR MINIMUM COMPRESSIVE) STRESSES IN PINE FLAT DAM ON RIGID FOUNDATION DUE TO S69E AND VERTICAL COMPONENTS, SIMULTANEOUSLY, OF TAFT GROUND MOTION: (A) HYDRODYNAMIC EFFECTS EXCLUDED, AND (B) HYDRODYNAMIC EFFECTS INCLUDED.



FIG. 9.11 STRESSES AT SELECTED LOCATIONS OF PINE FLAT DAM ON RIGID FOUNDATIONS DUE TO THE SEPARATE ACTION OF S69E AND VERTICAL COMPONENT OF TAFT GROUND MOTION. HYDRODYNAMIC EFFECTS INCLUDED. STATIC STRESSES EXCLUDED.

unbounded at the natural frequencies of water in the reservoir. As a result, structure-foundation interaction influences the response of the dam in a simpler manner than the hydrodynamic effects do (Fig. 9.5). The fundamental resonant frequency is decreased and the corresponding damping ratio is increased due to structure-foundation interaction (Fig. 9.5). This decrease in frequency is, however, smaller than the frequency reduction due to hydrodynamic effects. The ordinate of the pseudo-acceleration response spectrum for the S69E component of Taft ground motion is essentially unaffected by the increase in vibration period but would be reduced due to the increased damping (Fig. 9.6). This leads to increase in displacements (compare Figs. 5.7a and 5.12) and reduction in stresses (compare Figs. 9.8a and 9.13a). The displacements increase due to lengthening of the vibration period (Figs. 9.7a and 9.12). Flexibility of the soil permits motions at the base of the dam (Fig. 9.12) but these are much smaller than the motions at the crest of the dam. Comparison of Figs. 9.8a and 9.13a indicates that the stresses near the base of the dam are relaxed because of soil flexibility.

The effects of structure-foundation interaction on the response of the dam to horizontal and vertical components of ground motion acting simultaneously can be observed by comparing Figs. 9.9a and 9.14 with Figs. 9.10a and 9.15a, respectively. These effects are generally similar to those observed above in the responses to horizontal ground motion alone. The contributions of the vertical component of ground motion to the total response of the dam are rather small (compare Fig. 9.13a and 9.15a).

9.5 Dam-Water and Dam-Foundation Interaction Effects

Structure-foundation interaction affects the response of the dam in a similar manner, reducing the fundamental resonant frequency and increasing the effective damping, whether hydrodynamic effects are included or not (Fig. 9.5). The fundamental resonant frequency is reduced by dam-foundation interaction and by dam-water interaction. The two reductions are additive, resulting in further reduction in frequency when both interaction effects are considered (Fig. 9.5), and in considerably smaller resonant response as compared with either the response of the dam alone or the response including hydrodynamic effects.







ENVELOPE VALUES OF MAXIMUM PRINCIPAL (MAXIMUM TENSILE OR MINIMUM DUE TO S69E COMPONENT OF TAFT GROUND MOTION: (A) HYDRODYNAMIC COMPRESSIVE) STRESSES IN PINE FLAT DAM ON FLEXIBLE FOUNDATION EFFECTS EXCLUDED, AND (B) HYDRODYNAMIC EFFECTS INCLUDED. FIG. 9.13



INTERACTION BUT EXCLUDING HYDRODYNAMIC EFFECTS, TO S69E AND VERTICAL DISPLACEMENT RESPONSE OF PINE FLAT DAM, INCLUDING DAM-FOUNDATION COMPONENTS, SIMULTANEOUSLY, OF TAFT GROUND MOTION FIG. 9.14



ENVELOPE VALUES OF MAXIMUM PRINCIPAL (MAXIMUM TENSILE OR MINIMUM (A) HYDRODYNAMIC EFFECTS EXCLUDED, AND (B) HYDRODYNAMIC EFFECTS COMPRESSIVE) STRESSES IN PINE FLAT DAM ON FLEXIBLE FOUNDATION DUE TO S69E AND VERTICAL COMPONENTS OF TAFT GROUND MOTION: INCLUDED. FIG. 9.15

However, the displacements and stresses due to either excitation --horizontal ground motion only or horizontal and vertical ground motions simultaneously -- are considerably increased due to structure - foundation interaction, (compare Figs. 9.7b and 9.16, 9.8b and 9.13b, 9.9b and 9.17, 9.10b and 9.19b). Compared to the response of the dam including only hydrodynamic effects, the stresses in upper parts of the dam are increased due to structure-foundation interaction when the excitation is only the horizontal component of ground motion (Figs. 9.8b and 9.13b); also, when the excitation included the vertical component of ground motion (Figs. 9.10b and 9.15b). Stresses at the heel of the dam are increased to a much lesser extent because of the stress-relaxation due to foundation flexibility. The area enclosed by a particular stress contour increases due to dam-foundation interaction, indicating that the dam is stressed beyond that contour value over a larger portion. This increase in earthquake response occurs in spite of the decreased response indicated by complex frequency responses, because structure-foundation interaction shifts the fundamental resonant period to correspond with a peak of the response spectrum (Fig. 9.6). Thus, the effects of dam-foundation interaction and dam-water interaction on earthquake response of the dam depend partly on the relative ordinates of the response spectrum at resonant periods of the dam with and without these interaction effects.

The effects of structure-foundation interaction on the response of the dam are generally similar with or without the vertical component of ground motion. For reasons mentioned in Sec. 9.3, the contributions of the vertical component of ground motion to the response of the dam, including both sources of interaction, are rather small.

- 129 -

DISPLACEMENT RESPONSE OF PINE FLAT DAM, INCLUDING DAM-FOUNDATION INTERACTION AND HYDRODYNAMIC EFFECTS, TO S69E COMPONENT OF TAFT GROUND MOTION FIG. 9.16



-




•

- 131 -

. , · · · •

10. CONCLUSIONS

The method presented for two-dimensional linear analysis of earthquake response of gravity dam monoliths is effective for practical problems. Included in this analysis are the effects of dynamic interaction among dam, impounded water, and foundation rock on the dam response. The dam, impounded water, and foundation rock are considered as three substructures of the complete system. Displacements of the dam are expressed as linear combinations of generalized coordinates, which are selected as the normal modes of an associated dam-foundation system. Results can be obtained to any desired degree of accuracy by including the necessary number of generalized coordinates. The substructure approach, combined with transformation of displacements to generalized coordinates, leads to an efficient analysis procedure.

The analysis procedure is developed specifically for two-dimensional analyses of gravity dam monoliths supported on the surface of a viscoelastic halfplane and impounding water in the reservoir with horizontal bottom. The dam is discretized as a two-dimensional finite element system, but the foundation and fluid domains are treated as continua. The general approach and concepts embodied in the substructure method are also applicable to more complex systems where, because of irregular geometry and/or nonhomogeneous material properties, the foundation and/or fluid domains must also be discretized.

In general, dam-water interaction, including water compressibility and dam-foundation interaction, have significant influence on the response of the dam and should be considered in the analysis. Because the foundation impedances are smooth, slowly varying functions of excitation frequency, but the hydrodynamic terms are unbounded at the resonant frequencies of the fluid domain, structure-foundation interaction affects the response of the dam in a simpler manner than does structure-water interaction.

Dam-foundation interaction effects in the response of dams depend on several factors, the most important of which is the ratio E_f/E_s of the elastic modulii of the foundation and dam materials. As the E_f/E_s ratio decreases (which for a fixed E_s implies a decrease of the foundation modulus), the fundamental resonant frequency of the dam decreases; the response at the crest of the dam at this frequency decreases; and the frequency bandwidth at resonance increases, implying an increase in the apparent damping of the structure. The influence of decreasing the E_f/E_s ratio is similar at higher resonant frequencies; however,

- 133 -

the higher resonant frequencies are decreased to a lesser degree by structurefoundation interaction. The effects of structure-foundation interaction are essentially independent of the effects of dam-water interaction, except at the resonant frequencies of the fluid domain. The dam response at these frequencies is controlled by dam-water interaction and is essentially independent of the properties of the foundation rock.

The influence of structure-foundation interaction on the earthquake response of a dam depends in part on the change in the earthquake response spectrum ordinate due to the decrease in frequency and increase in damping. In addition, the stresses near the base of the dam are relaxed because of foundation flexibility.

The frequency response curves are complicated in the neighborhood of the natural frequencies of water in the reservoir. In particular, the response curves have a double resonant peak near the fundamental frequencies of the dam and fluid domain, considered separately. In comparison with the response behavior of a dam without water, the response in the fundamental mode exhibits highly resonant behavior when the reservoir is full; the peak response at the fundamental frequency is especially large compared to that at higher resonant frequencies. Because of dam-water interaction, the fundamental resonant frequency of the dam is decreased by an amount depending on the depth of water -with water in the upper parts of the dam having the most influence -- and modulus of elasticity of the dam. The higher resonant frequencies of the dam are reduced relatively little by dam-water interaction. Qualitatively, the effects of water on the dam response are generally similar whether the foundation rock is rigid or flexible. Hydrodynamic effects reduce the fundamental resonant frequency of the dam, including dam-foundation interaction, by roughly the same degree, independent of the foundation material properties. However, the apparent damping for the fundamental vibration mode is dominated by effects of structure-foundation interaction and varies little with the depth of water.

The displacements and stresses of Pine Flat Dam due to the Taft ground motion are increased significantly because of hydrodynamic effects.

The fundamental resonant frequency of the dam is reduced by dam-water interaction and by dam-foundation interaction. The two reductions are additive with the influence of the water usually being larger. This trend may or may not exist at higher resonant frequencies. Similarly there are no general trends regarding the comparative effects of the water and foundation on the resonant responses of the dam. Which effect is more significant depends on the $E_{\rm f}/E_{\rm s}$ ratio, the depth of water, the order of the resonant frequency (fundamental or higher), and the ground motion component (horizontal or vertical).

Compared to the response of Pine Flat Dam including only hydrodynamic effects, the stresses in the upper parts of the dam are significantly increased due to structure-foundation interaction. Stresses at the heel of the dam are increased to a much lesser extent because of the stress relaxation due to foundation flexibility.

The response of the dam, without water, to vertical ground motion is small relative to that due to horizontal ground motion, but it becomes relatively significant when hydrodynamic effects are included. However, the contribution of the vertical component of ground motion to the total response, including hydrodynamic effects, depends on the relative phasing of the responses to horizontal and vertical ground motion, which in turn depends on the phasing of the ground motion components and the vibration properties of the dam.

.

•

REFERENCES

- 1. "Koyna Earthquake of December 11, 1967", <u>Report of the UNESCO Committee</u> of Experts, New Delhi, April 1968.
- A. K. Chopra and P. Chakrabarti, "The Koyna Earthquake and the Damage to Koyna Dam," <u>Bulletin of the Seismological Society of America</u>, Vol. 63, No. 2, April 1973, pp. 381-397.
- 3. R. W. Clough, F. W. Sims, J. A. Rhodes, "Cracking in Norfork Dam", Proc. ACI Vol. 61, No. 3, March 1964.
- 4. R. W. Clough and J. M. Raphael, "Construction Stresses in Dworshak Dam", Structural Engineering Laboratory Report No. 65-3, University of California, April 1965.
- R. W. Clough and A. K. Chopra, "Earthquake Stress Analysis in Earth Dams," <u>Journal of the Engineering Mechanics Division, ASCE</u>, Vol. 92, No. EM2, April 1966.
- 6. A. K. Chopra and P. Chakrabarti, "The Earthquake Experience at Koyna Dam and Stresses in Concrete Gravity Dams," <u>International Journal of</u> <u>Earthquake Engineering and Structural Dynamics</u>, Vol. 1, No. 2, October-December 1972, pp. 151-164.
- H. M. Westergaard, "Water Pressures on Dams During Earthquakes," Transactions, ASCE, Vol. 98, 1933.
- S. Kotsubo, "Dynamic Water Pressures on Dam due to Irregular Earthquakes", <u>Memoirs, Faculty of Engineering</u>, Kyushu University, Fukoka, Japan, Vol. 18, No. 4, 1959.
- 9. J. I. Bustamante, E. Rosenblueth, I. Herrera and A. Flores, "Presion Hidrodinamica en Presas y Depositos", <u>Boletin Sociedad Mexicana de</u> Ingenieria Sismica, Vol. 1, No. 2, October 1963.
- 10. A. K. Chopra, "Hydrodynamic Pressures on Dams during Earthquakes", Journal of the Engineering Mechanics Division, ASCE, Vol. 93, No. EM6, Proc. Paper 5695, December 1967, pp. 205-223.
- E. Rosenblueth, "Presion Hidrodinamica en presas Debida a Acceleraction Vertical con Refraccion en el Fondo", <u>II Congreso Nacional de</u> <u>Ingenieria Sismica</u>, Veracruz, Mexico, 1968.
- 12. A. Flores, I. Herrera and C. Lozano, "Hydrodynamic Pressure Generated by Vertical Earthquake Component", <u>Proceedings</u>, Fourth World Conference on Earthquake Engineering, Santiago, Chile, 1969.
- 13. A. K. Chopra, E. L. Wilson and I. Farhoomand, "Earthquake Analysis of Reservoir-Dam Systems," Proceedings, Fourth World Conference on Earthquake Engineering, Santiago, Chile, January 1969.

- O. C. Zienkiewicz and R. E. Newton, "Coupled Vibrations of a Structure Submerged in a Compressible Fluid," <u>International Symposium on Finite</u> <u>Element Techniques</u>, Stuttgart, May 1969.
- 15. A. K. Chopra, "Reservoir-Dam Interaction During Earthquakes", <u>Bulletin</u> of the Seismological Society of America, Vol. 57, No. 4, August 1967, pp. 675-687.
- 16. A. K. Chopra, "Earthquake Behavior of Reservoir-Dam Systems," <u>Journal</u> of Engineering Mechanics Division, ASCE, Vol. 94, No. EM6, December 1968.
- 17. A. K. Chopra, "Earthquake Response of Concrete Gravity Dams," Journal of the Engineering Mechanics Division, ASCE, Vol. 96, No. EM4, August 1970.
- 18. P. Chakrabarti and A. K. Chopra, "Hydrodynamic Pressures and Response of Gravity Dams to Vertical Earthquake Component," <u>International</u> <u>Journal of Earthquake Engineering and Structural Dynamics</u>, Vol. 1, No. 4, <u>April-June 1973</u>, pp. 325-335.
- P. Chakrabarti and A. K. Chopra, "Earthquake Analysis of Gravity Dams Including Hydrodynamic Interaction," <u>International Journal of Earth-</u> quake Engineering and Structural Dynamics, Vol. 2, pp. 143-160, 1973.
- 20. P. Chakrabarti and A. K. Chopra, "A Computer Program for Earthquake Analysis of Gravity Dams Including Hydrodynamic Interaction," <u>Report</u> <u>No. EERC 73-7</u>, Earthquake Engineering Research Center, University of California, Berkeley, May 1973, 39 pages.
- 21. S. S. Saini, P. Bettess and O. C. Zienkiewicz, "Coupled Hydrodynamic Response of Concrete Gravity Dams Using Finite and Infinite Elements," <u>Earthquake Engineering and Structural Dynamics</u>, Vol. 6, No. 4, pp. 363-374, July-August 1978.
- P. Chakrabarti and A. K. Chopra, "Hydrodynamic Effects in Earthquake Response of Gravity Dams," <u>Journal of the Structural Division</u>, ASCE, Vol. 100, No. ST6, June 1974, pp. 1211-1224.
- 23. A. K. Chopra, "Earthquake Resistant Design of Concrete Gravity Dams," <u>Journal of the Structural Division</u>, ASCE, Vol. 104, No. ST6, June 1978, pp. 953-971.
- 24. A. K. Vaish and A. K. Chopra, "Earthqake Finite Element Analysis of Structure-Foundation Systems," Journal of the Engineering Mechanics Division, ASCE, Vol. 100, No. EM6, December 1974, pp. 1101-1116.
- 25. J. A. Gutierrez and A. K. Chopra," A Substructure Method for Earthquake Analysis of Structures Including Structure-Soil Interaction," <u>Inter-</u> <u>national Journal of Earthquake Engineering and Structrual Dynamics</u>, Vol. 6, No. 1, Jan-Feb. 1978, pp. 31-70.
- 26. W. D. Liam Finn, E. Varoglu and S. Cherry, "Seismic Water Pressure Against Dams," Chapter 21, pp. 420-442, in <u>Structural and Geotechnical</u>

Mechanics, W. J. Hall, editor, Prentice-Hall, Inc., 1977.

- 27. H. Barbat, V. Breaban, C. D. Ionescu, "Dam-Reservoir Interaction for a Dam with Flat Upstream Face during Earthquakes," <u>Proceedings, Sixth</u> <u>World Conference on Earthquake Engineering</u>, Vol II, New Delhi, India 1977, pp. 1301-1306.
- 28. D. Rea, C.Y. Liaw and A. K. Chopra, "Mathematical Models for the Dynamic Analysis of Concrete Gravity Dams," <u>International Journal of</u> <u>Earthquake Engineering and Structural Dynamics</u>, Vol. 3, No. 3, pp. 249-258, Jan-March 1975.
- 29. J. A. Gutierrez, "A Substructure Method for Earthquake Analysis of Structure-Soil Interaction," Report No. EERC 76-9, Earthquake Engineering Research Center, Univ. of Calif., Berkeley, April 1976.
- 30. R. W. Clough and J. Penzien, "Dynamics of Structures, <u>Mc Graw-Hill Book</u> Company, 1975.
- 31. J. W. Cooley and J. W. Tukey, "An Algorithm for the Machine Calculation of Complex Fourier Series," <u>Mathematics of Computation</u>, Vol. 19, April 1965, pp. 297-301.
- 32. G. Dasqupta and A. K. Chopra, "Dynamic Stiffness Matrices for Homogeneous Viscoelastic Half Planes," <u>Journal of the Engineering Mechanics</u> <u>Division</u>, ASCE, Vol. 105, No. EM5, October 1979; also see Report No. <u>UCB/EERC-77/26</u>, November 1977.
- 33. A. K. Chopra and J. A. Gutierrez, "Earthquake Response Analysis of Multistory Buildings Including Foundation Interaction," <u>International</u> Journal of Earthquake Engineering and Structural Dynamics, Vol. 3, 1974, pp. 65-67. Also see Report No. EERC 73-13, Univ. of Calif. Berkeley, June 1973.
- 34. A. K. Chopra, "Hydrodynamic Pressures on Dams during Earthquakes," Report No. 66-2, Structural Engineering Laboratory, Univ. of Calif., Berkeley, April 1966.

IDENTIFICATION

EAGD Earthquake Analysis of Gravity Dams

Programmed: P. Chakrabarti, Sunil Gupta, G. Dasgupta

PURPOSE

This computer program has been developed to determine the elastic dynamic response of monoliths of concrete gravity dams to earthquake ground motion. The effects of dam-water interaction and dam-foundation interaction on the response of a dam are included. The computer program is based on the substructure analysis procedure developed in Chapter 6, in which the dam is treated as a finite element system, the impounded water as continum, and the foundation rock or soil region as a finite element system or a viscoelastic halfspace, as appropriate for the site conditions.

The computer program is applicable to two-dimensional structural systems. Thus, the dam monolith and foundation must be idealized as a system in plane stress or plane strain. Compressiblity of water is recognized in the analysis. The excitation includes the transverse (to dam axis) and vertical components of free-field ground motion, assumed to be the same across the base of the dam.

In order to define the input to the computer program, a cross-section or monolith of the dam must be idealized as an assemblage of planar, quadrilateral finite elements as shown by the example later in this appendix. Elements in the idealization are identified by a sequence of numbers starting with one. All nodal points are identified by a separate numbering sequence starting with one.

Because of dam-foundation interaction, a dynamic stiffness matrix for the foundation appears in the equation of motion. This matrix depends on the excitation frequency and is defined with respect to the degrees-of-freedom of nodal points at the dam-foundation interface. This matrix must be determined by a separate analysis and provided to the computer program as an input.

The computer program is written in FORTRAN IV and was develped on the CDC 6400 Computer at the University of California, Berkeley. It can be executed on other computers with minimal changes. However, the Fast Fourier Transform package included is partly written in COMPASS, developed specifically for the CDC 6400 computer. In using this computer program on other computer systems, a comparable subroutine package for FFT computations should be provided.

INPUT DATA

The following sequence of punched cards and data on a tape numerically define the dam-water-foundation system, the ground motion and control parameters for the anlaysis.

- TITLE CARD (8A W) Α. Columns 1-80: Contain title or any information to be printed with results. CONTROL CARD (515, F10.0, 215, F5.0, 515) Β. Columns 1 - 5 NUMNP: Number of nodal points. 6 - 10 NUMEL: Number of elements. 11 - 15 NUMMAT: Number of different materials. 16 - 20NBASE: Number of nodal points at base of the dam. 21 - 25 NEV: Number of eigenvalues (NEV < MBAND + 1). If NEV = 0 and IGRAV \neq 0 only static analysis is performed. 26 - 35 WL: Water level in the reservoir, in feet. 36 - 40NPP: Number of nodal points on the upstream face of the dam affected by the water pressure (see G). 41 - 45 IGRAV: \neq 0, to perform static analysis. 46 - 50 PSP: = 0., if plane stress problem. = 1., if plane strain problem. 51 - 55 IRES: = 0, to perform dynamic response analysis, otherwise, only static analysis and mode shapes are computed. 56 - 60 IOPR: \neq 0, to skip calculation of frequencies and mode shapes. They are read from cards in
 - 61 65 IOPP: \neq 0, to punch frequencies and mode shapes on cards.

66 - 70 IRIG: \neq 0, if the foundation is rigid.

this case.

C. FOUNDATION CARD (3F10.0)

Omit this card if IRIG \neq 0, (i.e. for rigid foundation).

Columns 1 - 10 E: Modulus of elasticity of foundation, in Ksf. 11 - 20 RHO: Mass density of foundation, in K-sec²/ft 4. 21 - 30 RBASE: Spacing between equally-spaced nodal points at base of the dam.

D. DAM MATERIAL PROPERTY CARD (15,3F10.0)

The following cards must be supplied for each different material (NUMMAT cards).

Columns 1 - 5 Material identification number.

6 - 15 Modulus of elasticity, in Ksf.

16 - 25 Posson's ratio.

26 - 35 Mass density of material, in K-sec²/ft 4.

E. NODAL POINT CARDS (15, F5.0, 2F10.0, 215, 2F10.0)

Columns 1 - 5 Nodal Point Number.

6 - 10 Boundary condition code "p".

11 - 20 X-ordinate, in ft.

21 - 30 Y-ordinate, in ft.

31 - 60 Used for layer generation, otherwise leave blank.

Specification for Code "p":

p = 0. Both displacements unknown.

- p = 1. Zero displacement in the X-direction. Unknown displacement in the Y-direction.
- p = 2. Unknown displacement in the X-direction. Zero displacement in the Y-direction.
- p = 3. Zero displacement in the X-direction. Zero displacement in the Y-direction.

Nodal point cards must be in numerical sequence. If cards are omitted and Cols. 31 - 60 are left blank, the omitted nodal points are generated along a straight line between the defined nodal points (see Note 1). Or, if Cols. 31 - 60 are used they are generated in layers (see Note 2).

Note 1: Straight line generation.

If the (L-1) cards for nodal points N+1, N+2 N+L-1 are omitted and Cols. 31-60 of the card for nodal point N are left blank, the omitted nodal points are generated at equal intervals on the straight line joining nodes N and (N+L).

Note 2: Layer generation.

Layer generation may be used after two rows of nodal points are completely defined. If on the card for node N the following data is specified:

Columns 31 - 35 MOD: Module, m (> 0). 36 - 40 NLIM: Limit of generation (> N). 41 - 50 FACX: Amplification factor f_X . (if left blank, assumed to be 1) 51 - 60 FACY: Amplification factor f_Y . (if left blank, assumed to be 1)

the X-Y coordinates of points N+1, N+2 \ldots NLIM are generated by the formulas

 $x_k = x_{k-m} + f_X (x_{k-m} - x_{k-2m})$ $y_k = y_{k-m} + f_Y (y_{k-m} - y_{k-2m})$

for $k = N+1, \ldots$ NLIM. If NLIM = NUMNP no more nodal cards are needed. If NLIM < NUMNP, the card for point (NLIM+1) must follow.

The boundary condition code for generated nodal points is set equal to zero.

F. ELEMENT CARDS (615)

Columns

1 - 5 Element number

6 - 10	Nodal Point I	The maximum difference "b" between
11 - 15	Nodal Point J	these numbers is an indication of
16 - 20	Nodal Point K	the band width of the Stiffness Matrix. "b" may be minimized by a
21 - 25	Nodal Point L) judicious numbering of nodal points.
26 - 30	Material ident:	ification

For a right-hand coordinate system the nodal point number I, J, K, and L must be in sequence in a counter-clockwise direction around the element (see Fig. A.1). Element cards must be in element number sequence. If element cards are omitted the program automatically generated the omitted information by incrementing by one the preceeding I, J, K, and L. The material identification for the generated card is set equal to the corresponding value on the last card. The last element card must always be supplied. Triangular elements are also permissible; they are identified by repeating the last nodal number (i.e. I, J, K, K).

G. WATER PRESSURE CARDS (1615)

These cards are necessary only if NPP > 0.

Columns 1 - 5 = 0, if the reservoir is to the left of the dam. = 1, if the reservoir is to the right of the dam.

> 6 - 10 11 - 15 Nodal point numbers affected by the water starting from the top as shown 16 - 20 etc.

H. BASE CONNECTION NODE CARDS (1615)

Omit if IRIG \neq 0 (i.e. for a rigid foundation).

Columns	1 - 6 - 3	5	Nodal point numbers at the base of the dam which
	11 - 1	15	are connected to the foundation rock, starting
	•		from left to right; a total of NBASE numbers, as shown in Fig. A.3.

I. FREQUENCY CARDS (I12,F15.8)

This array is necessary only if the control variable (Card B) IOPR \neq 0.

Columns 1 - 12 mode number.

13 - 22 frequency in rad/sec.

J. MODE SHAPE CARDS (I12,2E15.6)

This array is also necessary only if the control variable (Card B) IOPR $\neq 0$. The mode shapes of the associated dam-foundation system have to be normalized in the sense $\Psi^{T} \equiv I$, where Ψ is the mode shape matrix, m is the mass matrix, and I the identity matrix. If the array is the output of a previous run of this program on puched cards, then the mode shapes are already normalized. This array is as follows:





One card for each nodal point Columns 1 - 12 Nodal point number 13 - 27 X-ordinate 28 - 42 X-ordinate One set for each mode shape.

K. RESPONSE CONTROL CARD (315,2F10.0)

IHV: Code for ground motion component Columns 1 - 5 = 0, if only horizontal ground accelerations are to be included. = 1, if only vertical ground accelerations are to be included. = 2, if both the horizontal and vertical ground accelerations are to be included. 6 - 10 NEXP: Defines the number of values of excitation frequency for which complex frequency response is to be evaluated. The number of excitation frequency values is 2^{NEXP}. 11 - 15 IHYD: Code for hydrodynamic interaction. IHYD $\neq 0$ if hydrodynamic interaction is to be neglected. In this case only hydrostatic effects are included. (If NPP = 0 and WL = 0 in card B then this must be non-zero). 16 - 25 DT: Time interval at which dynamic response is generated. The frequency increment for computing complex frequency response is determined from DT and NEXP. (See note 3.) 26 - 35 ALPHA: Coefficient of reflection between water and ground rock below reservoir. This may be computed as ALPHA = (k-1)/(k+1), where $k = C_r w_r/C_W$ with w_r and w being the unit weights of rock and water respectively, Cr the P-wave velocity in rock and C the velocity of sound in water (4720 ft/sec). ALPHA < 1.

Note 3: Choice of DT and NEXP.

For using the FFT algorithm any earthquake ground motion input data (see Card Group M) is interpolated at equal intervals of time $\Delta t \equiv DT$. The total number of ordinates is determined by NEXP, and is given by $\frac{N}{2} = 2^{NEXP}$. Thus, the total duration of ground excitation used in the computations is $\frac{T}{2} = \Delta t \cdot \frac{N}{2}$ (See Fig. A.4.) A "grace band" of $\frac{N}{2}$ additional zeros are appended to the record to minimize the aliasing errors in the FFT computations. The uniform frequency increment Δf

- 147 ÷

and the frequency range $\frac{F}{2}$ for which the complex frequency response function has to be computed is given by the following relationships

$$\Delta f = \frac{1}{T}$$
$$\frac{F}{2} = \frac{N}{2} \cdot \Delta f$$

After the Fourier transform computations, which include the forward transform of the ground motion and the inverse transforms of the responses are obtained at the first $\frac{N}{2}$ points spaced at equal intervals of time Δt . By specifying TD less than T (see Card Group M), response may be obtained for a shorter duration. As is shown in Fig. A.4 the second part of the response contains free vibration response of the structure after the ground excitation has ceased.

It has been stated earlier that Δt (i.e., DT) and N (or NEXP) determine the values of T, Δf and F. It is crucial to choose these appropriately so that maximum accuracy in the results is ensured. The following considerations govern the selection of the time step Δt and total number of points $\frac{N}{2}$:

1. The frequency increment is small enough to permit an accurate description, especially near the resonant frequencies, of the frequency responses in the generalized coordinates. It is recommended that

$$\Delta f < \frac{f_1}{50}$$

where f_1 is the fundamental frequency of vibration of the dam in cps. An estimate of this may be obtained from a preliminary mode shape analysis of the structure using this computer program.

2. The frequency range should be large enough to include the significant portion of the frequency response function for generalized displacement in the highest mode of vibration included in the analysis. If f_{ℓ} is the frequency of vibration of the highest mode included in the analysis, it is recommended that

$$\frac{F}{2} > 2 f_{\ell}$$

3. The frequency range should include all the significant frequencies contained in typical earthquake records. Earthquake data processed by modern techniques accurately reproduces frequencies up to about 25 cps. Based on these considerations, in general, it is recommended that

$$\frac{F}{2}$$
 > 25 cps.

4. For good resolution of the time-variation of response, it is recommended that

$$\Delta t < \frac{T_{\ell}}{8}$$

where $T_{\ell} = \frac{1}{f_{\ell}}$ is the period of vibration of the highest mode of the the dam-foundation system included in the analysis. (If this criterion is satisfied then the one in (2) is automatically satisfied).

5. In order to reduce errors due to aliasing, inherent in the discrete Fourier transform computations, below acceptable limits,

$$T > \frac{1.5T_1}{n}$$

where

- T₁ = vibration period for first mode of associated dam-foundation system.
- n = constant hysteretic damping coefficient, same for all modes.
- ℓ = the highest mode number included in the analysis.

L. DAMPING CARD (F10.0)

Columns 1 - 10 n: Constant hysteretic damping coefficient.

M. GROUND ACCELERATION INFORMATION CARDS.

1. CONTROL CARD (215,F10.0)

Columns 1 - 5 NXUGH: Number of ordinates describing timehistory of horizontal ground acceleration. 6 - 10 NXUGV: Number of ordinates describing timehistory of vertical ground acceleration. 11 - 20 TD: Time duration for which response is desired. TD ≤ DT. (2^{NEXP}+ 1).

2. HORIZONTAL GROUND ACCELERATION CARDS (6(F6.3,F6.4))

These cards are to be omitted if IHV = 1. NXUGH time-acceleration pairs describing the time-history of transverse horizontal ground acceleration are to be specified on these cards, with six pairs per card. Time must be expressed in seconds and the acceleration as multiples of g, the acceleration due to gravity. 3. VERTICAL GROUND ACCELERATION CARDS (6(F6.3,F6.4))

These cards are to be omitted if IHV = 0. NXUGV time acceleration points describing the time-history of vertical ground acceleration are to be specified on these cards, with six pairs per card. Time must be expressed in seconds and the acceleration as multiples of g, the acceleration due to gravity.

N. OUTPUT INFORMATION CARDS

1. OUTPUT CONTROL CARD (515)

Columns 1 - 5 NPRINT: Print interval. Nodal point displacements and element stresses are printed every NPRINT time intervals. 6 - 10ICOMB: = 0, if dynamic response is needed separately. = 1, if the static and dynamic responses are to be combined. 11 - 15 ISEL: Selection code \neq 0, if output is desired for only selected nodal points and elements; otherwise displacements of all nodal points and stresses in all elements are printed. 16 - 20 NNODE: Total number of nodes for which dis-

21 - 25 NNELM: Total number of elements for which stresses are to be printed, > 1.

placements are to be printed, > 1.

NNODE and NNELM may be left blank if ISEL = 0.

2. NODAL POINT SELECTION CARDS (1615)

These cards are to be omitted if ISEL = 0 or NNODE = 0. List the NNODE nodal point numbers at which displacements are to be printed.

3. ELEMENT SELECTION CARDS (1615)

These cards are to be omitted if ISEL = 0 or NNELM = 0. List the NNELM element numbers for which stresses are to be printed.

O. FOUNDATION MATRIX TAPE

If IRIG = 0, the dynamic stiffness matrix for the foundation region should be available on a tape. To be compatible with this program, this data should have been generated by Computer Programs described in "Dynamic Stiffness Matrices for Homogenous Viscoelastic Half Planes," by G. Dasgupta and A. K. Chopra, Report No. UCB/EERC-77/26; otherwise the input statements in this program should be modified by the user.

The data read from the tape should then be copied onto a scratch file called TAPE90, which is read by the computer program. After execution, the data is destroyed.

OUTPUT

The following is printed by the program. (Note that some of these may be suppressed according to the options provided in card B.)

- 1. First set of input data: structural and material properties, foundation properties, options, etc.
- Hydrostatic Loads: i.e. equivalent nodal point loads due to hydrostatic pressure of water in the reservoir.
 Nodal point displacements and element stresses for static loads.
- 3. Frequencies and Mode Shapes.
- 4. Second set of input: response data including damping, time step, etc.
- 5. Absolute value of complex frequency responses for generalized accelerations of the dam due to horizontal and vertical ground motions separately.
- 6. Third set of input: earthquake acceleration data, etc.
- 7. Displacements of selected nodal points (according to ISEL and NNODE in Card Group N) and stresses in selected elements (according to ISEL and NNELM in Card Group N) at instants of time determined by the print interval (NPRINT in Card Group N).
- 8. The peak values of major and minor principal stresses in each element during the earthquake and the times at which they occur.
- 9. The following quantities are written on tape unit 3.

Logical Record 1: NUMNP, NUMEL, NEV, ND, DT.ND is the total number of time intervals of response determined from TD and DT using integer arithmetic.

Starting with Record 2, for each of the ND time intervals, two records are written using the following two statements:

WRITE (3) X

WRITE (3) STRES

where X and STRES are one-dimensional arrays, dimensioned properly, so that X(2 * I - 1) and X(2 * I) are the X-component and the

- 151 -

Y-component of the displacement at node number I, I = 1,2 NUMNP, Similarly STRES (3 * N - 2), STRES (3 * N - 1), and STRES (3 * N) are the three components of stress, i.e. σ_x , σ_y , and σ_{xy} in the element number N, N = 1,2 NUMEL. A physical tape has to be requested, with the file name TAPE3, in the program control cards if these results are to be saved for subsequent use. Plotting programs are available to read from this tape and plot time-history of nodal point displacements and element stresses.

EXAMPLE

The preparation of input data for this program is illustrated by an example. A very coarse finite element idealization of a cross-section of a concrete gravity dam is shown in Fig. A.5, followed by the input data required for the analysis of this structure subjected to horizontal earthquake motion.

COMPUTER PROGRAM INFORMATION, LIMITATIONS, AND TIMING

The program is written in FORTRAN IV and was developed on a CDC6400 computer at the University of California, Berkeley. The core storage requirements of the program are separated into fixed and variable parts with the fixed part consisting of instructions, non-subscripted variables, and those arrays which do not depend on the size of the individual problem. The variable part is stored in array A which appears in the blank COMMON statement.

The high speed storage requirements of the program can be changed depending on the size of the problem to be solved. This is done by changing two FORTRAN statements at the start of the program, i.e.

COMMON A (N)

MSTOR = N

The value specified for N must exceed each of the following.

1. No + 2*NUMNP*MBAND+6*NB2

2. N_0 + 41*NUMEL

3. $N_0 + 6 \times NUMNP + NEV \times NEV$

4. N_O + 3*NNB+2*NB2+NPP+4*NTERM+NEV*NEV*(4+NTERM) * +NEV*(2*NUMNP+3*NCOMP+2*NWR*NCOMP+2*NTERM+2)

5. NDATA* (4+2*NEV) + NEV* (1+2*NWR) + NXUG

6. NUMNP*(5+NEV)+NEV*(2*NWR+1)+42*NUMEL

where,

 $N_{O} = 6*NUMNP+NBC+NPP+NEV+NNB+NB2$



FIG. A.5 EXAMPLE FINITE ELEMENT MESH FOR A CONCRETE GRAVITY DAM TO ILLUSTRATE PREPARATION OF INPUT DATA FOR COMPUTER PROGRAM

45 1152000 157 1 2 3 4 5 6 7 8 9 10 11 12 13 14		78 • 5 400 • 400 • 400 • 400 • 400 • 400 • 375 • 375 • 375 • 375 • 350 • 350 • 350 •	004814	9	1		
1152000 157 1 2 3 4 5 6 7 8 9 10 11 12 13 14	<pre>• •00316 6000• •2 16.75 24.75 32.75 40.75 48.75 16.75 24.75 32.75 40.75 16.75 24.75 32.75 40.75 32.75 50.15.</pre>	400. 400. 400. 400. 375. 375. 375. 375. 350. 350. 350. 350.	004814				
1 2 3 4 5 6 7 8 9 10 11 12 13 14	16.75 24.75 32.75 40.75 48.75 16.75 24.75 32.75 40.75 48.75 16.75 32.75 32.75 32.75 32.75 32.75 32.75 32.75 32.75 32.75 32.75 32.75	400. 400. 400. 375. 375. 375. 375. 375. 350. 350. 350. 350.	004014				
1 2 3 4 5 6 7 8 9 10 11 12 13 14	10.75 24.75 32.75 48.75 16.75 24.75 32.75 40.75 48.75 16.75 24.75 32.75 32.75 40.75 32.75 50. 15.	400. 400. 400. 375. 375. 375. 375. 350. 350. 350. 350.					
2 3 4 5 6 7 8 9 10 11 12 13 14	24,75 32,75 40,75 48,75 16,75 24,75 32,75 40,75 48,75 32,75 32,75 32,75 50, 15,	400. 400. 400. 375. 375. 375. 375. 350. 350. 350. 350.					
3 4 5 6 7 8 9 10 11 12 13 14	32.75 40.75 48.75 16.75 24.75 32.75 40.75 48.75 32.75 32.75 32.75 40.75 32.75 32.75 32.75 32.75 32.75 32.75 32.75 32.75 32.75 32.75 50.1 15.	400. 400. 375. 375. 375. 375. 375. 350. 350. 350. 350.					
4 5 6 7 8 9 10 11 12 13 14	40.75 48.75 16.75 24.75 32.75 40.75 48.75 16.75 24.75 32.75 40.75 50. 15.	400. 375. 375. 375. 375. 375. 350. 350. 350. 350.					
5 7 8 9 10 11 12 13 14	46.75 16.75 24.75 32.75 40.75 48.75 16.75 24.75 32.75 32.75 50. 15.	400. 375. 375. 375. 375. 350. 350. 350. 350.					
6 7 9 10 11 12 13 14	10.75 24.75 32.75 40.75 48.75 16.75 24.75 32.75 40.75 50. 15.	375. 375. 375. 375. 350. 350. 350. 350.					
8 9 10 11 12 13 14	24.75 32.75 40.75 48.75 16.75 24.75 32.75 32.75 50. 15.	375. 375. 375. 350. 350. 350. 350.					
9 10 11 12 13 14	40.75 48.75 16.75 24.75 32.75 40.75 50. 15.	375. 375. 350. 350. 350. 350.					
10 11 12 13 14	40.75 48.75 16.75 24.75 32.75 40.75 50. 15.	375. 375. 350. 350. 350.					
10 11 12 13 14	48.75 16.75 24.75 32.75 40.75 50. 15.	350. 350. 350. 350.					
11 12 13 14	16+75 24+75 32+75 40+75 50+ 15+	350. 350. 350. 350.					
12 13 14	24.75 32.75 40.75 50. 15.	350. 350. 350.					
13	32+75 40+75 50+ 15+	350.					
14	40.75 50. 15.	350.					
	15.						
15	15.	350+					
16		325.					
17	29.	325.					
18	43.	325.					
19	55.	325 •					
20	64.	325.					
21	15.	300.					
22	32.	300.					
23	51.5	300.					
24	69.	300 .					
25	80.	300.					
26	12.	250.					
27	40.5	250.					
28	70.	250.					
29	98.	250 .					
30	120.	250.					
31	9.	180.					
35	51+5	180.					
33	94.5	180.					
34	137.5	180 .					
35	175.	180.					
30	5.	100+					
37	64.	100.					
38	1234	100+					
39	180.	100.					
40	200.	100.					
41		0.					
42	/0.00 j	0.					
43	1074 976 6	.					
4944 4 - E	23243	0.					
43	314.	, v•	,				
1	1 0	10 5	1				
4	4 4 2 1	10 5	4 •				
5	0 11	14 /	1				
8	9 14	15 10	1				
9	11 16	17 12	· 1				4 -
12	14 19	20 15	1				

.

16	19	24	25	20	1									
17	21	26	27	22	ĩ									
20	24	54	30	25	1									
21	56	31	32	27	t									
24	29	34	35	30	1									
25	31	36	37	32	1									
28	34	39	40	35	l									
29	36	41	42	37	1									
32	39	44	45	40	1									
	1	6	11	16	21	26	31	36	41					
41	42	43	44	45										
	9		02											
• 1														
246	0 10	•						-						
0	-14	0.04	. .	59 1	0.080	149	0.140	• 5	9 6.200	97	0+260	+ 1127F5	2NW 1	
0.320	44	0.35	0 •	21		1/4	0.440-	-• 1 2	/ 0+500-4	. 17	0.540	• 113785	SUM S	
0.0600	144	0.00		114		141	0.740	• •	1 0.000-	267	0.050	• 192155	C WAS	
1 090-1	234	1.14	0.	114	1 20.0	120	1 260	• •	< 1.020-0 0 1.290-	405 (1 720	כשואיבו אי השורמי	2NW 4	
1.390 -	207	1.44	0	132	1.520 -	224	1.580-	• 9 •• 15	A 1.620-	. 41 . 110	1.700		2 NW 4	
1.760	240	1.82	0 -	12	1.880	327	1.040	. 11	7 2.020-	. 01	2.060	- 3991E8	2 NW 7	
2.120-	340	2.14	0	143	2.200-1	287	2.280	. 30	1 2.340-	A70	2.380-	• 392103	2 NW 8	
2.400	142	2.44	0	158	2.480 .	92	2.520-	29	9 2.560-	181	2.620-	299TE5	2NW Q	
2.640 -	93	2.69	0.	119	2.740 .	237	2.780	. 17	2 2.820	185	2.860-	- 37TES	2NW10	
2.880	67	2.92	č .	67	2.940 .	120	2.980-	- A	9 3.040	434	3.120	• 186TES	2NW11	
3.200 .	775	3.24	ο.	474	3.340 .	605	3.420	. 6	9 3.500 .	488	3.680-	-1514TF5	2NW12	
3.820 .	750	3.86	ο.	646	3.900 .	410	3.940	. 73	7 4.020-	191	4.120	. 581TE5	2NW13	
4.200	1015	4.26	ο.	254	4.320	897	4.360-	70	0 4.400-	648	4.420	. 529TES	2NW14	
4.480 .	411	4.52	0-•1	485	4.600 .	451	4.640	. 43	8 4.680 .	661	4.780-	. 817 TES	2NW15	
4.900 .	1329	4 . 99	0	542	5.040~~	5	5.080-	. 8	4 5.140 .	609	5.200	 112TF5 	2NW16	
5.300 .	754	5.40	0- • 1	182	5.560 .	610	5.600	. 29	6 54700 4	1016	5.760-	. 593TF5	2NW17	
5.820 .	297	5.88	0-•1	077	5.940 .	399	5.980-	• 30	4 6.060	474	6.160	 1797E5 	2NW1 8	
6.220	487	6.28	۰. ۵	206	5.340	115	5.500	•156	8 6.560 -	1332	6.600	+1411TF5	SNATO	
6.740	997	6.80	•••	338	6. 920	643	7.000	. 70	5 7.180	1179	7.260	 469765 	SNASO	
7.300	132	7.36	0.	705	7.420 .	51	7.460	• 52	2 7.580	1191	7• 62 C-	• 943TF5	2NW21	
7.700	1087	7.80	0 + 1	230	7.920	65	7.960	• 39	2 8.000	942	8+080	▲10737E5	2NW22	
8.140	104	8.22	••••	209	8.240 .	52	8.300	• 27	5 8.340	601	8+480	• 982TF5	SNA53	
8+580-+	171	8.62	0	274	8.540	326	8.720-	⊂• ⊃	1 8.760~4	745	8.860	• 459155	2N#24	
9.000	1333	9+12	0.1	302	9.240 0.620-	001	9.320-	•• 20 70	7 9+300-4 7 0 960	102	9.400	· 0431F5	2NW25	÷ 1
9452V +	1021	V . 30	0 •	510	9.020-1	7561	9.009	• 70 • 30	7 9.000 .	6 1 3 9	94969	• (0/1F0	2111221	
10.340-	2321	0.43	a .	1371	0.440	1801	0.480-	. 18	010-540	625	10-600-	• 3991F3	2NW28	
10.640-	3631	0.68		291	0.740-	5591	0.760-	- 14	010.820-	4021	0.960	- QORTES	2NW20	
11.080	8991	1.16	ō .	8151	1.200 .	5791	1.260-	- 45	411.340-	6371	11.420	- 292TE5	2NW30	
11.480	741	1.50	G .	561	1.560	5191	1.680	. 64	511-720	449	11.780	. 590TE5	2NV31	
11.860	3621	1.96	0	471	2.020	6231	2.180	• 58	012.240 .	. 41	12.300	. 318TF5	2NW32	
12.360	7021	2.44	0.	5671	2.480 .	2401	2.500	. 13	512-560-	283	12.600-	· 152TF5	2NW33	
12.660	4791	2.76	ο.	3191	2.500 .	2271	2+860	+ 47	612.980	426	13.020-	. 191TF5	2NW 34	
13.100	5571	3.22	0.1	1 30 1	3.340	6491	3.440	. 33	213.500	3341	13.600	 136TE5 	2NW35	
13.640	71	3.70	0	1511	3.740	991	3.780-	- 51	813-840 .	267:	13, 880	 70TE5 	2NW36	
13.920 .	441	3.98	0	4261	4.080 .	9081	4.120	• 39	814.160	398	14+220-	+ 229TF5	2NW37	
14.260 .	1881	4.32	0	1381	4.340 .	1621	4.400	• 26	714.440 .	398	14-480	 306TF5 	2 NW3A	
14.500 .	5931	4.56	0	4651	4.620	4651	4.660-	29	514.700-	6553	14+840	• 843TE5	2NW39	
15.000	6871	5.12	0	71	5.160	71	5.200	• 31	915.260 .	841	5.320	• 607TE5	2 NW4 0	
15.400	5701	5.52	•••	2931	5.600	6351	5.660-	• 79	215.780	280	15:860-	 282TF5 	2N¥41	
. 10	1	I	2	- 1										•
1	21												· · .	
ð														•
												1.		

,

- 155 -

2.

NUMNP	z	Number of nodal points
NUMEL	=	Number of elements
NBC	=	Number of displacement constratints
NPP	-	Number of nodal points affected by water pressure
NCOMP	=	Number of components of earthquake included in the analysis.
NDATA	=	2 ** NEXP
NEV	=	Number of generalized coordinates included.
NEXP	=	Exponent for the number of ordinates to define complex requency responses.
MBAND	=	2 * (MB + 1)
MB	-	$\max_{i} MB_{i}, i = 1, NUMEL$
MBi	=	difference between the largest and smallest nodal point numbers for element i.
NTERM	=	50
NXUG	ż	max (NXUGH, NXUGV)
NXUGH	=	Number of horizontal ground acceleration ordinates
NXUGV	=	Number of vertical ground acceleration ordinates
NBASE	= .	Number of nodal points at base of the dam.
NB2	=	NBASE*2
NNB	=	NB2*NB2*2
NWR	=	max(100,NDATA+1)

If only frequencies and mode shapes are desired along with static analysis it suffices to check (1) and (2) above.

The computer time required for solution depends on a number of factors. The more important ones are the number of time ordinates (which is determined by NEXP) the number of nodal points, the bandwidth of the stiffness matrix (the nodal points should be numbered in a manner which minimizes the bandwidth), the number of elements, the number of generalized coordinates to be included, and the interval for printing and writing the output.

Some representative execution times on a CDC 6400 computer are given for four cases of analysis (Tables A.1 and A.2) for Pine Flat dam (Chapter 8),

with NUMNP = 162, NUMEL = 136, MBAND = 22, and NEXP = 10, NEV = 5 for cases 1 and 2, 10 for cases 3 and 4, NBASE = 9. Table A.1 pertains to analysis for transverse ground motion only. Table A.2 is for analyses considering transverse and vertical ground motions, simultaneously. The results include the complete history of displacements and stresses during the earthquake.

Tables A.1 and A.2 provide some indication of the increase in computational effort required to include dam-water and dam-foundation interaction effects in the analysis. Considering that these interaction effects complicate the analysis greatly, the additional computer time required is modest. Comparatively, more computational effort is required to include dam-foundation interaction effects, primarily because a larger number of generalized coordinates need to be included in the analysis. The efficiency of the analysis lies in use of the substructure method along with transformation to generalized coordinates. TABLE A.1 COMPUTATION TIMES: PINE FLAT DAM, HORIZONTAL GROUND MOTION ONLY

CASE	FOUNDATION	HYDRODYNAMIC EFFECTS	NO. OF GENERALIZED COORDINATES	CENTRAL PROCESSOR TIME (SEC.)	REMARKS
1	RIGID	EXCLUDED	5	314	Eigenvalues, Eigenvectors,
2	RIGID	INCLUDED	5	474	and Dynamic Foundation
3	FLEXIBLE	EXCLUDED	10	810	Stiffness Matrix are read
4	FLEXIBLE	INCLUDED	10	1313	as input

TABLE A.2 COMPUTATION TIMES: PINE FLAT DAM, HORIZONTAL AND VERTICAL GROUND MOTION

.

CASE	FOUNDATION	HYDRODYNAMIC EFFECTS	NO. OF GENERALIZED COORDINATES	CENTRAL PROCESSOR TIME (SEC.)	REMARKS
1	RIGID	EXCLUDED	5	325	Eigenvalues, Eigenvectors
2	RIGID	INCLUDED	5	500	and Dynamic Foundation
3	FLEXIBLE	EXCLUDED	10	832	Stiffness Matrix are read
4	FLEXIBLE	INCLUDED	10	1356	as input

- 158 -

APPENDIX B

LISTING OF EAGD COMPUTER PROGRAM

÷

•

96 50 501 103 010 55 501 101 105 105 105 10.7 808 500 213 construction of the second of EAGN FAGN EAGD AGD : EAGD G V d D E A GD 5 AGD 5 AGD F AGD 5 A G D E A G D E A G D EAGD EAGD FAGD FAGD ⊭ 4Gn EAGD F AGD E A GO EAGD FAGD FAG0 EAGD EAGD FAG0 EAGD FAGN EAG0 EAGO 6949 1949 FAGN 05 V 3 GOV -EAGD AGD AGD. - 4GU GDA : F AGD AGD EAGD CALL STIFFLAMED, ATM29, ATM31, ATM91, ATM51, ATM51, ATM71, ASP) WRTTE (6.2105)HED.DAO.DVS.AOS.DVL.NV.NAASE Neo#2+Nump FORM TOTAL STRFFNESS AND MASS WATRICES READ DATA AND FORM ELEMENT PROPERTIES FF(NST)FF.LE.MSTOR) GO TO 110 Write(6.2010) TF(NTDTAL.LE.MSTOP) 40 TO 120 WRITE(6.2010) IF (1GRAV.E0.01 GO TO 140 NFOT AL =N1 2 + N12 4 3 NACOM=M X O E NACOM ° N TO TAL 9 VS=E+0.5/(PH0+(1.+FNU)) VS=SARTVS) WRITE(6.21041VS DVS=DA0+VS/PRASE SOLVE FOR STATIC LOADS NLO=N9 + (NEO*MBAND) N6=N5+NJ4HAT N 7=N6+NUMMA T N5T IFF=N7+NUMMAT A 05=A0+V5/RBASE N 13=N 12+ NUMEL N 14= N 13+NUMEL N 15=N 14 + NUMEL N 14= 3+NUMEL E+28N+1 [N=2 IN 5+26N+26N= UNN NI2=NI1+NUMEL NRCOM=NSY 1FF NI ±1 N2=N1+NUMNP N3=N2+MUMNP N4=N3+NEA NS =NA +NPP N6 =N5 +NEA N11=N10+NE 0 N 5=N 4 +NUMNP NB2 =NBAS E * 2 DM. =NW#DWS 800+10=80 260+60=60 DEN TENS NOC N7=N6+NEV 50 i 110 120 000 υu U. U. υu U, L ¢ 2 NN 12 22 e. 5.0 20 50 33 ę. ŝ 2 5 80 4 S S S 2 * Ŧ 35 2 2 -1 ÷ 5 Ş 2 A. 10 EAGD 5 4 5 0 F 4 6 0 ♦ E A GO #E 460 EAGD EAGD EAGD EAGD EAGD EAGD EAGD EAGD EAGD FAGD EAGD E A GD F A GD EAG0 EAG0 EAG0 #EAG0 EAGD E AGD EAGN EAGD FAGD EAGD EAGD FAGD EAGR EAGB FAGN +E AGD EAGD EAGD 05 V 2 EAGD EAGD EAGD EAGD EAGN FAGD PEAGD #F 460 COMMAN /CNTPL1/ NUMMP,MBAMD,NUMEL,NBC,1GRAV,NPP,ML,TRACE,10PP,10PPFEAGD E AGU FAGD EAGD FAGD COMMON/LABEL/HED(A) Common /Cntrl2/ Ndati,Nya,df,Nrest,Nbuf,Nblock,1HV,Ncomp,1Sel, Icoma, Vcntrl2/ Ndati,Ndatint,Nd NEV.M. NPP. NEV. N. WDP. PRUGRAM IDENTIFICATION - EATD PRUGRAM DESCRIPTION - EARTHANAKE ANALYSIS OF GRAVITY DAM-PRUGRAMMED OY - EOUNDATION - WATER SYSTEMS. PROGRAMMED OY - P. CHARRABARTI MODIFICATIONS RY - SUNIL GUNTA. GAUTAM DASGUPTA - UNIVERSIY DE CALIFERNIA. HERKELEY PRDTRAM EA ODETAPEL, TAPEZ, TAPEJ, TAPEGO, TNPUT, OUTPUT, PUNCH x, tapes=1 NPUT, TAPEA = OUTPUT 1 CHANGE THE FOLLOWING TWO CARDS TO MODIFY STORAGE READ(90) A0.MCODE.ES.RNU.RHAS.DAMP.DA0.NW.NBASE READ (5,1000) HED.NUMIP.NUMEL.MUMMAT.NHASE. 1684/959.1885.1098.1919.1815 2009 HED.NUMU.NUUMEL.NUMMAT.NHASE. 1684/959.1829.1098.1098.1919.1815 READ STANDARD FOUNDATION PROPERTIES FROM TAPE 1FtMCODE.c0.0) WRITF(6.2101) 1FtMCODE.c0.1) WRITE(6.2102) 1FtPR.cc.0.155=54(1.424401)/(1.44001)442. 5504.E*675 WBASE, IR RG. XJUNKE 221 E .RNU .WIG. DANP. RBASE READ AND PRINT CONTPOL INFORMATION [F {PSP-E0.0) № IF (6.2020) [F {PSP-NE.0.0) № IF (6.2021) [F {IPSC-E0.0) GD YO 100 N9ASE=0 100 READ (5,1050) E.AHO,PBASE WRITE (6.2050)HED. CALL SECUND (VI) COMMON A (20000) CALL NUBLOK (99) WRITE (6.2100) 4510R=20000 R E M IND 90 READ (90) GO TO 105 ພະບິ 00000 000 υu 0000 u 000

- 161 -

NI 6=NI 5+ NUM	EAGD	611	0= NN
N 17=N 16+24 PNUMEL	EAGD	•	NTERM=0
N16=N17+2*MJM6L	EAGN		157 N21=N20+(NEVI WTEPM)
N 5 TA T I C = N 1 8+ B4 NUMEL	1971 H		N22=N21+ (NEVENTERH) N23=N214NEVANEVANTERH)
	E A GD		N 23-M 2 2 4 112 4 4 14 2 4 1 10 10 10 10 10 10 10 10 10 10 10 10 1
	E AGD	611	
WH [] 7 [0, 2010]	EAGN	120	N26= N25 + NE ERW
30 CALL STATIC (A(N3), A(N) 0), A(N9), A(N) 1, A(N) 2), A(N) 3), A(N) 4),	(NIS 16 460	121	NFH =NDATA+1
a(N16), a(N17), a(N18), NFQ, MBAND, NUM, NUMEL)	E AGO	221	IF (NWR "GT."NFR) NWR =NFR
	EAGD	621	NGUF =NER NEV
MODESHAPES AND FREQUENCIES	E A GO	124	N2 7= N26+2+(N3UF + NCOMP)
	EAGD	621	NKOMFL X= N279NN
140 CALL SECOND (T2)			NGCONTMAND (NICUM, NAUPPEA)
IF (NEV.LE.0) GD YD 155			IF (NKUMPLA.FF.MSJUK) 50 10 100
N12#N11+NEG		021	
× 1 3= N 1 2 + NEG			STOP
	FAGO		100 CALL RUMPLEAIAIMINIAIMINISIAMIAIAIMIAIAIMIAIA
	EAGO	201	A 4 4 4 4 4 4 4 4 4 4 4 4 4 4 4 4 4 4 4
NUCCHARMAN INTELOMINE TO NEW CONTRACTION OF TO NEW	FAG0	661	AL PHA. S GN. NN. NEV . NEV I. NEQ. NT ERM. N OUF. DWS.
IT (NEIGENER, MANUT,	EAGO	134	
	EAGD	135	C TIME RESPONSES OF MODAL DISPLACEMENTS
sture fittent at NTL at NSL at N61 . A (N2) . A (N9) . A (N9) . A (N10) .	(N11)EAGD	961	
LOUTE FLOREN AND A STATE AND A ST	E 460	137	READ (5.1030) NXUGH,NXUGV,TD
LESS CALLER STEPANON (13)	EAG0	130	WRITE (6.2090) HED.
Laurante activity in the second se	EAGO	139	UNUGH . NXUGY . TD
	EAGD	0 • 1	NXIJG= MAXO (NXUGH, NXIJGV)
	EAGD	1.1	1F (NKUG.LE.0) GO TO 180
COMPLEX FREQUENCY RESPONSES ASSUMING WATER AS COMPRESSIBLE	EAGD	142	N 2=N 1+2+(NFR +NE V)
	E 4 G D	EVI	N3=N2+2+{2+{2+NDATA}
A EAD (5,1020) 144, NE XP,11470, AT, ALPHA	EAGD		17 87 48 VB VI
ND AT A=2+ # NEX P	EAGD	145	9 NX N N N N N N N N N N N N N N N N N N
WRITE (6, 2070)HED.	EAGN	146	N F RNS F M = NS + NXUG
144 . MDATA. 1HYD. GT. ALPHA	EAGO	1 4 7	NGCOM=MA XO[NGCOM.NTANSFM]
1.1ERM=50	EAGO	148	15 (NTANSFN.LE.MSTOR) 50 10 170
NWR # 100			WRITE(6,2010)
NCOMP= I	EAGD		STOP
[F [[[HV→1].6[.0] NCOMP=2 →- →	6965 F 400	151	TTO CALL THUST HAR AINT LINE TO THE
	EAG0	153	C SUPERPOSITION OF MODES AND EVALUATION OF
UDITE (6. 2095)	EAGD	154	
156 N10=N9 + (NEOF NEV)	EAGD	155	READ (5, 10 40) NPP INT, ICOMB, ISEL, NNODE , NN
	EAGN	156	WRITE (6.2040) HED.
NI 2= NI 1 + NNB	E 460	157	. NPRINT, COMP. I SEL , NNODE , NN
61N 421N=E1N	. FAGD	156	N2 =N1 +NED
2 GN+2 +2 TN=+1 N	6 4 4 0	651	NJ =N2 + NEO
N15=N14+2+1 NEV + NC 0MP)	EAG0	160	CA MARCONIA EN # 42
NI 6=NI 5+ 2+ (NE V+NE V)	EAGD	191	NS ER4 PNEV
NI 7= NI 6+NEV	E 460	162	
N18=N17+2+(NEV+NEV)	FAGD	163	N7 = N5 +NUL4E
HIGHNEY HIGHNEY	E A 60	•	NB =N7 + NUMFL
050 = N 1 6 + NEA & NG 040	EAGD	165	NO # HB + NUMFL.
1daN=NN	E 460	160	NIO=Ng +NUMEL
NEV1≭NEV+L	19 6 1	7.61	N 11=N 10+NNDRE
IF (NPP.GT.0) 60 10 157	1945	EC 1	• N12= N11 + NNEL.M

FAG0 169 FAG0 171 FAG0 173 FAG0 177 FAG0 187 FAG0 187 FAG0 187 FAG0 197 FAG0 201 FAG0 211 FAG0 221 FAG0 2 N6J , A(N7J , A(NBJ , A(N9J , 51. A(N16J, A(N17J, A(N18J), 41. A(N25J , A(N26J , A1N27J), 5. ESCAL E. NB 2J

102 283 205 205 284 280 292 E 6 2 ₽u č 205 296 262 208 999 30.0 10.2 303 104 ŝ 306 308 90 Q 010 312 115 315 105 IIE Ê Ě 7 I F 319 320 322 53 E 324 325 327 328 326 6 4 GD F 4 GD F 4 GD FAGD EAGN EAG0 EAG0 EAG0 FAG0 EAGD EAGD EAGD EAGD FAGD EAGD FAGD FAGD E 460 E 460 E 460 EAGD EAGD EAGD EAGD F AGD EAGD FAGD EAGD EAGN X, 241 114)/254 FNUMDATION SPECIFICATION /1X, 241144)// 5540 SPECIFICATIONS TOR FOUNDATION USED IN ANALYSIS// 184 VOUNGS MODLUS = E12.4.7/1 X/SF1.// 1844 POISSONS PATIN = E12.45.7 SOL 10771 50HD REFRACTION CONSTANT FOR WATER AND GROUND ROCK ... FID. 5///) /1×°16(1H#1/// = F12.5// * F12.5,7HRAD/SEC// * F12.5,7HRAD/SEC// * F12.5,7HRAD/SEC// * F12.5,7HRAD/SEC// = FI2.5, 7H FT/SEC//1 TY = F12.5.15H K-SEC442/F1444// Damping Capacity = F12.5.1/ NODE Spacing = F12.5.4H F1.//P //-----32HCONTROL INFORMATION FOR RESPONSE 16(1H+1/17H DYNAMIC ANALYSIS 15//) JIH FOUNDATION NODE SPACING 30H SHEAR WAVE VELOCITY THE MASS DENSE TY · JIH MEASURE OF HSE .XI * 50H0 REFRACT 2060 F 0RMAT (1H1/ 20 70 FORMAT (1H1. FORMAT (1H1. 20X.8A10// 2105 FORMATCH1. 20X . A A 10// 20X .8 A10//

20.90

Ð

• •

 10.00
 FORMAT
 715.
 F10.01

 10.00
 FORMAT
 715.
 F10.01

 10.00
 FORMAT
 515.
 F10.01

 10.00
 FORMAT
 515.
 F400.34

 10.00
 FORMAT
 515.
 F400.34

 10.00
 FORMAT
 515.
 F400.34

 2000
 FORMAT
 515.
 F400.34

 2000
 FORMAT
 610.01
 F400.34

 5000
 FORMAT
 610.01
 F400.245

 5000
 NUMBER
 F
 F400.245

 5000
 NUMBER
 F
 F400.245

 5000
 NUMBER
 F
 F400.255

 5000
 CODE
 F00.404
 F400.255

 5000
 CODE
 F
 F400.255

 5000
 CODE
 F
 EAGD 225 FAGD 226 EAGD 227 EAGD 227 EAGD 227 FAGD 229 FAGD 231 FAGD 232 EAGN 236 EAGN 237 EAGN 237 EAGN 239 FAGN 239 FAGN 239 FAGN 265 Fagn 267 CALL RESPONS(A(NI),A(N2),A(N3),A(N4),A(N5),A(N6),A(N7),A(N1),A(N9)EATD 233 1.4(N10),14(N11),A(N12),A(N13),A(N14),A(N15),NNODE,NNEL4,NF7,NEV, EAGD 234 200461,NUM} CALL SECOND (14) EAGD 265 268 EAGN 269 EAGO 270 273 NEEN EAGN 275 042 240 271 278 EAGO 272 FAGD 274 EAGO 276 ACCN MAS REFN CONSTRERED----EAGD 217 FAGD FAGD F AGD GROUND ACCN HAVE A 36HO NUMBER OF GROUND ACCN VALUES -----. 14/ (/ ---------2051 FORMAT (///884 DNLY HOR IZONYAL GROUND CALL JOBLOG (MSTOR, NRCOM, T1. T2. T3. T4) 13640 NUMBER OF THE INCREMENTS 23640 PRINT INTERVAL NBCOM=MAX01NBCOM,NRESPON) If (NRESPON,LE.4570R) GO TO 175 Write(K,2010) NRESPON=N15+ B*NUMEL 5 (1++1/2/) N 13=N12+ 34NUMFL N14=N13+244NUMFL N15=N14+ 24NUMFL 2030 FURMAT (1412./ 2060 FORMAT (1H1. · 20X .A AIO / 510P 5 1UP 175 1 80 U U

•

ACEDONEG . RNFG SURROUTINE - BANEIG(NN°NM°NEV°NBC¢EV,NERC¢SMASS,A+B,V+F,CM,SN+TRACF Dimens IOM - Evil)+Nebc(1), Smass(1)+ - A(NN+1)+Af1)+Af1)+SN11)+CM12)+

- 163 -

Ξ 211 ANEG ANEG ANEG PNFG PNFG ANFG ANFG ANFG ANFG ANFG ANFG RNEG RNEG RNEG ANES ANES ANES BNFG NNFG BNEG BNEG RNFG FINEG 6 ANEG ANFG ANFG 0 JNE C PNFG BNP G 9 AN EG **BNEG** RNEG RNEG BARG DAFG RNEG BINE **PINFG** PINFG A NEG **BNEG** BINEG Daned RNFG BNEG RNEG **BAFG** ANFG BINE Baneg SANE Paves DINEG FINFG 200 2 8 DQ 220 | = 1,NR |F (A95(B(1)).GT.A95(E)) E = 8(1) 00 300 1 ≠ 1.NR 1F (ABS(P(1)),GT.ANS(E)) E = B(1) COMTINUE (ABS (EPS) . GT . L . . AND . NS . L T . 15) CALL BANSOL INR ,MM. NN. V. B. KKKI CALL BANSOL (NR. MM. NN. V. B. I) CALL BANSOL (NR.MM.NN.V.B.0) D0 170 1 = NR.NN 200 WRITE (2) (V(1),1=1,NW) 2 E = 1./E EPS = (E-E1)/E4100. D9 230 1 = 1.NR 0 A(1) = A(1)4E 00 320 1 = 1.NR. H(1) = 0(1)/E D = A9S(H(1)-R(1)) IF (NBC.LE.01 GO DO 190 N = 1.NBC) NEIG = NEIG + 1 E1 = 0. Shift = 0. NS = 0 260 1 = 1.NR 1 = 8(1) DO 160 J = 1.4M 1 = 1.NR = NLOOP - 3 40.11A = I + SN = 5N I = NERC(N) • - • ¥ 1 = ¥ • • B(1) = 0. • SN = SN 220 CONTINUE KKK = 1 N = XXX 100 •0 E I = E ; " T XAMU (1) H CIMOS 1190 001 160 V(K) 110 0/11 ¥. 50 <u>0</u> ¥ w ш. 150 1000 002 230 260 006 υu 000 ں u ANE G BNE G nneg Bneg Bneg Bneg Bneg DENE 5 JNB BINE 9 JNE C ANEG ANEG ANEG 03NU RNFG DANEG RNEG BINEG Dang 9 ANE G RNFG PINEG R NFG NEG nue q Rue g BINE DINEG PINEG DINEG **PNFG** BINEG BINEG 5 JNB DEND BINEG BINEG RNEG PINE G DAVEG DINEG 5 JNU PNEG DENE **BINEG NNEG RNEG** BINEG BNEG DENE BINEG BINEG **DINEG** 5 JNN D I N I C JMPDSE HIJUNDARY CONDIIIDNS ON A аядеераластатататараластараларараластата жилоперисталастата X = \$4455(1) 1F (x.67.0.) GD TD 110 PRIMT 12.1 FORMAT (30PQNEG, FR ZERD MASS, EQUATION = 15) 1FLAG = 1 A(1.) = A(1.) + SMASS(1) + SMASS(K) 1F (NEC.L.E.O) GN TO 150 Г (FLAG.NE.0) STOP 10130 I = 1.4N L = 1 - 1 HR = MINO (РМ.NN-101) DD 130 J = 1.4M K = L + J G0 T0 120 SMASS(L) = 1./SORT(K) CONTINUE L=1-J+1 1= (L.LE+0) GO TO 140
 b0
 1.0
 N = 1.NRC

 1
 1
 1

 1
 1
 1

 1
 1
 1

 0
 1
 1

 0
 1
 40
 NN+1 = 1 021.00 NNR = NN - 1 PEWIND 2 TOL=1.0E-12 IFLAG = 0 NLOOP = 5 NSFAX = 50 DSHIFT = 0. NM = NMM NFIG = 0 NT = NN IN IT IAL IZE A11.J]=0. CONT INUE (L]) 110 1 30 140 2

- 164 -

00000

ں رہ ں

00000

υυv

υu

199 200 202 203 206 207 209 209 205 ۹ I N 223 222 ANEG ANEG ANEG ANEG ANEG ANEG ANEG V(1) = A110C2 + A22052 - X V(K) = A1206C2 + A11052 - X V(K) = A1206C2 - S2) + (A11-A220050 HR = M1N0 (1, MH) F (MR.LE.L) GG TO 570 IF (1.6E.NK) NEDC(N) = 1 - 1 V(LL) = A1+C - A2+5 V(L2) = A1+C - A2+5 HR = M1+0 (NHL,NR1-1) FF (MP+LT-3) GD TD 600 ų į V(L1) = A14C - A245 V(L2) = A24C + A145 LI = 1 DO 560 J = 2,44 LI = LI + NNA AI = V(LI) AZ = V(L2) 00 500 J = 3,44 L2 = L2 + NN L1 = L2 + NND A1 = V(2.1) A2 = V(2.2) C = n(k)/S0n1(0) IF (F9.L1.0.) C S = 50n1(S2) A12 = V(L) X = 2.4412454C SN(1) = S (N(1) = C O(= 0 L = NN + 1 60 TO 1000 (1)A = 1)V A22 = V(K| REWIND 2 500 V(L2) = A 600 CONTINUE 040 CONT INUE 560 V(L2) ч С 2 г 550 620 υ BNEG 127 BNEG 127 BNEG 127 BNEG 129 BNEG 130 BNEG 131 BNEG 133 BNEG 133 BNEG 133 BNEG 133 BNE G 136 BNE G 137 BNEG 139 BNEG 139 ANEG 139 BNEG 141 BNEG 142 BNEG 143 BNEG 143 BNEG 143 BNES 151 ANEG 152 BNEG 154 BNEG 155 ANFG 168 BNEG 169 ANFG 170 163 ANEG 115 ANEG 117 BNEG 117 5 N N N 148 BNEG 149 150 153 151 50 \$2 5.2.5 ANEG 145 ANEG 147 156 151 BNEG 159 44 101 091 5 JUG 9160 167 121 23NU ANEG 161 ANEG 1 BNEG BARG DINEG BANEG BNEG BAEG DENR BNEG ANEG BINEG tiant DINEG **UFG** 00 340 1 × 1.4N × = × 6011404(1) × = × 6011404(1) SHFF = SH1FF + AMAX(11.-4.\$SUMD.0.9)\$X/(Y*E) 10 350 1 = 1.4N V(1) = V(11 - 5H1FT CALL PANSOL (NP.WN.N.\$v.0.0) GO VD 400 IF (0.GT.OMAX) DMAX = D SIMM = SUMO + D + + Z 1 F (10.4X.LE.T QL.DR.+S.GE.NSWAX) G NL = NL + 1 F (11.4.1.+4.01P + 70 TO 250 REMID 2 (V(1), 1=1.NM) REMIZ (V(1), 1=1.NM) DO 430 1 × 1.NN R{11 = 841/7 IF (NEIG.GE.NEV) GO TO 530 REWIND 2 DEAD(2) {V(!), {±}, {M}} DO 450 NX = 1, NP FB = A(NX) F {FB, AFC, 0, } G1 TO 400 К и I + 1 0 и 01 + Rik)++2 1F (1,LT.NX) 4Л 70 550 50 и 01/0 V(1) = V(1) - SHIFT 00 600 1 × 8.NRS K u 1 + 2 00500 1 ≈ 1, MR L = NW ↔ 1 01 = R(1)##2 52 # 0. C = 1. 0 + 20 = 300 0 + 30 = 300 500 VIL) = 0. 60 10 250 CONT INUE о 0 н н х х NL * 0 • • ¢ ñ 450 480 320 0 V E 350 400 420 020

- 165 -

.....

000000000000000000000000000000000000	650 00 1 = 1,00 1 = 1,00 00 1 = 1,00 700 11.0EV) = 110 00 100 100 100 100 11.0EV = N(T) 00 100 100 100 100 100 100 11.0 00 00 100	NHEG 220 NH - 1 NHEG 731 100 DG 200.300.400 NHEG 731 100 DG 200.300.400 NHEG 731 110 DG 200.300.400 NHEG 731 11 11.00.300.400 11.411 NHEG 731 11 11.01.610.01 A(N.11) 11.0E-16 NHEG 733 11 11.01.610.01 A(N.11) 11.0E-16 NHEG 733 11.001 2.411.01 A(N.11) 1.0E-16 NHEG 733 000 200 L 2.411 NHEG 731 11 2.411 1.00 NHEG 741 11 2.411 2.411 NHEG 741 11 2.411 2.411 2.41	AANS AANS AANS AANS AANS AANS AANS AANS	
$ \begin{array}{cccccccccccccccccccccccccccccccccccc$	700 A11.MEV F (10 D0	NHEG 2:30 IF (KKK-1) 100.300.400 NHEG 2:31 NHE 2:20 NHEG 2:31 NHE 1.00.300.400 NHEG 2:31 NHE 2:00.0 NHEG 2:31 NHE 2:00.0 NHE 2:33 NHE 2:00.0 NHE 2:33 NHE 1.00.300.400 NHE 2:33 NHE 2:00.0 NHE 2:35 NH = 1.00.1 NHE 2:35 NH = 1.01.01 NHE 2:35 NH = 1.01.01 NHE 2:37 D1 2:00 L = 2.4N NHE 2:37 D1 1:0 N L NHE 2:27 D1 1:0 N L NHE 2:27 D1 1:0 N N NHE 2:27 D1 1:0 N N	AANS AANS AANS AANS AANS AANS AANS AANS	
$ \begin{array}{cccccccccccccccccccccccccccccccccccc$	IF (NEV.LE.1) GO TO 00 00 00 00 00 00 00 00 00 00 00 00 00	BVEG 231 100 Dd 200 N = 1.NB BVEG 233 $r * N = 1$ BVEG 234 $r * N = 1$ BVEG 235 $r * N = 1$ BVEG 234 $r * N = 1$ BVEG 235 $r * N = 1$ DO 00 L = A(N,L) / r * 1 / r N) PVEG 237 $r = A(N,L) / r * 1 / r N)$ PVEG 237 $r = A(N,L) / r * 1 / r N)$ PVEG 239 $1 = (r \in 0, 0, 1) < r + 1 / r N)$ PVEG 230 $1 = (r \in 0, 0, 1) < r + 1 / r N)$ PVEG 230 $1 = (r + 1, L) - r + 1 / r N)$ PVEG 231 $1 = (r = 0, 0, 1) < r + 1 / r N)$ PVEG 240 $0 = 1 = (r + 1, R)$ PVEG 241 $0 = 1 = (r + 1, R)$ PVEG 243 $1 = A(1, L) = r + 1 / R$ PVEG 243 $1 = (r + 1, R)$ PVEG 245 $1 = A(1, L) = r + 1 / R$ PVEG 245 $1 = N = 1 / R$ PVEG 240 $1 = N = 1 / R$ PVEG 241 $1 = 2 / R$ PVEG 251 $0 = 1 = N = 1 / R$ PV	TANS TANS TANS TANS TANS TANS TANS TANS	
$ \begin{array}{cccccccccccccccccccccccccccccccccccc$	$ \begin{array}{c} (1 = MEV - 1) \\ 0 = MEV - 1 \\ 0 = MEV - N \\ MEG = 232 \\ MEG = 233 \\ ME$	NHEG 2.32 M = M = 1 RNEG 2.31 IF (AIN, I).EG(0.0.1) A(N, I) RNEG 2.35 IF (AIN, I).EG(0.0.1) A(N, I) RNEG 2.35 RN = MINO MN-MINO RNEG 2.30 IF (C.50.0.1) Gn RNEG 2.30 IF (C.50.0.1) Gn IN RNEG 2.40 DO 10 MN L RNEG 2.41 L MN-MINO MN-MINO RNEG 2.41 L L MN-MINO RNEG 2.41 L L MN-MINO RNEG 2.41 L L L RNEG 2.41 L L L RNEG 2.43 R	AMNS AMNS <td></td>	
	$\begin{array}{cccccccccccccccccccccccccccccccccccc$	BVEG 231 IF (A(N, 1), E(n, 0, 1) A(N, 1) $1,0E-16$ RVEG 234 WIR $a \times 100$ $a \times 100$ $a \times 100$ $a \times 100$ RVEG 235 WIR $a \times 100$ $a \times 100$ $a \times 100$ $a \times 100$ RVEG 235 WIR $a \times 100$ $a \times 100$ $a \times 100$ $a \times 100$ RVEG 237 C $a \times 100$ $a \times 100$ $a \times 100$ $a \times 100$ RVEG 230 I $a \times 100$ $a \times 100$ $a \times 100$ $a \times 100$ RVEG 240 $b \times 100$ $a \times 100$ $a \times 100$ $a \times 100$ $a \times 100$ RVEG 241 $b \times 100$ $a \times 100$ $a \times 100$ $a \times 100$ $a \times 100$ RVEG 243 100 110 $a \times 100$ $a \times 100$ $a \times 100$ RVEG 245 100 110 $a \times 100$ $a \times 100$ $a \times 100$ RVEG 240 00 100 $a \times 100$ $a \times 100$ $a \times 100$ RVEG 240 00 100 $a \times 100$ $a \times $	1 1 <td></td>	
$ \begin{array}{c} \mbox{rel} \mb$	$ \begin{array}{c} \mathbf{H} = \mathbf{H} \mathbf{V} - \mathbf{H} \\ \mathbf{H} = \mathbf{H} \mathbf{V} + \mathbf{H} \\ \mathbf{H} = \mathbf{V} \mathbf{V} + \mathbf{H} \\ \mathbf{H} = \mathbf{H} \mathbf{V} + \mathbf{H} \\ \mathbf{H} = \mathbf{H} \mathbf{V} + \mathbf{H} \\ \mathbf{H} + \mathbf{H} \mathbf{V} + \mathbf{H} \mathbf{V} \\ \mathbf{H} + \mathbf{H} \mathbf{H} \\ \mathbf{H} + \mathbf{H} \mathbf{H} \\ \mathbf{H} \\ \mathbf{H} + \mathbf{H} \mathbf{H} \\ \mathbf{H} \\ \mathbf{H} + \mathbf{H} \mathbf{H} \\ \mathbf{H} \\ \mathbf{H} \\ \mathbf{H} \mathbf{H} \mathbf{H} \\ \mathbf{H}$	RHEG 234 $PIVOI = A(N_1)$ RHEG 235 HH = MINO (MAINH-MI) PHEG 237 D0 200 L = 2,443 RHEG 239 1F (C.EC.0.1) GAI (120) 100 110 RHEG 239 1F (C.EC.0.1) GAI (120) 100 110 RHEG 239 1F (C.EC.0.1) GAI (120) 100 110 RHEG 240 D0 100 111 1 RHEG 241 D0 100 1 1 RHEG 241 D0 111 1 1 RHEG 243 D0 1 1 1 1 RHEG 243 D0 1 1 1 1 1 RHEG 243 D0 1 1 1 1 1 1 1 1 0 RHEG 245 D0 350 1 1 1 1 0 1 1 1 0 1 1 1 0 1 1 1 0 1	11 12 24 24 24 24 24 24 24 24 25 24 26 24 26 24 27 24 26 24 27 24 26 24 27 24 28 24 29 24 20 24 26 24 27 24 28 24 29 24 20 24 21 24 26 24 27 24 28 24 29 24 20 24 21 24 26 24 27 24 28 24 29 24 20 24 20 24 21 24 26 24 27 24 28 24 29 24 20 24 20 24 21 24 26 24 27	
$ \begin{array}{cccccccccccccccccccccccccccccccccccc$	NIS $NIV - H$ NIS $NIV - H$ NIS $NIV - HS$ NIS $IVS - HS$ NIS $IVS - HS$ NIS $IVS - HS$ HACKSPACE <	RNFG 235 HR = MINO (MM.NN-M) RNFG 234 D0 200 L = 2.4MR RNFG 234 D1 200 L = 2.4MR RNFG 230 L = 2.4MR RNFG 230 I = 4.14.1.1.1.1.1.1.1.1.0 RNFG 240 J = 4.1 RNFG 241 D0 100 K = L.4MR RNFG 241 D0 100 K = L.4MR RNFG 241 D0 100 K = L.4MR RNFG 242 D0 100 K = L.4MR RNFG 243 D0 100 K = L.4MR RNFG 244 D0 100 K = L.4MR RNFG 245 D0 01500 RNFG 245 D0 00 350 N = 1.4MR RNFG 240 D0 00 350 N = 1.4MR RNFG 250 MR 10.1 RNFG 251 D0 00 350 N = 1.4MR RNFG 253 D0 00 350 N = 2.4MR RNFG 253 D0 00 350 L = 2.4MR RNFG 253 D0 00 350 L = 2.4MR	TI NNE PNNE NNE </td <td></td>	
No. No. <td>Mr = Mrs + 1 $Mrs = 23x$ $Prs = 1$ $Prs = 11$ $Prs = 237$ $Prs = 237$</td> <td>NFG 23A D0 200 L = 2.40 NHEG 237 C = $4(4.L)/PI'VIT$ F NHEG 230 I = $(C.FC0.0.)$ Gn ID 200 NHEG 230 I = $4.4.L$ PI VIT NHEG 230 I = $4.4.L$ Gn ID 200 NHEG 230 I = $4.4.L$ PI VIT NHEG 240 J = J + 1 PI VIT NHEG 240 J = J + 1 PI VIT NHEG 241 D0 170 K = L, MR NHEG 241 D1 A1L, J = A(1, J) - CAAN, KI NHEG 243 D11L, J = A(1, J) - CAAN, KI NHEG 243 D10 11L, J = A(1, J) - CAAN, KI NHEG 244 ANL, J = C NHEG 245 D0 350 N = 1, NR NHEG 250 NA NHEG 251 NA NHEG 251 NA NHEG 253 D1 350 NHEG 254 D1 350 NHEG 255 D1 350 NHEG 254 D1 14.1</td> <td>PANK PANK PANK PANK PANK PANK PANK PANK</td> <td></td>	Mr = Mrs + 1 $Mrs = 23x$ $Prs = 1$ $Prs = 11$ $Prs = 237$	NFG 23A D0 200 L = 2.40 NHEG 237 C = $4(4.L)/PI'VIT$ F NHEG 230 I = $(C.FC0.0.)$ Gn ID 200 NHEG 230 I = $4.4.L$ PI VIT NHEG 230 I = $4.4.L$ Gn ID 200 NHEG 230 I = $4.4.L$ PI VIT NHEG 240 J = J + 1 PI VIT NHEG 240 J = J + 1 PI VIT NHEG 241 D0 170 K = L, MR NHEG 241 D1 A1L, J = A(1, J) - CAAN, KI NHEG 243 D11L, J = A(1, J) - CAAN, KI NHEG 243 D10 11L, J = A(1, J) - CAAN, KI NHEG 244 ANL, J = C NHEG 245 D0 350 N = 1, NR NHEG 250 NA NHEG 251 NA NHEG 251 NA NHEG 253 D1 350 NHEG 254 D1 350 NHEG 255 D1 350 NHEG 254 D1 14.1	PANK PANK PANK PANK PANK PANK PANK PANK	
$ \begin{array}{cccccccccccccccccccccccccccccccccccc$	$ \begin{array}{c} \mbox{figstrate} 1 & \mbox{figstrate} 2 & \mbox{figstrate} 1 & \mbox{figstrate} 2 & \mbox{figstrate} 1 & \mbox{figstrate} 2 & \$	NHFG 237 $C = A(N_{*}L/P) VOT$ NHEG 239 $I \neq (C, C_{*}O_{*}O_{*})$ G_{*} NHEG 239 $I \neq (C, C_{*}O_{*}O_{*})$ G_{*} $I_{*} = 0$ NHEG 240 $J = 0$ $D_{*} = 0$ $D_{*} = 0$ NHEG 241 $D_{*} = 0$ $D_{*} = 0$ $D_{*} = 0$ NHEG 243 $D_{*} = 0$ $D_{*} = 1 + H$ $D_{*} = 0$ NHEG 243 $D_{*} = 0$ $D_{*} = 1 + H$ $D_{*} = 0$ NHEG 243 $D_{*} = 1 + H$ $D_{*} = 1 + H$ $D_{*} = 0$ NHEG 243 $D_{*} = 1 + H$ $A(N_{*}, I) = C$ $A(N_{*}, I)$ NHEG 243 $D_{*} = 1 + H$ $D_{*} = 0$ $D_{*} = 1 + H$ NHEG 245 $D_{*} = 1 + H$ $D_{*} = 0$ $D_{*} = 0$ NHEG 245 $D_{*} = 1 + H$ $D_{*} = 0$ $D_{*} = 0$ NHEG 240 $D_{*} = 1 + H$ $D_{*} = 0$ $D_{*} = 0$ NHEG 240 $D_{*} = 0$ $D_{*} = 1 + H$ $D_{*} = 0$ NHEG 250 $D_{*} = 0$ $D_{*} = 1 + H$ $D_{*} = 0$ NHEG 250 $M_{*} = 1 + H$ $M_{*} = 0$ $M_{*} = 0$ NHEG 250 <td< td=""><td>100 100 100 100 100 100 100 100</td><td></td></td<>	100 100 100 100 100 100 100 100	
Contraction Contraction <thcontraction< th=""> <thcontraction< th=""></thcontraction<></thcontraction<>	$ \begin{array}{c} P N F C \\ P A C C T A C C A C C A A A C A A A A A A A A$	BMEG 23A IF (C.E0.0.) GA IA 200 BMEG 240 I = M + L BMEG 241 J = 0 BMEG 241 J = 1 BMEG 243 J = 1 BMEG 243 BO 100 K = L.MR BMEG 243 J = 1 BMEG 243 BO 100 K = L.MR BMEG 243 BO 10.0 K = L.MR BMEG 243 BO 11.1 = A(1) - CANN.KI BMEG 245 BO CONTIUCE BMEG 245 BO CONTIUCE BMEG 240 J (AN.L) = C BMEG 240 J (AN.L) = C BMEG 240 J (AN.L) = C BMEG 251 GO 10 500 BMEG 252 MA = MINO (MM.N-M) BMEG 253 MA = MINO (MM.N-M) BMEG 254 DO 350 N = L.MN BMEG 255 DO 350 L = 2.01 BMEG 254 DO 350 L = 2.01 BMEG 255 DO 350 L = 2.01 BMEG 255 DO 350 L = 2.01	25 20 25 20 26 20 20 20 20 20 20 20 20 20 20 20 20 20 20 20 2	
$ \begin{array}{cccccccccccccccccccccccccccccccccccc$	$ \begin{array}{c} HACK SPACE \ I \\ KK = W + I \\ KK = W + I \\ T = VR - L \\ T = VR - R \\ T = V \\ T = $	NHEG 270 $1 = M + L$ RHEG 240 $J = 0$ $s = L + HR$ RHEG 241 DD 100 K = L + HR RHEG 242 $J = J + 1$ RHEG 243 DD 11(J) = J + 1 RHEG 243 DD 11(J) = A(1,J) - CAA(N,K) RHEG 243 100 A(1,J) = A(1,J) - CAA(N,K) RHEG 245 100 A(1,J) = A(1,J) - CAA(N,K) RHEG 245 200 CUNTINUE RHEG 240 JF (A(1N,1)) = G.O.O.O.) RHEG 240 MM-1 RHEG 240 MM-1 RHEG 240 MM-1 RHEG 250 MM-1 RHEG 251 C.C.B(1) RHEG 253 DD 350 N = 1, NP RHEG 251 C.C.B(1) MM-N-M RHEG 253 DD 350 N = 1, NP RHEG 251 C.C.B(1) MM-N-M RHEG 253 DD 350 N = 1, NP RHEG 254 DD 350 N = 1, NP RHEG 253 DD 350 L = 2, NR RHEG 254 DD 350 L = 2, NR	2011 2011 2011 2011 2011 2011 2011 2011	
$ \begin{array}{cccccccccccccccccccccccccccccccccccc$	$ \begin{array}{c} K K = U + I \\ D D O O L = I, M S \\ D D O O L = I, M S \\ D D O O L = I, M S \\ D D O O O L = I, M S \\ D D O O O L = I, M S \\ D D O O O L = I, M S \\ D D O O O D J = N S \\ D D O O O J = N S \\ D D D O O D J = N S \\ D D D O O J = N S \\ D D D O D J = N S \\ D D D D D D D \\ D D D D D D \\ D D D D D D \\ D D D D D D D \\ D D \\ D D D D D D D \\ D D \\ D D D D \\ D \\ D \\ D D \\ D \\$	0.0456 > 40 $J = 0$ $0.046 > 41$ $0.0 + 100 + x = 1.443$ $0.0476 > 243$ $0.0 - 140 + x = 1.443$ $0.045 > 243$ $100 - 14.41 = 4$ $0.045 > 243$ $100 - 14.41 = 4$ $0.045 > 243$ $0.041.41 = 4$ $0.045 > 243$ $0.041.41 = 4$ $0.045 > 243$ $0.041.41 = 4$ $0.045 > 243$ $0.041.41 = 4$ $0.045 > 243$ $0.041.416$ $0.045 > 243$ $0.00 - 350 - x = 1.047$ $0.045 > 243$ $0.00 - 350 - x = 1.047$ $0.045 > 243$ $0.00 - 350 - x = 2.041$ $0.045 > 253$ $0.01 - 350 - x = 2.041$ $0.045 > 253 - 0.01 - 0.01 - 0.01$ $0.01 - 0.01$ $0.045 > 253 - 0.01 - 0.01 - 0.01$ $0.01 - 0.01$ $0.045 > 0.01 - 0.01 - 0.01$ $0.01 - 0.01$ $0.045 > 0.01 - 0.01 - 0.01$ $0.01 - 0.01$ $0.045 > 0.01 - 0.01 - 0.01$ $0.01 - 0.01$ $0.045 > 0.01 - 0.01 - 0.01$ $0.01 - 0.01$ $0.045 > 0.01 - 0.01 - 0.01$ $0.01 - 0.01$ $0.045 > 0.01 - 0.01 - 0.01$ $0.01 - 0.01$ $0.045 > 0.01 - 0.01 - 0.01$ $0.01 - 0.01$ $0.045 > 0.01 - 0.01$	22 200 2	
$ \begin{array}{cccccccccccccccccccccccccccccccccccc$	p_1 p_0 $t = 1, \text{MS}$ p_1 p_1 p_1 p_2 $1 = 1, \text{MS} - L$ $1 = 1, \text{MS}$ p_1 p_1 p_1 p_2 $1 = 1, \text{MS} - L$ p_1 p_1 p_1 p_1 p_1 p_2 $x = 1 + 1$ p_1 p_1 p_1 p_1 p_1 p_1 $x = 1 + 1$ p_1 p_1 p_1 p_2 p_1 p_1 $x = 1, 1, 1$ p_1	NFG PAL DOI DOI X X HA DHFG 242 J J J L HA DHFG 242 J J J L L DHFG 243 J J J L L DHFG 243 JA L L C C L DHFG 243 JA A(N,L) = C C A(N,L) = C L DHFG 245 JA A(N,L) = C A(N,L) = C L L DHFG 245 JA JA L L L L DHFG 250 N = 1,NP N N N L L DHFG 250 N = 1,NP N N N L L DHFG 250 N = 1,NP N N L N L DHFG 250 N = 1,NP N N L L	22 SNAR 23 SNAR 25 SNAR	
$ \begin{array}{c} 1 \cdot 1 $	$ \begin{bmatrix} 1 & \text{MR} - L \\ 1 & \text{MR} - L \\ 0 & 000 & 3 & \text{KK,WEV} \\ 0 & 000 & 3 & \text{KK,WEV} \\ 0 & 000 & 3 & \text{KK,WEV} \\ 1 & \text{AII} & \text{AII} & \text{AIII} \\ 0 & 000 & 3 & \text{KK,WEV} \\ 1 & \text{AIII} & \text{AIII} & \text{AIII} \\ 2 & \text{AIII} & \text{AIII} \\ 2 & \text{AIII} & \text{AIIII} \\ 2 & \text{AIII} & \text{AIIII} \\ 2 & \text{AIIIII} \\ 2 & \text{AIIIIII} \\ 2 & \text{AIIIII} \\ 2 & \text{AIIIIII} \\ 2 & \text{AIIIIII} \\ 2 & \text{AIIIIII} \\ 2 & \text{AIIIIIII} \\ 2 & AIIIIIIIIIIIIIIIIIIIIIIIIIIIIIIIIIIII$	BVFG 242 $J = J + 1$ BVFG 243 BO A1(J) = A(1,J) = C4A(N,K) BVEG 245 BO A1(J) = A(1,J) = C4A(N,K) BVEG 245 200 CUNTINUE BVEG 247 300 D350 N = 1.NP BVEG 250 N = 1.NP BVEG 251 C = B(N) BVEG 252 NM = N = 1 BVEG 253 D = 20 D1 350 N = 1.NP BVEG 251 C = B(N) BVEG 253 D = 20 D1 350 L = 2.MIN BVEG 254 D = 0.2.2 D1 01 350 L = 2.MIN BVEG 254 D = 0.2.2 D1 01 250 L = 2.MIN	AANS AANS AANS AANS AANS AANS AANS AANS	
$ \begin{array}{cccccccccccccccccccccccccccccccccccc$	x = 1 + 1 $x = 1 + 1$ $y = 1 + 1 + 1 + 1$ $y = 1 + 1 + 1 + 1$ $y = 1 + 1 + 1 + 1 + 1$ $y = 1 + 1 + 1 + 1 + 1 + 1$ $y = 1 + 1 + 1 + 1 + 1 + 1 + 1$ $y = 1 + 1 + 1 + 1 + 1 + 1 + 1 + 1 + 1 + 1$	NHEG 243 100 A(1, J) = (J, J) - C4A(N, K) NHEG 244 A(N, L) = C 0.00 200 A(N, L) = C 0.00 200 A(N, L) = C 0.00 200 COUTUNE C 0.00 200 COUTUNE C C 0.00 200 COUTUNE A(NU-1) 1.00E-16 0.00 200 300 300 300 300 0.00 300 300 300 300 M. NU-M) 0.00 250 M M. NU-M) M. NU-M) 0.016 250 M. NU-M) M. NU-M) M. NU-M) 0.017 2.017 2.017 1.017 M. NU-M) 0.016 250 1.2.2.017 M. NU-M) M. NU-M) 0.017 2.017 2.017 2.017 M. NU-M) 0.017 2.017	AANS 25 AANS 25 AANS 26 AANS 26 AANS 26 AANS 20 AANS 2	
$ \begin{array}{cccccccccccccccccccccccccccccccccccc$	$ \begin{array}{cccccccccccccccccccccccccccccccccccc$	DNEG 244 A(N,L) = C DNEG 245 200 CUNTINE DNEG 245 200 CUNTINE DNEG 245 17 (NN.1) = 1.0E-16 DNEG 240 300 D0 350 N = 1.NP DNEG 240 300 D0 350 N = 1.NP DNEG 240 N = 1.NP DNEG 240 N = 1.NP DNEG 250 N = 1.NP DNEG 251 C = 81(N) DNEG 252 D(N = 2.2,NP) DNEG 253 D(N = 2.2,NP) DNEG 254 N = 4.1 DNEG 255 D(N = 2.2,NP)	25 25 26 20 20 20 20 20 20 20 20 20 20 20 20 20	
$ \begin{array}{cccccccccccccccccccccccccccccccccccc$	Al = $A(1, J)$ Al = $A(1, J)$ Al = $A(1, J)$ = $A1000(1)$ - $A2950(1)$ Al = $A100(1)$ - $A100(1)$ - $A100(1)$ BOD - $A100(1)$ - $A100(1)$ BOD - $A100(1)$ - $A100(1)$ BOD - $A100(1)$ BOD - $A100(1)$ - $A100(1)$ BOD - A	0.865 245 200 CUNTINUE 0.875 0.5 1° $(1.0.1)$ 0.875 2° 1° $(1.0.1)$ 0.875 2° 1° $(1.0.1)$ 0.875 2° 1° $(1.0.1)$ 0.875 2° 1° 1° 0.865 2° 1° 1° 0.865 2° 0.0 0.0 0.865 2° 0.0 0.0 0.865 2° 0.0 0.0 0.865 2° 0.0 0.0 0.865 0.0 0.0 0.0 0.876 0.0 0.0 0.0 0.876 0.0 0.0 0.0 0.01 0.01 0.01 0.01	214 SVAN 214 SVAN 214 SVAN 215 SVAN	
11.1 × 11.001 11.2 × 11.01 11.2 × 11.01 11.0 × 11.01 10.001 20.01	$ \begin{array}{c} \lambda 2 = \lambda(k,J) \\ \lambda(1,J) = \lambda + \lambda(u(1) + \lambda 2 + SM(1)) \\ \lambda(1,J) = \lambda + \lambda(u(1) + \lambda 2 + SM(1)) \\ 000 \ \lambda(k,J) = \lambda + SCM(1) - \lambda + SM(1) \\ 000 \ \lambda(k,J) = \lambda + SCM(1) - \lambda + SM(1) \\ 000 \ \lambda(k,J) = \lambda + SCM(1) - \lambda + SM(1) \\ 000 \ \lambda(k,J) = \lambda + SCM(1) - \lambda + SM(1) \\ 000 \ \lambda(k,J) = \lambda + SCM(1) - \lambda + SM(1) \\ 000 \ \lambda(k,J) = \lambda + SCM(1) - \lambda + SM(1) \\ 000 \ \lambda(k,J) = \lambda + SCM(1) - \lambda + SM(1) \\ 000 \ \lambda(k,J) = \lambda + SCM(1) - \lambda + SM(1) \\ 000 \ \lambda(k,J) = \lambda + SM(1) \\ 000 \ \lambda(k,J) \\ 000$	BNEG 245 IF (A(NN,1).EQ.0.) A(NN,1] = 1.0E-16 RNEG 247 GO TO 500 RNEG 249 300 DO 350 N = 1.NP RNEG 240 N = 1.NP RNEG 240 N = 1.NP RNEG 250 N = N - 1 RNEG 250 N = N - 1 RNEG 251 C = 81N0 RNEG 253 B(N) = C/A(N,1) RNEG 254 B(N) = C/A(N,1) RNEG 255 B(N) = C/A(N,1) RNEG 255 B(N) = C/A(N,1)	8 - 5 - 5 - 5 - 5 - 5 - 5 - 5 - 5 - 5 -	
$ \begin{array}{c} 0.1 (1,1) = 1 (1 + (1,1) + (1,$	$\Lambda(1, J) = \Lambda(CM(1) + \Lambda 2^{5}SM(1))$ $\Lambda(1, J) = \Lambda_2(CM(1) - \Lambda 1^{5}SM(1))$ $\Lambda(1, J) = \Lambda_2(CM(1) - \Lambda 1^{5}SM(1))$ $\Lambda(1, J) = \Lambda_2(CM(1) - \Lambda 1^{5}SM(1))$ $\Lambda(1, J) = \Lambda(1, J)$	RMEG 2.47 G0 T0 500 RMEG 2.40 300 b0 50 w w w RMEG 2.40 300 b0 50 w w w w RMEG 2.50 W w w w w w RMEG 2.51 C w W w w w RMEG 2.52 B(1) w w w w w RMEG 2.53 D(1) 3.50 L w w w RMEG 2.54 D(1) M w	840 840 840 840 840 840 840 840 840 840	
000 010 010 010 1	000 A(K,J) = A1*SN(I) A1*SN(I) A1*SN(I) A1*SN(I) A0 D0 D1	RNEG 2.40 300 D0 350 N = 1.40 RNEG 2.40 M = N = -1 MM_NN-M BNEG 2.51 C 8144 BNEG 2.51 C 8144 RNEG 2.53 D1 300 RNEG 2.53 D1 300 RNEG 2.54 D1 2.414	SE SAVE	
$ \begin{array}{cccccccccccccccccccccccccccccccccccc$	$ \begin{array}{cccccccccccccccccccccccccccccccccccc$	BNEG 240 M = N - 1 BNEG 250 MR = M100 (MM,NN-M) BNEG 251 C = 8(N) BNEG 253 B(N) = C/A(N,1) BNEG 253 D(1) = C/A(N,1) BNEG 253 D(1) = C/A(N,1) BNEG 253 D(1) = C/A(N,1) BNEG 254 D(1) = M + L	10 20 20 20 20 20 20 20 20 20 20 20 20 20	
$ \begin{array}{cccccccccccccccccccccccccccccccccccc$	M = LL - K + 1 M = L - K + 1 M = 0 D0 050 J = 1,4 D0 620 J = 0 M = 0 E1 = EV(J) E2 = EV(J+1) N = 0 E2 = EV(J+1) N = 0 N = 0 E1 = E1.1-E2 G0 f1 0 N = 0 E1 = E(J+1) N = 0 N = 0 E2 = EV(J+1) N = 0 N = 0 E1 = E(J+1) N = 0 N = 0 EV(J) = E1 E1 E1	BNEG $SS1$ $MR = MINO$ $MN-M$ $BNEG$ $SS1$ $C = BINO$ $MN-M$ $BNEG$ $SS1$ $C = BINO$ $MN-M$ $BNEG$ $SS2$ $BIN1$ $MN-M$ $BNEG$ $SS3$ $BI330$ $L = 2.MR$ $BNEG$ $SS6$ $L = 2.MR$ $BNEG$ $SS6$ $L = 2.MR$	SURE SURE Sure Sure Sure Sure	
$ \begin{array}{cccccccccccccccccccccccccccccccccccc$	00 0.0 0.1 0.1 0.1 0.1 E1 E1 E(1) 0.1 0.1 0.1 E2 E(1) 0.1 0.1 0.1 0.1 E1 (E1) E1 E1 0.1 0.1 EV(J) E1 E1 0.1 0.1 0.1 EV(J) E1 E1 0.1 0.1 0.1	DNEG 251 C = 8(N) ANET 252 B(N) S C / (N, 1) ANEG 253 D1 350 L = 2.4R ANEG 253 D1 = M + L = 2.4K + 1) ANEG 255 D = 0.0 D1 = M + L	50 5 NVG 50 5 NVG 51 5 NVG	
$ \begin{array}{c} \text{E} = \text{FU11} \\ \text{E} = \text{FU11} $	E1 = EV(J) E2 = EV(J) BNEG 252 B(N) E2 = EV(J+1) BNEG 253 D0 3 BNEG 253 D0 3 F1 E(LIJLE2) G0 Th BJO BNEG 253 J50 B(1) EV(J+1) = E1 BNEG 255 350 B(1) EV(J) = E2 BNEG 256 A00 B(N)	ANEG 257 B(N) = C/A(N,1) Aneg 253 D(1 = 2,4R Aneg 254 1 = M + L Aneg 255	RANS 34	
$ \begin{array}{c} \label{eq:constraints} & \begin{tabular}{lllllllllllllllllllllllllllllllllll$	E2 = EV(J+1) B1 = E2 = EV(J+1) B1 = E2 = EV(J+1) = E1 =	ANEG 253 DA 350 L = 2,4R ANEG 254 L = 4 L ANEG 255 - 20 AND - 244 AND	HANS JE	
Fr(14) F(14,1/E2) GO TO B30 1 * * * 1 Ansis 37 <	IF (E1.4.T.6.2) GO TO 830 EV(J+1) = E1 EV(J+1) = E2 EV(J) = E2 A00 8(N)	RNFG 254 I = M + L RNFG 255 - 55 - 55 - 51 - 21 - 21 - 21 - 21		
$ \begin{array}{c} \mathbb{E} V(J_{1}) = \mathbb{E} \\ \mathbb{E} V(J_{1}) = \mathbb{E} V(J_{1}) = \mathbb{E} \\ \mathbb{E} V(J_{1}) = \mathbb{E} V(J_{1}) = \mathbb{E} V(J_{1}) = \mathbb{E} V(J_{1}) = \mathbb{E} \\ \mathbb{E} V(J_{1}) = \mathbb{E} V(J$	EV(J) = EI BNEG 255 350 0(1) EV(J) = E2 A00 8(N)	RNFG 255 350 350 350 3 20 1 - 400 3 140	ANS 36	
$ \begin{array}{cccccccccccccccccccccccccccccccccccc$	EV(J) = E2 A00 B(N)		PANS 37	
$ \begin{array}{cccccccccccccccccccccccccccccccccccc$		HEG 256 400 B(NN) = B(NN)/A(NN, 1)	BANS 38	
$ \begin{array}{c} \label{eq:constraint} \mbox{160 Constraint} $	00 820 1 ± 1.NN	UNES 257 DO 450 K # 2,NN	OF SNVB	
20 ATLJJI TANT ATLJJI ATLJJI ATLJJI ATLJJI ATLJJI ATLJJI ATLJJI ATLJJI ATLJJII ATLJJIII ATLJJIII ATLJJIII ATLJJIII ATLJJIII ATLJJIII ATLJJIII ATLJJIII ATLJJIII ATLJJIIII ATLJJIII ATLJJIIII ATLJJIIIIII ATLJJIIII ATLJJIIIII ATLJJIIIIIIIIIIIIIIIIIIIIIIIIIIIIIIIIIII			HANS AU	
720 CM11.011 = TEM 720 CM11.011 TEM 110 4.0.1 110 4.0.1 1100 5.0.0 110 4.0.1 1100 5.0.0 11000 5.0.0 110000 5.0.0 110000 5.0.0 11000000 5.0.0	A(1,1) ± A(1,1+1) = (1,1+1			
910 COMPTUNE 01 450 L -2.40 mmss -3.40 910 0.0 0.0 2.10 mms -4.1 <t< td=""><td>920 A(1, 541) = TEMP</td><td></td><td>HANS 42</td><td></td></t<>	920 A(1, 541) = TEMP		HANS 42	
90 00 <td< td=""><td>A30 CONTINUE</td><td>THE 251 DO 450 L = 2.46</td><td>HANS 43</td><td></td></td<>	A30 CONTINUE	THE 251 DO 450 L = 2.46	HANS 43	
3435(1) = 1, xto 3435(1) = 000 REUNN 450 NUN = 0100 FLUAN 20 020 0 = 1, NEV 00 920 0 = 1, NEV 11, J) = 4(1, J) + X(1, J) +			44 5NVI	
920 ASS(1) = 1.7X* 00 ASS(1) = 1.7X* 0	X = 54/35(1) 450 B(X)	HILG 203 450 B(N) = B(N) - A(N,L)+B(1)	04 SNV6	
00 020 01 1 411.01) * 411.016 x 410 410 020 01 1 411.016 x 410 416 410 416 410 020 010 1 411.016 x 410 416 410 410 410 410 010 010 1 410	SWASS(1) = 1./X442 - 200 RETL		Ce SNYH	
920 ATT.JT =	00 920 J = 1, NEV END	ENG 205 END	HANS 4	
HETUNA HETUNA END SUDROUTINE RANSOL FUN, 444, 400 NA. A. P. KKK) SUDROUTINE CINTER (X, Y, 444 X, 0X, A) SUDROUTINE CINTER (X, Y, 444 X, 0X, A) SUDROUTINE CINTER (X, Y, 444 X, 0X, A) SUDROUTINE (X, 444 X, 0X, 444 X, 0X, A) SUDROUTINE (X, Y, 444 X, 0X, 0X, 444 X, 0X, 444 X, 0X, 0X, 444 X, 0X, 0X, 0X, 0X, 0X, 0X, 0X, 0X, 0X,				
GINT SUMPOUTINE CINTER(X,Y,WMAX,DX,A) CINT C SUPPOUTINE BANSOL (NN,WW,NDIN,A,B,KKK) BANS E SUBROUTINE (INTERPOLATE A FUNCTION AT FOUAL INTERVALS IF X CINDINCINI 3 SUPPOUTINE BANSOL (NN,WW,NDIN,A,B,KKK) BANS I C SUBROUTINE (INTERPOLATE A FUNCTION AT FOUAL INTERVALS IF X CINDINCINI 3 C NUMB BANSOL (NN,WW,NDIN,A,B,KKK) BANS I C SUBROUTINE (INTERPOLATE A FUNCTION AT FOUAL INTERVALS IF X CINDINCINI 3 C NUMB I C SUBROUTINE (INTERPOLATE A FUNCTION AT FOUAL INTERVALS IF X CINDINCINI 3 CINT 3 C NUMB I C NUMB INCREMENTARY CINT 3 C NUMB I C NUMB I C NUMB 3 C NUMB CINT 3 C C NUMB C				
SUMPOUTINE ANSOL TNN, WA.M.D.H., A.B.KKK) SUMPOUTINE TO INTERPOLATE A FUNCTION AT FOUAL INTERVALS IF & CINPDITCINI 3 SUMPOUTINE TO INTERPOLATE A FUNCTION AT FOUAL INTERVALS IF & CINPDITCINI 3 C NUMERATION STATEMENT TO THE ALTERVALS IF & CINPDITCINI 3 C NUMERATION STATEMENT TO THE ALTERVALS OF E CONTREST THE STATEMENT FOUAL INTERVALS IF & CINT 5 C NUMERATION STATEMENT TO THE ALTERVALS IF & CINPTICINI 3 C NUMERATION STATEMENT TO THE ALTERVALS OF E CONTREST OF BE GENERATED AT FOUAL INTERVALS IF & CINT 5 C NUMERATION STATEMENT TO THE ALTERVALS AT FOUAL INTERVALS IF & CINT 5 C NUMERATION STATEMENT TO THE ALTERVALS AT FOUAL INTERVALS IF & CINT 5 C NUMERATION STATEMENT TO THE ALTERVALS AT FOUAL INTERVALS IF & CINT 5 C NUMERATION STATEMENT TO THE ALTERVALS AT FOUAL INTERVALS AT FOUAL INTERVALS OF A STATEMENT AT A STATEME				
GINT DUTINE BANSOL THN, MM. MD IM, A.B.KKK) GINT 2 SUBROUTINE BANSOL THN, MM. MD IM, A.B.KKK) BANS 1 C SUBROUTINE TO INTERPOLATE A FUNCTION AT FOUAL INTERVALS IF X CINDIDICINI C C C C SUBROUTINE BANSOL THN, MM. MD IM, A.B.KKK) BANS 1 C SUBROUTINE TO INTERPOLATE A FUNCTION AT FOUAL INTERVALS IF X CINDIDICINI C C C C C C C C C C C C C C C C C C C C C C C				
SUBROUTINE FANSOL THN.44.40 HA.A.F.KK) SUBROUTINE CINTERTX,Y.444X,DX.A) SUBROUTINE FANSOL THN.44.40 HA.F.KK) SUBROUTINE TO INTERPOLATE A FUNCTION AT FOLAL INTERVALS OF CINTER CONTERT A FUNCTION AT FOLAL INTERVALS OF CONTINUENT CONTERT A FUNCTION AT FOLAL INTERVALS OF CINTER CONTERT A FUNCTION AT FOLAL AT FOLAL INTERVALS OF CINTER CONTERT A FUNCTION AT FOLAL AT FOLAL AT FOLAL INTERVALS OF CINTER CONTERT A FUNCTION AT FOLAL AT				
$ \begin{array}{c} SUBROUTINE BANSOL THN, WW, WD IH, A, B, KKK) \\ \text{SUBROUTINE BANSOL THN, WW, WD IH, A, B, KKK) \\ \text{SUBROUTINE BANSOL THN, WW, WD IH, A, B, KKK) \\ \text{SUBROUTINE BANSOL THN, WW, WD IH, A, B, KKK) \\ \text{SUBROUTINE BANSOL THN, WW, WD IH, A, B, KKK) \\ \text{SUBROUTINE BANSOL THN, WW, WD IH, A, B, KKK) \\ \text{SUBROUTINE BANSOL THN, WW, WD IH, A, B, KKK) \\ \text{SUBROUTINE BANSOL THN, WW, WD IH, A, B, KKK) \\ \text{SUBROUTINE BANSOL THN, WW, WD IH, A, B, KKK) \\ \text{SUBROUTINE BANSOL THN, WW, WD IH, A, B, KKK) \\ \text{SUBROUTINE BANSOL THN, WW IH, A, B, KKK) \\ \text{SUBROUTINE BANSOL THN, WW, WD IH, A, B, KKK) \\ \text{SUBROUTINE BANSOL THN, WW, WD IH, A, B, KKK) \\ \text{SUBROUTINE BANSOL THN, WW, WD IH, A, B, KKK) \\ \text{SUBROUTINE BANSOL THN, WD IH, A, B, KKK) \\ \text{SUBROUTINE BANSOL THN, WD IH, A, B, KKK) \\ \text{SUBROUTINE BANSOL THN, WD IH, A, B, KKK) \\ \text{SUBROUTINE BANSOL THN, WD IH, A, B, KKK) \\ \text{SUBROUTINE BANSOL THN, WD IH, A, B, KKK) \\ \text{SUBROUTINE BANSOL THN, WD IH, A, B, KKK) \\ \text{SUBROUTINE BANSOL THN, WD IH, A, K, CTOR II \\ \text{SUBROUTINE BANSOL THN, WD IH, A, K, CTOR II \\ \text{SUBROUTINE BANSOL THNE SUBROUTINE BANSOL THNE BANSOL THAN BANSOL MA THAN FLEWENTS) \\ \text{SUBROUTINE BANSOL THNE SUBROUTINE BANSOL THNE BANSOL MA THAN FLEWENTS) \\ \text{SUBROUTINE BANSOL THNE SUBROUTINE BANSOL MA THAN BANSOL MA THAN FLEWENTS) \\ \text{SUBROUTINE BANSOL THNE SUBROUTINE BANSOL MA THAN BANSOL MA THAN FLEWENTS) \\ \text{SUBROUTINE BANSOL MA THAN BANSOL MA THAN BANSOL MA THAN BANSOL MA THAN THAN BANSOL MA THAN FLEWENTS) \\ \text{SUBROUTINE BANSOL MA THAN BANSOL MA THAN BANSOL MA THAN BANSOL MA THAN FLEWENTS) \\ \text{SUBROUTINE BANSOL MA THAN FLEWENTS) \\ \text{SUBROUTINE BANSOL MA THAN BANSOL MA THAN BANSOL MA THAN BANSOL MA THAN THAN BANSOL MA THAN THAN THAN THAN THAN BANSOL MA THAN BANSOL MA THAN BANSOL MA THAN THAN THAN THAN THAN THAN THAN BANSOL MA THAN BANSOL MA THAN THAN THAN THAN THAN THAN THAN THA$	HUNS	SUBPOUTINE CINTER(X,Y,MMAX,DX,A)	CINT 1	
GUARDITINE BANSOL FUN, WW.MDTN, A.A.A.A.A.A.A.A.A.A.A.A.A.A.A.A.A.A.A.		C CONTRACTOR OF A CONTRACTOR AT COUNT IN COURSE OF A		
$ \begin{array}{cccccccccccccccccccccccccccccccccccc$	C SURDALITINE RANCOR FAM. MM. MM. MM. MM. MM. MM. C SURDALI		CINF 4	
C Intervention intervention intervention and the state with the main of the main and the definition of the main of the main and the state of the st		$\eta ANS = 2$ $r = x \cdot y = DFFINES THE GIVEN FUNCTION$	CINT 5	
C IN COPE LINEAR EQUATION SQLVER FOR SYMMETPIC RAND WATPICES DANS A C DX = X SPACING BETWEEN GENERATED OPDINATES C CINT 7 C KKK = 0 TRIANGQLARTIZE SHAD MATRIX A C A = AURAY OF OPDINATES AT EQUAL SPACING 1 YMAX ELEMENTS) C LNT 9 C KKK = 1 REDUCES NUD BICKSUMSTITUTES VECTOR 01 ANNS 5 C A = AURAY OF OPDINATES AT EQUAL SPACING 1 YMAX ELEMENTS) C LNT 9 C KKK = 2 RECKSUMSTITUTES VECTOR 03 ANNS AT N=1 A ALIDEX CONTRACTIVITES VECTOR 04 CONTRACTION CONTRACTION AT CONTRACTION CONTRACTOR 04 CONTRACTION CONTRACTOR CONTRACTION CONTRACTING CONTRACTION CO	C 944494894949494949484848448484888888888	CONTRACT OF DEPINATES TO BE GENERATED AT EQUAL INTERVALS	CINT 6	
C KKK = 0 FRIANGULARIZES HAMA MATRIX A RAVIS 5 C A = AURAY OF OPDINATES AT EQUAL SPACING 1 VMAX FLEMENTS) CINT B C KKK = 1 REUNCES AND BACKSUNSTITURES VECTOR 10 TANS 6 D HENSION X(1), V(1), A(1) C KKK = 2 RACKSUNSTITURES VECTOR 8 TANS 7 N=1 C KKK = 2 R	C IN CORE LINEAR EQUATION SOLVER FOR SYMMETPIC RAWD MATPICES DANS A C DX	RANS A C DX = X SPACING BETWEEN GENERATED DIVINATES	C 141 7	
C KKK = 1 RENGCES AND BACKSHISTIUFES VETOR 10 RANS 6 DIMENSION X(1), V(1), A(1) V(1), A(C KKK ≈ 0 TRTANGULARTZES HAND MATRIX A RAVE 5 C A	RAVS 5 C A * ARRAY OF OPDINATES AT EQUAL SPACING 1 44AX ELEMENTS)	C INT B	
C KKK = 2 MACKSUNSTITUTES VECTUR B 94NS 7 N=1 CINT 10 C ++++++++++++++++++++++++++++++++++++	C KKK = 1 REINICES AND BACKSUNSTITUTES VECTOR 0	HANS & DIMENSION X(1), Y(1), A(1)	CINT 9	
C ************************************	C KKK = 2 RACKSUNSTITUTES VECTOR 8 N=1	1=Z . 1 SNV5	CINT 10	
			CINT 11	

- 166 -
| DO 300 MF2.MMAX
EM=N
00 TO 300
301 N=N1
101 N=N1
102 A(2*M-1)=A(1)-(FM-1.0)*NX] 301.302.303
102 A(2*M-1)=X(1)-(FM-1.0)
102 A(2*M-1)=Y(1)-(X(1)-1)(X(1)-1)(1)
103 S(1)PE=(Y(M)-Y(1)-1)(X(1)-1)(1)(1)
103 S(1)PE=(Y(M)-Y(1)-1)(1)(1)(1)(1)(1)(1)(1)(1)(1)(1)(1)(1)(1 | CINT CINT | | | 0.000 |
|---|-----------|----------|---|--------|
| КИЗМ
КИ ТО 304
304 М-МН - ККИУ-КИЛ-КИ-Г.О)ФЛХ 301.302.303
304 П - ККИУ-ХИЛ-Г.С.О)ФЛХ 301.302.303
302 АГ264-П, - КИУ-Г.С.С.И-Г.) - ККИ-Г.
303 SLUPE=[Y(M)-Y[N-F]]/(X(N)-2.4K1)
ADDX=[F4-1.0795N-C]/-K(N-F]) - КС]
4.10-4.10-4.0755N-C] - КС]
4.10-4.10-4.0755N-C] - КС]
4.10-4.10-4.0755N-C]
4.10-4.10-4.0755N-C]
4.10-4.10-4.0755N-C]
4.10-4.10-4.0755N-C]
4.10-4.10-4.0755N-C]
4.10-4.10-4.0755N-C]
4.10-4.10-4.0755N-C]
4.10-4.10-4.0755N-C]
4.10-4.10-4.0755N-C]
4.10-4.10-4.0755N-C]
4.10-4.10-4.0755N-C]
4.10-4.10-4.0755N-C]
4.10-4.10-4.0755N-C]
4.10-4.10-4.0755N-C]
4.10-4.10-4.0755N-C]
4.10-4.10-4.10-755N-C]
4.10-4.10-4.10-755N-C]
4.10-4.10-755N-C]
4.10-4.10-755N-C]
4.10-4.10-755N-C]
4.10-4.10-755N-C]
4.10-4.10-755N-C]
4.10-4.10-755N-C]
4.10-4.10-755N-C]
4.10-4.10-755N-C]
4.10-755N-C]
4.10-755N-C]
4.10-755N-C]
4.10-755N-C]
4.10-755N-C]
4.10-755N-C]
4.10-755N-C]
4.10-755N-C]
4.10-755N-C]
4.10-755N-C]
4.10-755N-C]
4.10-755N-C]
4.10-755N-C]
4.10-755N-C]
4.10-755N-C]
4.10-755N-C]
4.10-755N-C]
4.10-755N-C]
4.10-755N-C]
4.10-755N-C]
4.10-755N-C]
4.10-755N-C]
4.10-755N-C]
4.10-755N-C]
4.10-755N-C]
4.10-755N-C]
4.10-755N-C]
4.10-755N-C]
4.10-755N-C]
4.10-755N-C]
4.10-755N-C]
4.10-755N-C]
4.10-755N-C]
4.10-755N-C]
4.10-755N-C]
4.10-755N-C]
4.10-755N-C]
4.10-755N-C]
4.10-755N-C]
4.10-755N-C]
4.10-755N-C]
4.10-755N-C]
4.10-755N-C]
4.10-755N-C]
4.10-755N-C]
4.10-755N-C]
4.10-755N-C]
4.10-755N-C]
4.10-755N-C]
4.10-755N-C]
4.10-755N-C]
4.10-755N-C]
4.10-755N-C]
4.10-755N-C]
4.10-755N-C]
4.10-755N-C]
4.10-755N-C]
4.10-755N-C]
4.10-755N-C]
4.10-755N-C]
4.10-755N-C]
4.10-755N-C]
4.10-755N-C]
4.10-755N-C]
4.10-755N-C]
4.10-755N-C]
4.10-755N-C]
4.10-755N-C]
4.10-755N-C]
4.10-755N-C]
4.10-755N-C]
4.10-755N-C]
4.10-755N-C]
4.10-755N-C]
4.10-755N-C]
4.10-755N-C]
4.10-755N-C]
4.10-755N-C]
4.10-755N-C]
4.10-755N-C]
4.10-755N-C]
4.10-755N-C]
4.10-755N-C]
4.10-755N-C]
4.10-755N-C]
4.10-755N-C]
4.10-755N-C]
4.10-755N-C]
4.10-75 | CINT 1 | | | |
| 00 10 10 20 20 20 20 20 20 20 20 20 20 20 20 20 | | | | CSEO |
| 02. N2M IF (KN)-X(1)-(FW-(.0)*0X) 301.302.303
302 A(2**-1)-Y(N)-(| CINT | | 1F (N) 700.550 | CSFO |
| 302 47 (2414)24(1) (2414)40 (2414)
302 47 (241-1) = Y(N)
303 5L (10E=1Y (N) - Y (N-1) 1 / Y (N) - X (N-1) 1
ADDX=[FN-1 0 + 0 + 1 / Y (N-1) - X (N-1) 1
ADDX=[FN-1 + Y (N-2) + X (N-2) - X (N-1) 1
ADDX=[FN-1 + Y (N-2) + X (N-2) - X (N-1) 1
ADDX=[FN-1 + Y (N-2) + X (N-2) - X (N-1) 1
ADDX=[FN-1 + Y (N-2) + X (N-2) - X (N-2 | CINI D | | | CSFQ |
| СС 717.1.1.1.1.1.1.1.1.1.1.1.1.1.1.1.1.1.1 | CINT 1 | | 350 D0 600 L=1.LL | CSFO |
| 303 SLORE //// // И/-У И/-У 91// К(N)-К (N-8)
Аррузс[см-10/04/2/ с(XN-8) -/41)
А солосстветель // манолов спре | CINT BE | _ | N0 00 00 00 00 | CSFO |
| ADDX=[FH-1,0]+DX-4X4N-8]-X41]
44344-14-444-054ADDX443ADDF | CINT 19 | _ | 600 BIN.L)=BIN.L)-AIN. J)*R[J.L] | CSFO |
| A 1 24M-1 1 - V (M-D 1 4 4 00 X 4 7 00 F | CINT 20 | _ | | 0.55.0 |
| | CINT 2 | | 60 10 500 | CSFO |
| 300 CONTINUE | | | | |
| PETURN | CINI 2 | | 700 RETURN | CSFO |
| | | | | CSEO |
| | | | | |
| | | | | |
| | | | | |
| | | | | |
| SUBROUTINE CSYMFO(A.B.MN.LL) | CSF0 1 | | SLIBBOUT ING STARM - NEAC, SWASS, FV. AKED, TMASF. 4. W1. W2. W3. W4. (C. | Nula |
| AL ANDER BE NO F AL HONGL | CSE0 3 | | escale.web.web.web.web.web.web. | E 1GN |
| | C 5E 0 A | | | N51 4 |
| DU 4 15 MER ONN | CSEG . | | | н 10 м |
| | CSE0 A | | | NO LL |
| | | | | |
| FORK DIN.L. | | | C GUARDIE AUDITORIA NO AND | E I GN |
| | CSE0 10 | _ | | *EIGN |
| 50 81N.L.)=8(N.L.)< | CSEO 11 | | | E I GN |
| | C 5F.0 12 | <u>.</u> | DIMENSION NEEC(1), SMASS(1), A(NEO, MBAN), EV(1), W1(1), W2(1), W3(1), | 201 |
| CHECK FOR LAST FOUATION | CSF0 12 | , | WALP .CCINEV INASELL) | |
| | | | COMPLEX AREONDSANDSANDSANDSANDSANDSANDSANDSANDSANDSA | NE LGN |
| 002°000°000°100-41 | C550 16 | • - | CUMBUR / CNINCLY NORMAGNEDATION CLEANCE AND | NG 1 4 |
| 00 DG A50 \$=N\$.NN | C 5E0 11 | | • • × JUNK (12) | NCIS |
| | CSF0 15 | _ | COMMON/LABEL/HED(D) | EIGN |
| | CSE0 19 | _ | | |
| | CSED 20 | _ | TF (TRIGANE.0) GUTU 125
Destand an | L I GN |
| TE ECADERATER CITY COCORDON CO | 0.560 | | | といい |
| | CSE0 23 | _ | READ 1901 | ERGN |
| MDD1FY A(1°.) | CSF0 24 | _ | READ (90) AKFO | NULL |
| | CSE0 24 | | PEWIND 40 | E16N |
| DG 300 1×1.00 | CSF0 26 | _ | 00 110 [=1°445 | NG L |
| A E 8 . J) = A (8 . J) + A E 8 . J + A E N . J) | C SFO 2 | | 50 110 J= 1. NA2 | 201 |
| 300 A(J.1)=A(1.J) | CSF0 2P | | AXF0(f.,J)=AXF0(f.,J)=ESCALE | |
| - | | | | NUL |
| MODIL A HCE OF D | | | tor is from we at 60 to 300 | 200 a |
| | C SF 0 32 | | PED IT ALTRANET OF A CONTRACTOR | NC 4 L |
| 00 H(0.8 P=B(P.1 P-A(J.N)#B(N.L) | CSF0 3 | | BACKSPACE | ELGN |
| | CSF0 34 | _ | REAT | Nelis |
| 155 CONTINUE | CSF0 JF | | SF SIGPAV.NE.GP SEAD (1) | ELGN |
| | CSEO 36 | _ | te tipic.Ne.Of GT TO 135 | E 8 GN |
| HACK-SUBSTETUTICN | 14 D 35 J | | | N50 ユ |

167 can

``

٥. m ic ¢ E I GN E 1 GN FORM FORM FORM F 0UN F 0UN F 0UN F 0UN EIGN E I GN SUDROUTINE FOUNDLAKFI,4KF2,4KF5,1EASE,4,0UMMY,5,W,0WS,ESCALF,NTEMP,FOUN IRIX)-MATRIX A) 2008 FORMAT (4944) VERIFICATION OF ORTHOGONALITY RELATION-AT+M4A=1) 2009 FORMAT (1112,F15.8) SUMADUTINE TO COMPUTE FOUNDATION STIFFNESS CONTATINITION TO Coefficient Matuix SUBHOUTTNE FORMKLAKF.ETG.IBASE.OUMMY.S.NEV.ND2.NEOL COMPLEX AKF(ND2.ND21,OUMMYLMD2).SLNEV.NEV) DIMENSTON ETG(NEO.1).18ASE(1) TRANSFORM KFIWT TO MODAL COORDINATES 110 DUMMY(J)=DUMMY(J)+EIG(L, 1)*AKF(K,J) L=[AASE(J) 120 5(1,K)=5(1,K)+DUMMY(J)+E1G(L,K) 130 5(K,t)=5(1,K) 1000 FORMAT (1615) 1010 FORMAT (122,F15,0) 1020 FORMAT (112,2E15,6) 2020 FORMAT (1112,2E15,6) 2006 FORMAT (141) DUMMY(J)=0.0 DO 110 K=1.NB2 L=EBASE(K) 00 130 1=1 NEV 00 110 J=1 NH2 00 130 K=1.NEV 00 120 J=1.NR2 J=1 , NH2 NA2, NEV. NEQ 1 HETURN QN 3 END U ç 0000 53 59 83 84 85 86 -----5 • ۲. ۲ ê 81 5 ê E 16N 516N 516N FIGN E 1 GN E 1 GN EIGN E 16N E 15N E 16N E 16N EIGN E 1 GN E I GN E 1 GN F 1 GN EIGN N513 EIGN A LON ELGN EIGN E 1 GN EIGN ELGN E I GN ELGN F 1GN E I GN E 1 GN E 1 GN E 1 GN E 1GN N513 F 1 GN ELGN € 1 GN EIGN F IGN E16N E I GH E16N E I GN FLGN ELGN F16N E I GN 1 1 GN F 16N CALL BANEIG(NEQ,MDAND,NEV,NBC,EV,NEBC,SMASS,A,W1,A,W2,W3,W4,TRACE)E1GN F LGN DD 110 [=1,MP2 [=10.86(1) DD 130 J=1,MP2 K=1MASE(J)_4.41 K=1MASE(J)_4.41 K=1KALE-01.0TL(K,GT,MBAND1) GD TO 130 ALL,KJAALL,KJ)HEAL(AKFO(1,J)) CONTHUE OUTPUT ETGENVALUES AND ETGENVECTORS PUNCH FREDUFNCIES AND MODESHAPES READ (5.1020) M.ARL, M. ARL+1.N) WRITE (6.2004)HED READ FREQUENCIES AND MODESHAPES SUM=SUMEA(K, 1)#SMASS(K)#A(K, J) CC(1, J)=SUM READ (5.1010) (FU(N).N=1.NEV) PUNCH 1020.M.ALL.N.ALL.41.N) Return DD 140 1=1.MEV EVIJ=SORTEV(1)) 0 MRITE (6.2005) 1.EV(1) WHITE (6.2007) CALL MPHINT (A.NEQ.NEV) CALL MPHINT (CC, NEV , NEV) IF (INPP.E0.0) AFYUGN DD 200 N=1.NEV 200 PUNCH 2009NN FVIN) D1 210 N=1.NEV D1 210 N=1.NUNNP DO 240 [=1.NEV Waite (5.2005) 7.EV(1) Waite (6.2007) CALL MPRINT CA.NED.MF VI IF (11) 190.190.170 WRITE (6, 2006)HED 00 230 N=1,NU 00 230 N=1,NUMP 00 190 L=1.11 CC(1.1)=CC(1.1) WAITE (6.2008) UN 190 1∓1.NEV DO 160 J=1.NEV DO 150 K=1,NEQ 5UM=0.0 CONTINUE L-M+M=] -1=11 210 220 130 135 140 150 160 071 061 2 30 240 J ບບບ υu

υv

	。 DIFF DIMENS (DN 18ASE(1),A(NEO,1) DIMENS(DN HED(8)			5			
	DIMENSION PRASETS, ALMER, IV DIMENSION HEDLAB					5767	,
	DIMENSION HED (8)		N 00 -	-			2
			NUOR	0	5. 2 + 2 = 2 = 4 = 2 = 2	SHI	5
			F CUN	=	wri ye (6 • 2000) 14ED •	JBLG	-
	5MQ/M# #1		F OUN	2	**************************************	J .01.6	-
	N# #14		FOUN	m	RETURN	JALG	*
NUMBER NUMBER<			= GUN	U a		JALG	-
F CONTRACT CO	NINC #NW~NTEMP		FOUN	2	000 FORMAT (171.	งายก	-
			F GUN	5	· 20% 8410//	JALG	-
	1F (NINC, E0,0) GO TO 160		FOUN	•	25X.31(1H+)/	3766	Ξ
1 7000 10 2 7000 10 2 2000 10 2000 10 10000 10000	IF (NINC.LE.2) GO TO 120	•	FOUN	, E	- 26x,114,0 8 , 0 6 /	JBLG	-
0 0	NT INC -2		E CUIN	0		3 14 0	ň
No. No. <td></td> <td></td> <td>NINA</td> <td></td> <td>THE ADDRESS OF ANY TRUCK ON THE SUPERSTRUCT OF S</td> <td>0 101 //</td> <td>•</td>			NINA		THE ADDRESS OF ANY TRUCK ON THE SUPERSTRUCT OF S	0 101 //	•
No. No. <td></td> <td></td> <td></td> <td></td> <td>A CALMER AND AND COMMON COMMON STRUCTURE AND CALCULATE AND CALMER TO A STRUCTURE AND CALMER AND CAL</td> <td></td> <td>2 0</td>					A CALMER AND AND COMMON COMMON STRUCTURE AND CALCULATE AND CALMER TO A STRUCTURE AND CALMER AND CAL		2 0
NUMBER NUMBER<				e F			(; e
F 000000000000000000000000000000000000	READ (90)				. IOX, 44 HFURM STIFFNESS AND SULVE FUP STATIC LOADS ". FB.	79166	N
F FUNCTION F FUNCTION <td>IF (NINC.LE.I) BACKSPACE 90</td> <td></td> <td></td> <td></td> <td>. 10x.44HCCMPUTE FREQUENCIES AND MODE SHAPES 4.FR.0</td> <td>9 H C</td> <td>N I</td>	IF (NINC.LE.I) BACKSPACE 90				. 10x.44HCCMPUTE FREQUENCIES AND MODE SHAPES 4.FR.0	9 H C	N I
IF FUNC.LEARLY BACKSPACE 90 IF FUNC.FUNC.FUNC.FUNC.FUNC.FUNC.FUNC.FUNC.	IF (NINC.LE.I) RACKSPACE 90		FOUN	2	■ IOX, 44HEARTHOUAKE RESPONSE	e Ter s	ς.
REGOUND FERONO	JF (NINC.LE.1) BACKSPACE 90		FOUN	ç	tox.44HT0TAL EXECUTION TIME *.FR.3	J.91.G	~
# 60100 # 60100 # 60100 # 60100 # 60100 # 60100 # 60100 # 60100 # 60100 # 60100 # 600000 # 600000 # 60100 # 600000 # 600000 # 6000000 # 60100 # 6000000 # 6000000 # 6000000 # 60100 # 6000000 # 60000000 # 6000000000000000000000000000000000000	5 40 (od)		F GUN		END	JULG	ŝ
Control Control <t< td=""><td>READ1 90)</td><td></td><td>FOUN</td><td>c</td><td></td><td></td><td></td></t<>	READ1 90)		FOUN	c			
FEO1001 FEO100101 FEO100101 FEO100101<	READ (90) AKFI		E CUN	¢			
FENDIGI FOUNTINE	PE AD (90)		FOUN	0			
RE110101 ME2 COURD 110 COURD 114 COURD 114 <td>READ1 901</td> <td></td> <td>FOUN</td> <td>=</td> <td></td> <td></td> <td></td>	READ1 901		FOUN	=			
00 10 11 30000114 SOM00114	READ (90) AKF?		FOUN	N			
00 10	00 120 1 = 1 ° M92		NOC #				
AFTIL:JUNCFETI:JU			N004	• 1	SURPOUT NE KONPLEXTR. 2. I JOC 0 STASS. EV 2 AKTO 0 BUASS. 2 AKT 0 2 BUASS. 2 AKT 0		
AFESTIJJANEZALE F000 30 -EILELISTICTESATERITIA COM 100 110 111,000 DITFARETIJJANEZALE F000 30 -EILELISTICTESATERIADOA VUOLTERGO ALMADA VUO	AKFI (1 • J) = AKFI = [• J] = F5CALE			5 1	•DUMMY. WV. S. YEMP . SFU. A 2. 0 .		
00170 J11.000 0170 J11.000 00170 J11.000 </td <td>AKF2([,J)=AKF2([,J)#F5CALE</td> <td></td> <td></td> <td>•</td> <td>• EI • EI • EI2• CTEMPI • CTEMPZ•LAMBDA• VIN F• TERMQ • ALPHA• SGM• NN•NFV• NEV</td> <td>ather a second</td> <td></td>	AKF2([,J)=AKF2([,J)#F5CALE			•	• EI • EI • EI2• CTEMPI • CTEMPZ•LAMBDA• VIN F• TERMQ • ALPHA• SGM• NN•NFV• NEV	ather a second	
01 TO JI JI JI VECTI JI ACTI JI JI VECTI JI ACTI JI VECTI JI ACTI JI JI JI VECTI JI ACTI JI JI VECTI JI JI VECTI JI ACTI JI JI VECTI JI ACTI JI JI VECTI JI JI VECTI JI ACTI JI JI JI JI VECTI JI JI VECTI JI JI VECTI JI JI JI VECTI JI JI JI VECTI JI JI VECTI JI JI JI VECTI JI JI VECTI JI JI VECTI JI JI VECTI JI JI JI JI VECTI JI	00 1 70 1=1,ND2		100M	-	• NEQ.NTERM.NBU.DMS.E SCALE .NB 2)		
Diff=JunkT(1,J) Comute Contract Annual MinkL and C	24W 1 1 0 1 1 1 0 1 0 0						
ACTIT.JITARTI.J	D [FF= AKF241, J - AKF241, J]		FOUN	0	\$\$\$\$\$\$		
ACTULATION FOUN 42 COMMON 42 COMMON 42 COMMON 44000, 1000 (AKF (ں ع	COMPLIE COMPLEX FREQUENCY RESPONSES		
COMMULE FOUNATION STIFFNESS CONTRINUTION TO FINA 32 COMMULE FOUNATION STIFFNESS CONTRINUTION TO FINA 33 COMMULE FOUNATION STIFFNESS CONTRINUTION TO FINA 34 COMMULE FOUNATION STIFFNESS FOUNATION STORE COMMULE FOUNATION STIFFNESS FOUNATION STORE COMMULE FOUNATION STORE AND		•	-		**************************************		
Constructor			NIG L			d and a set	
THE TERMS UP THE CLEPTICLENT WARKET	COMPUTE FURMATION STIFFMESS CONTRATION TO A				COMMON / CNINCL/ NAMP, MIGAN, NUMEL, NHIC, SERAV, NPP		-
CALL FGRIMK(AKF.A.11A3E, DUMWY, 9, NEV, NB2, NEO) FOUN > , 4XUWK(12) , 4XUWK(12) xxum xxum RETURN FEURN FOUN 7 , YUWK(14) xxum xxum RETURN FEURN FOUN 7 , YUWK(14) xxum xxum RETURN FOUN 7 , YUWK(14) 2 , YUWK(14) xxum RETURN FOUN 7 0 10				• •			• •
CAIL FORMKIAKEAALINASE, DUMAY, SARE VIRUS, AND			L OOK	c 1	• • XJUME 1 2)	K Child	
RF URN FUNN - VJUWKIA - VJUWKIA - VJUWKIA - VJUWKIA END END - CUMMVIA - VJUWKIA - VJUWKIA - VJUWKIA - VJUWKIA END END - CUMMVIA - CUMMVIA - VJUWKIA - VJUWA - VJUWKIA - VJUWKIA - VJUWKIA - VJUWKIA - VJUWKIA - VJUWA	CALL FORMK (AKF.A. (BASE, DUMMY, S.MEV, NB2, NEO)			¢ I	COMMON /CNTRLZ/ NDATA,NWR,DT,NREST,NDUF,NRLOCK,TWV,NCOMP		
RETURN FILM 40 COMMON/LAEEL/HEORA) KOUN KOUN KOUN END TOUN 49 C DIMENSION KOUN KOUN KOUN SUBROUTINE_JORI.OGTMSTOP.WECOM.II.TZ.II.TZ.II.GEMPOLID.SERVEL/HEORA JAE 1 JAE KOUN KOUN SUBROUTINE_JORI.OGTMSTOP.WECOM.II.TZ.II.TZ JAE JAE JAE KOUN KOUN SUBROUTINE_JORI.OGTMSTOP.WECOM.II.TZ.II.TZ JAE JAE JAE KOUN KOUN SUBROUTINE_JORI.OGTMSTOP.WECOM.II.TZ.II.TZ JAE JAE KOUN KOUN KOUN SUBROUTINE_JORI.OGTWSTOP.WECOM.II.TZ JAE JAE JAE KOUN KOUN SUBROUTINE_JORI.OGTWSTOP.WECOM.II.TZ JAE JAE KOUN KOUN KOUN BLANC.COM.INE_JORI.OFTWSTOP.WECOM.INCOM.I			FOUN	•	• • ×1 UNK 4 5		1
END 014643104 8113,2113,1202113,684.5113,61462,11 67040 8040 SUBROUTINE_JOHLOG(48708,44504,11,52,13,17) 01464113,51640111,51640111,518404,11,4470451,5447615 8040 8040 SUBROUTINE_JOHLOG(48708,44504,11,52,13,77) 0145 01464113,51640111,518404,11,4674051,447161,5 8040 8040 SUBROUTINE_JOHLOG(48708,44504,11,512,13,77) JALG JALG 2 0040EX 447402,113,518404,11,467402,113,44672403,4071,0404 8040 BLANK COMMON STORAME AND E FECUTION TIME JALG 3 A5501412,11,4464402,113,44672402,4071,0424,113,44672402,4071,0404 8040 BLANK COMMON STORAME AND E FECUTION TIME JALG 3 A5501412,11,4464402,113,4467402,113,44724042,4071,0444113,447402,4071,0444113,4474402,4071,0444113,4474402,4071,0444113,4474402,4071,0444113,44744041,044444444444444444444444444444	RETURN		MEL		COMMON/F A BEL/HED(8)		
Buildouti ME_JONLOGIMSTOP «MICOM.T1., F2. VIJ. «TATASTUT. FACTIL SERVICE VILLENT CONTRACT AND CONTRACT A			NOD	0			
TURN TERMINING(MSTOR/WFCOM/11,TZ,T3,T3,T4) JPL TERMINING(MSTOR/WFCOM,11,TZ,T3,T4) JPL T COMPLEX NY (MEV.11), GENOL(11, GENOL(BIMENS 10M A (1), 2(1), 170C(1), 2MA SS(1), EV(1), A(NE 0, NE V).		
COMPACTING COMPACTING <td></td> <td></td> <td></td> <td></td> <td>+TEMP(1) + A2(1 % O(NEV *) + EI(NEV *) + EI(NEV *) +</td> <td></td> <td>•</td>					+TEMP(1) + A2(1 % O(NEV *) + EI(NEV *) + EI(NEV *) +		•
SUBPOUTINE JORI.DGTWSTOR_WARCOM_TI_ST2_ST3_TA JALG COMPLEX NYTHEV_TI_STNET_STNET		<i></i>			.CTEMP1(1).CTEMP2(1).TEPH0(1).EI2(NEV.NEV.1).HACC(6.2).FMT(6).		* 1
SUBROUTINE JONLOGIMSTOPANFCOMATLATATATATATATATATATATATATATATATATATAT					。 IBASE (1)		N (
SUBBUTINE JORI DGIMSTOR-WHCOM-TI-FZ-0T3.TA) JALG 1 AKFOLUNTINE-KITASTATE ALLANGTANGTANGTANGTANGTANGTANGTANGTANGTANGT				U	VIE 11 HONDER VERSEN VERSEN VERSEN VERSEN VERSEN VERSEN		~ 6
4 2 5F01(WEV.1), HW1 KOUP 4 2 5F01(WEV.1), HW1 KOUP 4 4 2 5F01(WEV.1), HW1 4 4 4 4 4 4 4 4 4 4 4 4 4 4 4 4 4 4 4 4 4 4 4 4 4 4 4 4 4 4 4 4 4 4 4 4 4 4 4 4 4 4 4 4 4 4 4 4 4 4 4 4 4 4 4 4 4 4 4 4 4 4 4 4 4 4 4 4 4 4 4 4 4 4 4 4 4 4 4 4	SUBDOUTINE ROPIOCIMSTOP NACON . 11 . 12 . 11 . 14)		9 6 7		CUMPLEX ANTINEVELTANNEVELTATEMAGETURETAND ALTATAT Avenaums-ti-averand-ti-aveltand-modians/averatand-angle.commy(1),	K OWP	< n
 COMON/LAEL/HED/R) 			JALG		SFOLMEVID AND AND AND AND AND AND AND AND AND AN	K OWD	~
BLANK COMMIN STORAGE AND EXECUTION TIME JALG 4 REAL LAMIDATI) ALLENDING STORAGE AND EXECUTION TIME STIFFUESS PROPORTIONALITY CONSTANTS KOUP 2 ALLENDING AND PRINT MASS AND STIFFUESS PROPORTIONALITY CONSTANTS KOUP 2 COMMON/LABEL/PEDTAT COMMON/LABEL/PEDTAT	*************************	*************	8787+0	U E		K GM2	ĉ
KONP 2 COMMONLAEL/MEDIA) COMMONLAEL/MEDIA) COMMONLAEL/MEDIA) COMMONLAEL/MEDIA) COMMONLAEL/MEDIA) COMMONLAEL/MEDIA)	BLANK CI3MMON STORAGE AND EXFCUTION TIME		J RL 6	•	REAL LAMBDA(1)	K D V D	ŧ.
COMMONLAER.PEDIA) JALG 7 C READ AND PRINT MASS AND STIFFNESS PROPORTIONALITY CONSTANTS KOUP 2 COMMONLAER.PEDIA) XOUP 2 KOUP 2	******************************	************	9.181.44	U S		dwD X	è.
COMMON/LAEEL/HEOVA) COMMON/LAEEL/HEOVA	•		18,6	U v	READ AND PRINT MASS AND STIFFNESS PROPORTIONALITY CONSTANTS	K ONP	٠,
	COMMON/L ABEL/ HEDIA }		JBLG	v	FAR DAMPING WATRIX	K OND	Ň

- 169 -

non bulk f f F F F F non bulk f F F F F F non bulk f F F F F F F F F F F F F F F F F F F		40MP	90	130 SI=CLEMPL(N)	KON
ment of a manufactor ment of a manufactor ment of a manufactor ment of a manufactor ment of a manufactor ment of a manufactor ment of a manufactor ment of a manufactor ment of a manufactor ment of a manufactor ment of a manufactor ment of a manufactor ment of a manufactor ment of a manufactor ment of a manufactor ment of a manufactor ment of a manufactor ment of a manufactor ment of a manufactor ment of a manufactor ment of a manufactor ment of a manufactor ment of a manufactor ment of a manufactor ment of a manufactor ment of a manufactor ment of a manufactor ment of a manufactor ment of a manufactor ment of a manufactor ment of a manufactor ment of a manufactor ment of a manufactor ment of a manufactor ment of a manufactor ment of a manufactor ment of a manufactor ment of a manufactor ment of a manufactor ment of a manufactor ment of a manufactor ment of a manufactor ment of a manufactor ment of a manufactor ment of a manufactor ment of a manufactor ment of a manufactor ment of a manufactor					
	READ AND WILLE & CM ELA	KOMP		140 AM GSLAMBDAIN FFLCTJF-CILTFF ra-rintage	NO X
entrementant control control 0.000 0.000 0.000 0.000 0.000 0.000 0.000 0.000 0.000 0.000 0.000 0.000 0.000 0.000 0.000 0.000		KOMP	4		WCX
NUMBER Construction Construction Construction 0.0.0.0.0.0.0.0.0.0.0.0.0.0.0.0.0.0.0.		KOMP	35		KON
$ \begin{array}{cccccccccccccccccccccccccccccccccccc$		4 DMD	96		NOX
0 0		40MP	37		K0Y
0. 50101.100.1 1. 100.000 1. 100.00001.100.0000101 1. 100.00001.0000101 1. 100.000 1. 100.000 1. 100.00000.0000 1. 100.00000.0000 1. 100.000 1. 100.000 1. 100.00000.0000 1. 100.000000000 1. 100.000 1. 100.000 1. 100.0000000000 1. 100.00000000000000000000000000000000		K CHR	38	00 150 K=1.NEV	NO X
If functional of it of additional additiona	50 SFD(1.1)=0.	KONP	¢.	TEAM=(a(11-1,K)+S1-a(JJ-1,K)+S2+(a(11-1,K)~a(JJ-1,K))+	NC 3
IF (non-resol) of (10 256 000 01 01 01 01 01 01 01 01 01 01 01 01		4 UMB	40	I C3/ (024), AMBDA(N)))/LAMBDA(N)	KON
	IF (NPP.ED.0) GO TO 250	K OMP	41	150 E1(K+1.N)=E1(K+1.N)+TERM	NOX
Construction Construction<		KOMP	~~	160 CONT INUE	KON
Constraint Constra	CC=4 720.	KOMO	E 4	170 CONTINUE	NGX
11 Section Constrained Constrained <thconstrained< th=""> <thconstrained< th=""></thconstrained<></thconstrained<>	. RHOT .0625/32.2	d NO X	<		мОм
Litute(CAP) Control Contro Control Control	CONSTRUCTOR (1 ~ + ALPHA) + SGN	KOMP	45	C RESET VALUE OF A AT FIRST FACE POINT	KOM
$ \begin{array}{cccccccccccccccccccccccccccccccccccc$	L T= L JBC (NPP)	K GMP	. 94		КОМ
$ \begin{array}{cccccc} 100 & 0.0 & 0.0 & 11111 \\ 100 & 101$	11=ML-Z1LT)	KONP	× 7	[= { J NC (1)	КQH
MODEY VALUE OF A AF FIGST FACE POIN COM CO CITUATION COM	H= 0. 5+P(/H	KOMO	48		¥Û¥
MOLEY VALUE (F. A. AT FIRST FACE POLAT COD DO 100 ECT, MEV COD DO 100 ECT, MEV COD DO DO 100 ECT, MEV COD DO DO 100 ECT, MEV DO DO <thdo< th=""> DO <thdo< th=""></thdo<></thdo<>		KOMP	49	2 (1) = A TEMP	NC X
$ \begin{array}{cccccccccccccccccccccccccccccccccccc$	NUDIFY VALUE OF A AT FIRST FACE POINT	KOMP	50	DQ 180 K=1.NEV	KOM
11.10(1) COUNT		K OMP	51	180 A(11-1,K)=7EMP(K)	HC X
j:i:0.000 j:i:0.000 <t< td=""><td>I = 1 JBC (1)</td><td>KOMP</td><td>52</td><td></td><td>N() M</td></t<>	I = 1 JBC (1)	KOMP	52		N() M
$ \begin{array}{cccccccccccccccccccccccccccccccccccc$	J= 196 (2)	K OMD	53	C SCALE I(K,N) BY SORT (2. WHO+C/H) AND LAMBDA(N) BY C	КÜМ
$ \begin{array}{cccccccccccccccccccccccccccccccccccc$		KONP	5.		K DM
00 00 200 20 00 20		KOMP	55	FAC=SQRF(2,+RH0+CC/H)	KON
$ \begin{array}{cccccccccccccccccccccccccccccccccccc$	00 100 K=1.NEV	KONP	56	DD 240 N=1,NTEAM	КŪМ
100 011-0.4 -0.0 <	1 ENP4K) = 4 (1 1 - 1 - K)	K OMP	57	L AMBDA(N)=LAMB)A(N) +CC	KOH
VIENT_2(1) VIENT_2(1) VOID 50 200 E(K,N) = E(K,	100 1 1 1 2 1 2 1 2 1 2 1 2 1 2 1 2 1 2	40.4b	58	DD 240 K*1.NEVI	KOM
Z(1)=M. Z(1)=M. COD CO COD CO COD	Y T EMP= 2 (1)	40MD	59	240 El(x,v)=El(x,v)+EAC	WU X
INTILIZE ICON 61 D0 246 HIL CON	241) 244	KOMP	60	C. C	MOM
INTILATE		K OMP	41	DO 246 N=1.NTERM	нах
IDE IDE <td>INTTALSZE ITK.NJ</td> <td>KOMP</td> <td>62</td> <td>DD 245 M=1, NEV</td> <td>KON</td>	INTTALSZE ITK.NJ	KOMP	62	DD 245 M=1, NEV	KON
00 245 COU		A DMD	63	EII (M.N)×EI (M+I.N)*EI(I.N)	WO X
ENERVEL1 ENERVEL1 ENERVEL1 ENERVEL1 ENERVEL1 ENERVEL1 ENERVEL1 LANDARL ENERVEL1 ENERVEL1 ENERVEL1 ENERVEL1 ENERVEL1 ENERVEL1 ETTLINSTELLEN ETTLINSTELLEN ETELEN ETELEN ETELEN ENERVEL1 ETTLINSTELLEN ETTLINSTELLEN ETELEN ETELEN ETELEN ETELEN ETTLINSTELLEN ETTLINSTELLEN ETELEN ETELEN ETELEN ETELEN ETTLINSTELLEN ETELEN ETELEN ETELEN ETELEN ETELEN ETELEN ETELEN ETELEN ETELEN ETELEN ETENEN ETELEN ETELEN ETEREN ETELEN ETEREN ETENEN ETELEN ETELEN ETEREN ETEREN ETEREN ETENEN ETELEN ETEREN ETEREN ETEREN ETENEN ETENEN ETELEN ETENEN ETENEN ETENEN ETENEN ETENEN ETENEN E	00 120 W=1,NTERM	KONP	44	DD 245 K#4.4EV	W() X
Lumbarth Kound of Eliti		K DMD	65	E 12 (W. K. ×13+ E 14 + 3 • V) + E 14 + 2 • V)	HOX
E1(1, h) = (-1, ++ (+-1)/L, AMED.A(M) 01 07 245 CONTINUE 245 CONTINUE 010 01 0 010 0 0 0 00	L. AMBDA(N)=EN+B	KOMP	66	E12(K.W.N)=E12(V.K.N)	NC X
00 110 K=1.NEV KOMO 64 246 CONTINUE KOM KOM 110 ETKULINI-0. KOMO 69 246 CONTINUE KOM KOM 120 ETKULINI-0. KOMO 73 250 DO 260 HIL KOM KOM 10 LITUE CONSTANTS TKINI XOM KOM	EI(I.N)=(-[.)**(N-1)/LAMEDA(N)	KOWD	67	245 CONFINUE	AOX X
IIO E 1(K 11, M)=0. KOND CO G ENERAL ISED CONSTANTS AND VECTOR O KOND KOND <t< td=""><td>00 110 K=1.NEV</td><td>K DMP</td><td>68</td><td>246 CONTINUE</td><td>NUX</td></t<>	00 110 K=1.NEV	K DMP	68	246 CONTINUE	NUX
120 CONTINUE KOMP 70 C GENERALISED CONSTANTS AND VECTOR 0 KOM CALCULATE CONSTANTS FIK.NJ KOUP 71 250 DD 260 N=1.NEV KOM 00 170 LEULATE CONSTANTS FIK.NJ X0M Y0 KOM 01 170 LEI.NN X1M1 EV X0M KOM 1=1JN5(L1) X0M Y5 SUM=0. KOM KOM 1=1JN5(L1) X0M KOM Y6 X0 KOM 1=1JN5(L1) X0M KOM Y6 X0M KOM <	110 E f (K +1 • N)=0 •	KOMP	69	C	*() X
CALCULATE CONSTANTS ITK.NI KOW TI C 00 170 L=1.NN 00 170 L=1.NN 230 00 260 H=1.NEV (M) KOW 1=1 JHC(L1) X0H 73 230 00 260 H=1.NEV (M) KOW 1=1 JHC(L1) X0H 73 SUM=0. KOW 1=1 JHC(L1) X0H 75 SUM=0. KOW 1=1 JHC(L1) X0H 75 SUM=0. KOW 1=1 JHC(L1) X0H 75 JJ=241 KOW J=1JHC(L1) J=1JHC(L1) X0H 77 JJ=241 J=1JHC(L1) JJ=241 KOW ZOW KOW J=1JHC(L1) J=1JHC(L1) JJ=241 KOW KOW J=1JHC(L1) J=1JHC(L1) J=1JHC(L1) KOW KOW J=241 KOW ZSS JHC(M) KOW KOW J=241 KOW KOW	120 CONTINUE	A CMP	2	C GENERAL ISED CONSTANTS AND VECTOR O	NCX
CALCULATE CONSTANTS F(K,N) CALCULATE CONSTANTS F(K,N) <td></td> <td>dwD X</td> <td>5</td> <td></td> <td></td>		dwD X	5		
00 170 124104 K0140	CALCULATE CONSTANTS I (K.N.)			250 00 260 M=1, NEV	
10170 L=1.NN 10170 L=1.NN 0.0040 73 5.0441=0. K044 1-10170 L=1.NN 5.0442 75 5.0442 8.042 1-1111 1-11-1 1.1-1 1.1-1 K044 1-11-1 1.1-1 1.1-1 K044 75 5.042 1-1-1 1.1-1 1.1-1 K044 76 5.042 K044 1-1-1 1.1-1 1.1-1 K044 76 5.042 K044 1-1-1 1.1-1 1.1-1 K044 76 5.042 K044 11-1-1 1.1-1 K044 76 5.042 K044 11-1 K044 76 5.042 K044 K044 10-1 K044 71 1.1-1 2.54 K044 11 K1 K044 73 2.57 K044 11 K1 K044 73 2.57 K044 11 K1 K044 73 2.54 K044 11 K1					
J=1JBC(L1) J=1JBC(L1) KUW KUW KUW J=1JBC(L+1) LI LI LI KUW J=2J+J LI LI LI KOW J=2J+J KOW Z J=JI KOW KOW J=2J+J KOW Z J=JI KOW KOW J=2J+J KOW Z J=JI KOW KOW J=2J+J KOW Z LI KOW KOW J=2J+J KOW Z SUMPSKILL KOW KOW D=2D LO N=LNEAM KOW KOW KOW KOW D0<160 N=LNEAM	00 170 L=1.NN	KON0		SUM 1 = 0.	
J=LJAC(L+1) J=LJAC(L+1) J=LJAC(L+1) J=L+1 J=L+1 D=Z(1)-Z(J) D=Z(1)-Z(J) D=Z(1)-Z(J) D=Z(1)-Z(J) D=Z(1)-Z(J) D=Z(1)-Z(J) D=Z(1)-Z(J) D=Z(1)-Z(J) D=Z(1)-Z(J) D=Z(1)-Z(J) D=Z(1)-Z(J) C (I) Z(J) D=Z(J)-Z(J) C (I) Z(J) D=Z(J)-Z(J) C (I) Z(J) C (I) Z(J)	I = I JI3C (T)	A DMP	5	5UM2#0.	MOX
J1=1-1 J1=1-1 J1=1-1 VOUN V1 J1=1-1 VOUN V1 J1=1-1 J1=1-1 KOUN KOUN KOUN KOUN KOUN D2=2(1)-2(1) D2 SUU1=SUU1AAL Unseurastill KOUN KOUN D2=2(1)-2(1) D2 SUU1=SUU1AAL Unseurastill KOUN D1 Labola D0 255 SUU2=SUU254L KOUN D1 Labola D0 255 SUU2=SUU254L KOUN D1 Labola D0 255 SUU2=SUU254L KOUN D1 Labola D1 D1 SUU2557 KOUN D1 Labola D1 SUU2557 KOUN KOUN S1=SU(ARG) S1 S1 S1 S1 KOUN S1=SU(ARG) S1 S1 S1 S1 KOUN S1=SU(ARG) S1 S1 S0 S1 S1 S1=SU(ARG) S1 S0 S1 S0 S1 S1=SU(ARG) S1 S0 S1 S0 S1 S1=SU(ARG) S1 S0 S1 S0 S1 S1=SU(ARG) S1 S1 S1 S1 S1 <td>J=1766(L+1)</td> <td>A GHE</td> <td>e</td> <td>DD 255 J=1 NUMNP</td> <td>NC Y</td>	J=1766(L+1)	A GHE	e	DD 255 J=1 NUMNP	NC Y
JJ=JFJ JJ=JFJ D2=Z(1)=Z(1)=Z(1)=Z(1)=Z(1)=Z(1)=Z(1)=Z(1)					HO Y
D2=2(1)-2(1) D2=2(1)-2(1) KOM KOM KOM D10=2(1) D10 D2 D2 D3 D3 D3 C0 D10 D10 D2 D3 D4 D4 <td< td=""><td></td><td>LINCX</td><td>52</td><td>1 () (; = .)</td><td>KON</td></td<>		LINCX	52	1 () (; = .)	KON
00 160 N=1,NFGM 0 255 51W254X423411) NIM 0 10 0 160 N=1,NFGM 0 255 51W254X423411) N (NIM 0 130 N) N (NIM 0 131 15 (L. ME. 1) 130, 171 130 N) N (NIM 0 13 257 014, 21-54,259,257 N) (NIM 0 131 15 (L. ME. 1) 160 171 130 N) N (NIM 0 13 255 014, 21-54,259,257 N) (NIM 0 131 15 (L. ME. 1) N (NIM 0 13 255 014, 21-54,259,257 N) (NIM 0 131 15 (L. ME. 1) N (NIM 0 13 255 014, 21-54,259,257 N) (NIM 0 131 15 (L. ME. 1) N (NIM 0 131 15 (L. ME. 1) N) (NIM	$D_2 = 2(1) - 2(1)$	KGND	19	Sum J≖Sum I+A(L° M)*SMASS(L.)	HC X
IF (LNE.1) G0 T0 130 K0W 01 IF (IHV-1) 25H,259,257 K0M AGG=LANDDA(N)*(7(1)-2(LT)) K0W 72 257 0(H,2)=-5UM2 K0M AGG=LANDDA(N)*(7(1)-2(LT)) K0W 73 259 0(H,1)=-5UM1 K0M C1= C61ARG1 C1 C1 259 0(H,1)=-5UM1 K0M G0 T0 140 C1= C61ARG1 C1 259 0(H,1)=-5UM1 K0M	DO 160 N=1.NTERM	K DMD	80	555 SUM2#SHW2+7(7)*HS+1W/7)*+2M2#SZ005	NGX
AGG=LANDOA(N)+(7(1)-7(LT)) KOM ^O A? 257 Q(H, 2)=-SUH? KOM S152(HARG) S151(HARG) KOM A3 259 Q(H, 1)=-SUMI KOM C1= C054(ARG) G0 T0 260 KOM KOM A5 259 Q(M, 1)=-SUH2 KOM	IF (L. NE . I) GO TO 130	4 OND	10	IF (IHV-1) 254,257	KOM
SI=SIN(APG) SI=SIN	ARG=LAMDD4(N) + { Z (1 } - Z (L Y) }	K OMP	55	257 0(M, 2)=-SUM2	NCX
C1=C0S(ARG) C1=C0S(ARG) G0 T0 140 C0=C0 K0MP P3 C0 T0 140 C0 T0 140 K0M	S I = 51 N (ARG)	K DMD	83	259 Q[M,1]×-SUMI	¥0×
60 T0 140 ×10 ×10 ×10 ×10 ×10 ×10 ×10 ×10 ×10 ×1	C1 = COS (AR G)	d NOX	8 4 1	G0 T0 260	XOX.
	60 TO 1 & 0	TINE X	5 H 2	250 Q(M, 1)= -5UM2	NOX

- 170 -

260 CONT INUE

` 1

•

e.

•

Ų

0000

....

260 CONT SNUE		541 dNDX
		Set dwDX
C SET [MAGINARY PART OF AKFO TO BE ZEPO		KOWP 146
C IN SURSEQUENT COMPUTATIONS THE SAME STORAGE	IS USED AS COMPLEX	KOND 247
		KOMP 150
1 AKFO(18.0J)=HFAL(AKFO(26.0J)) +(10.)		KOMP 151
IF (IPIG.EQ.U) CALL FORMKIAKFO, A.IBASE.DUN	IV. SFO.NEV.NB2.NEO)	KOMP 152
261 CONTINUE		651 unDX
C WRITE WODE SHAPE ARRAY ON TAPE		KOMP 155
		K:DMP 156
MARTE (1) A		KOMP 157
C SET UP FREMJENCY INCREMENTS FOR FREQUENCY RESP	NSES	KOM0 158
C FKN - NDATA		KONP 150
0F = 1 °/(2,4EN+DT)		KOMP 151
1004 = 2.401 MTF		K 0MP 162
WR (TE (6°2010 8 HED		KONP 163
Ū		KUNP 164
NNTV = NEV		KUMP 165
IF (NEV.GI.6) ANEV=6		Sol duty
1 re_ ANK # (MME V+ 1) # 10		ACHU ISA
ENCODE (50,2020,014) TUXANK		KOMP 169
WHITE TOURNEY TOUGHT FUNCTION TO A COMPLETE CONCERNS OF THE CO		KOMP 170
LL =0		KOMP 171
0= WW		K 0MP 172
NPL.DCK =0		KCMP 173
NTEMP = 0		KIMP 174
		KOWP 175
		NOMO 120
REWIND J If Itals we alson to 26.2		KONP 1 PO
DEVIND 40		101 dw0.4
		KONP 192
263 DO 570 KKK+1.NFR		KOMP 183
DM 2= DMEGA \$ \$ 2		A R CMDX
BW1 = CMPL, X (1 DFAC1)		KOND BAS
1 F (NPP.E0.0) GO VO 295.		K()MP 185
r check fub timiting values of omega		40MD 188
		KDWD 169
DQ 265 N=8.NTERM		Dol dMUX
IF LANS(LAMBDA(N)-OMEGA).LF.1.0F-061 DMEGA=(FGA+1 *0 E-05	161 awux
205 CUN NUC.		EGI UMUN
C INFTALIZE SUMEGAT AND LOAD VECTOR		Vol unux
Ų		KDMD 195
295 DO 320 M=1 . NEV		KOMP 196
DG 305 I=I.NCDMP		KIJT 141

305 MY(M.\$)=CMPLK(Q(M.{)).00)	4 UMD	061	
0 4 2° 2 51 - 0 2 3 4 2 3 4 2 3 4 2 4 2 4 2 4 2 4 2 4 2		60 f	
1F (M.GE.NEV) GO VO 320	CINC X	200	
N 2-11 4-12		102	
ABN * IWEN 012 DU	ŝ	205	
	a MD	602	
10 S(X, K)=S(X, X)	40wD¥	204	
J20 CONTINUE	GM() X	205	
	K OMD	50 E	
IF (NPP.FA.0) GO TO 500	UNDX X	202	
IF (1MV.FO.D) GU TU 450	K OMP	208	
DMC= DMEGA/CC	K () M D	50d	
1F (QMEGA.GT.0) GO TO 410	cim() X	010	
	K OMD	112	
FORCE COFFFICIENT FOR VERTICAL MOVION, DMEGA, ED.O	r Ovn	212	
FIRST FACE Philint	K GWD	110	
	dw0 ×	214	
[=] JBC(])	d OnD a		
Ja 1 JBC (2)	dw0x	510	
022=MF=2(7)	C NOX	21.7	
(f)2-(1)2=520	dh() X	518	
T ERMO { }= D22**3/{6.+CC*D2.4			
	d N D X		
SECUND FACE PDIME	av0 x	666	
1974 31	4 UMD	EZZ	
	KONP	224	
	4 U MD	225	
+ 204 E 20	1 MUX	525	
024=2(1)-2(1)	un0x	100	
₹ERM0(2}*(-D21**3*(D24/D23+f °\$+B25+))(0×0*(C+D24))	K NMP	822	
1F (NPP.60.3) 73 430	X OND	529	
	k Owb	230	
0.01.11.12 July 2	A DMD	E.C.	
	c a O X	212	
NN*E=W 00¢ 00	dw0 x	EEC.	
	A CMC	53.	
	dy() x	512	
1 ZG=0ZO	d HOX	2.16	
021=022	0404	237	
		0000	
	d NO X	044	
())))))))))))))))))))))))))))))))))))	KOMP	4	
	KOMU	242	
60 10 430	e fumus	243	
	k OMP	244	
FORCE COEFFICIENTS FUR VERTICAL MOTIEN, OMEGA.GT.0	dw() X	245	
,	K CIMP	245	
A10 DM3=DM664+0MC++2	KOWD	247	
	UNO X	248	
F1R5T FACE POINT	aw0X	C 4 2	
		000	
J=1JHC(2) D22=H - Zf D	KOMP	553	

....

υ 000 . . .

- 171 -

1040 FORMAT 1 2F10.01 105 505 520 u U ų J υU J υu Ċ c 415 940 875 940 875 940 269 279 101 550 259 292 265 26.6 260 270 275 277 280 282 593 205 295 28.7 2.83 200 202 E02 \$62 505 900 300 000 255 ç 260 l vž 503 264 267 27.8 1 42 2.84 5 505 100 994 30.2 401 101 27 anti) KUND KOMP dWD X dWD X dWD X CHOX CHOX CHOX CHOX CMUX CMUX UND X KNMP KOMP div Ü X CWG X 40MD K NWD CM() K () M () 4 (JMD 4 OWD dMD X ciw() X KUMD GMD C UMD (UMD) UNG 940 dNU. Cw0 CIMC) 4.30 ARG=IMC+H FACY=CONST/1(1,++ALPHA)+COStARG1+10+,1,1+11+-ALPHA)+StHtARG1) J=J1+J1 F (1142.60.1) 114(4.1)=114(4.1)+A(JJ-1,4)+TEPM0(J)+FACV F (1142.60.2) 114(4.2)=114(4.2)+A(JJ-1,4)+TEPM0(J)+FACV 1 ERM0 (2)= { -(14 C+020 /02 3+52/023+(52-53) /074) /0M3 ADD CONFRIGHTION OVE TO FOUNDATION R.EXINILITY D0 A60 4=1.NEV |F {{HV.ME.1}} AY{M.1}=4Y{M.1}-E41{M.N}/TEM2 D0 460 k=4.NEV ANG= (L AMBDA(N) - EMEGA)+ (L AMBDA(N)+0MEGA) 420 1ERM0(M)=((S2-53)/NZ4+(52-51)/0Z3)/0M3 SECOND FACE POINT SET UP MATRIX SIGNESA) AND LOAD VECTOR S(4,K)=S(M,K)-CU26E12(M,K,N)/TERM2 460 S(K,4)=S(4,K) 410 CONTINUE 500 IF (IR16,ME,0) G0 TD 501 D Z4 = Z ([) - Z (J) S 1= S(N(DZ 2+0)4C) T ER40 (() = (04C+DZ2-S 1)/(043+DZ 4) TERM2=C SOR T(CMPL X(ARG +0.1) 1F (NOP.E0.3) 60 TO 430 450 DO 470 N=1,NTERM 53=51N(022#04C) 5 3=51 N (DZ24 0MC) 00 440 M=1.NFV 00 440 J=1.NN (7)2-(()2=620 DO 420 M=3.NN (()2-(1)2=620 L =1JTC(3) D20=WL-2(1) D21=D22 D21=D22 t =1 .59C(M) J= 1.58C(M+1.) 022=#L-2(J) 023=024 11-12061=16 440 CONTINUE A70=014 55=25 51=52 15=25 ں ں ں U U υu 000 L

128 124 125 126 310 315 611 6 I E 02E 363 364 365 31.3 ۲IF 717 225 B SE 0.46 000 111 132 331 466 335 336 117 33.6 000 042 100 542 EVE 445 245 146 918 6.61 056 35 352 151 154 355 356 157 354 955 160 361 362 101 Ì K ()MD K ()WD uhù X Uhù X KOMP KUMP с () МП К () МП К () МП 40nu K () ND K ()MP 40MD A KOWD K ÜND к Омп KOMD 4 Own cwDX CNDX CNDX CNDX K (JMD K UVP K () W D K OMP k gun uhû X r i)wP k N40 KOWD K DWD K OMD кОмр K DMP K OHD к Өмр K CIMP K OWD KUWP r Own K OHP k (JMD K OMD K UND UM(I) X K ÜMD K OMP K D WD 400 AKE2(1,J)=AKE2(1,J)#FSCALE 400 CALL FUNNINIAKF1,AKE2,AKE4,19ASE,A,DUMMY,S,OMEGA,NWS,ESCALF,MTF40, D0 505 w=1.WNEV hacciw.l)=i)w24cans(hvim.l)) kmite (6.2040) [wega.(ihaccim.l).w=1.MNEV).[=1.NcnNP] SOLVE FOR GENERAL 1240 DISPLACEMENTS 00 525 1=1.MCOMP FF (1.60.11 11APE=2 FF (1.60.2) 11APE=3 525 CALL MR1TEY(YOUF(1.11.MN,1TAPE) [F (1.60.1) TTAPE=2 F (1.60.2) TTAPE=3 590 CALL WPITEV(VIUF(1.1).NR.ITAPE) CALL CSYMEO(S.HY .NEV. NCOMP) AKFI (1. J)= AKFI (1. J) + ESCALE 1F (DMEGA.GT.0) GO TO 490 DD 520 [=1.NCCMP YRUF(4M.) =HY(K.1) IF (LL.NE.NWR) GD TO 530 NREST=NFR~NWH #NULOCK IF (NREST.E0.0) RFTURN 00 590 1=1.NCOMP D0 505 1=1.NCOMP LL=LL+1 BD 520 KK=1.NEV 530 04664=04664+004 570 CONTINUE NALOCK=NALOCK 1 REAU(90) READ (90) AKF? DA 490 1=1, NR? DO 490 J=1, NR? READ (90) AKF1 Read(90) NR=2 #NREST #NF V **.**NB2 .NEV .NEQ) READ(50) HE AD [90] CONT INUE 1 + WM= WM NRUT 34 0=11 11=0

.

umu x umu x

2080 FORMAY UNI. . 2014.raio//		0 L		51 UMB	a
. ADM ANSOLUTE VALUE OF FREQUENCY RESODNSES FOR GENERA	ALIKUMP 34	, e ,		0040 17	
«ZED ACCELERATIONS DF THE DAM /	KOMP 3C		00 15 J=1.12	0140	en i
• X+00	KUMG	0	0,0=(1)4	61 QYN0	•
[//	LE ANDX	-	00 10 (= .3	0740 20	•
2020 FORMAT (194(10%,10440R2.6XCTT.13,337HK,104VERT.6XC1T//.54%5HOMEGA	E SMOX .	23	10 9(1.1)=0.0	12 UV/0	-
	f ahOx		00 15 1×1.12	22 0110	A
2030 FOHMATISTIO DAPANA COEFTICENT,/X./UII//4.4 CUNSTAWI YV31CHE 1 5.4444112.444544144444444444444444444444			15 00[1.])=0.0		•
OLDARTING LOCATING AT LOCATING AND	L dwox	ں يو ج			
	E UNDX	1		97 44 00	
				12 0VN0	
				DUAND 28	~
		•	T 268 T 2-R 4	01140 29	~
				DE UVNU	~
			27-12=212	IE GVNO	
•			21.3= 21-23	0000 32	
SUDROUTINE MPRINT (A.MR.MC)	1 bulk	-		EE GVND	-
	Ludw	e.	2.23=22-23	DUAN 34	
C	1018444	F)	524 = 22-24	DUAD 35	
C SUBROUTINE TO PRINT MATRICES	Ladw	• •	¥ 2 −E Z = 4 2 Z	90.40 36	
C	1 Volme e a	v,	Y QL=A1 34224-A244213	0040 37	
	Loom	e '	RM =[A1 + A2 + R3 + R4 } /4 • 0	0040 38	~
	T ROM	~ 1	ZM=(Zt2;Z+Z;+Z;)=WZ	96 0M00	
					-
	T T T T				
	1 HOOM				
			929681692066115°0±(11)11=1		
		4 F			
	1 BOM	. •			
	MDR T			0000 47	
1000 FORMAT (/// AX, 8114)	MPRT 1	•		0440 40	-
1001 FURMAT 1X.14.3X.RE14.7)	1 ddw	7		00 AD 49	~
	Mppt	E		0140 50	~
			H2=Sideline Internet interne	00.40 51	
				CAN DAVA	
			614 × SM4 16	0000 23	
		U		OUAD SA	
			0° 4/4 222 4d 54 42 44 51 = 1 4 °C		~
	0000000			20 000	•
9 0 0 1 1 0 0 1 1 0 0 1 0 0 0 0 0 0 0 0		- 0	P 25= - (F 40 212 - FP 4234)/ 4 ° 0		
		. •			
ዾ ዀዀዀዀዀዀዀዀዀዀዀዀዀዀዀዀዀዀዀዀዀዀዀዀዀዀዀዀዀዀዀዀዀዀዀዀ	UPOD40				
r FIDAS STIFFNESS MATPIN OK, FENTROIDAG STRESS MATPIN OF	OUAD	. 10			
C AND LOAD VECTOR FOR AN INCOMPATIBLE SOPARAMETRIC	0 UAD			DUAD 62	
C QUADR ILATERAL ELEMENT USING A FOUR POINT ENTEGRATION FORMULA	ON NO	*	SX/S HDH SX	ES OVO	-
Construction and the second	640044	с в		OUAD 64	
U	01/1D	••		10 UAD 65	
DIMENSION OK (8, 8),00(13, 8),01 3,31,11(8, 813,12),55(8),000(18,12)	1040		STRAIN DISPLACEMENT TRANSFORMATION	0010 66	
2	0000	1		15 GAUG	
DATA 55/ -8 R R. / . YT /-8 I I /	ou An	€6: I	86 1 ° 1) ±- 1M+X2 - SM+X2	00400 48	~
		P2 4		00 01 00 00 00 00 00 00 00 00 00 00 00 0	
		ł	B(1.5) a TOAXI SOAX2	n :	

- 173 -

• .

. •

DF NIE ELEMENTOURD 127 01400 127 01400 129 01400 129 01400 131 01400 133 01400 133 01400 133 01400 133 01400 134 01400 135 01400 153 01400 153 01400 155 01400 155	REDA REDA REDA REDA REDA REDA REDA REDA
0.00 70 1=1.0 0.011=01(1.) 0.011=01(1.) 0.011=1.1 0	SUDRINTINE READD IA.LA.N.NUMEL) 01 MENSION AILA. 01 MENSION AILA Commun.vue.784 2220 1 f 0
0140 71 0140 71 0140 73 0140 73 0140 73 0140 73 0140 73 0140 73 0140 73 0140 73 0140 73 0140 73 0140 73 0140 73 0140 73 0140 73 0140 74 0140 73 0140 73 0140 73 0140 74 0140 74 0140 74 0140 74 0140 74 0140 74 0140 74 0140 74 0140 74 0140 74 0140 74 0140 74 0140 74 0140	0.400 116 0.400 117 0.400 119 0.400 129 0.400 128 0.400 128 0.400 128 0.400 128 0.400 128
→ + +	
<pre>x115444X2 x115444X4 x115444X4 x115464X4 x115544X4 x115544X4 x115544X4 x115544X4 x115544X4 x115544X4 x115544X4 x115544X4 x115544X4 x115 x115 x115 x115 x115 x115 x115 x11</pre>	LT.D(L.11/1000000.) GA TO A0 L Ga(m,m) L ([1,1]-C+R0(m,J) L ([1,1]-C+R0(m,J) Gencate The Stiefness matrix and LDAD Vector

- 174 -

េះ AUBA

200 A(1)=A(1)						r.	_
		PEDR 12					
		RED8 13					<u> </u>
IF (N .EQ. NUMEL) GO TC 400		RED8 14					
IF ((MH_A) .LE. LB) GO TO 400		9608 15					
300 READ (1) B		REDR 16					_
2 # O		PE00 17			:	•	_
400 RETURN		9 EUB		SUARDUTINE REAL VIY, YREAL NC NEV . I TAPE)	DELY	-	
END		REDB 19		COMPLEX Y (NC. 1)	RELV	N	_
				DIMENSION VREAL(L)	R5L ¥	Ħ:	_
				COMMON /CNTPLZ/ NDATA,MWR,DT,NREST,NBUF,NBLOCK,14V,NCOMP.15FL.	PELY	*	
				f CDM9, NP41NT , ND	RELY	r	-
			U		P.H.Y	¢	
· ·			U	WRITE ON TAPE REM. PART OF RESPONSES IN BLOCKS	net.v	•	
			U		RFLY	æ	
SUBROUTINE PEADSS (SST , XY , LLM, NUMEL)		PEDS 1		REWIND LIAPE	PELY	c :	1
Ū		PEOS 2		1904×24 NM 0	PEL Y	0	
C SUBROUTINE TO READ ELEMENT STRESS TRANSFORMATION MA	ATRICES.	RE05 3		NBCOCK=N0 /NROW	PFLY	11	
C CENTROIDAL COORDINATES AND LM ARRAY FROM TAPE 1. AND	D STORE	RE05 4		IF (NOLDCK.EG.O.) NRCM#ND	nelv	€V #	
C THEM IN CORE AS ONE-DIMENSION ARRAYS		REDS 5		NB = NG QM¢ NE A	PEL V	5	
		REDS 6		NDEST×ND	RELV	•	
DIMENSION SST(1) XX(1) LLM(1)		PE05 7		0=fr	PELY	5	
COMMON /ELDATA/ LM(M).SS[309]0,KC+VC+ELMASS[41+319+5	0 1 4 1 1 1 C 2 4 1 1 1 0 1 C			1 BLOCK # 0		4	
			U		DEL Y		
		REDS 11	-	00 IN CACKET BLUCK I	DFL Y	0	
		REDS 12			A 144	00	
		RED 5 13			PELV	~	
		0505 14			DELV	ŝ	
		RE05 15			PFL V	5	
		PE05 16	·		RELY	**	
		RE05 17	•••		730 730	52	
		REOS IN	°.		PELY	56	
		REDS 19	J	A CAL WOLFFY (YDFAL, NG, 11APF)	PELY	27	_
800 SSTER Desst 1. J b		REDS 20	Ļ		RFL Y	28	
		REDS 21	,	IF A 184.0CK & T.T. WELDCK 9 GO TO 100	vjja	6	
N 2=N+N		55 25 22	U		ል ገታተ	0E	
KY 4N2-1 9= X C		REDS 23	ļ		PB,Y	36	
X Y & W ? D = YC		RE05 24		NB= NR E S T # N E V	A T B B	32	
U		REDS 25		IF (IRLOCK.GT.NRLOCK) RETURN	RELY		
DO 810 J=1.R		RE05 26		IF (NREST.GT.0) GO TO 100	RELY	4 19	
8 4 X # X		PENS 27		RETURN	RELV	17.	
8 8 G LEMEND= E.M. 6.3 9		PEDS 24	U		P FL Y	2	
120 CONTINUE		REDS 29		END	PFLY	È.	
RETURN		REDS 30					
							_
· · · · · · · · · · · · · · · · · · ·							
•							
				SUBROUT INE RESPONSING X. XS. A. Y. VHUE SIGMAX. TMAX. SIGNIN. THEN. NODE. NET	1. MAESP	ono E	
		050 4		* STAES. SST "XY "L.M. WNODE: WHELM. NEU." WITH WITH WHE	10230	C P	
SUGRUUTINE HEADTET CMONTANT	•	DEDY 2		٩٩٥٥٥٥٥٥٥٥٥٥٥٥٥٥٥٥٥٥٥٥٥٥٥٥٥٥٥٥٥٥٥٥	4495.50	•	
READ (ITAPE) Y	•	REAV 3	ن ر	SUPPORTOSI TI ON OF NODES AND TI ME RESPONSES OF DISPLACEMENTS	8 F S D	ŝ	
PE TURN		4 A038	, U	AND STRESSES	RESP	¢	
-							
							-

- 175 -

2 NANA SANA 5.0 \$ 0 20 522 ¢ 2 2 ٤ ŝ ŝ 222 2 2 5 5 ٤ ŝ 23 5 Ŀ e F 5 225 1 n<u>e</u> So Resp RESP Resp Resp RE 50 RE 50 neso Peso Resp RESP RESP RE50 RE50 RFSN RFSN 0E 50 8 E 50 965.00 RESP RESP RESP RESP RESP RESP RESP nes n Ne so 0530 46.50 PE SP PE SP PE SP nE Sn Reso AES P 0530 465 D RES P nESP RESP n 5 3 d nE S.o nrisn 0530 OT WENSTON XINFOLXSINEO),AINEO,NEV),YINEVI,YENEUFII),SIGMAXINUME J, PESP OT WAXII),SIGMINIUWE J,TMINII,NODEII),NELWII),SIRFSIMUMIRESP , SSTII),XYII),LMII 0E SD OF SID RFSIN 0530 n 5 3 a , XJUNK(19) COMMCN /CNTRL2/ NDATA.NMP.DT.NMESI.NRUF.NRLDCK.IHV.NCDMP.15FL. ICAMP.NPRINT.ND ••••• HHITIALIZE DISPLACEMENTS AND STRESSES 4ATRIX FROM TAPE sseeseeseeseeseeseeseesee READ ELEMENT STRESS TRANSFORMATION MATRICES READ SELECTED NIDAL POINT AND ELEMENT NUMBERS FOR WHICH DESPONSES ARE TO HE PRINTED READ STATTC DISPLACEMENTS AND STRESSES FROM TAPE IF [1GHAV.E0.0.04.1CHMB.E0.01 GD 10 100 FF [NNODE.EC.0] GN 10 145 READ (5,1000) [NUDE[1],1=1.NNODE] 145 FF [NNELM.EC.0] GD 10 160 READ (5,1000) [NFLM[1],1=1,NNELM) 60 TO 160 REWIND 2 REWIND 3 WATTE (3) NUMMP,NUMFL,NEV,ND,DT CALL READSSISST, XY.LLM, NUMELJ HACKSPACE I BACKSPACE I READ (1) XS,SIGMAX,SIGMIN GD TD 130 IF (1560.60.0) 60 TO 150 COMMON/LAREL/HED(N) DO 120 1=1.NUMEL 130 00 140 1=1.000451. TMAXII)=0.0 100 DO 110 1=1.NEO SIGMAN(1)=0. 120 SIGMIN(1)=0. 9 ACKSPACE 1 READ MODESHAPE NUDDE = NUM NP NNEL M= NUM EL. READ (1) A 140 TMINED=0. 110 X5(1)=0. . . 150 160

READ MUDAL RESPONSES 205 10 6 IF (1961MT.NE.NPRINT) GN TN 220 IF (10NUDE.LE-0).AND.(NNELM.LE-0)) URTIE (6.2010) T 10RINT=0 TF (15FL.E0.0) N=1 TF (15FL.ME.0) N=1 TF (15FL.ME.0) N=NDDE41) Waite (6.20201 N.X(N+N-1).X(N+N) (NNUDE.LE.0) GR TR 230 1F {NBLACK+E0.0} NROM=NO NB=NRAW4NEV CALL READY (Y BUF, NB, 2) DO 235 NT=1,NROW DO 190 K=1,NEV 00 190 K=1,MEV SUM1=SUM1+YK)#A(1,K) SUM2=SUM2+Y(K)#A(1,K) SUPERPOSITION OF MODES 00 200 N=1,NUMNP J=1+N DG 210 1=4.NNDE | PR| N| = | PR| N | P| X [] = 20M] + XS[]) 1 BLOCK=10LOCK+1 WRITE COLX 4 AN + 2 = AQ4N 230 1 = J--1 SUM 1=0. 5UM2=0 . 0=x30N1 I =X30N1 5 H = M = M 0 = W 8 ÷

RESP 115 RESP 116 RESP 117 RESP 117 4650 100 PFS0 100 A650 102 P650 103 P650 104 N650 105 P650 105 104 107 21 112 --RESP 1 9650 9650 8650 0650 0650 RE50 RESP RFSP 8650 0650 0650 0650 0650 0650 0650 a23a PESP RES 0 RES 0 8650 8650 8650 8650 8650 RESO 4534 RE SP RFSP RESP 053d a£ So 05,30 0530 oE Sp 05 3U 96 Sn nesn ofso RESD PE So o£3a 053d ofice n€∃u

- 176 -

000

U υu J υu υ.

170 U ں ں

000

υ

υU

ບູບູ

- - 180 V(K)=YRUF(M) ບບບ
- - - 190
- C15X+2M05=(1) X 002
- υv ں
- U
- PRINT NODAL DISPLACEMENTS υu
- - - 205
- 210

U

- 220
- CUMPUTE ELFMENT STRFSSES AND PRINT

2 **** 4 4 4 e e 40 3 2 <u>۱</u> 5 ំ ហ លំ លំ លំ លំ 233 53 22 091a 051a 8160 8161 8170 051 d 91.00 P 1.60 0914 0914 R 160 R 130 P 160 P 160 R 1 GD P 160 0514 8160 8 160 8 160 8 160 8160 8160 6160 8160 8160 8160 P160 R 1 GN 8 1 G D A 1 GD 05 t q R 160 R 16n P 7 GD 0914 R 1 GO 691 a R 16D 1999 09 l d 091d u i gn R 1GD 60 J 60 40 1 C D R 1 GD P 1 GD P 1 GD P 1 GD 09 8 M CALCULATE RIGID BODY STIFFNESS MATRIX AND STORE DNLY THE UPPER U(1) = U(1) + XS(1X) U(2)=U(2) + XS(1X) U(2)=U(2) + XS(1X)+(2(1)-XCG)+XS(1Y)+(U(1)-XCG) 00 130 K=1,NR2 Abt1,J)=Abt1,J}+REALfAKF0([,K)]+BfK,J} Comtinue DISPLACEMENT THANSFORMATION MATHEN B CALCULATE RO AND ADD TO LOAD VECTOR SOLVE FOR RIGIO BODY DISPLACEMENTS CALL BANSOL (3.3.3.4KP.U.0) Call Bansol (3.3.3.4KP.U.8) H({ Y .3) = (R(] 1) - XCG) CONTINUE L=1845E(1) DO 180 J=1,3 PO=80486(1,J)+U(J) XS(L)=XS(L)-PO DD 120 1=1.NASE 50 140 J=1.00 11=18ASE1171/2 DO 190 [=1.NR2 00 150 K=1,NB2 AKR(1,J)=0. D0 170 1=1.3 D0 170 J=1.3 00 150 1=1.3 00 150 3=1.3 611 X A(1X .2)=0 . AB(1.J)=0. 0=(1 . . I)= 0 . B([Y,29=1. TREANGLE. 8 (1X+ 3)=0. 1-A1=X1 8 - A 3 = H 3 1+3-5=1 RETURN A 0=0. ŝ 150 064 064 064 0 4 3 0 4 3 0 4 3 8 N N V 0 8 8 130 000 000 υu 000 PES# 151 PES# 152 120 128 129 130 9000 0000 0000 140 541 145 949 949 - N D ii) * ¢ c 2 221 127 5 5 **4** 8 243 1 ** 4. • 1 ::0 121
 0501
 0501
 11 ME
 RESONSE
 CF ADDAL
 D1SPLACEMENTS
 AND ELEMENT
 RESO

 1STRESSES
 1NULUDING
 STATIC
 CF ECTS
 CF MATIL
 NI
 RESO

 2010
 F0RMATL//IAH
 TIME
 T(SC)
 T(SC)
 RESO
 RESO

 2010
 F0RMATL//IAH
 TIME
 T(SC)
 T(SC)
 RESO
 RESO

 2010
 F0RMATL//IAH
 TIME
 T(SC)
 T(SC)
 RESO

 2020
 F0RMAT
 T(UO, XX, 2E(0.4)
 RESO
 RESO

 2030
 F0RMAT
 T(UO, MX, 2E(0.4)
 RESO

 2030
 F0RMAT
 T(UO, FELC, 1)
 RESO
 RESP PESP RESP RESP RESP RESP RESP b€ 3u RESP 9539 9539 9659 nESa PESa RESO P E SP RESP RESP RESP 8 1 G D R 1 G D в 160 в 160 в 160 в 160 в 160 491 af e 160 e 160 e 160 0160 0160 0160 0170 0214 **6160** uS au 051 N • 954 TIME RESONSE OF MODAL DISPLACEMENTS AND ELEMENT Istresses...Not including statte effects......xx.94(114-1/2) 240 WRITE (6.2030) WRITE (6.2040) (1,TWAX(1),SIGMAX(1),TWW(1),SIGMIN(1),1=1,NUWEL) SUBROUTINE RIGIO (R.Z.X5.18455.4KF0.8.4B.XC4.YC4.BL.NUPNP.NRASE. DIMENSION U(3) AKR(3,3) A(1) X(1) X(1) A(02,3) IBASE(1) ARANRY ç Z3U CALL STRESS(X,SST,XY,LLH,S\GMAX,IMAX,S\GMUN,TMIN,STRFS,T,MELM, INNELM,ISEL,NUM,INDEX) 235 T=T+D1 TO CALCULATE FORCES CCRRFSPONDING TO PIGIO PNDY DISPLACEMENT The mase PRINT STRESS ENVELOPS RESULTANT FORCES ON THE FOUNDATION 1- 1 10LUCK.GT.NNLUCK) GO TO 240 1- (NREST.GT.O) GO TO 170 IF (IRLOCK.LT.NALOCK) GO. TO 170 COMPLE X AKFOLNIS .1) NMMN.1=1 011 001 1000 FORMAT (1615) 2000 FORMAT (141. NROM=NREST NG=NRE ST* NE V 00 100 1=1.3 U(11=0. 2001 FORMAT (1H1. · 20%.8410// NETURN 4 SBN . END 00 υU

> 177 -830

υ

......

U U U

F 4 6 6 F 63 55 55 STIF STIF STIF STIF STIF 511F STIF 5116 STIF STIF STRE STIF STIF 511F ST 1F ST 1F ST 1F 5115 5116 5146 STIF 511F 511F 511F 5116 5116 5116 51 I F STIF 5115 3115 5116 STIF 51 I F SHF SILF 511F STLF 511F 5116 STLF SHF 5115 511F STIF "XJIJNKI 19] Common /Floata/ Lmib),ssi3.b1,xc,vc,flmassia1,sib.mtvpf.0ia1. D MENSION R(1)*2(1)*NEHC(1)*CODE(1)*1 X(4)*1E(2)*YMD(1)*EMI(1)* SUBROUTTNE STIFFTR, Z.NEBC, CODE, YMOD, ENU.AO, NUMMAT, PSP1 READ (5.1001) MIYDE,YMDDIMIYDE),EAU(MYPE),EAU4YPE) WHITE (6.2006) MIYDE,YMDDIMIYDE),EAU4MIYDE),ADUMIYDE) IF (ENU4MIYDE),LI,0.4999) GO TO 105 (5.1002) N.CODE(N).R(N).Z(N).MOD.M.IM.FACR.FACZ COMMON /CNTRL1/ NUMNP, MAAND, NUMEL, NBC, IGPAV Y WADA M TYPE) = Y NADA (M TYPE) / (1 ~ = ENJ (N TYPE) + + 2) ENJ/M TYPE } = ENJ/(M TYPE) / (1 ~ = ENJ(M TYPE)) READ AND PRINT MATERIAL PROPERTIES READ AND PRINT OF NODAL POINT DATA IF (FACP.LE.0.0) FACP=1.0 IF (FAC2.LE.0.0) FAC7=1.0 WPITF (6.2100) M00.NL1M.FACR.FAC2 60 TO 110 IF (PSP.E0.0.0) 50 TO 110 IF (N-L1) 190.170.130 • COMMON/LARGE ZHERER 0f= {R {N}-R{L} }7/0 V D2={2{N}-2{L} }7/0 V WR EVE (6. 2000 1160 WRITE 16. 2004 MED PO 110 H=1.NUMMAF Z(K)=Z(K-1)+NZ GI TA 170 L1=N+1 C 00E (K) = 0 × 0 H (K) = H (K - 1) + 04 00 140 K=LI.M ENU(MTYPE)=1. F = V NOD (NTY PE) 110 CONTINUE D E V=N-L 10.03.31 11=1.11 (1)041 N-9C=0 READ 1 + N=N e L 105 • 150 001 120 130 160 : : : : υu J ų J U. ¢ ¢ e 0125459 σ STAT STAT STAT STAT STAT STAT STAT S TA T STAT STAT S IA T STAT STAT STAT SYAT SYAT STAT STAT STAT STAT SVAT STAT SYAT 1 V 1 5 STAT SIAT STAT STAT 1 41 2 1112 FS FAT TAT 5 1 4 1 1 VIS STAT STAT STAT

WATTE (6,2020)HEA Call Stressix9,551,XY,LLM,SIGHAX,TMAX,SIGMIN,TMIN,STRES.TT,NELM. INUWEL 0,NUM,0) 231 [144] //244 STATIC ANALYSIS PESULTS/1X,231(144)/// 424 DISPLACEMENTS WIE TIJ STATIC LOADS...1N FT / DI 4F NSI DN NE NC (11) , XS (NE D) , A (NE G) , MAAN | 1, S I GMAX (MUME), T MAX (1), IS I GM (MUME), T M (1), S T R ES (1) , S Y (1) , L (1) , L (1) SUARADUTI NE STATECTNERC, XS. A. SIGWAX, FMAX, SIGMIN, FMIN, STRES. ISST. XY (LLM, NEQ, MAN , NUM, NUME) WRITE (6.2009) HED WHITE (6.2010) [N.X5(2+N-1],X5(2+M],N=1,NUMNP) HITE (6.2010) [N.X5(2+N-1],X5(2+M],N=1,NUMNP) Di 100 N=1,NUMEL COMMEN ZENTRLIZ NUMNP, MEAND, NUMEL, MAE CALL BANSOL (NEQ. MBAND, NEG. 4, X5, 0) CALL BANSOL (NEQ. MBAMD, NEQ. 4, X5, 1) CALL READSS(SST, XY.LLM, NUMEL) WRITE (1) XS.SIGMAX.SIGMIN . «XJUNKĮ 201 COMMON/LABELZ FERIAI * 66666-=(N) XVW51 S S 164 IN(N)=95999. 1MAX(N)=0. FORMAT (1H1. 10= (N) NIM1 20X,8A10// (1). OV 38 RETURN END 800z 100 J U J ų J J

- 178 -

IF ((CUD.EQ.1.0).DR.(COD.EQ.3.0)) GO TO 171 1F [(COD.E0.2.0).0R.(CUD.E0.3.0)) GO TO 173 WRITE (6,2002)(K,CODE(K),M(K),Z(K),K=L1,N) 1F {N1.LE.0.08.N2.LE.0} GO TO 200 CNDE{N1=0.0 READ AND PRINT ELEMENT PROPERTIES The second secon READ(5,1003) M.(1E(1),1=1,5) N=N+1 175 [F (MCD.GT.0) GO TO 150 [F (N-NUMAP) 220.220.210 [F (N-NUMAP) 220.220.210 BF(MB.GT.MAAND) MDANN=MD 210 WRITE (6.2160) N.NUMNP 00 290 [#[.4 10 290 J=1,4 MM=1APS[1X[])-7X[J]} FF(MM.GT.MR) MB=MM DETERMINE BAND W 107 M IFIM.EQ.N) GO TO 260 WR [YE (6. 2001)HED 190 WRITE (6, 2140) N NEBC (NBC) =2*N-1 200 WRITE (6.2150) 0.0 250 6=1.4 60 10 290 60 270 1=1.4 1x(1)=16(1) NBC=NRC +1 NEGC[NBC]= 24W C 0D=C 0DE (N) NTYPE = 1E (5) N1=N-WU) 50 TO 175 60 10 172 NBC=NBC +1 48=2+HH+2=8# 220 CONTINUE CONT INUE 0=(INVIJ# 0=0UM 5 1 Up 5700 510P STOP 0=6 M Ų= j 0= 14 260 [270 170 12 £13 042 250 1 280 290

STIF STIF

ST 1F

STIF 51-1E 511F 3115 3115 ST 15 ST IF STIF STIF 51 IF \$11E STIF 51 JF

3115

ບູບ

51 1F

3115

STIF STIF

J=11121 J=11121 K=1K[3] L=1K[3] R=1C[4] R=1C 20(1H4)/21H NATERIAL INFORMATION/1%,20(1H4)/// ZI(100)/220 ELEMENT SPECIFICATION/LX,21(100)/ f No. 1 J K L Materiat WRITE (6+2003) Notix(13.1=1.4). WYPE STRESS-STRAIN TRANSFERMATION MATRIX 1F (ENUIMTYPE).LE.0 .999) GD TO 300 1002 FORMAT (15,F5.0,2F10.0,215,2F10.0) 1003 FORMAT (615) STORE ELEMENT INFORMATION ON TAPE GO TO 305 F=YMOD(MTYPE)/(1,-FNU(MTYPE)+42) FORM ELEMENT STRFFNESS MATRICES C(3.3)=0.5+F*(1.-ENU(MTYPE)) CALL VRITERLW.111.N.MUMEL) If(N.EQ.NUMEL) GO TO 320 If(N.EQ.M) GO TO 230 GO TO 240 C (1 . 1) =F C(1. 2) =F+ENU(MTYPE) 1001 FORMAT (15.3F10.0) C(1,3)=0.0 C(2,1)=C(1,2) C(2,2)=C(1,1) €.9×51+52 €.4€.5K.3×15 €.4€1K-23=53-8 2000 FARMAT (1H1. 2001 FORMAT (1+1. 0G 310 1=f.& . 20X.AA10// . 2014.8A10// C(2,3)=0.0 C(3,1)=0.0 C(3,2)=0.0 8 t= 3×6 t b **310 CONFINUE** (1) XI = 1 1 + 1 = ¥ 1 305 ບບບ 000 000 ÷., Ē 90 32 9 × 8 5 × 5 5 52 22222 10 5 99 56 001 501 601 100 110 ŧ 86 ē 107

> STIF STIF STIF 311 S STLF STIF STIF

STLF

5115

u 느느

STLF

STIF

STIF

STIF

STIF STIF STIF STIF STIF STIF 51 IF

STIF STIF

STIF STR

STIF

STIF

ST 15 STIF ST IF 511F

20 .1 15 20 22 124 4 45 361 147 40 99 53 65 23 960 162 166 2 168 163 154 STIF ST IF 1 511F 511F STIF STIF STIF STIF 5115 5115 5115 STLF STIF STIF 511F STEF 51 IF ST IF STIF 3115 3115 5116 51 I F STEF ST15 STEF STIF STIF STIF 3818 2112 STIF STIF 3115 STIF STIF 5116 3115 ST 1F 5115 5115 571F STIF S115 "

u

¢ e e 2 = 222 5186 5186 9418 94 9418 94 STRES 5106 51 RE A MAS I MAS STRF STRF STRF STRF STRF STRF STRF 5196 3 T PE STRF GHTAURXYSTPF S FDF × 5HS 1 GAY 1F ((BB.EQ.0.0).AND.(5/6(3).EQ.0.0)) 6C TO 255 2000 FORMAT(1) + 234ELEMENT STRESSES IN PSI/ 1640EL-840 7x 147 7x 147 7x 545164x 7x 2640 25x 7851674AX 5x 74816441N 5x 541840EL /) 2001 FORMAT (15,1X,2FE,2,5FE12,4,F10,2) 255 IF ([INDEX.NF.0].0D.(NVELM.E0.0)) GO TO 370 IF (15EL.En.0) GO TO 310 IF (N.GT.NNELM) GO TO 330 IF (N.G.NELM) GO TO 330 310 MAITE (6.2001) N.XC.VC.(51G(11).1=1.6) 350 IF (SIGMIN(N)-SIG(SI) 300,300,360 360 SIGMIN(N)=SIG(S) fPIN(N)=TT 330 [F (SIGMAX[N)-SIG(Å)) 340,350,350 340 Sigmax[N)=SIG(Å) SUBRIUTINE SYMINUTA,MDIM,NMAX) Cumelex A(ndim,1) Cumelex D D0 200 N=1,NMAX \$ 19(6)=28.64841AN2(516(3).88) CALCULATE PRINCIPAL STRESSES H8= (\$ 16(1)-\$16(2)}/2 * CP=50R1(H8**2+516(3)**2) 0 = A{N+N} 2 A{N+N} = C4PLx(-1+0+0+0) CC=(5|6(1)+S(6(2))/2+0) A(N.J) = A(N.J)/0 DD 150 T=1.NMAX TF (N.€0.1) GD TD 150 DD 140 J=1.NMAX WRITE (3) STRES XAMN . I=L 001 (10) S 1615)=CC+CP XC=XY(N2-1) VC=XY(N2) T={N}XANT 300 CONTINUE 320 RETURN 1 + X = X END 201 Ų υ U 0000 Ų STIF 171 STIF 172 STIF 173 STIF 173 5115 178 5115 179 5115 179 5115 180 5115 182 3 440 RT= F16.6.164 K-SEC442/F1.4447) STJE 175 2100 F00MAT (214 GENERATION MITH 4405-13.3X.64 NL M=14.3X.64 FAC9=F10.5.5TJE 176 13X.64 FAC2=F10.51 STTE 177 18 0 1 1 2 2 20202222222222 222 169 169 170 Ξ 2 5 ev F 511F 511F 511F STRE , xJUNK121)
 DIMENSION X0(1),STG(6),STGMAX(1),TMAX(1),STGMIN(1),TMIN(1),NELM(1)STGE
 LENSION X0(1),STG(1),XY(1),LLM(1)
 STRESONUN,SST(1),XY(1),LLM(1)
 STRESONUN,SST(1),XY(1),XY(1),STG(1) STRE 5195 3012 3195 3185 510F SUIPOUTINE STRESSEXX0.555.X7.LM.SIGMAX.TMAX.SIGMIN.TMIN.SERES.TT. STRE INELM.NNELM.15EL.NUM.INDEX1 5TRE STRE STRE StpE STRE STRF STRE STRE SIRF 5 1 RE 1015 2130 FORMAT (2010-01147 JOHNE CARD IS MISSING) 2140 FORMAT (3140-01091)147E CARD FOR JEINE DO. 15.164 NOT IN SFOUENCE) 2150 FORMAT (31401030154164 CHENT NFORMATION IN GENERATE MESA) 2150 FORMAT (3140103161 NUMBER 15.214 EXCEEDS GIVEN NUMNP=15) 25(1441/264 NODAL PCINT SPECIFICATTOW/1X,25(144)// 2005 FOPMAT (2640400AL ODINT CATD P-ORD //) 2006 FOPMAT (2640400AL ODINT CATD READT N -15) 2006 FOPMAT (1640404FF1AL NUMUER 1)/ 1 400 E = F16.67 + K/SFT./ 3 400807 F10.66 IF ([INDEX.E0.0].AND.(NNELM.GI.0)) WPITE (6.2000) CCMMCN / CNFRL1 / NUMNP, MEAND, NUMEL 516(1)=S16(1)+S51(11)+X0(JJ) 00 190 J=1+3 COMPUTE ELEMENT STRESSES STRES(NN-2)=\$16(1) STRES(NN-1)=\$16(2) STRES(NN) =\$15(3) 00 300 N=1,NUMEL N?=N+N 00 180 1×1×3 0.1=0.0 0 1A0 J=1.0 NB={ N-1) + 8 (M) ATT=(M) 1+11=11 No Cin NN **C+UN=W** 11=0 EN0 i i X 1 80 190

> _ 180 -

ບບບ

υ

READ (5.8010) LR.(1JBC(K),K=1,NPP) IF (LR.E0.0) MRITE (5.2010)HED IF (LR.ME.0) WRITE (5.2011)HED MRITE (5.2020) I=1JBC(1) J=1JBC(2) PRESSURE BOUNDARY CONDITIONS | |=[M(201) MASS (11)=MASS (11) (ELMASS (1) MASS (11-1)=MASS (11) CONT (NUE ANTE (6, 2030) (1, MI1, MI2) A(81,JJ)=A(81,J)}FS(1,J) CONTINUE CONTINUE #12±-0J*F*#2。*0#/6。 2J1= PJ*F#(1-F/3,)*02/2。 a J2 =- P J+F + { 1 - F / 3 • } + DA/2 • 10 200 GO TO 160 XS(11)=1)=XS(11)+R12 CS(11)=XS(11)+R12 IF (LR.NE.0) SGN=-SGN SEANC B-SEISX=(1-FFISX R11= PJ+F++2.#DZ/6. 15 [X1.LE.0.] GO \$C 84 (Cf)SX=(C)SX NN=NPP-1 DD 150 L=NF,NN P1=P.9 IF (NPP.E0.0) 1= { J = { (L) J = { JGC { (L) J = { JGC { (L + 1) J = J + J + J 00 130 fat.4 0 2=2(1)-2(J) 0R=A(1)-A(J) WR)TE (1) A x=ML−Z{ J} P_J=0 .0624#K X1=4L-211) P J=P J# 56N 60 10 210 NDS#Fd=Fd 1 2 = 1 + 1 81 = 8 JE 15 = 8 J2 F = X/DZ 1+1=11 5 GN=1 . 21=0 -42=0. NF = 2 1 = U N 1 001 120 200 210 **U U U** 8 ñ fN T 21 2 ŝ 2 20 5 I MAS TOTL IMY2 IMY2 IMY2 IMY2 IMY2 IMY2 I MAS I MAS SURPOUTINE TITTAL (R.Z.MERC.1. JOC, MASS, AKFO, ITASE, A.XS, B, AB, WEQ, MAAN, TOTL .SGW.ME2.ESCALE) TOTL TOTL 101 TU II 1011 1011 rout 10 11 ron 101 TUTL LOLL DIL 7071. 1011 1011 1011 TOTAL STIFFMESS AND MASS MATHICES AND LDAD VECTOR ARE FORMED BY TOTL Reading the rlement stiffness and mass etc.from tade 1.000al pointfot T011, ro 1L TOTL 110 101 1016 101 1016 1011 FOTI. FOTL TUU LOTE 101 СОЧИЛИ /СИГАLI/ NUMNP,MBAND,NUMEL,NBC,IGRAV,NPP,WL,IFACE ..1016,1079,1074.0656.78.17 ..2104K123 .Cumhon /Eldata/ Lmt B1,5513,01.XC,YC,6ELma55141,513,01,MTVPE,0161. DIWENSTON MT11,Z(1).NERC(1). MASS(1),XS(1).AINEO.WBAN J.(JRC(1) DIMENSION TRASE(1).BR(MD2.3).AR(AB2.3) COMPLEX AKFO (492.MB2) CALCULATED. FORM TOTAL MASS AND STRFFNESS MATRICES LF {N.ED.J} GO TA 140 A(1,J) = A(1,J) - A(1,N)\$A(N,J) A(J,1) = A(1,J) A(1,L) = A(1,J) A(1,N) = A(N, 1) A(1,N) = A(N, 1) CONTINUE **UUE TO WATER PRESSURE ARE** CALL READOLLM. 111. N. NUMELS UJJ.LT.IP GO TO 110 00 120 1±1,8 11≠LM(1) XS(11)=KS(11)+0(1) DO 110 J=1,8 COMMON AL ABEL MED 1.8) HASS(1)=0.0 XS(1)=0.0 D0 100 J=1.44AND D11.J)=0.0 DG 300 1=1,0MAX DG 300 J=1,0MAX Af1,J) = -Af1,J) Af1,J) = -Af1,J) CONTINUE RETURN END UD 140 N=1.NUMEL DO 100 1=1.NEO 13=6.66.9=68+8 85 € J.J.e. 7=85 REAL MASS 10 (3.3) LOADS 040 300 1 00

202 2 ŝ 5 5 ŝ 53 504 2 C 9 65 ŝ 2 ŝ ŝ 20 Ξ ~ 2 ŝ 35 č 1 8 79 2 5.5 é 95 ê 10 5 90 0 e 5 2 ŝ ë 101L 101L 101L 101L TOTL 701L 701L 701L 701L 7 016 7 016 7 016 7 016 7 016 7 016 1011 TUTE TOT 101

- 181 -

000

υU

ųųų

161 161 189 190 1011 TPNS TPNS 70 TL 70 TL TUT 1101 TOTL TOTE 1018 101 CALL RIGIO (R.2,XS,19ASE,AKFO,8.AD,XCG,YCG,AL,NUMMP,NNASE,MR2) Confinue SURPOUTINE TRASFRATY, UG, YBUF, XT, XUG, NC, ME V, TO , NXUGH, NXUGV) ADD FOUNDATION STATIC STIFFNESS TO STRUCTUPE STIFFNESS 46M THE RESERVOIR IS TO THE LEFT OF THE 07 240 1=1.NF2 L=18ASE(1) DO 240,1=1.NF2 K=18ASE(1)4.1 1F(KLLE+0)+07* [K+6T,ABAND]) GO TO 240 A(L+1)=A(L+X)+HFAL(AKFO(1,J)) CONTINE CONTINE D ISPLACEMENT BOUNDARY CONDITIONS X-LOAD COMPUTE RO AND ADD TO XS DD 170 1=1.NE0 18ACE=18ACE+ANS(A(1.11) 1F (NEC.L.E.0) GD TD 190 DG 180 K=1,NBC L=1-J+1 1F (L.LE.Q) GO TO 180 A(L.J)=0. 1324 NODAL POINT > 2030 FORMAT (112,2410.3) I=NERC(K) A(1,1)=1,0 X5(1)≠0, DD 190 J=2,MAAND 1010 FORMAT (1615) 2010 FORMAT (1111. 2011 FURMAT 1141. · 20X, AA10// · 20% . A 10// A(1,J)=0. 190 CONTINUE 190 RETURN TRACE=0. END 170 8 8 8 0 0 0 0 0 0 0 0 0 0 8 1000 U 000 0000

119

104

105 109

105 10.7 104 110 Ξ 2 5 -115 115 117 611 2 122 123 124

r 014.

35 29 79

5 ŝ 1 00 101 10.7

ADD STATIC FOUNDATION STIFFNESS AND FOUTVALENT FORCES FOR RIGID DODY MOTION OF FLEXIBLE FOUNDATION EQUATION NUMBERS CORRESPENDING TO BASE DOF. FORM STATIC FOUNDALION STIFFNESS READ(5, 1010)(10ÅSE(1),1 =1.NN2.2) COURDINATES OF CENTROLD OF BASE WRITE (6.2030) (1.011.012) 150 CONTINUE WRITE (6.2030) (J.RI.R2) AKFOT1.J=AKFO(1.J)#FSCALE AKFOTJ.1)=AKFOT1.J [BASE(11) =2*J BASE(11-1)=1AASE(11)-1 FF(1816.NE.0) GO TO 250 READ EASE NODE NUMBERS tR =t 0A SE (M12~1) BL = t 0A SE (M12~1) XCG = (A (TA) + R (TL) 1/2 • YCG = (A (TA) + Z (TL)) /2 • רא+(ו-(ר) sx=(רר) sx ורא+(ור) sx=(רר) sx 11 h+(1-11) \$X=(1-11) \$X H12=-{2*#P1#P1}#DR/6* BJ1= {2*#P3#P1}#D2/6* RJ2=-{2.40J+P1)408/6. -9/20+1(-0+)04-51 =118 P.J=0. U6244(W.-2 (J)) 00 230 1=1. NASE 1=1,N112 J=1 . NN2 []=]+[J= 18A5E(11-1) AKFO 1L= 1 BASE(1) RT 1=RT 1+P 1 RT2= RT2 +R2 RT=PJT N5 Stid=fd REW IND 90 READ(90) READ(90) DE UNIMER 06)0V30 160 CONTINUE 00 220 8 Z=R J 2 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0 2.30 0000 000 000

621

1 2 A

130 133

125 126 127

136 136 136 136 136

Ī 42 EVI •• • 545

		TONS	×		29 SNat
		L DNG		CT IN COLUMN TO COLUMN THE COLUMN	ES SNUL
		TRNS	¢		T RNS 64
DYMENSION AT (1), XUGAT PUBLY MBEET - MBHE, N	ABI DCK - 1HV - NCOMP - 15F1 -	TRNS	•	TALE DESAUTION TO AND AT APE D	10NS 65
COMMUN ZUNIHLEZ MUALA WARGU SUNESI STOUTS		T DAIG	-		TRNS 66
°		TRNS	2 =		TRNS &7
	•	TONS			19 SNa1
F AC= 47.4 2*17 =		TRNS		COLUCE AND	TRNS 69
		TPNS	-		1 SNU1
NA+0+00 - 00 - 00 - 00 - 00 - 00 - 00 - 0		T RNS	15	120 Y[JJ,KK]=YBUF[MM]	1 SNOT
		SNUL	9	LOU CONTINUE	21 SN61
TRANSFORM COMPUTATIONS	•	7 RNS	1	U	T SWIT
		SNS1		C FORM Y (w) 40.06 (w)	SI SNUL
dh(JN)* [#] 00 * 00		TRNS	0	C DO LEO MEV	TRNS 76
DEAD FARTHORIAKE DATA		TRNS	51	DO 160 J=1.NC	TRNS 77
	-	SNdA	22	160 Y1J,K1=Y1J,K1+UG1J1/DT	TRNS 7A
11 ADE=2		SNB1	53	U	T RNS 79
IF (1.60.2) GO TO 50		SNAL		C WRITE ARRAY Y ON TAPE	DE SNOL
IF (1HV-1) 40.50.40		SNY 1	5 X		
40 READ (S.1000) (AT(K). XUG(K).K #1. NXUGH)		SN91	5	REVINO ILAPE	CH SNUL
WRITE I 6.2000 / KIIKY / KUKKY / KEI / MAGGAY		SNat	5	120 CALI WRITFY(Y(1,K),W2,ITAPE)	TRNS RA
		TRNS	29	400 CONTINUE	TRNS A5
50 [TAPE=3		T RNS	96	REMIND 2	TRNS R6
60 READ (5,1000) [XT(K], XHG(K], K=1, NXUGV)		TRNS	E	REWIND 3	TONS AT
WRITE (6.2001) CATERS.XUG(K).K+1.NXUGV)		SNUL	έ£	3 € ∧ = 0	THNS AB
NHTNKUGA		SNUL		190 HFV#IEV+1	TRNS A9
70 00 140 K=1.NX				16V1 = 16V + 8	
140 XUG(K)=XUG(K)#FAC		T RNS	36	CALL READY(Y(1,1EV),M2,2)	TO SNH
ONA Share to the share succession to the second	A TTS ALLASTING	TRNS		TF (TPV «NE«278 519 519 200 A stationarys vis . Fevel D. ND . 41	TRNS 93
UNA FRANKLARE FAMILY NAME AND ALL ALL ALL AND		TRNS	F		A P SNA T
		TRNS	ê	1 0 0 1 1 1 1 1 1 1 1 1 1 1 1 1 1 1 1 1	TPNS 95
50 HCK 1=0		SNdA	04		TRNS 96
101=(70/07+=5)+1		SNal	Ĩ	C AL LASING OF ANRAY Y	TP 2NAT
EF (NO.GT. NDATA) ND=NDATA		SNBA	₹	U	BC SNAL
CALL CINTERIXT, XIPG. ND. DT. UG)		SNUL	E -	200 DO 210 J=2,MJATA	1945 949
		2 NO 1			TON SMAL
FOUR LER TRANSFORM OF EARTHOUAKE ACCELERATIO				210 V(JJ,1EV)=CUNJG(V(J,1EV)]	CUI SNAL
		SNAT		Y (NC, YEV)= CMPLX(HEAL #YINC, HEV)FOGOTYC. Caid Eeffiyyydd y Revd ami	TRNS 103
		TRNS	41		TRNS 104
BEAD FIMPLEX FREDUENCY RESPONSE FREW TAPE A	AND A	TRNS	64	CALL REALTY V VIF. NC. NEV. 2)	TRNS 805
STOPE AS 2-DIMENSIONAL ARRAY	0	SNa 1	50	AF, TURN	101 2021
		2NHT	5		HOI SNUL
REWIND FRAFF		SNUL	6	JUUU FURMAT TOLTOSISTO 24 PP 2010 FORMAT T//27H HIDPLZINHTAL FARTHOUAKE DATA /	TRNS 109
DO 110 N=1 . NHLOCK	•	SNUL	45	////	VANS ILO
0=WW		T RNS	ц,	-100H TIME ACCN-G TIME ACCN-G TIME ACCN-G	TI SNULL
CALL READY (YENJF, NB, ITAPE)		SNat	99	WE ACCN-G TIME ACCN-G TIME ACCN-G //	288 SNAT
00 100 LL = 1 .NWR		SNHL	2	.[6[4x,F6.J.2X,F6.4]]	WIT SNUL
		C NO A		2001 FURMAN //2/H VEWALCAL EAWINGOARD DATA /	SII SNHA
00 100 KK=1. MEV		ARNS	2	LINE ACCN-G TIME ACCN-G TIME ACCN-G	TETENS 116
A REAL FOR STATE OF A DURING A DURING A DURING A STATE OF A DURING A STATE OF A DURING A DURING A		Shal	÷	WE ACCN-G TIME ACCN-G TIME ACCN-G //	TANS 117
					×.

- 183 -

.

U U U

0000

υuu

0.00 000

υ

ų

•

.

1957650 FFTM 40 FFTM 40 FFTM 41 FFTM 42 FFTM 43 FFTM 43 FTM 44 F F T M F F T M F F T M LABEL END OF UV APRAV AETURN Paint 30, NTX Format (+0 Frandr Defected Hy &Fftwre≠+ 5x, 4ntx = + 110) Format (+0 Frandr Defected Hy &Fftwre≠+ 5x, 4ntx = + 110) Ford SUBROUTINE FFTCSULJT) COMBON COSNLY UV15123, INFIN, IAD COMMON YFFIP, N1, 1HEIA DD 10 N = 7, 512, 2 TH = THETA + FLOAT(JJ) UV (N = 1) = COS(TP) UV (N = 1) = COS(TP)
UV (N = 1) = COS(TP) UV (N = 1) = COS(TP)
UV (N = 1) = COS(TP) JF = 0 Theta = - 1WOP1 / FLAAT(NF) 00.20 L = 256, NT = 256 Call FFTC 201J1 Call FFTC 201J1 Cantinue FND FLE 90 IF (JT .NE. 0) GO TO 25 NIM = NT NT = NT IDENT FFIM . END KEEP Term 1. X X ON 1 5 3 * 20 30 0 TRNS 119 TRNS 119 4 6 010515555 - ~ - + -¢ ⊂ ∂ 9 - 2 -6 P c c 4 81 9 4 9 19 4 9 19 4 9 19 4 9 19 4 9 19 4 9 19 4 9 19 4 9 19 4 9 19 4 9 19 4 9 19 4 9 19 4 1 ыяту ыяту ыяту ыяту ыяту N 133 667W 677W 677W 677W FFTM F F T 4 SUDROUTINE FFTAKETNIX) COMMON / CSMD/ UV (5121, INFTN, IND COMMON / FFT2 N1, NTM COMMON / FFT2 N1, NTM COMMON / CSM3111751779586/ REWIND 90 NTY = NTX IND = 1 IND = 1 SURFINITINE WRITEB (A.LA.N.NUWEL) DIMENSION A(LA) COMMONYNUF/UL2220) L8=2220 IF (N.,ME. 1) GO TO 100 1F & N . FA. MUWEL) GA 10 300 1F & (N+LA) .4.E. LB) GA 10 400 300 MR11F (1) B IF (MT -GT. 256) GO TO 3 MT = 256 1ND = 0 IND = 0 INFIN = 3777777777777777 SUBROUTINE WRITEYLY "LN"ITAPEL Divension Y(LN) Write(Itape) y Write(On Edon . { 6{ 4 4, F6, 3, 2X, F6.4) } END 00 200 8=1.LA 11=1+M N=NM M=NM N 14 M= M 001 M=0 RE FURN 0 ≈ N (IN J

184 -_

400

m

гети 104 гети 104 гети 104 гети 111 213 021 120 2 1 54.4.4 5 M L L L WA 31 MA 3 3 1111 W # 13 13 AEGIN NEW ALNCK Sumply of Cosinisains finismed FIRST PART OF INIT FOLIOWS XA÷IA - XI+ CX Dx ×++in + ×++in + 0.4 0.4 XATONI UI+XK VI+XK VI+XK - VI+YK 10 XK U1+YK V1+XK V1+YK + V1+XK INTERTOR LOOP KK T() ×0 U2\$XK ×0 ◊ U2\$XK V2 4 YK XQ & U24 XK TU YO U2≠YK YO + U2+YK V2 4 X K 35 ŝŠ ž

541 A2+05 012 X1,440 924 X1 924 A1+05 925 A1+05 925 A1+05 925 A1+05 925 A1+05 925 A1+05 928 A1+05 94 543 43005 FX6 73774 D76 - 75 546 43 594 43905 F0 43905 F0 8417,808 541 84 541 84 SAJ A4.05 554 A3005 575 X14X3 776 X14X3 776 X14X3 90,405 90,405 90,405 725 A545 725 A547 725 A547 725 X24 725 A5417 725 A5417 725 X24 725 A5417 725 A547 725 A547

2 Z Z

. BLK IT

RNR

8 A 8 N 8

~ F. 5	e v c v	50	 	 		52	. 63	4.7	65	\$÷	57	5	10	11	72	53	44	77	76	79	А0	Ē	28	28	99	A 7	80 80	00	10	42	66	46	95	96	10	86	60	00	5	 40	05	90	10
					WL 11	CFT W	FF 14	FF 7 4	FF TN	11 L L L	1 1 1 N	FFY	FF TH	FF TM	* 1 3 3	FF 74	11 1 1 1 1 1 1 1 1 1 1 1 1 1 1 1 1 1 1	EF TN	M1 11	H A S S	8F 1M	FF14		FF TN	FFTM	1 1 1 1 1	6 F TM	AT TH	MA 34	NLJJ	WL 11		FF 7 M	W1 11	11 H H	1111	FFTW	FFYM		FFTM	FF 14	FLW	1 7133
· ·	athone of country of the sheater								WEPSE TRANSFORM									P		LECK NY 15 A DOMER OF ?	•			- 110N 15 NT = 1				IT FLAST SED OF CSNDS INTO UV		3 = FIRST ADDRESS OF XX ARDAY						RAST LOCATION AFTER XX ARRAY			JARER OF FLEWENIS OIV 10FD HY 2				

VFB 24/0146 T1,36/1 855 1143 4415 9035 T143, T415 5005 T8401, T840 90155 761401, R040 50155 761401, R040 60 660 VED 19/04/FFT.42/1 155 1 50051 1 50051 1 5015151 1 50151 1 50151 1 50151 1 50151 1 50151 1 500 576 BI 546 Lax 548 Lax 548 Lax 548 Lax 548 Lax 132 X14X1 132 X14X1 548 Tax 545 Team 545 Team 545 Team 545 Team 545 U 250 U 0,00,004 WACPO LA.LB USE + 858 0 USE /FFTP/ 855 2 USE + 24: X1 .NDT NG X1 .NDT CX2 X1 AX2 1 AX2 1 NZ X2.NDT AX6 1 Z4 X5 .NF Z4 X5 .NF 543 83 FX6 X3/X4 SA6 43 SAL LA Ex5 XI SA6 LN ENDM NT * SUUST + + 7 H 15 NEXI а • в • в GEN

- 185 -

 •
 •
 •
 •
 •
 •
 •
 •
 •
 •
 •
 •
 •
 •
 •
 •
 •
 •
 •
 •
 •
 •
 •
 •
 •
 •
 •
 •
 •
 •
 •
 •
 •
 •
 •
 •
 •
 •
 •
 •
 •
 •
 •
 •
 •
 •
 •
 •
 •
 •
 •
 •
 •
 •
 •
 •
 •
 •
 •
 •
 •
 •
 •
 •
 •
 •
 •
 •
 •
 •
 •
 •
 •
 •
 •
 •
 •
 •
 •
 •
 •
 •
 •
 •
 •
 •
 •
 •
 •
 •
 •
 •
 •
 •
 •
 •
 •
 •
 •
 •
 •
 •
 •
 •
 •
 •
 •
 •
 •
 •
 •
 •
 •
 •
 •
 •
 •
 •
 •
 By what and show the set in the term of the set in the 192 75.0 FFTM 257 FITM S 111 4 L 3 3 GF H4,412, PASS LT 82, 94, PASS BC 82, PA, CGNJ BX X1 SAT A2 BAT A2 BAT A2 SAT A2 SAT A2 SAT A2 SAT A2 SAT A2 SAT A3 EG N0.90,RNR PX4 80.x1 NXA 80.x4 SA1 UV MREAN THERM NURY NOFRM 847 ×2 547 45 541 41+81 542 42+83 8×7 ×1 547 42 RJ FFTERR PS 546 04484 546 46483 541 86487 841 86487 841 - X1 547 41 547 41 547 41 VU INS CND. ROUNDI 111 S U UNVU INANI 1 21 41 PET * CONJ 55¥d TRAN 1 cN FFTH 145 FFTH 147 FFTH 147 FFTH 147 FFTH 200 FFTH 201 FFTH 202 FFTH 203 FFTH 203 FFTH 203 DIVIDE B6 (SEMIRLOCK SLZF) AV 2 -NEXT SEQUENCE OF COSINUSOIDS INITIAL VALUE DE J J.Le. Ji Swap be XX(JT) And de XX(J) FIPST LOCATION OF XX ADDAY LOCATION OF JT FOR NDFDM FNITTAL VALUE OF JT PENTIND TAPE OF COSINUSOTOS RUN IN NEW SEQ CSNDS PEMUTATION OPERATOR CONTINUE TTERATION XA + UZAYK + VPAXK SAME WATREX 10 VK FX6 X5+X6 FX6 X5+X6 546 A4 546 A4 145 145 144 15 146 145 144 17 513 A5 15 148 155 X1 542 16 745 15 542 16 544 16 5 911 JT 911 JT 910 JU 916 JU 916 JU 917 JU 1. X X 1 x 1 85 587 x1 587 x1 835 O

ž

- 186 -

POUND +

× .

11 11<
111 111
Store Bit First store First store C0 Person First store First store F0 Person First store First store Store store
En Prefeit En Participa de la compacta de la compac
Summary Ling Frame. Consols Archary Wrighs), Harles, Hol Consols Archary Wrighs), Harles, Hol Consols Archary Wrighs), Harles, Hol Consols Archary Wrighs) (10) - 1 - 1 - 1 - 1 - 1 - 1 - 1 - 1 - 1 -
Submut life Free Free Free Free Free Free Free Fr
OPDIUTING FFTORE FFTORE CONCOL ING FFTORE FFTORE CONCOL AND AND FFTORE FFTORE FFTORE CONCOL AND FFTORE FFTORE FFTORE
Gundal Kerner Free
$ \begin{array}{c} \begin{array}{c} \begin{array}{c} \begin{array}{c} \begin{array}{c} \begin{array}{c} \begin{array}{c} \begin{array}{c}$
100.1.11 110.1.11 110.1.11 100.1.11 110.1.11 110.1.11 110.1.11 110.1.11 111.11 110.1.11 111.11 111.11 110.1.11 111.11 111.11 111.11 111.11 111.11 111.11 111.11 111.11 111.11 111.11 111.11 111.11 111.11 111.11 111.11 111.11 111.11 111.11 111.11 111.11 111.11 111.11 111.11 111.11 111.11 111.11 111.11 111.11 111.11 111.11 111.11 111.11 1111.11 111.11 111.11 1111.11 111.11 111.11 1111.11 111.11 111.11 1111.11 111.11 111.11 1111.11 111.11 111.11 1111.11 111.11 111.11 1111.11 111.11 111.11 11111.11 111.11 111.11 11111.11 111.11<
Image: Solution of the cost was in the set of the cost was in the set of the se
HED (20) UV HE (10) HE (10) HE (10) </td
Elitor FETA FUT - FETA FUT - FETA FUT - FETA FUT - FUT -
$ \begin{array}{c} \text{EFW} \text{CAL FFWEIGUT} \\ \text{EFW} \text{CAL FFWEIGUT} \\ \text{EFW} \text{CAL FFWEIG} \\ CAL FF$
F (140) - F(
IND = 0 FFTM 311 IND = 0 FFTW 312 EVENT 00 = 0X FFTW 313 FVT 40 = NT FFTW 313 FVT 40 = NT FFTW 314 FVM 1 = 0 = 0X FFTW 314 FVM 1 = 0X FFTW 314
60.10 2 10.11 FTERP FILL 11 FTERP FILL 10. NUT = 110. NUT A DOWER OF 2.11 FFILL 311 FFILL 311
ENTRY FF TERP FORMAN 1400 FOX - MIT A POWER OF 201 FORMAN 1400 FOX - MIT A POWER OF 201 FTTW 317 FTTW 317 FTTW 317 FTTW 310 FTTW 310
FUNNI 40 NU 30 FUNNI 40 NU 30 FUNNI 40 NU 310

- 187 -

. .

EERC REPORT LISTING

•

EARTHQUAKE ENGINEERING RESEARCH CENTER REPORTS

NOTE: Numbers in parenthesis are Accession Numbers assigned by the National Technical Information Service; these are followed by a price code. Copies of the reports may be ordered from the National Technical Information Service, 5285 Port Royal Road, Springfield, Virginia, 22161. Accession Numbers should be quoted on orders for reports (PB -----) and remittance must accompany each order. Reports without this information were not available at time of printing. Upon request, EERC will mail inquirers this information when it becomes available.

- EERC 67-1 "Feasibility Study Large-Scale Earthquake Simulator Facility," by J. Penzien, J.G. Bouwkamp, R.W. Clough and D. Rea - 1967 (PB 187 905)A07
- EERC 68-1 Unassigned
- EERC 68-2 "Inelastic Behavior of Beam-to-Column Subassemblages Under Repeated Loading," by V.V. Bertero 1968 (PB 184 888)A05
- EERC 68-3 "A Graphical Method for Solving the Wave Reflection-Refraction Problem," by H.D. McNiven and Y. Mengi-1968 (PB 187 943)A03
- EERC 68-4 "Dynamic Properties of McKinley School Buildings," by D. Rea, J.G. Bouwkamp and R.W. Clough 1968 (PB 187 902)A07
- EERC 68-5 "Characteristics of Rock Motions During Earthquakes," by H.B. Seed, I.M. Idriss and F.W. Kiefer 1968 (PB 188 338)A03
- EERC 69-1 "Earthquake Engineering Research at Berkeley," 1969 (PB 187 906)All
- EERC 69-2 "Nonlinear Seismic Response of Earth Structures," by M. Dibaj and J. Penzien 1969 (PB 187 904)A08
- EERC 69-3 "Probabilistic Study of the Behavior of Structures During Earthquakes," by R. Ruiz and J. Penzien 1969 (PB 187 886)A06
- EERC 69-4 "Numerical Solution of Boundary Value Problems in Structural Mechanics by Reduction to an Initial Value Formulation," by N. Distefano and J. Schujman - 1969 (PB 187 942)A02
- EERC 69-5 "Dynamic Programming and the Solution of the Biharmonic Equation," by N. Distefano 1969 (PB 187 941)A03
- EERC 69-6 "Stochastic Analysis of Offshore Tower Structures," by A.K. Malhotra and J. Penzien 1969 (PB 187 903) A09
- EERC 69-7 "Rock Motion Accelerograms for High Magnitude Earthquakes," by H.B. Seed and I.M. Idriss 1969 (PB 187 940) A02
- EERC 69-8 "Structural Dynamics Testing Facilities at the University of California, Berkeley," by R.M. Stephen, J.G. Bouwkamp, R.W. Clough and J. Penzien - 1969 (PB 189 111)A04
- EERC 69-9 "Seismic Response of Soil Deposits Underlain by Sloping Rock Boundaries," by H. Dezfulian and H.B. Seed 1969 (PB 189 114)A03
- EERC 69-10 "Dynamic Stress Analysis of Axisymmetric Structures Under Arbitrary Loading," by S. Ghosh and E.L. Wilson 1969 (PB 189 026)Al0
- EERC 69-11 "Seismic Behavior of Multistory Frames Designed by Different Philosophies," by J.C. Anderson and V. V. Bertero - 1969 (PB 190 662)Al0
- EERC 69-12 "Stiffness Degradation of Reinforcing Concrete Members Subjected to Cyclic Flexural Moments," by V.V. Bertero, B. Bresler and H. Ming Liao 1969 (PB 202 942)A07
- EERC 69-13 "Response of Non-Uniform Soil Deposits to Travelling Seismic Waves," by H. Dezfulian and H.B. Seed 1969 (PB 191 023)A03
- EERC 69-14 "Damping Capacity of a Model Steel Structure," by D. Rea, R.W. Clough and J.G. Bouwkamp 1969 (PB 190 663) A06
- EERC 69-15 "Influence of Local Soil Conditions on Building Damage Potential during Earthquakes," by H.B. Seed and I.M. Idriss - 1969 (PB 191 036)A03
- EERC 69-16 "The Behavior of Sands Under Seismic Loading Conditions," by M.L. Silver and H.B. Seed 1969 (AD 714 982) A07
- EERC 70-1 "Earthquake Response of Gravity Dams," by A.K. Chopra 1970 (AD 709 640)A03
- EERC 70-2 "Relationships between Soil Conditions and Building Damage in the Caracas Earthquake of July 29, 1967," by H.B. Seed, I.M. Idriss and H. Dezfulian 1970 (PB 195 762)A05
- EERC 70-3 "Cyclic Loading of Full Size Steel Connections," by E.P. Popov and R.M. Stephen 1970 (PB 213 545) A04
- EERC 70-4 "Seismic Analysis of the Charaima Building, Caraballeda, Venezuela," by Subcommittee of the SEAONC Research Committee: V.V. Bertero, P.F. Fratessa, S.A. Mahin, J.H. Sexton, A.C. Scordelis, E.L. Wilson, L.A. Wyllie, H.B. Seed and J. Penzien, Chairman - 1970 (PB 201 455)A06

- EERC 70-5 "A Computer Program for Earthquake Analysis of Dams," by A.K. Chopra and P. Chakrabarti 1970 (AD 723 994) A05
- EERC 70-6 "The Propagation of Love Waves Across Non-Horizontally Layered Structures," by J. Lysmer and L.A. Drake 1970 (PB 197 896)A03
- EERC 70-7 "Influence of Base Rock Characteristics on Ground Response," by J. Lysmer, H.B. Seed and P.B. Schnabel 1970 (PB 197 897)A0.3
- EERC 70-8 "Applicability of Laboratory Test Procedures for Measuring Soil Liquefaction Characteristics under Cyclic Loading," by H.B. Seed and W.H. Peacock - 1970 (PB 198 016)A03
- EERC 70-9 "A Simplified Procedure for Evaluating Soil Liquefaction Potential," by H.B. Seed and I.M. Idriss 1970 (PB 198 009)A03
- EERC 70-10 "Soil Moduli and Damping Factors for Dynamic Response Analysis," by H.B. Seed and I.M. Idriss 1970 (PB 197 869)A03
- EERC 71-1 "Koyna Earthquake of December 11, 1967 and the Performance of Koyna Dam," by A.K. Chopra and P. Chakrabarti 1971 (AD 731 496)A06
- EERC 71-2 "Preliminary In-Situ Measurements of Anelastic Absorption in Soils Using a Prototype Earthquake Simulator," by R.D. Borcherdt and P.W. Rodgers - 1971 (PB 201 454)A03
- EERC 71-3 "Static and Dynamic Analysis of Inelastic Frame Structures," by F.L. Porter and G.H. Powell 1971 (PB 210 135)A06
- EERC 71-4 "Research Needs in Limit Design of Reinforced Concrete Structures," by V.V. Bertero 1971 (PB 202 943)A04
- EERC 71-5 "Dynamic Behavior of a High-Rise Diagonally Braced Steel Building," by D. Rea, A.A. Shah and J.G. Bouwhamp 1971 (PB 203 584)A06
- EERC 71-6 "Dynamic Stress Analysis of Porous Elastic Solids Saturated with Compressible Fluids," by J. Ghaboussi and E. L. Wilson 1971 (PB 211 396)A06
- EERC 71-7 "Inelastic Behavior of Steel Beam-to-Column Subassemblages," by H. Krawinkler, V.V. Bertero and E.P. Popov 1971 (PB 211 335)A14
- EERC 71-8 "Modification of Seismograph Records for Effects of Local Soil Conditions," by P. Schnabel, H.B. Seed and J. Lysmer 1971 (PB 214 450)A03
- EERC 72-1 "Static and Earthquake Analysis of Three Dimensional Frame and Shear Wall Buildings," by E.L. Wilson and H.H. Dovey 1972 (PB 212 904)A05
- EERC 72-2 "Accelerations in Rock for Earthquakes in the Western United States," by P.B. Schnabel and H.B. Seed 1972 (PB 213 100)AC3
- EERC 72-3 "Elastic-Plastic Earthquake Response of Soil-Building Systems," by T. Minami 1972 (PB 214 868)A08
- EERC 72-4 "Stochastic Inelastic Response of Offshore Towers to Strong Motion Earthquakes," by M.K. Kaul 1972 (PB 215 713) A05
- EERC 72-5 "Cyclic Behavior of Three Reinforced Concrete Flexural Members with High Shear," by E.P. Popov, V.V. Bertero and H. Krawinkler - 1972 (PB 214 555)A05
- EERC 72-6 "Earthquake Response of Gravity Dams Including Reservoir Interaction Effects," by P. Chakrabarti and A.K. Chopra 1972 (AD 762 330)A08
- EERC 72-7 "Dynamic Properties of Pine Flat Dam," by D. Rea, C.Y. Liaw and A.K. Chopra-1972 (AD 763 928)A05
- EERC 72-8 "Three Dimensional Analysis of Building Systems," by E.L. Wilson and H.H. Dovey 1972 (PB 222 438) A06
- EERC 72-9 "Rate of Loading Effects on Uncracked and Repaired Reinforced Concrete Members," by S. Mahin, V.V. Bertero, D. Rea and M. Atalay - 1972 (PB 224 520)A08
- EERC 72-10 "Computer Program for Static and Dynamic Analysis of Linear Structural Systems," by E.L. Wilson, K.-J. Bathe, J.E. Peterson and H.H.Dovey 1972 (PB 220 437)A04
- EERC 72-11 "Literature Survey Seismic Effects on Highway Bridges," by T. Iwasaki, J. Penzien and R.W. Clough 1972 (PB 215 613)A19
- EERC 72-12 "SHAKE-A Computer Program for Earthquake Response Analysis of Horizontally Layered Sites," by P.B. Schnabel and J. Lysmer - 1972 (PB 220 207)A06
- EERC 73-1 "Optimal Seismic Design of Multistory Frames," by V.V. Bertero and H. Kamil-1973
- EERC 73-2 "Analysis of the Slides in the San Fernando Dams During the Earthquake of February 9, 1971." by H.B. Seed, K.L. Lee, I.M. Idriss and F. Makdisi - 1973 (PB 223 402)Al4

- EERC 73-3 "Computer Aided Ultimate Load Design of Unbraced Multistory Steel Frames," by M.B. El-Hafez and G.H. Powell 1973 (PB 248 315)A09
- EERC 73,-4 "Experimental Investigation into the Seismic Behavior of Critical Regions of Reinforced Concrete Components as Influenced by Moment and Shear," by M. Celebi and J. Penzien - 1973 (PB 215 884)A09
- EERC 73-5 "Hysteretic Behavior of Epoxy-Repaired Reinforced Concrete Beams," by M. Celebi and J. Penzien 1973 (PB 239 568)A03
- EERC 73-6 "General Purpose Computer Program for Inelastic Dynamic Response of Plane Structures," by A. Kanaan and G.H. Powell 1973 (PB 221 260)A08
- EERC 73-7 "A Computer Program for Earthquake Analysis of Gravity Dams Including Reservoir Interaction," by P. Chakrabarti and A.K. Chopra - 1973 (AD 766 271)A04
- EERC 73-8 "Behavior of Reinforced Concrete Deep Beam-Column Subassemblages Under Cyclic Loads," by O. Küstü and J.G. Bouwkamp 1973 (PB 246 117)Al2
- EERC 73-9 "Earthquake Analysis of Structure-Foundation Systems," by A.K. Vaish and A.K. Chopra 1973 (AD 766 272)A07
- EERC 73-10 "Deconvolution of Seismic Response for Linear Systems," by R.E. Reimer 1973 (PB 227 179)A08
- EERC 73-11 "SAP IV: A Structural Analysis Program for Static and Dynamic Response of Linear Systems," by K.-J. Bathe, E.L. Wilson and F.E. Peterson - 1973 (PB 221 967)A09
- EERC 73-12 "Analytical Investigations of the Seismic Response of Long, Multiple Span Highway Bridges," by W.S. Tseng and J. Penzien - 1973 (PB 227 816)AlC
- EERC 73-13 "Earthquake Analysis of Multi-Story Buildings Including Foundation Interaction," by A.K. Chopra and J.A. Gutierrez 1973 (PB 222 970)A03
- EERC 73-14 "ADAP: A Computer Program for Static and Dynamic Analysis of Arch Dams," by R.W. Clough, J.M. Raphael and S. Mojtahedi 1973 (PB 223 763)A09
- EERC 73-15 "Cyclic Plastic Analysis of Structural Steel Joints," by R.B. Pinkney and R.W. Clough 1973 (PE 226 843) A08
- EERC 73-16 "QUAD-4: A Computer Program for Evaluating the Seismic Response of Soil Structures by Variable Damping Finite Element Procedures," by I.M. Idriss, J. Lysmer, R. Hwang and H.B. Seed - 1973 (PB 229 424)A05
- EERC 73-17 "Dynamic Schavior of a Multi-Story Pyramid Shaped Building," by R.M. Stephen, J.P. Hollings and J.G. Bouwkamp 1973 (PB 240 718)A06
- EERC 73-18 "Effect of Different Types of Reinforcing on Seismic Behavior of Short Concrete Columns," by V.V. Bertero, J. Hollings, O. Küstü, R.M. Stephen and J.G. Bouwkamp - 1973
- EERC 73-19 "Olive View Medical Center Materials Studies, Phase I," by B. Bresler and V.V. Bertero 1973 (PB 235 986)A06
- EERC 73-20 "Linear and Nonlinear Seismic Analysis Computer Programs for Long Multiple-Span Highway Bridges," by W.S. Tseng and J. Penzien 1973
- EERC 73-21 "Constitutive Models for Cyclic Plastic Deformation of Engineering Materials," by J.M. Kelly and P.P. Gillis 1973 (PB 226 024)A03
- EERC 73-22 "DRAIN 2D User's Guide," by G.H. Powell 1973 (PB 227 016)A05
- EERC 73-23 "Earthquake Engineering at Berkeley 1973," (PB 226 033)All
- EERC 73-24 Unassigned
- EERC 73-25 "Earthquake Response of Axisymmetric Tower Structures Surrounded by Water," by C.Y. Liaw and A.K. Chopra 1973 (AD 773 052)A09
- EERC 73-26 "Investigation of the Failures of the Olive View Stairtowers During the San Fernando Earthquake and Their Implications on Seismic Design," by V.V. Bertero and R.G. Collins - 1973 (PB 235 106)A13
- EERC 73-27 "Further Studies on Seismic Behavior of Steel Beam-Column Subassemblages," by V.V. Bertero, H. Krawinkler and E.P. Popov - 1973 (PB 234 172)A06
- EERC 74-1 "Seismic Risk Analysis," by C.S. Oliveira 1974 (PB 235 920)A06
- EERC 74-2 "Settlement and Liquefaction of Sands Under Multi-Directional Shaking," by R. Pyke, C.K. Chan and H.B. Seed 1974
- EERC 74-3 "Optimum Design of Earthquake Resistant Shear Buildings," by D. Ray, K.S. Pister and A.K. Chopra-1974 (PB 231 172)A06
- EERC 74-4 "LUSH A Computer Program for Complex Response Analysis of Soil-Structure Systems," by J. Lysmer, T. Udaka, H.B. Seed and R. Hwang - 1974 (PB 236 796)A05

- EERC 74-5 "Sensitivity Analysis for Hysteretic Dynamic Systems: Applications to Earthquake Engineering," by D. Ray 1974 (PB 233 213)A06
- EERC 74-6 "Soil Structure Interaction Analyses for Evaluating Seismic Response," by H.B. Seed, J. Lysmer and R. Hwang 1974 (PB 236 519)A04
- EERC 74-7 Unassigned
- EERC 74-8 "Shaking Table Tests of a Steel Frame A Progress Report," by R.W. Clough and D. Tang-1974 (PB 240 869)A03
- EERC 74-9 "Hysteretic Behavior of Reinforced Concrete Flexural Members with Special Web Reinforcement," by V.V. Bertero, E.P. Popov and T.Y. Wang 1974 (PB 236 797)A07
- EERC 74-10 "Applications of Reliability-Based, Global Cost Optimization to Design of Earthquake Resistant Structures," by E. Vitiello and K.S. Pister - 1974 (PB 237 231)A06
- EERC 74-11 "Liquefaction of Gravelly Soils Under Cyclic Loading Conditions," by R.T. Wong, H.B. Seed and C.K. Chan 1974 (PB 242 042)A03
- EERC 74-12 "Site-Dependent Spectra for Earthquake-Resistant Design," by H.B. Seed, C. Ugas and J. Lysmer 1974 (PB 240 953)A03
- EERC 74-13 "Earthquake Simulator Study of a Reinforced Concrete Frame," by P. Hidalgo and R.W. Clough 1974 (PB 241 944)Al3
- EERC 74-14 "Nonlinear Earthquake Response of Concrete Gravity Dams," by N. Pal 1974 (AD/A 006 583)A06
- EERC 74-15 "Modeling and Identification in Nonlinear Structural Dynamics I. One Degree of Freedom Models," by N. Distefano and A. Rath 1974 (PB 241 548)A06
- EERC 75-1 "Determination of Seismic Design Criteria for the Dumbarton Bridge Replacement Structure, Vol.I: Description, Theory and Analytical Modeling of Bridge and Parameters," by F. Baron and S.-H. Pang - 1975 (PB 259 407)Al5
- EERC 75-2 "Determination of Seismic Design Criteria For the Dumbarton Bridge Replacement Structure, Vol. II: Numerical Studies and Establishment of Seismic Design Criteria," by F. Baron and S.-H. Pang-1975 (PB 259 408)All (For set of EERC 75-1 and 75-2 (PB 259 406))
- EERC 75-3 "Seismic Risk Analysis for a Site and a Metropolitan Area," by C.S. Oliveira 1975 (PB 248 134)A09
- EERC 75-4 "Analytical Investigations of Seismic Response of Short, Single or Multiple-Span Highway Bridges," by M.-C. Chen and J. Penzien - 1975 (PB 241 454)A09
- EERC 75-5 "An Evaluation of Some Methods for Predicting Seismic Behavior of Reinforced Concrete Buildings," by S.A. Mahin and V.V. Bertero - 1975 (PB 246 306)Al6
- EERC 75-6 "Earthquake Simulator Study of a Steel Frame Structure, Vol. I: Experimental Results," by R.W. Clough and D.T. Tang 1975 (PB 243 981)A13
- EERC 75-7 "Dynamic Properties of San Bernardino Intake Tower," by D. Rea, C.-Y. Liaw and A.K. Chopra-1975 (AD/A008 406) A05
- EERC 75-8 "Seismic Studies of the Articulation for the Dumbarton Bridge Replacement Structure, Vol. I: Description, Theory and Analytical Modeling of Bridge Components," by F. Baron and R.E. Hamati - 1975 (PB 251 539)A07
- EERC 75-9 "Seismic Studies of the Articulation for the Dumbarton Bridge Replacement Structure, Vol. 2: Numerical Studies of Steel and Concrete Girder Alternates," by F. Baron and R.E. Hamati - 1975 (PB 251 540)Al0
- EERC 75-10 "Static and Dynamic Analysis of Nonlinear Structures," by D.P. Mondkar and G.H. Powell 1975 (PB 242 434)A08
- EERC 75-11 "Hysteretic Behavior of Steel Columns," by E.P. Popov, V.V. Bertero and S. Chandramouli 1975 (PB 252 365) All
- EERC 75-12 "Earthquake Engineering Research Center Library Printed Catalog," 1975 (PB 243 711)A26
- EERC 75-13 "Three Dimensional Analysis of Building Systems (Extended Version)," by E.L. Wilson, J.P. Hollings and H.H. Dovey - 1975 (PB 243 989)A07
- EERC 75-14 "Determination of Soil Liquefaction Characteristics by Large-Scale Laboratory Tests," by P. De Alba, C.K. Chan and H.B. Seed - 1975 (NUREG 0027) A08
- EERC 75-15 "A Literature Survey Compressive, Tensile, Bond and Shear Strength of Masonry," by R.L. Mayes and R.W.
 Clough 1975 (PB 246 292)Al0
- EERC 75-16 "Hysteretic Behavior of Ductile Moment Resisting Reinforced Concrete Frame Components," by V.V. Bertero and E.P. Popov 1975 (PB 246 388)A05
- EERC 75-17 "Relationships Between Maximum Acceleration, Maximum Velocity, Distance from Source, Local Site Conditions for Moderately Strong Earthquakes," by H.B. Seed, R. Murarka, J. Lysmer and I.M. Idriss - 1975 (PB 248 172)A03
- EERC 75-18 "The Effects of Method of Sample Preparation on the Cyclic Stress-Strain Behavior of Sands," by J. Mulilis, C.K. Chan and H.B. Seed - 1975 (Summarized in EERC 75-28)

- EERC 75-19 "The Seismic Behavior of Critical Regions of Reinforced Concrete Components as Influenced by Moment, Shear and Axial Force," by M.B. Atalay and J. Penzien - 1975 (PB 258 842)All
- EERC 75-20 "Dynamic Properties of an Eleven Story Masonry Building," by R.M. Stephen, J.P. Hollings, J.G. Bouwkamp and D. Jurukovski 1975 (PB 246 945)A04
- EERC 75-21 "State-of-the-Art in Seismic Strength of Masonry An Evaluation and Review," by R.L. Mayes and R.W. Clough 1975 (PB 249 040)A07
- EERC 75-22 "Frequency Dependent Stiffness Matrices for Viscoelastic Half-Plane Foundations," by A.K. Chopra, P. Chakrabarti and G. Dasgupta - 1975 (PB 248 121)A07
- EERC 75-23 "Hysteretic Behavior of Reinforced Concrete Framed Walls," by T.Y. Wong, V.V. Bertero and E.P. Popov 1975
- EERC 75-24 "Testing Facility for Subassemblages of Frame-Wall Structural Systems," by V.V. Bertero, E.P. Popov and T. Endo 1975
- EERC 75-25 "Influence of Seismic History on the Liquefaction Characteristics of Sands," by H.B. Seed, K. Mori and C.K. Chan-1975 (Summarized in EERC 75-28)
- EERC 75-26 "The Generation and Dissipation of Pore Water Pressures during Soil Liquefaction," by H.B. Seed, P.P. Martin and J. Lysmer - 1975 (PB 252 648)A03
- EERC 75-27 "Identification of Research Needs for Improving Aseismic Design of Building Structures," by V.V. Bertero 1975 (PB 248 136)A05
- EERC 75-28 "Evaluation of Soil Liquefaction Potential during Earthquakes," by H.B. Seed, I. Arango and C.K. Chan-1975 (NUREG 0026)A13
- EERC 75-29 "Representation of Irregular Stress Time Histories by Equivalent Uniform Stress Series in Liquefaction Analyses," by H.B. Seed, I.M. Idriss, F. Makdisi and N. Banerjee - 1975 (PB 252 635)A03
- EERC 75-30 "FLUSH A Computer Program for Approximate 3-D Analysis of Soil-Structure Interaction Problems," by J. Lysmer, T. Udaka, C.-F. Tsai and H.B. Seed 1975 (PB 259 332)A07
- EERC 75-31 "ALUSH A Computer Program for Seismic Response Analysis of Axisymmetric Soil-Structure Systems," by E. Berger, J. Lysmer and H.E. Seed 1975
- EERC 75-32 "TRIP and TRAVEL Computer Programs for Soil-Structure Interaction Analysis with Horizontally Travelling Waves," by T. Udaka, J. Lysmer and H.B. Seed 1975
- EERC 75-33 "Predicting the Performance of Structures in Regions of High Seismicity," by J. Penzien 1975 (PB 248 130)A03
- EERC 75-34 "Efficient Finite Element Analysis of Seismic Structure Soil Direction," by J. Lysmer, H.B. Seed, T. Udaka, R.N. Hwang and C.-F. Tsai - 1975 (PB 253 570)A03
- EERC 75-35 "The Dynamic Behavior of a First Story Girder of a Three-Story Steel Frame Subjected to Earthquake Loading," by R.W. Clough and L.-Y. Li - 1975 (PE 248 841)A05
- EERC 75-36 "Earthquake Simulator Study of a Steel Frame Structure, Volume II Analytical Results," by D.T. Tang-1975 (PB 252 926)Al0
- EERC 75-37 "ANSR-I General Purpose Computer Program for Analysis of Non-Linear Structural Response," by D.P. Mondkar and G.H. Powell - 1975 (PB 252 386)A08
- EERC 75-33 "Nonlinear Response Spectra for Probabilistic Seismic Design and Damage Assessment of Reinforced Concrete Structures," by M. Murakami and J. Penzien - 1975 (PB 259 530)A05
- EERC 75-39 "Study of a Method of Feasible Directions for Optimal Elastic Design of Frame Structures Subjected to Earthquake Loading," by N.D. Walker and K.S. Pister - 1975 (PB 257 781)A06
- EERC 75-40 "An Alternative Representation of the Elastic-Viscoelastic Analogy," by G. Dasgupta and J.L. Sackman 1975 (PB 252 173)A03
- EERC 75-41 "Effect of Multi-Directional Shaking on Liquefaction of Sands," by H.B. Seed, R. Pyke and G.R. Martin 1975 (PB 258 781)A03
- EERC 76-1 "Strength and Ductility Evaluation of Existing Low-Rise Reinforced Concrete Buildings Screening Method," by T. Okada and B. Bresler - 1976 (PB 257 906)All
- EERC 76-2 "Experimental and Analytical Studies on the Hysteretic Behavior of Reinforced Concrete Rectangular and T-Beams," by S.-Y.M. Ma, E.P. Popov and V.V. Bertero 1976 (PE 260 843)A12
- EERC 76-3 "Dynamic Behavior of a Multistory Triangular-Shaped Building," by J. Petrovski, R.M. Stephen, E. Gartenbaum and J.G. Bouwkamp = 1976 (PB 273 279)A07
- EERC 76-4 "Earthquake Induced Deformations of Earth Dams," by N. Serff, H.B. Seed, F.I. Makdisi & C.-Y. Chang 1976 (PB 292 065)A08

- EERC 76-5 "Analysis and Design of Tube-Type Tall Building Structures," by H. de Clercq and G.H. Powell 1976 (FB 252 220) Alo
- EERC 76-6 "Time and Frequency Domain Analysis of Three-Dimensional Ground Motions, San Fernando Earthquake," by T. Kubo and J. Penzien (PB 260 556)All
- EERC 76-7 "Expected Performance of Uniform Building Code Design Masonry Structures," by R.L. Mayes, Y. Omote, S.W. Chen and R.W. Clough - 1976 (PB 270 098)A05
- EERC 76-8 "Cyclic Shear Tests of Masonry Piers, Volume 1 Test Results," by R.L. Mayes, Y. Omote, R.W. Clough 1976 (PB 264 424)A06
- EERC 76-9 "A Substructure Method for Earthquake Analysis of Structure Soil Interaction," by J.A. Gutierrez and A.K. Chopra 1976 (PB 257 783)A08
- EERC 76-10 "Stabilization of Potentially Liquefiable Sand Deposits using Gravel Drain Systems," by H.B. Seed and J.R. Booker-1976 (PB 258 820)A04
- EERC 76-11 "Influence of Design and Analysis Assumptions on Computed Inelastic Response of Moderately Tall Frames," by G.H. Powell and D.G. Row 1976 (PB 271 409)A06
- EERC 76-12 "Sensitivity Analysis for Hysteretic Dynamic Systems: Theory and Applications," by D. Ray, K.S. Pister and E. Polak - 1976 (PB 262 859)A04
- EERC 76-13 "Coupled Lateral Torsional Response of Buildings to Ground Shaking," by C.L. Kan and A.K. Chopra 1976 (PB 257 907)A09
- EERC 76-14 "Seismic Analyses of the Banco de America," by V.V. Bertero, S.A. Mahin and J.A. Hollings 1976
- EERC 76-15 "Reinforced Concrete Frame 2: Seismic Testing and Analytical Correlation," by R.W. Clough and J. Gidwani 1976 (PB 261 323)A08
- EERC 76-16 "Cyclic Shear Tests of Masonry Piers, Volume 2 + Analysis of Test Results," by R.L. Mayes, Y. Omote and R.W. Clough - 1976
- EERC 76-17 "Structural Steel Bracing Systems: Behavior Under Cyclic Loading," by E.P. Popov, K. Takanashi and C.W. Roeder 1976 (PB 260 715)A05
- EERC 76-18 "Experimental Model Studies on Seismic Response of High Curved Overcrossings," by D. Williams and W.G. Godden 1976 (FB 269 548)A08
- EERC 76-19 "Effects of Non-Uniform Seismic Disturbances on the Dumbarton Bridge Replacement Structure," by . F. Baron and R.E. Hamatj - 1976 (PB 282 981)A16
- EERC 76-20 "Investigation of the Inelastic Characteristics of a Single Story Steel Structure Using System Identification and Shaking Table Experiments," by V.C. Matzen and H.D. McNiven - 1976 (PB 258 453)A07
- EERC 76-21 "Capacity of Columns with Splice Imperfections," by E.P. Popov, R.M. Stephen and R. Philbrick 1976 (PB 260 378)A04
- EERC 76-22 "Response of the Olive View Hospital Main Building during the San Fernando Earthquake," by S. A. Mahin, V.V. Bertero, A.K. Chopra and R. Collins - 1976 (PB 271 425)A14
- EERC 76-23 "A Study on the Major Factors Influencing the Strength of Masonry Prisms," by N.M. Mostaghel, R.L. Mayes, R. W. Clough and S.W. Chen - 1976 (Not published)
- EERC 76-24 "GADFLEA A Computer Program for the Analysis of Pore Pressure Generation and Dissipation during Cyclic or Earthquake Loading," by J.R. Booker, M.S. Rahman and H.B. Seed - 1976 (PB 263 947)A04
- EERC 76-25 "Seismic Safety Evaluation of a R/C School Building," by B. Bresler and J. Axley 1976
- EERC 76-26 "Correlative Investigations on Theoretical and Experimental Dynamic Behavior of a Model Bridge Structure," by K. Kawashima and J. Penzien 1976 (PB 263 388)All
- EERC 76-27 "Earthquake Response of Coupled Shear Wall Buildings," by T. Srichatrapimuk 1976 (PB 265 157)A07
- EERC 76-28 "Tensile Capacity of Partial Penetration Welds," by E.P. Popov and R.M. Stephen 1976 (PB 262 899) A03
- EERC 76-29 "Analysis and Design of Numerical Integration Methods in Structural Dynamics," by H.M. Hilber 1976 (PB 264 410)A06
- EERC 76-30 "Contribution of a Floor System to the Dynamic Characteristics of Reinforced Concrete Buildings," by L.E. Malik and V.V. Bertero 1976 (PB 272 247)Al3
- EERC 76-31 "The Effects of Seismic Disturbances on the Golden Gate Bridge," by F. Baron, M. Arikan and R.E. Hamati 1976 (PB 272 279)A09
- EERC 76-32 "Infilled Frames in Earthquake Resistant Construction," by R.E. Klingner and V.V. Bertero 1976 (PB 265 892)Al3

- UCB/EERC-77/01 "PLUSH A Computer Program for Probabilistic Finite Element Analysis of Seismic Soil-Structure Interaction," by M.P. Romo Organista, J. Lysmer and H.B. Seed - 1977
- UCB/EERC-77/02 "Soil-Structure Interaction Effects at the Humboldt Bay Power Plant in the Ferndale Earthquake of June 7, 1975," by J.E. Valera, H.B. Seed, C.F. Tsai and J. Lysmer 1977 (PB 265 795)A04
- UCB/EERC-77/03 "Influence of Sample Disturbance on Sand Response to Cyclic Loading," by K. Mori, H.B. Seed and C.K. Chan - 1977 (PB 267 352)A04
- UCB/EERC-77/04 "Seismological Studies of Strong Motion Records," by J. Shoja-Taheri 1977 (PB 269 655)Al0

UCB/EERC-77/05 "Testing Facility for Coupled-Shear Walls," by L. Li-Hyung, V.V. Bertero and E.P. Popov - 1977

- UCB/EERC-77/06 "Developing Methodologies for Evaluating the Earthquake Safety of Existing Buildings," by No. 1 -B. Bresler; No. 2 - B. Bresler, T. Okada and D. Zisling; No. 3 - T. Okada and B. Bresler; No. 4 - V.V. Bertero and B. Bresler - 1977 (PB 267 354)A08
- UCB/EERC-77/07 "A Literature Survey Transverse Strength of Masonry Walls," by Y. Omote, R.L. Mayes, S.W. Chen and R.W. Clough 1977 (PB 277 933)A07
- UCB/EERC-77/08 "DRAIN-TABS: A Computer Program for Inelastic Earthquake Response of Three Dimensional Buildings," by R. Guendelman-Israel and G.H. Powell 1977 (PB 270 693)A07
- UCB/EERC-77/09 "SUBWALL: A Special Purpose Finite Element Computer Program for Practical Elastic Analysis and Design of Structural Walls with Substructure Option," by D.Q. Le, H. Peterson and E.P. Popov - 1977 (PB 270 567)A05
- UCB/EERC-77/10 "Experimental Evaluation of Seismic Design Methods for Broad Cylindrical Tanks," by D.P. Clough (PB 272 280)Al3
- UCB/EERC-77/11 "Earthquake Engineering Research at Berkeley 1976," 1977 (PB 273 507)A09
- UCB/EERC-77/12 "Automated Design of Earthquake Resistant Multistory Steel Building Frames," by N.D. Walker, Jr. 1977 (PB 276 526)A09
- UCB/EERC-77/13 "Concrete Confined by Rectangular Hoops Subjected to Axial Loads," by J. Vallenas, V.V. Bertero and E.P. Popov - 1977 (PB 275 165)A06
- UCB/EERC-77/14 "Seismic Strain Induced in the Ground During Earthquakes," by Y. Sugimura 1977 (PE 284 201)A04
- UCB/EERC-77/15 "Bond Deterioration under Generalized Loading," by V.V. Bertero, E.P. Popov and S. Viwathanatepa 1977
- UCB/EERC-77/16 "Computer Aided Optimum Design of Ductile Reinforced Concrete Moment Resisting Frames," by S.W. Zagajeski and V.V. Bertero - 1977 (PB 280 137)A07
- UCB/EERC-77/17 "Earthquake Simulation Testing of a Stepping Frame with Energy-Absorbing Devices," by J.M. Kelly and D.F. Tsztoo 1977 (PB 273 506)A04
- UCB/EERC-77/18 "Inelastic Behavior of Eccentrically Braced Steel Frames under Cyclic Loadings," by C.W. Roeder and E.P. Popov - 1977 (PB 275 526)A15
- UCB/EERC-77/19 "A Simplified Procedure for Estimating Earthquake-Induced Deformations in Dams and Embankments," by F.I. Makdisi and H.B. Seed - 1977 (PB 276 820)A04
- UCB/EERC-77/20 "The Performance of Earth Dams during Earthquakes," by H.B. Seed, F.I. Makdisi and P. de Alba 1977 (PB 276 821)A04
- UCB/EERC-77/21 "Dynamic Plastic Analysis Using Stress Resultant Finite Element Formulation," by P. Lukkunapvasit and J.M. Kelly 1977 (PB 275 453)A04
- UCB/EERC-77/22 "Preliminary Experimental Study of Seismic Uplift of a Steel Frame," by R.W. Clough and A.A. Huckelbridge 1977 (PB 278 769)A08
- UCB/EERC-77/23 "Earthquake Simulator Tests of a Nine-Story Steel Frame with Columns Allowed to Uplift," by A.A. Huckelbridge - 1977 (PB 277 944)A09
- UCB/EERC-77/24 "Nonlinear Soil-Structure Interaction of Skew Highway Bridges," by M.-C. Chen and J. Penzien 1977 (PB 276 176)A07
- UCB/EERC-77/25 "Seismic Analysis of an Offshore Structure Supported on Pile Foundations," by D.D.-N. Liou and J. Penzien 1977 (PB 283 180)A06
- UCB/EERC-77/26 "Dynamic Stiffness Matrices for Homogeneous Viscoelastic Half-Planes," by G. Dasgupta and A.K. Chopra -1977 (PB 279 654)A06
- UCB/EERC-77/27 "A Practical Soft Story Earthquake Isolation System," by J.M. Kelly, J.M. Eidinger and C.J. Derham -1977 (PB 276 814)A07
- UCB/EERC-77/28 "Seismic Safety of Existing Buildings and Incentives for Hazard Mitigation in San Francisco: An Exploratory Study," by A.J. Meltsner - 1977 (PB 281 970)A05
- UCB/EERC-77/29 "Dynamic Analysis of Electrohydraulic Shaking Tables," by D. Rea, S. Abedi-Hayati and Y. Takahashi 1977 (PB 282 569)A04
- UCB/EERC-77/30 "An Approach for Improving Seismic Resistant Behavior of Reinforced Concrete Interior Joints," by B. Galunic, V.V. Bertero and E.P. Popov - 1977 (PB 290 870)A06

UCB/EERC-78/01 "The Development of Energy-Absorbing Devices for Aseismic Base Isolation Systems," by J.M. Kelly and D.F. Tsztoo - 1978 (PB 284 978)A04 UCB/EERC-78/02 "Effect of Tensile Prestrain on the Cyclic Response of Structural Steel Connections, by J.G. Bouwkamp and A. Mukhopadhvav - 1978 UCB/EERC-78/03 "Experimental Results of an Earthquake Isolation System using Natural Rubber Bearings," by J.M. Eidinger and J.M. Kelly - 1978 (PB 281 686) A04 "Seismic Behavior of Tall Liquid Storage Tanks," by A. Niwa - 1978 (PB 284 017)Al4 UCB/EERC-78/04 UCB/EERC+78/05 "Hysteretic Behavior of Reinforced Concrete Columns Subjected to High Axial and Cyclic Shear Forces," by S.W. Zagajeski, V.V. Bertero and J.G. Bouwkamp - 1978 (PB 283 858)A13 UCB/EERC-78/06 "Inelastic Beam-Column Elements for the ANSR-I Program," by A. Riahi, D.G. Row and G.H. Powell - 1978 UCB/EERC-78/07 "Studies of Structural Response to Earthquake Ground Motion," by O.A. Lopez and A.K. Chopra - 1978 (PB 282 790)A05 UCB/EERC-78/08 "A Laboratory Study of the Fluid-Structure Interaction of Submerged Tanks and Caissons in Earthquakes," by R.C. Byrd - 1978 (PB 284 957) A08 UCB/EERC-78/09 "Model for Evaluating Damageability of Structures," by I. Sakamoto and B. Bresler - 1978 UCB/EERC-78/10 "Seismic Performance of Nonstructural and Secondary Structural Elements," by I. Sakamoto - 1978 UCB/EEBC=78/11 "Mathematical Modelling of Hysteresis Loops for Reinforced Concrete Columns," by S. Nakata, T. Sproul and J. Penzien - 1978 UCB/EERC-78/12 "Damageability in Existing Buildings," by T. Blejwas and B. Bresler - 1978 "Dynamic Behavior of a Pedestal Base Multistory Building," by R.M. Stephen, E.L. Wilson, J.G. Bouwkamp UCB/EEBC-78/13 and M. Button - 1978 (PB 286 650)A08 UCB/EERC-78/14 "Seismic Response of Bridges - Case Studies," by R.A. Imbsen, V. Nutt and J. Penzien - 1978 (PB 286 503)A10 UCB/EERC-78/15 "A Substructure Technique for Nonlinear Static and Dynamic Analysis," by D.G. Row and G.H. Powell -1978 (PB 288 077)ALO UCB/EERC-78/16 "Seismic Risk Studies for San Francisco and for the Greater San Francisco Bay Area," by C.S. Oliveira -1978 UCB/EERC-78/17 "Strength of Timber Roof Connections Subjected to Cyclic Loads," by P. Gülkan, R.L. Mayes and R.W. Clough - 1978 UCB/EERC-78/18 "Response of K-Braced Steel Frame Models to Lateral Loads," by J.G. Bouwkamp, R.M. Stephen and E.P. Popov - 1978 "Rational Design Methods for Light Equipment in Structures Subjected to Ground Motion," by UCB/EERC-78/19 J.L. Sackman and J.M. Kelly - 1978 (PB 292 357)A04 "Testing of a Wind Restraint for Aseismic Base Isolation," by J.M. Kelly and D.E. Chitty - 1978 UCB/EERC-78/20 (PB 292 833)A03 UCB/EERC-78/21 "APOLLO - A Computer Program for the Analysis of Pore Pressure Generation and Dissipation in Horizontal Sand Layers During Cyclic or Earthquake Loading," by P.P. Martin and H.B. Seed - 1978 (PB 292 835)A04 UCB/EERC-78/22 "Optimal Design of an Earthquake Isolation System," by M.A. Bhatti, K.S. Pister and E. Polak - 1978 (PB 294 735) A06 UCB/EERC-78/23 "MASH - A Computer Program for the Non-Linear Analysis of Vertically Propagating Shear Waves in Horizontally Layered Deposits," by P.P. Martin and H.B. Seed - 1978 (PB 293 101)A05 "Investigation of the Elastic Characteristics of a Three Story Steel Frame Using System Identification," UCB/EERC-78/24 by I. Kaya and H.D. McNiven - 1978 "Investigation of the Nonlinear Characteristics of a Three-Story Steel Frame Using System Identification," by I. Kaya and H.D. McNiven - 1978 UCB/EERC-78/25 UCB/EERC-78/26 "Studies of Strong Ground Motion in Taiwan," by Y.M. Hsiung, B.A. Bolt and J. Penzien - 1978 UCB/EERC-78/27 "Cyclic Loading Tests of Masonry Single Piers: Volume 1 - Height to Width Ratio of 2," by P.A. Hidalgo, R.L. Mayes, H.D. McNiven and R.W. Clough - 1978 UCB/EERC-78/28 "Cyclic Loading Tests of Masonry Single Piers: Volume 2 - Height to Width Ratio of 1," by S.-W.J. Chen, P.A. Hidalgo, R.L. Mayes, R.W. Clough and H.D. McNiven + 1978 UCB/EERC-78/29 "Analytical Procedures in Soil Dynamics," by J. Lysmer - 1978

EERC - 8

- UCB/EERC-79/01 "Hysteretic Behavior of Lightweight Reinforced Concrete Beam-Column Subassemblages," by B. Forzani, E.P. Popov, and V.V. Bertero - 1979
- UCB/EERC-79/02 "The Development of a Mathematical Model to Predict the Flexural Response of Reinforced Concrete Beams to Cyclic Loads, Using System Identification," by J.F. Stanton and H.D. McNiven - 1979
- UCB/EERC-79/03 "Linear and Nonlinear Earthquake Response of Simple Torsionally Coupled Systems," by C.L. Kan and A.K. Chopra - 1979
- UCB/EERC-79/04 "A Mathematical Model of Masonry for Predicting Its Linear Seismic Response Characteristics," by Y. Mengi and H.D. McNiven - 1979
- UCB/EERC-79/05 "Mechanical Behavior of Light Weight Concrete Confined by Different Types of Lateral Reinforcement," by M.A. Manrique and V.V. Bertero - 1979
- UCB/EERC-79/06 "Static Tilt Tests of a Tall Cylindrical Liquid Storage Tank," by R.W. Clough and A. Niwa - 1979
- UCB/EERC-79/07 "The Design of Steel Energy Absorbing Restrainers and Their Incorporation Into Nuclear Power Plants for Enhanced Safety: Volume 1 - Summary Report," by P.N. Spencer, V.F. Zackay, and E.R. Parker - 1979
- UCB/EERC-79/08 "The Design of Steel Energy Absorbing Restrainers and Their Incorporation Into Nuclear Power Plants for Enhanced Safety: Volume 2 - The Development of Analyses for Reactor System Piping," "Simple Systems" by M.C. Lee, J. Penzien, A.K. Chopra, and K. Suzuki "Complex Systems" by G.H. Powell, E.L. Wilson, R.W. Clough and D.G. Row - 1979
- UCB/EERC-79/09 "The Design of Steel Energy Absorbing Restrainers and Their Incorporation Into Nuclear Power Plants for Enhanced Safety: Volume 3 - Evaluation of Commerical Steels," by W.S. Owen, R.M.N. Pelloux, R.O. Ritchie, M. Faral, T. Ohhashi, J. Toplosky, S.J. Hartman, V.F. Zackay, and E.R. Parker - 1979

UCB/EERC-79/10 "The Design of Steel Energy Absorbing Restrainers and Their Incorporation Into Nuclear Power Plants for Enhanced Safety: Volume 4 - A Review of Energy-Absorbing Devices," by J.M. Kelly and M.S. Skinner - 1979

UCB/EERC-79/11 "Conservatism In Summation Rules for Closely Spaced Modes," by J.M. Kelly and J.L. Sackman ~ 1979

- UCB/EERC-79/12 "Cyclic Loading Tests of Masonry Single Piers Volume 3 - Height to Width Ratio of 0.5," by P.A. Hidalgo, R.L. Mayes, H.D. McNiven and R.W. Clough - 1979
- UCB/EERC-79/13 "Cyclic Behavior of Dense Coarse-Grain Materials in Relation to the Seismic Stability of Dams," by N.G. Banerjee, H.B. Seed and C.K. Chan - 1979
- UCB/EERC-79/14 "Seismic Behavior of R/C Interior Beam Column Subassemblages," by S. Viwathanatepa, E.P. Popov and V.V. Bertero - 1979
- UCB/EERC-79/15 "Optimal Design of Localized Nonlinear Systems with Dual Performance Criteria Under Earthquake Excitations," by M.A. Bhatti - 1979
- UCB/EERC-79/16 "OPTDYN A General Purpose Optimization Program for Problems with or without Dynamic Constraints," by M.A. Bhatti, E. Polak and K.S. Pister - 1979
- UCB/EERC-79/17 "ANSR-II, Analysis of Nonlinear Structural Response, Users Manual," by D.P. Mondkar and G.H. Powell - 1979
- UCB/EERC-79/18 "Soil Structure Interaction in Different Seismic Environments," A. Gomez-Masso, J. Lysmer, J.-C. Chen and H.B. Seed - 1979
- UCB/EERC-79/19 "ARMA Models for Earthquake Ground Motions," by M.K. Chang, J.W. Kwiatkowski, R.F. Nau, R.M. Oliver and K.S. Pister - 1979
- UCB/EERC-79/20 "Hysteretic Behavior of Reinforced Concrete Structural Walls," by J.M. Vallenas, V.V. Bertero and E.P. Popov - 1979
- UCB/EERC-79/21 "Studies on High-Frequency Vibrations of Buildings I: The Column Effects," by J. Lubliner - 1979

UCB/EERC-79/22 "Bond Deterioration of Reinforcing Bars Embedded in Confined Concrete Blocks," by S. Viwathanatepa, E.P. Popov and V.V. Bertero - 1979

UCB/EERC-79/23 "Shaking Table Study of Single-Story Masonry Houses, Volume 1: Test Structures 1 and 2," by P. Gülkan, R.L. Mayes and R.W. Clough - 1979

- UCB/EERC-79/24 "Shaking Table Study of Single-Story Masonry Houses, Volume 2: Test Structures 3 and 4," by P. Gülkan, R.L. Mayes and R.W. Clough - 1979
- UCB/EERC-79/25 "Shaking Table Study of Single-Story Masonry Houses, Volume 3: Summary, Conclusions and Recommendations," by R.W. Clough, P. Gulkan and R.L. Mayes

UCB/EERC-79/26 "Recommendations for a U.S.-Japan Cooperative Research Program Utilizing Large-Scale Testing Facilities," by U.S.-Japan Planning Group - 1979

- UCB/EERC-79/27 "Earthquake-Induced Liquefaction Near Lake Amatitlan, Guatemala," by H.B. Seed, I. Arango, C.K. Chan, A. Gomez-Masso, and R. Grant de Ascoli - 1979
- UCB/EERC-79/28 "Infill Panels: Their Influence on Seismic Response of Buildings," by J.W. Axley and V.V. Bertero - 1979
- UCB/EERC-79/29 "3D Truss Bar Element (Type 1) for the ANSR-II Program," by D.P. Mondkar and G.H. Powell - 1979
- UCB/EERC-79/30 "2D Beam-Column Element (Type 5 Parallel Element Theory) for the ANSR-II Program," by D.G. Row, G.H. Powell, and D.P. Mondkar
- UCB/EERC-79/31 "3D Beam-Column Element (Type 2 Parallel Element Theory) for the ANSR-II Program," by A. Riahi, G.H. Powell, and D.P. Mondkar - 1979
- UCB/EERC-79/32 "On Response of Structures to Stationary Excitation," by A. Der Kiureghian - 1979
- UCB/EERC-79/33 "Undisturbed Sampling and Cyclic Load Testing of Sands," by S. Singh, H.B. Seed, and C.K. Chan - 1979
- UCB/EERC-79/34 "Interaction Effects of Simultaneous Torsional and Compressional Cyclic Loading of Sand," by P.M. Griffin and W.N. Houston - 1979
- UCB/EERC-80/01 "Earthquake Response of Concrete Gravity Dams Including Hydrodynamic and Foundation Interaction Effects," by A.K. Chopra, P. Charkabarti, and S. Gupta - 1980

.

· ·

. ఆర